

Barwell, Nicholas Dominic (2010) *Studies of the interaction between diabetes family history, exercise, adiposity and metabolic health.* PhD thesis.

http://theses.gla.ac.uk/2072/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

# Studies of the interaction between diabetes family history, exercise, adiposity and metabolic health

# Nicholas Dominic Barwell

Submitted for the degree of Doctor of Philosophy

November 2009



#### ABSTRACT

The rising tide of obesity and type 2 diabetes has been recognised to have reached epidemic proportions. There is a significant burden of mortality and morbidity associated with the development of these conditions and current estimates suggest that the burden of disease in the next two decades is likely to place considerable strain upon healthcare systems, particularly in the developing world. The development of insulin resistance is a key contributor to the pathogenesis of type 2 diabetes, however the origin of insulin resistance is complex and it is currently unclear precisely how the inter-related components of this metabolic dysfunction are triggered. Obesity is also implicated in the pathogenesis of insulin resistance and represents a risk factor which is potentially modifiable by lifestyle interventions such as exercise and weight loss. Observational and prospective studies have also shown the benefits of lifestyle intervention in reducing the incidence of diabetes in those judged to be at greater risk.

People with a parental history of type 2 diabetes have an increased lifetime risk of diabetes and frequently display metabolic abnormalities which, despite persisting normoglycaemia, are evidence of a 'pre-diabetic' state and which may themselves carry increased morbidity and mortality. Observational studies suggest a greater difference in insulin sensitivity between active and sedentary offspring, compared to the difference between active and sedentary individuals with no diabetes family history. This is thought to represent an interaction between positive energy balance, a sedentary lifestyle and a 'thrifty genotype'. These observations suggest that individuals with a parental history of diabetes are more susceptible to the deleterious health effects of a sedentary lifestyle, but that they may be more responsive to an increase in physical activity. Exercise interventions can be expected to have positive effects upon metabolic health and adiposity however, the individual response to exercise is extremely variable. Other factors such as lifestyle alteration have been implicated in the difference between the observed and predicted response to exercise. Therefore, the aim of this thesis was to examine the physical, metabolic and lifestyle differences between sedentary pre-menopausal women with a parent with type 2 diabetes and matched control subjects with no family history of the condition. In particular, this thesis aimed to explore the effect of an aerobic exercise intervention on metabolic health and body composition, whether the response to exercise is dependent upon a familial history of diabetes and the processes by which exercise might induce any observed changes.

In order to explore the impact of a sedentary lifestyle on women with, and without a family history of type 2 diabetes, thirty four pre-menopausal, sedentary women with a parental history of diabetes (Offspring) and thirty six matched women without a familial history of diabetes (Controls) were recruited. Assessments of body composition, insulin sensitivity, adipose tissue-derived hormone concentration, substrate utilisation, endothelial function by carotid-radial pulse wave velocity, cardiorespiratory fitness, diet and habitual physical activity were performed. Twenty eight Offspring subjects and thirty four matched Controls participated in a seven-week aerobic exercise intervention, training at 65-80 % of predicted maximal heart rate with incremental increases in training duration on a weekly basis. The previously described assessments were performed before and 15-24 hours after the intervention and in a subgroup of 19 Controls and 17 Offspring subjects, further assessments of insulin sensitivity, adipose tissue-derived hormone concentration, substrate utilisation and endothelial function were performed after a further three day period without exercise. In order to determine potential mediators of exercise-induced fat loss fifty five women participated in measurements of substrate utilisation, body composition, endothelial function, insulin sensitivity, cardiorespiratory fitness, dietary intake and habitual physical activity prior to, and after the seven week exercise intervention.

The findings from these studies confirmed that sedentary women with a family history of type 2 diabetes displayed lower insulin sensitivity than those without a parental history of

diabetes. In addition, insulin resistance in this group appears to be related to a greater sensitivity to the influence of adipose tissue, particularly circulating non-esterified fatty acids and adipose tissue-derived inflammatory cytokines. In Offspring alone, baseline insulin sensitivity was associated with plasma adiponectin concentration and negatively associated with circulating non-esterified fatty acid concentration. These associations may represent physiological attempts to compensate for developing insulin resistance. Offspring also displayed an augmented metabolic response to the exercise intervention in comparison to Controls. This study showed a 23% increase in post-intervention insulin sensitivity in Offspring with no significant increase in insulin sensitivity in Controls despite a similar improvement in cardiorespiratory fitness and adherence to the exercise regime. Improved post-intervention insulin sensitivity was accompanied by reduced circulating leptin, increased fat and decreased carbohydrate oxidation in both fasting and post-glucose states. No change in diet was observed but Offspring appeared to increase their level of habitual physical activity. The magnitude of change in insulin sensitivity was associated with a parental history of diabetes, but stronger associations were observed between baseline insulin resistance and an ability to reduce circulating leptin in response to exercise. Wide individual variation in fat mass change was observed in the response to exercise, and as expected the strongest predictor of exercise-mediated fat mass reduction was the net energy cost of the intervention. However, a change in fasting respiratory exchange ratio (RER), suggesting an increase in fat oxidation was also independently associated with reduced fat mass.

The combined findings of this thesis suggest that sedentary pre-menopausal daughters of people with type 2 diabetes are more insulin resistant and that this state is, in part, a consequence of heightened sensitivity to fatty acid and inflammatory cytokine release from adipose tissue. However, it would also appear that they represent a high-risk group who are susceptible to the insulin-sensitising effects of exercise and that this may be mediated by a metabolic pathway which involves reductions in circulating leptin concentrations. Finally, the ability to lose fat mass in response to exercise is related to the energy deficit incurred by the

activity but also by an individual's ability to shift fasting substrate utilisation towards fat oxidation. Public health strategies have traditionally focused upon lifestyle interventions which are directed at the population in general. However, awareness of the risks conferred by obesity and familial history of type 2 diabetes and the potential benefits of intervention may suggest that targeting public health resources towards these high-risk groups is a more appropriate and effective strategy.

# TABLE OF CONTENTS

Abstract	2
Table of Contents	6
List of Tables	16
List of Figures	19
Acknowledgements	24
Declaration of Publications	26
Abbreviations	27
Chapter 1 – Introduction	29
1.1 Introduction	29
1.2 Type 2 diabetes mellitus and other states of impaired glucose homeostasis	29
1.3 The rising tide of obesity and type 2 diabetes	33
1.4 Populations at risk of type 2 diabetes	33
1.5 Cardiovascular risk and insulin-resistant states	37
1.6 Common metabolic cofounders of cardiovascular risk in insulin resistant states	39
1.7 Type 2 diabetes: insulin resistance or beta-cell dysfunction?	41
1.8 The aetiology of insulin resistance	41
1.8.1 Insulin resistance and adipose tissue	42
1.8.2 Insulin resistance and skeletal muscle	45
1.8.3 Insulin resistance and endothelial dysfunction	47
1.8.4 Insulin resistance and the liver	49

1.9 Insulin resistance in obesity and lipodystrophy ('dysadiposity')	51
1.10 The effect of insulin resistance on people with a family history of T2D	53
1.11 Insulin resistance in women	54
1.12 Insulin resistance, cardiovascular risk and low grade inflammation	55
1.13 Adipose tissue hormones and insulin resistance	58
1.14 Genetic predisposition to insulin resistance	61
1.15 Environmental impact on genetic predisposition	63
1.16 Physical activity and diabetes risk	66
1.16.1 Metabolic impact of a single exercise session	67
1.16.2 Metabolic impact of repeated exercise sessions	70
1.16.3 Other insulin sensitising effects of exercise	71
1.16.4 Exercise in persons at risk of type 2 diabetes	73
1.17 Lifestyle interventions for weight loss	73
1.17.1 Diet and weight loss	74
1.17.2 Exercise and weight loss	75
1.18 Clinical approaches to treatment of insulin resistance	76
1.19 Summary	78
Chapter 2 – General Methods	81
2.1 Introduction	81
2.1.1 Inclusion and exclusion criteria	81

	2.1.2 Recruitment, screening and selection	82
	2.1.3 Study design	87
2.	2 Anthropometry and body composition	89
	2.2.1 Height	89
	2.2.2 Body mass	89
	2.2.3 Skinfold thickness	89
	2.2.4 Waist and hip circumference	90
	2.2.5 Dual energy x-ray absorptiometry scanning	91
2.	3 Metabolic testing	91
	2.3.1 Protocol of metabolic testing	91
	2.3.2 Metabolic rate and energy substrate utilisation	93
	2.3.3 Indirect calorimetry calculations	94
	2.3.4 Blood pressure	96
	2.3.5 Pulse wave velocity	96
	2.3.6 Fasting blood sampling	99
	2.3.7 Oral glucose tolerance test	99
2.	4 Plasma preparation and analysis	100
	2.4.1 Plasma preparation and storage	100
	2.4.2 Enzyme colorimetric methods	100
	2.4.3 Enzyme-linked immunoassays	101

2.5 Establishment of relationships of heart rate to $\dot{V}O_2$ and $\dot{V}CO_2$ during rest and	
exercise	102
2.5.1 Relationship of heart rate to $\dot{V}O_2$ and $\dot{V}CO_2$ during low intensity activities	102
2.6 Assessment of cardiorespiratory fitness	103
2.6.1 Assessment of heart rate	105
2.6.2 Expired air collection and analysis	106
2.6.3 Estimation of $\dot{V}O_{2max}$	107
2.6.4 Capillary lactate sampling	108
2.7 Exercise intervention	108
2.7.1 Exercise intensity and frequency	108
2.7.2 Exercise activities and facilities	110
2.8 Dietary monitoring and analysis	110
2.9 Evaluation of habitual physical activity	111
2.9.1 Heart rate recording	111
2.9.2 Evaluation of energy expenditure of exercise sessions	112
2.9.3 FLEX pilot study	113
2.9.3.1 Methodology for the evaluation of the FLEX method	114
2.9.3.2 Results	116
2.9.3.3 Discussion	117
2.10 Indices of physical activity	118

2.11 Power calculation	118
2.11.1 Statistical analysis	119
2.11.2 Index of insulin sensitivity calculation	119
Chapter 3 – Physical and metabolic characteristics of sedentary women with and	
without a family history of diabetes	121
3.1 Introduction	121
3.2 Methods	124
3.2.1 Volunteers	124
3.2.2 Study design	124
3.2.3 Energy and macronutrient intake	125
3.2.4 Establishment of relationships of heart rate to $\dot{V}O_2$ and $\dot{V}CO_2$	125
3.2.5 Evaluation of habitual physical activity	125
3.2.6 Evaluation of cardiorespiratory fitness	125
3.2.7 Anthropometric assessment	126
3.2.8 Metabolic testing	126
3.2.9 Blood analysis	126
3.2.9.1 Calculation of glucose, insulin, NEFA and 3-OHB concentrations and in	sulin
sensitivity during oral glucose tolerance test	127
3.2.9.2 Statistical analysis	127
3.3 Results	129

3.3.1 Concentrations of glucose, insulin, NEFA and 3-OHB	129
3.3.2 Insulin sensitivity	132
3.3.3 Physical characteristics and body composition	133
3.3.4 Metabolic rate and substrate utilisation	136
3.3.5 Blood pressure and peripheral pulse wave velocity	137
3.3.6 Plasma lipids	137
3.3.7 Liver enzymes	138
3.3.8 Adipokines and C-reactive protein	138
3.3.9 Cardiorespiratory fitness and habitual physical activity	143
3.3.10 Energy intake and dietary composition	146
3.4 Discussion	147
Chapter 4 – Physical and metabolic responses to a structured exercise progr	amme in
women with and without a family history of diabetes	156
4.1 Introduction	156
4.2 Methods	158
4.2.1 Volunteers	158
4.2.2 Study design	158
4.2.3 Fitness testing	159
4.2.4 Metabolic testing	159
4.2.5 Body composition assessment	159

4	.2.6 Assessment of diet and habitual physical activity	160
4	.2.7 Exercise intervention	160
4.3 \$	Statistical analysis	160
4	.3.1 Calculations of glucose, insulin, NEFA and 3-OHB and insulin	
	sensitivity during OGTT	161
4.4 1	Results	161
4	.4.1 Changes during the 'usual lifestyle' period for the Delayed Exercise	
	group	161
4	.4.2 Compliance to exercise intervention	162
4	.4.3 Post-intervention data	162
	4.4.3.1 Physical characteristics and body composition	162
	4.4.3.2 Concentrations of glucose, insulin, NEFA and 3-OHB	165
	4.4.3.3 Insulin sensitivity index	166
	4.4.3.4 Plasma lipids	168
	4.4.3.5 Blood pressure and peripheral pulse wave velocity	168
	4.4.3.6 Liver enzymes	169
	4.4.3.7 Adipokines and C-reactive protein	169
	4.4.3.8 Metabolic rate and substrate utilisation	171
	4.4.3.9 Cardiorespiratory fitness and habitual physical activity	172
	4.4.3.10 Energy intake and dietary composition	174

4.4.4 Chronic effects of exercise intervention	175
4.4.4.1 Three-day 'detraining' results	175
4.5 Discussion	177
Chapter 5 – Determinants of the exercise mediated change in insulin sensitivity	183
5.1 Introduction	183
5.2 Methods	189
5.2.1 Volunteers	189
5.2.2 Study design	189
5.2.3 Fitness testing	190
5.2.4 Metabolic testing	190
5.2.5 Body composition assessment	190
5.2.6 Exercise intervention	191
5.3 Statistical analysis	191
5.4 Results	192
5.4.1 Major influences on exercise-mediated change in insulin sensitivity	192
5.4.2 Baseline influences on exercise-mediated change in insulin sensitivity	193
5.4.3 Exercise-mediated changes influencing insulin sensitivity	195
5.4.4 Multivariate analysis	205
5.5 Discussion	207
Chapter 6 – Determinants of fat loss in response to exercise in previously	
sedentary women	220

6.1 Introduction	220
6.2 Methods	223
6.2.1 Volunteers	223
6.2.2 Study design	223
6.2.3 Metabolic assessment	223
6.2.4 Body composition assessment	224
6.2.5 Fitness test and calibration of heart rate versus $\dot{V}O_2$ relationship	224
6.2.6 Physical activity and dietary assessment	224
6.2.7 Exercise intervention	225
6.2.8 Reproducibility and statistical power	225
6.2.9 Statistical analysis	226
6.3 Results	227
6.4 Discussion	236
Chapter 7 – General Discussion	242
7.1 Experimental chapter summaries	242
7.2 Adipose tissue – distribution, function and insulin sensitivity	248
7.3 Lifestyle factors and implications for weight loss and insulin sensitivity	255
7.4 Results in women – implications for future studies	258
7.4.1 General considerations	258
7.4.2 Insulin resistance in Offspring	259

7.4.3 Augmented metabolic response to exercise	260
7.4.4 Determinants of exercise-mediated fat loss	262
References	264
Appendices	313

# LIST OF TABLES

# Chapter 1 – Introduction

Table 1.1 Diagnostic criteria for diabetes mellitus	30
Table 1.2 Definitions of the metabolic syndrome	32
Chapter 2 - General Methods	
Table 2.1 Description of exercise intervention	109
Table 2.2 Total daily energy expenditure at baseline and after intervention. Post-	
intervention energy expenditure calculated by different methods	116
Chapter 3 – Physical and metabolic characteristics of sedentary women with and	
without a family history of diabetes	
Table 3.1 Fasting and OGTT values for glucose, insulin, NEFA and 3-OHB	130
Table 3.2 Physical characteristics and body composition	133
Table 3.3 Correlations between indices of adiposity and insulin sensitivity	135
Table 3.4 Metabolic rate and substrate utilisation in fasted and post-glucose state	136
Table 3.5 Blood pressure and peripheral pulse wave velocity	137
Table 3.6 Fasting lipids	138
Table 3.7 Fasting adipokines and C-reactive protein	139
Table 3.8 Correlations between adipokines and insulin sensitivity	141
<b>Table 3.9</b> Indices of cardiorespiratory fitness and habitual physical activity	143
Table 3.10 Correlations between indices of cardiorespiratory fitness, habitual physical	
activity and unadjusted insulin sensitivity and insulin sensitivity adjusted	

for fat mass	145
Table 3.11 Energy and macronutrient intake	146
Chapter 4 – Physical and metabolic responses to a structured exercise programme	in
women with and without a family history of diabetes	
<b>Table 4.1</b> Absolute change in physical characteristics and body composition	164
Table 4.2 Absolute change in fasting and OGTT values for glucose, insulin, NEFA and	1
3-OHB	166
<b>Table 4.3</b> Absolute change in blood pressure and peripheral pulse wave velocity	168
Table 4.4 Absolute change in fasting adipokines	169
<b>Table 4.5</b> Absolute change in metabolic rate and substrate utilisation	171
<b>Table 4.6</b> Absolute change in cardiorespiratory fitness and indices of habitual activity	173
Table 4.7 Absolute change in energy and macronutrient intake	174
Table 4.8 Absolute change in 3-day 'detraining' metabolic variables	176
Chapter 5 – Determinants of the exercise mediated change in insulin sensitivity	
Table 5.1 Correlations between baseline insulin sensitivity and the change in insulin	
sensitivity after exercise	194
Table 5.2 Correlations between post-exercise variable change and change in: unadjusted	ed
insulin sensitivity, insulin sensitivity adjusted for change in fat mass, insulin	1
sensitivity adjusted for baseline insulin sensitivity	199
Table 5.3 Correlations between post-exercise variable change and change in: insulin	
sensitivity adjusted for total net energy expenditure of intervention, insulin	

sensitivity adjusted for net energy expenditure of last week of interventon,

insulin sensitivity adjusted for net energy expenditure of last session.	203
Chapter 6 – Determinants of fat loss in response to exercise in previously sedentar	ry
women	
Table 6.1 Volunteer characteristics at baseline and in response to exercise	228
Table 6.2 Correlations between interventions, behavioural compensation and physiol	ogical
variables and change in fat mass in response to exercise training	231

# LIST OF FIGURES

# Chapter 1 – Introduction

Figure 1.1 Gene-environment interaction in development of type 2 diabetes	35
Figure 1.2 Interaction between insulin resistance, obesity and	
inflammation in the development of type 2 diabetes	40
Chapter 2 – General Methods	
Figure 2.1 Protocol of Bruce Exercise Tolerance Test	85
Figure 2.2 Flowchart of volunteer recruitment, randomisation and retention	86
Figure 2.3 Study design for Delayed Exercise group	88
Figure 2.4 Study design for Exercise group	88
Figure 2.5 Timeline illustrating metabolic testing protocol	92
Figure 2.6 Photograph of male volunteer using Deltatrac Metabolic Monitor	94
Figure 2.7 Illustration of measurement of peripheral pulse wave velocity	97
Figure 2.8 Protocol of submaximal exercise test	105
Figure 2.9 Illustration of relationship between oxygen consumption and heart rate at h	nigh
and low intensity calculated by (A) heart rate <sup>3</sup> and (B) separation of 'active	,7
and 'inactive' heart rates as required by the FLEX method	113
Figure 2.10 Differences in pre- and post-intervention resting heart rates measured by	
different methods	114

Chapter 3 – Physical and metabolic characteristics of sedentary women with and	
without a family history of diabetes	
Figure 3.1 Glucose, insulin, NEFA and 3-OHB curves during OGTT	131
Figure 3.2 Insulin sensitivity index	132
Figure 3.3 Scattergrams illustrating relationship between total fat mass and baseline	
insulin sensitivity	134
Figure 3.4 Individual value plots of baseline adipokines and C-reactive protein	140
Figure 3.5 Scattergrams illustrating relationship between adiponectin and fat	
mass-adjusted insulin sensitivity	142
Figure 3.6 Scattergrams illustrating relationship between C-reactive protein and	
fat mass-adjusted insulin sensitivity	142
Chapter 4 – Physical and metabolic responses to a structured exercise programme	
in women with and without a family history of diabetes	
Figure 4.1 Insulin sensitivity index measured prior to and after exercise	167
Figure 4.2 Absolute change in insulin sensitivity index after exercise	167
Figure 4.3 Circulating leptin concentration prior to and after exercise	170
Figure 4.4 Absolute change in circulating leptin after exercise	170
Figure 4.5 Relationship between heart rate and $\dot{V}O_2$ before and after intervention	172
Chapter 5 – Determinants of the exercise mediated change in insulin sensitivity	
Figure 5.1 Individual changes in insulin sensitivity ranked in order of unadjusted ISI	192

Figure 5.2 Scattergram illustrating relationship between baseline insulin sensitivity

and change in insulin sensitivity after exercise	193
Figure 5.3 Scattergram illustrating relationship between change in leptin and change	
in insulin sensitivity after exercise.	196
Figure 5.4 Individual changes in insulin sensitivity after adjustment for change in	
fat mass, ranked in order of unadjusted insulin sensitivity	197
Figure 5.5 Individual changes in insulin sensitivity after adjustment for baseline insuli	n
sensitivity, ranked in order of unadjusted insulin sensitivity	198
Figure 5.6 Individual changes in insulin sensitivity after adjustment for: total net energy	;y
expended during intervention, net energy expended during last week of	
intervention, ranked in order of unadjusted insulin sensitivity	202
Chapter 6 – Determinants of fat loss in response to exercise in previously sedentary	,
women	
Figure 6.1 Individual changes in fat mass after exercise intervention, ranked in order of	f
change in unadjusted fat mass	
change in unadjusted fat mass	229
<b>Figure 6.2</b> Individual changes in fat mass adjusted for net energy expenditure of	229
<b>Figure 6.2</b> Individual changes in fat mass adjusted for net energy expenditure of intervention, after exercise intervention, ranked in order of change in	229
<b>Figure 6.2</b> Individual changes in fat mass adjusted for net energy expenditure of intervention, after exercise intervention, ranked in order of change in unadjusted fat mass	229 232
<ul> <li>Figure 6.2 Individual changes in fat mass adjusted for net energy expenditure of intervention, after exercise intervention, ranked in order of change in unadjusted fat mass</li> <li>Figure 6.3 Individual changes in fat mass adjusted for net energy expenditure of</li> </ul>	229 232
<ul> <li>Figure 6.2 Individual changes in fat mass adjusted for net energy expenditure of intervention, after exercise intervention, ranked in order of change in unadjusted fat mass</li> <li>Figure 6.3 Individual changes in fat mass adjusted for net energy expenditure of intervention and change in energy intake, after exercise intervention,</li> </ul>	229
<ul> <li>Figure 6.2 Individual changes in fat mass adjusted for net energy expenditure of intervention, after exercise intervention, ranked in order of change in unadjusted fat mass</li> <li>Figure 6.3 Individual changes in fat mass adjusted for net energy expenditure of intervention and change in energy intake, after exercise intervention, ranked in order of change in unadjusted fat mass</li> </ul>	229 232 233

residual change in fat mass, adjusted for the effects of net energy cost of intervention and change in energy intake 234

# **Chapter 7 – General Discussion**

Figure 7.1 Division of subcutaneous adipose tissue into superficial and deep subcutaneous

adipose tissue pools

All people dream, but not equally. Those who dream by night in the dusty recesses of their minds wake in the day to find it was vanity. But the dreamers of the day are dangerous people, for they may act on their dreams with open eyes to make it possible.

T.E Lawrence.

In the beginning of the malady it is easy to cure but difficult to detect, but in the course of time, not having been either detected or treated in the beginning, it becomes easy to detect but difficult to cure.

Niccolò Macchiavelli (1513).

#### ACKNOWLEDGEMENTS

The fact that this thesis exists at all is a testament to the support that I have received throughout the years of both the study periods and the protracted write-up. Moving from an entirely clinical career to a research environment was something of a culture shock for me, particularly given the steep learning curve which I faced in relation to performing genuine scientific experiments. I could not have achieved this without the expertise and patient understanding of the technical staff in the exercise science laboratories of the Institute of Biomedical and Life Sciences at Glasgow University. I would particularly like to thank John Wilson, Heather Collin, Paul Paterson and Ian Watt. Not only did you support me with the 'nuts and bolts' of the experiments, but with your time and experience. I would also like thank Dr Francesca Burton and Dr Robert Scott who provided unpaid tutoring and moral support.

I was fortunate enough to have a number of undergraduate students who provided vital manpower for experiments to be conducted. Jennifer Graham, Martyn Glancy, Eileen Barr, Gemma Brown, Iosif Alexandridis, Billy Murphy, Jo Kirkby, Melanie Leggate and Oonagh Markey were instrumental in keeping the study timetable on schedule. Particular thanks must go to Jo, Melanie and Oonagh who really engaged in the study and gave freely of their time and their intellect. Josephine Cooney, and all the staff in the Vascular Biochemistry labs at Glasgow Royal Infirmary were extremely efficient and very supportive. Carol Stewart and all the staff at the Sports and Recreation Service were instrumental in facilitating free exercise facilities and an informal meeting place for the volunteers. Jane Logan's expertise with the Yorkhill DEXA scanner was vital to the study. The members and moderators of www.kerrydalestreet.co.uk are to be thanked for allowing me to recruit through their website. The British Heart Foundation supported this work financially, through the media, and their backing was vital to recruitment and public interest in this work. My heartfelt thanks goes to all the wonderfully tolerant and enthusiastic ladies who participated in these studies. I put

them through a punishing schedule of testing and exercising, and they bore the time constraints, frequent lab visits and the exercise regime with grace and patience. I apologise to them all for my terrible chat and 'motivational' mottos!

I would also like to thank Drs Dalia Malkova and Jason Gill who co-supervised me through this process. You have been encouraging, patient and tolerant and most importantly you have provided me with the benefit of your extensive experience and knowledge. Your support and direction has been invaluable, and the balance of your personalities has allowed me to push myself harder than I thought I could, yet retain a sense of humour and perspective. You are both wonderful scientists and excellent people. My clinical supervisor, Dr Steve Cleland has been a very valuable friend, as well as an example of how to meld pure science with everyday clinical practice.

Finally I must thank those long-suffering friends and family who have had to endure my moaning and increasing social isolation. Steve, Nicki, Greg, Sheona, Big Dave, Lesley, Casino Dave, Esther, Stu, Helena, Mackie, Kathryn, Mark and Robert have encouraged and cajoled (and provided IT support). My parents, Margaret and Iain are truly wonderful people. I could not imagine a more supportive couple, and I am in no doubt that I could not have managed to achieve anything of note in my life without your love, friendship and example. Finally, I must thank my wonderful wife Lorraine. The wee Sausage Mole has had to put up with a part-time husband for most our married life, but has always been there for me, and has kept me focussed and rational when hysterics were the next option! I love you more than you'll ever know.

#### **DECLARATION OF PUBLICATIONS**

Unless otherwise indicated by acknowledgement or reference to published literature the work contained herein is that of the author.

The findings of some of the studies described in this thesis have been published as follows:

#### **Published research papers**

Barwell ND, Malkova D, Moran CN, Cleland SJ, Packard CJ, Zammit VA, Gill JM. (2008). Exercise training has greater effects on insulin sensitivity in daughters of patients with type 2 diabetes than in women with no family history of diabetes. *Diabetologia 51(10):1912-9*.

Barwell ND, Malkova D, Leggate M, Gill JM. (2009)

Individual responsiveness to exercise-induced fat loss is associated with change in resting substrate utilization. *Metabolism.* 58(9):1320-8.

### LIST OF ABBREVIATIONS

T2D	Type 2 diabetes mellitus		
VO <sub>2max</sub>	Maximal oxygen uptake		
3-ОНВ	3-hydroxybutyrate		
NEFA	Non-esterified fatty acid		
PWV	Pulse wave velocity		
BP	Blood pressure		
TG	Triglyceride		
HDL	High density lipoprotein		
LDL	Low density lipoprotein		
VLDL	Very-low density lipoprotein		
sdLDL	Small, dense low density lipoprotein		
BMI	Body mass index		
RMR	Resting metabolic rate		
PGMR	Post-glucose metabolic rate		
RER	Respiratory exchange ratio		
СНО	Carbohydrate		
SFA	Saturated fatty acids		
MUFA	Monounsaturated fatty acids		
PUFA	Polyunsaturated fatty acids		

RPE	Rate of perceived exertion		
IFG	Impaired fasting glycaemia		
IGT	Impaired glucose tolerance		
PPAR	Peroxisome proliferator-activated receptor		
GNG	Gluconeogenesis		
GL	Glycogenolysis		
HGP	Hepatic glucose production		
AMPK	Adenosine monophosphate-kinase		
LPL	Lipoprotein lipase		

#### **CHAPTER 1**

#### **INTRODUCTION & LITERATURE REVIEW**

#### **1.1 Introduction**

This chapter aims to provide the relevant scientific background to the studies in this thesis, additionally it aims to establish the theoretical basis for these studies. This chapter begins with an overview of type 2 diabetes mellitus, insulin resistance, obesity and the public health impact of these conditions. Following this is a more detailed consideration of the role of insulin resistance and obesity as co-founders in the development of premature cardiovascular disease. Finally the aetiology of insulin resistance is explored, including mechanisms by which insulin resistance can be modified and which may lead to the reduction in risk of both the development of type 2 diabetes and cardiovascular disease.

#### 1.2 Type 2 diabetes mellitus and other states of impaired glucose homeostasis

Diabetes mellitus is a disease of metabolic dysfunction, characterised by chronic hyperglycaemia with disturbances of carbohydrate, fat and protein metabolism resulting from defects in insulin secretion, insulin action or both (WHO Consultation 1999) (Table 1.1). Type 2 diabetes (T2D) is a disease of insulin resistance, although defects in pancreatic betacell function, mass and insulin secretion are generally also required to achieve the degree of hyperglycaemia which is diagnostic of the condition (Ferrannini 1998;Kahn 2003;McGarry 2002;Petersen & Shulman 2006;Reaven 1988). Chronic hyperglycaemia is associated with the development of microvascular disease which, through an effect on the microvascular beds of the retina, renal parenchyma, and both peripheral and autonomic nervous systems may give rise to symptomatic or occult complications (The DCCT Research Group 1993;UKPDS Study Group 1998). However, the process of microvascular disease is not confined to these vascular beds (Rizzoni et al. 2001). Indeed, microvascular disease is recognised to be present early in the pathogenesis of T2D (Jaap, Shore, & Tooke 1997), at the level of the skeletal muscle microcirculation and is closely associated with the degree of insulin resistance (Sydow, Mondon, & Cooke 2005). Of additional clinical relevance, macrovascular diseases such as coronary artery disease, cerebrovascular disease and occlusive peripheral vascular disease are more common in patients with diabetes (Stamler et al. 1993).

Other states of impaired glucose homeostasis have been categorised and although these states do not share the chronic hyperglycaemia of T2D, many of the fundamental metabolic abnormalities found in T2D are common to these 'pre-diabetic' conditions. Impaired Glucose Tolerance (IGT) and Impaired Fasting Glycaemia (IFG) are both associated with an increased risk of developing T2D (Unwin et al. 2002). However although these conditions can exist separately, they may also overlap (Table 1.1) and represent stages on the continuum of insulin resistance and dysglycaemia (Bi et al. 2010;Ning et al. 2010). The risk of developing T2D from either isolated IFG, isolated IGT or combined IFG-IGT is dependent on the background risk within a given population: however, the development of altered glucose homeostasis is associated with a significant increase in the risk of progression to T2D and the combination of IFG and IGT may be associated with an increased relative risk of the order of 4-15 fold over a period of 5 - 10 years (Unwin et al. 2002).

	Fasting plasma glucose (mmol.l <sup>-1</sup> )	2 – hour post-glucose load plasma glucose (mmol. <sup>1-</sup> )
Diabetes mellitus	$\geq 7.0$	≥11.1
Isolated IFG	6.1 - 6.9	-
Isolated IGT	≤ 6.1	7.8 - 11.0
Combined IFG & IGT	6.1 - 6.9	7.8 - 11.0

Table 1.1 Diagnostic criteria for diabetes mellitus, IFG and IGT (WHO Consultation 1999).

The fasting plasma glucose threshold of 7.0 mmol.1<sup>-1</sup> is a diagnostic indicator of diabetes, but is also the threshold at which microvascular complications begin to occur (WHO Consultation 1999). There is no evidence for increased risk of microvascular complications in patients with IFG or IGT (Unwin et al. 2002), although subclinical microvascular dysfunction is present in these dysglycaemic states and is evident early in the pathogenesis of insulin resistance (Jessani, Millane, & Lip 2009). However, macrovascular complication risk does not appear to be limited by a glycaemic threshold. There is evidence from observational studies that glycaemia is independently associated with cardiovascular events and that post-prandial glycaemia is more strongly associated with cardiovascular events than fasting glycaemia (Hanefeld et al. 2000;The DECODE Study Group 1999;The DECODE Study Group 2001).

The term 'Metabolic Syndrome' (Ferrannini et al. 1991;Haller 1977;Reaven 1992) has been used to define a group of risk factors which are common to both the development of T2D and cardiovascular disease (Laws & Reaven 1993;Lebovitz 2006). The exact definition of the components of the Metabolic Syndrome has been debated, leading to a number of different definitions and a number of adjustments for ethnicity (Table 1.2). Regardless of the definition, emphasis is placed on the combination of central obesity, dyslipidaemia, hypertension and dysglycaemia (Alexander et al. 2003;International Diabetes Federation 2005;WHO Consultation 1999). Identification of the Metabolic Syndrome is of clinical benefit in raising awareness of both cardiovascular risk and risk of subsequent development of T2D (Grundy et al. 2006;Saely et al. 2006;Sattar et al. 2003).

	WHO	NCEP ATP III	IDF
Insulin resistance	↑ Insulin levels <u>or</u>	Fasting blood glucose	Fasting plasma glucose
	↑ Fasting plasma glucose <u>or</u>	$\geq$ 6.1 mmol.l <sup>-1</sup>	$\geq$ 5.6 mmol.l <sup>-1</sup> or
	↑ Post-prandial glucose		Known T2D
Dyslipidaemia	$TG > 1.7 \text{ mmol.l}^{-1}$	$TG > 1.7 \text{ mmol.l}^{-1}$	$TG > 1.7 \text{ mmol.l}^{-1*}$
	<u>or</u>	HDL < 1.03 mmol.l <sup>-1</sup> $\stackrel{\frown}{\bigcirc}$	HDL < 1.03 mmol.l <sup>-1</sup> $\circlearrowleft^*$
	$HDL < 1.94 \text{ mmol.l}^{-1}$	$< 1.29 \text{ mmol.l}^{-1} \uparrow$	$< 1.29 \text{ mmol.l}^{-1} \bigcirc^*$
Blood pressure	$\geq$ 140/90 <sup>*</sup>	> 130/85	$\text{SBP} \ge 130^*$
			$DBP \ge 85^*$
Obesity	Waist-Hip ratio > 0.9	Waist circ > 102 cm $\stackrel{?}{\bigcirc}$	Waist circ $\ge$ 94 cm $^{\sim}$
	BMI 30	$>$ 88 cm $\stackrel{\bigcirc}{\downarrow}$	$\ge$ 80 cm $\bigcirc$
	Waist circ > 94cm		

**Table 1.2** Defining criteria for Metabolic Syndrome.

Shaded boxes indicate that condition is required for diagnosis. For WHO & IDF classifications, additional 2 criteria must be met. For NCEP classification a combination of any 3 factors is diagnostic. <sup>\*</sup>or treatment for stated condition (Alexander et al. 2003;International Diabetes Federation 2005;WHO Consultation 1999).

Metabolic syndrome, IFG and IGT criteria allow identification of individuals 'at-risk' of developing premature cardiovascular disease, type 2 diabetes or both. Although not characterised as disease states in themselves, they serve to focus clinical perspective on aggressive risk factor management. Early lifestyle intervention with dietary and physical activity advice is recommended (Grundy et al. 2006;Knowler et al. 2002;Orchard et al.2005;Pan et al. 1997;Tuomilehto et al. 2001), with additional use of pharmacological agents to treat hypertension and dyslipidaemia. To date, use of pharmacological agents to treat dysglycaemia has been extensively investigated, with evidence to suggest that progression from IGT and IFG to T2D can be delayed and possibly prevented (Scheen 2007). There is however, a lack of long-term data on the efficacy of these interventions (Grundy et al.

al. 2006) and doubt about the cost-effectiveness of alternatives (Eddy, Schlessinger, & Kahn 2005;Herman et al. 2005;Johansson et al. 2009).

#### **1.3** The rising tide of obesity and type 2 diabetes

The global prevalence of obesity is increasing (Ford & Mokdad 2008), and from a long-term public health perspective, this situation is likely to deteriorate further given the burden of obesity in children and adolescents (Krassas et al. 2001). The obesity epidemic has been paralleled by a global increase in the prevalence of T2D (King, Aubert, & Herman 1998;Wild et al. 2004). Increases in diabetes may be partially related to the aging population in developed countries (Wild et al. 2004), however the largest increase in diabetes is projected to occur in the 45 – 64 year age group in developing countries which is related to increased urbanised lifestyle and associated obesity in this age group (King et al. 1998;Wild et al. 2004). Recent estimates of a global prevalence of 366 million patients with diabetes by 2030 may be conservative based on assumptions that the current prevalence of obesity remains constant (Wild et al. 2004). Obesity is a well recognised risk factor for the development of T2D (Barrett-Connor 1989;Edelstein et al. 1997;Ohlson et al. 1985;Ohlson et al. 1988) and may account for 80-90% of all T2D incidence (Astrup & Finer 2000), however not all those persons who become obese will develop T2D and both insulin resistant pre-diabetic states and T2D may occur in non-obese individuals (St Onge, Janssen, & Heymsfield 2004).

#### 1.4 Populations at risk of type 2 diabetes

The population risk of diabetes is not constant between ethnic groups. Ethnically diverse countries such as the USA allow the analysis of the impact of ethnicity on the risk of developing diabetes and comparison of the diabetogenic effects of a 'Westernised' lifestyle (sedentary physical activity levels, high energy diet rich in fat and refined carbohydrate) on ethnic groups by comparison with their parent populations (Nakanishi et al. 2004;Schulz et

al. 2006). McBean and colleagues (2004) observed a higher diabetes prevalence in Black, Hispanic and Asian Americans compared with Caucasian Americans, and recorded the greatest increases in rates of diabetes amongst the Asian Americans over a seven year period (McBean et al. 2004). Although some of the increased risk in ethnic groups may be related to greater obesity levels (Centers for Disease Control and Prevention (CDC) 2009;Lorenzo et al. 2009), compared to Caucasians, other ethnic populations continue to display increased risk in normal weight individuals (Zhang, Wang, & Huang 2009). Maskarinec and co-workers (2009) describe an approximate two-fold increase in diabetes prevalence in all ethnic groups compared to Caucasians, even after adjustment for body mass index (BMI) (Maskarinec et al. 2009).

As previously stated, obesity is not the sole determinant of T2D. A multiple regression analysis of over 100,000 person-years of observational data in the Physicians' Health Study suggests that a sedentary lifestyle may account for 25% of all diabetes incidence (Manson et al. 1992). In addition, comparative ethnic studies have identified increased risk of diabetes in populations exposed to a Westernised lifestyle, compared with genetically similar populations who have higher levels of habitual physical activity and lower energy diets (Nakanishi et al. 2004; Schulz et al. 2006). Since these populations are genetically identical, the increased risk of T2D must be the result of an interaction between a genetic susceptibility to T2D and an environment which promotes diabetes (Figure 1.1).



Figure 1.1 Gene-environment interaction in the development of T2D.

The paradigm of the gene-environment interaction is observed in the Pima Indian communities in the USA and in Mexico. Mexican Pima Indians lead a physically demanding non-mechanised, agricultural lifestyle, and have no greater prevalence of obesity or diabetes than non-Pima Mexicans living in the same area (Valencia et al. 1999). Five-fold increases in the prevalence of both obesity and T2D have been described in American Pima Indians, despite higher energy intake in the Mexican Pimas (Schulz et al. 2006;Valencia et al. 1999). A significant difference was observed between physical activity levels in the Mexican and American populations, with American Pima Indians participating in less than 5 hours of activity per week (a 4-fold difference) (Schulz et al. 2006;Valencia et al. 1999). More detailed assessment of the dietary habits of the American Pima Indians revealed a similar energy intake and macronutrient composition to the general American population but compared to the Mexican Pimas, a greater percentage of the calories consumed were derived from fat (Smith et al. 1996).

Further evidence of a gene-environment interaction can be seen in other groups. The Australian Aboriginals of the Torres Strait Islander community display rates of obesity and
diabetes which are 3-fold and 6-fold higher than the general Australian population (Leonard et al. 2002). Canadian Oji-Cree Indians, like American Pimas, display an extremely high prevalence of T2D and a close relationship between development of the condition and a Westernised lifestyle (Hegele et al. 2003). The common factor in all these geographically disparate communities is a relatively rapid change in lifestyle; from a traditionally high level of habitual physical activity and a diet composed of high fibre and low fat, to a sedentary lifestyle with equivalent total energy intake but a dietary composition high in refined carbohydrate and fat. Specific 'pro-diabetes' gene mutations have been observed in Oji-Cree (Hegele et al. 2003) and Pima Indians (Ma et al. 2008). The rapid increase in diabetes prevalence in these communities cannot be attributed to genetic factors which are likely to have been present within these groups for generations. However, given the 'sudden' rise in diabetes, it is likely that a longstanding genetic predisposition towards T2D (which presumably previously conferred a selection advantage) has combined with a 'diabetogenic' lifestyle. This so-called 'thrifty genotype' has previously been described (Neel 1962), and considered in the context of diabetes (Booth, Chakravarthy, & Spangenburg 2002).

Other groups who are at risk of T2D have been identified. First-degree relatives of patients with T2D have approximately three times the risk of developing diabetes than matched controls with no family history of the disease (Ohlson et al. 1988;Kobberling & Tillil. 1982). In addition, they appear to display metabolic manifestations of this risk, in the form of insulin resistance, long before the development of dysglycaemia (Ezenwaka, Davis, & Offiah 2001;Humphriss et al. 1997;Perseghin et al. 1997;Warram et al. 1990). This risk appears to be particularly evident in sedentary people with a first-degree relative with T2D (Higgins et al. 2005). Sedentary offspring of people with diabetes display 'pre-diabetic' metabolic abnormalities such as reduced insulin sensitivity (Ahn et al. 2004;Ekelund et al. 2009;Ekelund, Griffin, & Wareham 2007) and are more likely to manifest higher waist circumferences and dyslipidaemia than active offspring (Higgins et al. 2005). This increased metabolic risk appears to be partially attenuated in offspring who are active (Ahn et al.

2004;Ekelund et al. 2009) it is however unclear whether intervention in sedentary individuals with established pre-diabetic metabolic changes would lead to a significant alteration in metabolic profile and the subsequent risk of developing T2D. One of the aims of this thesis is to establish the extent to which an exercise intervention influences the metabolic profile which is associated with the combination of a sedentary lifestyle and a parental history of T2D.

# 1.5 Cardiovascular risk and insulin-resistant states

The diagnosis of T2D is understood to confer up to a four-fold increase in cardiovascular risk (Kannel & McGee 1979;Stamler et al. 1993). Indeed, it has been suggested that the diagnosis of T2D should be viewed as a cardiovascular risk equivalent to that of a patient with a previous myocardial infarction (Haffner et al. 1998;Whiteley et al. 2005). However, recent meta-analysis would not support this conclusion and in fact suggests that individual patient risk factors should be assessed, rather than a 'blanket' assumption that the onset of T2D automatically confers increased cardiovascular risk (Bulugahapitiya et al. 2009). The method of this meta-analysis is not able to assess the specific cardiovascular impact of T2D on women and ethnic groups (Bulugahapitiya et al. 2009) and there is some evidence that T2D may have a relatively greater cardiovascular impact upon women compared to men (Becker et al. 2003;Hu et al. 2001;Mak & Haffner 2003), and South Asians compared to Caucasians (Patel et al. 2008). For South Asians it is unclear whether the increased cardiovascular risk is related to a specific effect of diabetes in this population, or simply related to the increased prevalence of T2D (Tziomalos et al. 2008).

Whilst T2D may have a heterogenous effect on cardiovascular risk which is difficult to quantify, it is clear that diabetes does contribute to cardiovascular mortality and morbidity (Stamler et al. 1993) and that cardiovascular risk is likely to be related to the duration of the

condition (Shah et al. 2009a). In addition to increasing risk of cardiovascular events, T2D is also associated with higher first-event mortality (Miettinen et al. 1998) and high rates of recurrent events (Giorda et al. 2008). Miettinen and colleagues (1998) also observed that the increased risk conferred by co-existing diabetes and vascular events was greater in women than in men (Miettinen et al. 1998). Additionally, patients with diabetes are estimated to suffer from cardiovascular events at a much younger age, possibly 15 years before equivalent non-diabetic patients (Booth et al. 2006). Patients with diabetes also suffer from greater morbidity after cardiovascular events (Murcia et al. 2004;Oliveira et al. 2002;Stone et al. 1989;Yu et al. 2000).

Increased cardiovascular risk is also observed in other insulin resistant states. The metabolic syndrome is associated with increased prevalence of coronary heart disease (Alexander et al. 2003;Lakka et al. 2002) and increased incidence of all cardiovascular events (Lakka et al. 2002;McNeill et al. 2005;Rutter et al. 2005). Malik and colleagues (2004) showed that cardiovascular disease and total mortality was increased in patients with the metabolic syndrome, and that risk was increased even when only one or two of the diagnostic criteria were met (Malik et al. 2004).

Rutter and co-workers suggested that the increased cardiovascular risk observed in metabolic syndrome patients was independently associated with low insulin sensitivity, after correction for other diagnostic criteria (Rutter et al. 2005). This may suggest that the key metabolic abnormality which drives increased risk in these patients is insulin resistance (Saely et al. 2005). Liese and colleagues previously highlighted the role of insulin resistance (categorised as hyperinsulinaemia) as a major determinant in the subsequent development of metabolic syndrome (Liese et al. 1997). Confirmatory factor analysis by Shen and colleagues (2003) suggested that insulin resistance and obesity were the primary contributors to the development of metabolic syndrome, and by extension the underlying risk of subsequent

diabetes and vascular disease (Shen et al. 2003). More recent confirmatory factor analysis suggests that one key metabolic driver could be the foundation for the multiple abnormalities (Pladevall et al. 2006).

Further evidence that insulin resistance is an important cardiovascular risk factor is evident. The DECODE Study Group highlighted the association between IGT and mortality (The DECODE Study Group 1999;The DECODE Study Group 2001). Additionally, Blake and colleagues (2004) identified increased coronary heart disease (CHD) in individuals with IGT (Blake et al. 2004), whilst Brunner and co-workers (2006) observed a linear increase in CHD at post-challenge glucose concentrations of greater than 4.6 mmol.1<sup>-1</sup> (Brunner et al. 2006). Hu and colleagues (2002) also described greater risk of cardiovascular events in 'prediabetic' women who subsequently developed T2D (Hu et al. 2002). Even within patients with established T2D, insulin resistance is independently associated with cardiovascular risk (Bonora et al. 2002).

# 1.6 Common metabolic cofounders of cardiovascular risk in insulin resistant states

Although insulin resistance is a key cofounder of metabolic syndrome, T2D and the attendant cardiovascular risk, other abnormalities such as dyslipidaemia (Taskinen 2005), obesity (Grundy 2002;Han et al. 2002) and chronic low-grade inflammation (Pradhan et al. 2001;Sattar et al. 2003) are frequently present in affected individuals. These factors may interact with, or be dependent upon, insulin resistance producing a complex metabolic state which predisposes to both atherosclerosis and T2D (Figure 1.2).



**Figure 1.2** Obesity, insulin resistance and chronic inflammation have complex interactions in the development of atherosclerosis and T2D.

Appreciation of the importance of cardiovascular risk in patients with, or at risk of, T2D is recognised to be of paramount importance. A shift in diabetes care has occurred in the last two decades, from a glucocentric model with concentration on detection of early microvascular disease to a model which includes these traditional goals but also targets aggressive treatment of macrovascular risk. Clinical targets for the optimal management of T2D include early detection of cardiovascular risk markers and aggressive treatment (American Diabetes Association 2008). Further exploration of the role of diabetes in the development of cardiovascular disease is warranted, with specific focus on risks in women

and non-Caucasians. Pre-menopausal women have a low risk of cardiovascular events and are therefore relatively understudied with respect to metabolic risk factors which might contribute to heightened cardiovascular risk. In view of the increased relative risk of cardiovascular disease in women who develop diabetes, this thesis aims to examine pre-menopausal daughters of patients with T2D in an attempt to further characterise the metabolic processes which might contribute to later diabetes and cardiovascular disease.

#### 1.7 Type 2 diabetes: Insulin resistance or beta-cell dysfunction?

Both insulin resistance and reduced insulin secretion are recognised in T2D (Kahn 2003). The pancreatic beta-cell has been traditionally considered to display adaptive secretory responses to insulin resistance, by augmenting insulin secretion to maintain normoglycaemia (Cerasi 1995;Taylor, Accili, & Imai 1994). Beta-cell failure has therefore been thought to be a relatively late occurrence in T2D, signifying a failure of both this adaptive response and the ability to maintain normal blood glucose. It is now accepted that beta-cell dysfunction occurs earlier in the natural history of the development of T2D, although this toxicity is still a consequence of the complex metabolic state found in insulin resistance with direct effects mediated by circulating non-esterified fatty acids (NEFA) (Kahn 2003). It is likely therefore, that insulin resistance remains the key metabolic abnormality in the development of T2D and that this, in concert with the metabolic effects of obesity, drives early beta-cell toxicity (Groop 1999).

# 1.8 The aetiology of insulin resistance

Given the importance of insulin resistance in the development of type 2 diabetes and its independent associations with cardiovascular disease, early intervention to prevent or reverse its progression may have significant public health benefits by the reduction of these chronic diseases (Lopez & Murray 1998). The state of insulin resistance is characterised by a reduced

ability of insulin to suppress hepatic glucose production and to promote peripheral glucose disposal. Type 2 diabetes occurs when this condition is accompanied by a failure of the pancreatic beta-cell to secrete enough insulin to overcome the degree of insulin resistance (McGarry 2002). The insulin resistant state is associated with metabolic dysfunction in adipose tissue, hepatic, skeletal muscle tissue and the vascular endothelium. These changes are both a consequence of, and contribute to, insulin resistance and it is difficult to separate metabolic cause, from effect.

### **1.8.1** Insulin resistance and adipose tissue

Adipose tissue is a metabolically active site of energy balance and hormone synthesis (Havel 2004). The adipocytokines produced by adipose tissue have paracrine and endocrine function, therefore adipose tissue has the ability to influence other tissues such as liver and skeletal muscle through both the dynamic flux of fatty acids and tissue-derived hormones. An enlarged adipose tissue compartment is associated with increased lipolysis and increased circulating non-esterified fatty acids (NEFA) (Boden 1997; Delarue & Magnan 2007) and is indicative of adipose tissue insulin resistance (Delarue & Magnan 2007). Increased NEFA induces chronic skeletal muscle insulin resistance through both reduced carbohydrate oxidation and reduced insulin stimulated glucose uptake (Boden 1997;Boden & Shulman 2002). In an insulin resistant state, adipose tissue fails to suppress lipolysis in response to circulating insulin and may lead to inappropriate release of NEFA in the context of a postprandial state, producing further 'lipid overflow' to the liver and skeletal muscle (Arner 2005). In addition, NEFA induce hepatic insulin resistance characterised by unchecked hepatic glucose production (Roden et al. 2000), although there may be a greater contribution to hepatic glucose production by NEFA-induced elevation of circulating insulin and glucagon (Chen, Iqbal, & Boden 1999). However, since NEFA concentrations are not always elevated in insulin resistant states, the lipotoxicity may be related to relatively reduced fat oxidative capacity in target organs (Corpeleijn et al. 2009). Alternatively, post-prandial NEFA

concentrations may have relatively greater impact upon insulin sensitivity and therefore elevated circulating NEFA in response to dietary fats could be implicated in the 'lipid overflow' (Jackson et al. 2005).

Further contribution to 'lipid overflow' may occur in the post-prandial state in insulin resistant subjects. Lipoprotein lipase (LPL) is responsible for hydrolysis of circulating triglyceride, releasing fatty acids which can be taken up by adipose tissue or skeletal muscle. In health, LPL expression on the vascular endothelium of adipose tissue is enhanced by insulin, promoting uptake of fatty acids in the fed state. Skeletal muscle LPL is not influenced by insulin and uptake of fatty acids is dependent on circulating plasma concentrations (Frayn, Arner, & Yki-Jarvinen 2006). Type 2 diabetes, insulin resistance and carbohydrate-rich diets produce greater post-prandial elevations in triglycerides than in insulin sensitive and carbohydrate restricted subjects (Karpe 1997;Kim et al. 2001a). Higher post-prandial NEFA concentrations are attributable to unsuppressed lipolysis in adipose tissue and the insulin-resistant liver also fails to suppress VLDL-triglyceride secretion (Frayn, Arner, & Yki-Jarvinen 2006). Therefore insulin-resistant adipose tissue and liver combine to provide skeletal muscle with both NEFA and fatty acid-rich lipoproteins which increase intramyocellular triglyceride through the unrestricted actions of skeletal muscle LPL.

Adipo(cyto)kines have a complex relationship with insulin sensitivity and appear to influence a number of metabolic systems which can contribute to the modulation of insulin sensitivity (Havel 2004). Obesity has been described as a chronic low-grade inflammatory condition (Hotamisligil et al. 1995). Adipokines mediate stimulation of both inflammation and insulin resistance in a paracrine fashion in adipose tissue and by an endocrine effect on skeletal muscle, liver and endothelium (Gustafson 2010). Chronic overfeeding results in expansion of adipose tissue pools due to enlargement of adipocytes (Spalding et al. 2008). Enlarged adipocytes alter their adipokine secretory profile, increasing expression of interleukin-6 (IL- 6) which stimulates lipolysis and macrophage infiltration of adipose tissue (Heilbronn & Campbell 2008). Spalding and colleagues have shown increased adipocyte turnover in obese patients (Spalding et al. 2008) and it has been suggested that enlarged adipocytes which become hypoxic and subsequently necrotic also stimulate chronic inflammation within adipose tissue (Heilbronn & Campbell 2008). The combination of enlarged adipocytes and increased adipose tissue macrophages results in a positive feedback cycle of chronic inflammation characterised by increased IL-6, Tumour Necrosis Factor- $\alpha$  (TNF $\alpha$ ), plasminogen activator inhibitor-1 (PAI-1) and other cytokines which have local effects on adipose tissue and systemic effects (Vettor et al. 2005;Berg & Scherer 2005). Adipocytes release cell adhesion molecules such as inter-cellular adhesion molecule 1 (ICAM-1) and vascular cell adhesion molecule 1 (VCAM-1) which recruit circulating immune cells into the vascular wall (Andersson et al. 2008). Adipose tissue bound macrophages and adipocytes are responsible for TNFa production which mediates impaired insulin signalling through activation of the nuclear factor kappa-B (NF-KB) transcription factor (Wagenmakers et al. 2006). Activation of NF- $\kappa$ B by the same stimulus has further effects on the endothelium leading to increased expression of cell adhesion molecules and by a separate pathway, TNFamediated superoxide formation leads to reduced nitric oxide bioavailability and further endothelial damage (Wagenmakers et al. 2006).

Leptin is correlated with adipose tissue mass, has local pro-inflammatory effects and increased circulating leptin is associated with insulin resistance (Nyholm et al. 1997;Wannamethee et al. 2007). Additionally, obesity is associated with low circulating adiponectin and is correlated with insulin resistance (Hotta et al. 2000;Weyer et al. 2001;Yamauchi et al. 2001). Both leptin and adiponectin influence insulin sensitivity directly but also contribute to endothelial function and therefore influence insulin sensitivity indirectly through modulation of the chronic inflammatory state which co-exists with insulin resistance (Berg & Scherer 2005).

### **1.8.2** Insulin resistance in skeletal muscle

Skeletal muscle is the most important site of insulin-stimulated glucose disposal (DeFronzo et al. 1981) and is the main site of peripheral insulin resistance (DeFronzo et al. 1985). Skeletal muscle insulin resistance is associated with increased intramyocellular lipid accumulation (Krssak et al. 1999) and impaired insulin stimulated glucose uptake and oxidation (Kelley & Mandarino 2000;Turcotte & Fisher 2008). It is suggested that chronically increased circulating NEFA concentrations lead to increased intramuscular triglyceride and lipid intermediate deposition which in turn interfere with both the intracellular insulin signalling cascade (Wolf 2008) and subsequent GLUT4 dependent insulin-stimulated glucose uptake (Delarue & Magnan 2007). The so-called 'athlete's paradox' would appear to cast doubt over the role of intramyocellular lipid in insulin resistance, since trained subjects who are highly insulin sensitive have high levels of intramyocellular lipid (IMCL) (Goodpaster et al. 2001). However, it is suggested that it is the dynamic turnover of IMCL rather than the total volume which is the important factor in determining the impact of IMCL on insulin sensitivity (Corcoran, Lamon-Fava, & Fielding 2007).

The mechanism by which 'static' IMCL contributes to insulin resistance is incompletely understood but appears to be associated with reduced insulin signalling, GLUT4 expression, glucose uptake, lipid oxidation and glycogen synthesis (Corcoran, Lamon-Fava, & Fielding 2007;Zierath, Krook, & Wallberg-Henriksson 2000). Both increased circulating plasma NEFA and IMCL concentrations produced reduced insulin-stimulated glucose uptake and glycogen synthesis (Boden 1997;Roden et al. 1999;Savage et al. 2007). Specifically, lipid intermediate metabolites (ceramides, long-chain fatty acyl-CoA and diacylglycerol (DAG)) are implicated in the molecular processes which cause skeletal muscle insulin resistance. Diacylglycerol is understood to activate serine/threonine kinases (protein kinase C (PKC) family) which phosphorylate and subsequently reduce the activity of upstream components of the insulin signalling pathway such as the insulin receptor and insulin receptor substrate-1 (IRS-1) (Bonen, Dohm, & van Loon 2006; Petersen & Shulman 2006). A reduction in the activity of the insulin signalling cascade leads to lower insulin-stimulated glucose uptake, protein kinase C also inactivates glycogen synthase kinase 3 further reducing skeletal muscle glycogen storage (Delarue & Magnan 2007). Ceramides inhibit the activation of Akt/Protein Kinase B (Akt/PKB) at a point 'downstream' of phosphatidylinositol 3-kinase (PI3K) resulting in downstream disruption of the insulin signalling cascade (Stratford et al. 2004). Stratford and colleagues observed ceramide-mediated disruption of Akt/PKB activation by two mechanisms (firstly Akt/PKB membrane translocation, secondly Akt/PKB dephosphorylation) in 3T3-L1 preadipocytes and the same mechanism is likely in skeletal muscle (Straczkowski & Kowalska 2008). Inhibition of ceramide synthesis and improved insulin sensitivity in animals is associated with increased Akt/PKB activity and in humans, exercise-mediated reductions in skeletal muscle ceramide have been associated with improved insulin sensitivity (Straczkowski & Kowalska 2008). It has been hypothesised that IMCL-related increases in PKC activity may lead to activation of the transcription protein nuclear factor ĸ-B (NFĸB) which suppresses PI3K function, this has been observed in skeletal muscle in rodents but not, to date, in human skeletal muscle (Itani et al. 2002;Kim et al. 2001b).

Turcotte and Fisher (2008) suggest that individuals with insulin resistance may exhibit a constitutive increase in fatty acid binding proteins (such as CD36) leading to increased myocyte fatty acid incursion (Turcotte & Fisher 2008). In addition, fat oxidation is impaired in insulin resistant skeletal muscle, contrary to the state predicted by Randle's hypothesis. Insulin and glucose increase skeletal muscle content of malonyl CoA, which is synthesised in muscle from acetyl CoA by acetyl CoA carboxylase-2 (ACC2). The accumulation of malonyl CoA may reduce fatty acid oxidation by inhibition of carnitine palmitoyl transferase-1 (CPT1) and subsequent accumulation of long-chain fatty acids (such as palmitate) have been shown to induce skeletal muscle insulin resistance by inhibition of Akt/PKB (Bonen, Dohm, & van Loon 2006;Kelley & Mandarino 2000;Wolf 2008). In contrast with hepatic tissue,

skeletal muscle has low levels of fatty acid synthase (FAS) therefore despite increased concentrations of malonyl CoA, rates of fatty acid synthesis de novo are low in skeletal muscle (McGarry 2002). Accumulation of long-chain fatty acids or intermediates are therefore derived from circulating lipids or stored IMCL (Koves et al. 2008). Muscle biopsies from insulin resistant subjects treated with the peroxisome proliferator-activated receptor  $\gamma$ (PPAR $\gamma$ ) agonist rosiglitazone show restoration of malonyl CoA and ACC2 activity and increased fat oxidation (Bandyopadhyay et al. 2006). Insulin resistant subjects are recognised to have reduced density of skeletal muscle mitochondria and this has been considered to be important in the impairment of both fatty acid oxidation and ATP synthesis (Petersen, Dufour, & Shulman 2005). However, the idea of a primary mitochondrial deficiency being responsible for skeletal muscle insulin resistance appears to be inconsistent with the functional reserve of the mitochondria, the normal basal creatine phosphate/ATP ratio observed in insulin resistant subjects and apparently preserved oxidative function in severely insulin resistant diabetic subjects (Holloszy 2009; Nair et al. 2008). It has been suggested that reduced insulin-mediated ATP synthesis is in fact a consequence of local and central insulin resistance, without an underlying mitochondrial defect (Wagenmakers 2005).

#### **1.8.3** Insulin resistance and endothelial dysfunction

Insulin is a vascular hormone and in healthy individuals, it promotes vasodilatation (Baron 1994;Cleland et al. 1999). Therefore, insulin-mediated glucose uptake in skeletal muscle is partly dependent on the vascular actions of insulin, which maintains blood flow through the vascular beds as well as the direct metabolic effect of insulin on GLUT4 translocation and glucose uptake (Laakso et al. 1990;Serne et al. 2006). There is debate about the validity of the relationship between vascular and metabolic actions of insulin since the vasodilating effect of insulin appears to be balanced by a vasoconstricting effect in healthy rodent models (Serne et al. 2006). Additionally, the vascular effect of insulin is observed to differ depending on a variety of factors such as; the metabolic health (and the species!) of the subject, the

section of the skeletal muscle vasculature studied and the dose and timing of the insulin stimulus (Clerk et al. 2004). Skeletal muscle glucose uptake preceeds the insulin-mediated increase in total muscle blood flow (Clark et al. 2003). This statement might appear to be incongruous with the hypothesis that the vascular and metabolic effects of insulin on skeletal muscle are coupled however, insulin-mediated capillary recruitment and glucose uptake are closely linked and reveal a direct relationship between insulin-mediated vasodilatation and glucose uptake which is not found when other vasodilators are used (Clark et al. 2003). Further evidence to support this relationship is provided by rodent models where reduced capillary recruitment is observed when insulin resistance is induced (Clark et al. 2003;Vincent et al. 2005).

It is suggested that insulin has a critical flow-mediated action, not on large calibre conduction and resistance vessels, but on terminal arterioles and the capillary beds (Sydow, Mondon, & Cooke 2005). Insulin is thought to have this effect through a NO-dependent action – further linking the metabolic interaction between insulin and the endothelium (Eringa et al. 2002). Insulin is recognised to activate endothelial Nitric Oxide Synthase (eNOS) by a signalling cascade which utilises IRS-1, PI3K and PKB – similar to the insulin signalling pathway involved in GLUT4 expression (Eringa et al. 2002;Sydow, Mondon, & Cooke 2005;Vincent et al. 2003). In animal models, genetic knockout of NO or phamacological blockade of NOS activity both result in reduced skeletal muscle capillary recruitment and insulin resistance (Sydow et al. 2005;Vincent et al. 2003). In addition, human studies have shown that infusion of the eNOS inhibitor asymmetric dimethylarginine (ADMA), causes endothelial dysfunction and insulin resistance (Sydow et al. 2005) and circulating ADMA concentrations are higher in patients with T2D (Abbasi et al. 2001). Decreased NO bioavailability is considered to be the crucial factor in the aetiology of endothelial dysfunction (Imrie, Abbas, & Kearney 2010). Interstitial insulin concentrations are more closely associated with skeletal muscle glucose uptake than plasma insulin concentrations, suggesting that insulin delivery to the myocyte (which requires capillary recruitment and trans-endothelial transport across the vascular endothelium) is a key determinant of the metabolic action of insulin (Castillo et al. 1994). Impaired insulin action in skeletal muscle capillary endothelium prevents capillary recruitment and trans-endothelial insulin transport and is mediated by reduced NO bioavailability which is a result of Akt/PKB interference leading to diminished eNOS activity (Barrett et al. 2009). Failure of skeletal muscle glucose uptake results in increased carbohydrate delivery to the liver which promotes lipogenesis and subsequent NEFA delivery to skeletal muscle, further increasing IMCL and myocyte insulin resistance by the mechanisms described in the previous section. The initial 'uncoupling' of insulin-stimulated NO production has been demonstrated by TNF $\alpha$ -mediated impairment of insulin-induced activation of Akt and eNOS and implicates the low-grade inflammatory process as a potentially founding factor in endothelial insulin resistance (Serne et al. 2006;Barrett et al. 2009;DeFronzo 2010).

#### **1.8.4** Insulin resistance in liver

Carbohydrate and fat metabolism are highly influenced by circulating insulin, glucose and NEFA concentrations. In a state of chronic insulin resistance, accompanied by beta-cell dysfunction, NEFA stimulate hepatic glucose production through gluconeogenesis (Roden et al. 2000;Delarue & Magnan 2007). Under normal circumstances increased circulating NEFA does not increase hepatic glucose output due to down-regulation of hepatic glycogenolysis (hepatic autoregulation) however, in an insulin resistant state hepatic glucose production is increased by persisting glycogenolysis which may be mediated by increased circulating glucagon (Chen, Iqbal, & Boden 1999;Samuel et al. 2009) or may be mediated by chronically elevated concentrations of NEFA (Shah et al. 2002).

The failure of insulin to suppress hepatic glucose production is closely related to the degree of hepatic steatosis (Kotronen et al. 2007;Seppala-Lindroos et al. 2002). In addition, reduced hepatic insulin clearance is associated with hepatic steatosis (Kotronen et al. 2007). The source of fatty acids which lead to hepatic steatosis is debated, but is likely to be derived from chronic dietary excess and fatty acid release from visceral and/or subcutaneous adipose tissue. Hyperinsulinaemia, which is a compensatory response to peripheral insulin resistance, stimulates hepatic fatty acid synthesis from glucose and further exacerbates hepatic steatosis (Taylor 2008). In addition fatty acid oxidation is inhibited in the hepatocyte by the accumulation of malonyl CoA, which interferes with fatty acid transport into the mitochondria (Taylor 2008). Increased fatty acid oxidation in obese mice produces increased insulin sensitivity and decreased steatosis (Reid et al. 2008), and pioglitazone mediated stimulation of PPAR $\gamma$  in patients with T2D reduces both hepatic glucose production and hepatic fat content (Ravikumar et al. 2008).

However, hepatic steatosis *per se* is not necessarily a mediator of hepatic insulin resistance. In mouse models, hepatic steatosis can be induced by pharmacological blockade of  $\beta$ oxidation, without affecting insulin sensitivity (Grefhorst et al. 2005). Patients with familial
hypobetalipoproteinaemia are unable to export VLDL from the liver and are susceptible to
hepatic steatosis but do not display insulin resistance (Schonfeld et al. 2008). It is therefore
possible that hepatic fat, like IMCL, is closely associated with insulin resistance but not a key
mediator of its development. Thus, a combination of excess fatty acid delivery to the
hepatocyte, increased intrahepatic fatty acid synthesis and hyperinsulinaemia produce a state
hepatic insulin resistance and hepatic steatosis which, in turn promote hepatic glucose
production and reduced insulin clearance.

Chapter 3 of this thesis aims to examine the differences in insulin sensitivity between sedentary pre-menopausal women with a family history of diabetes and matched controls. In

addition the relationship between adiposity, adipose tissue derived factors, lifestyle and substrate utilisation is explored in an attempt to understand the aetiology of insulin resistance.

#### **1.9 Insulin resistance in obesity and lipodystrophy ('dysadiposity')**

Obesity is strongly correlated with the development of insulin resistance and subsequent progression to T2D and premature cardiovascular disease (Barrett-Connor 1989;Ohlson et al. 1985;Ohlson et al. 1988;Wang et al. 2005). Section 1.9 illustrates the importance of fatty acid metabolism in the aetiology of tissue specific and total body insulin resistance. In addition, high risk groups such as South Asians appear to manifest greater sensitivity to the effects of excess adipose tissue, displaying dyslipidaemia and dysglycaemia at lower levels of adiposity than Caucasians (Razak et al. 2007; Whincup et al. 2005). Chandalia and colleagues report that South Asians have an expanded deep subcutaneous adipose pool and larger adipocytes than matched Caucasians (Chandalia et al. 2007). This may suggest that the increase in circulating inflammatory proteins (CRP, IL-6, PAI-1) found in South Asians is a consequence of a greater proportion of metabolically active adipose tissue, for a given BMI (Hall, Sattar, & Gill 2008). Increased body mass, even within 'normal' range body mass index (BMI) is associated with increase risk of T2D in a Japanese population (Nagaya et al. 2005). All adipose tissue is not however, metabolically equal. Offspring of patients with T2D tend to display increased visceral adiposity compared with subjects with no family history of T2D who were matched for total adiposity, in addition offspring subjects displayed greater insulin resistance which was closely associated with the degree of visceral adiposity (Nyholm et al. 1997; Nyholm et al. 2004). As in South Asians, Australian Aboriginals and Chinese populations, offspring subjects may display enhanced sensitivity to adiposity exhibiting lower insulin sensitivity for the same fat mass as found in matched controls (Kriketos et al. 2004).

Visceral adipose tissue is thought to have greater influence on insulin sensitivity than subcutaneous fat (Macor et al. 1997). Use of direct imaging techniques such as computed tomography (CT) and magnetic resonance imaging (MRI) have allowed distinction between visceral and subcutaneous adipose pools and are more sensitive methods of assessing these compartments than traditional anthropometry (Jia et al. 2003;Pi-Sunyer 2000;van der & Seidell 1993). An enlarged visceral adipose compartment may increase NEFA delivery to the liver via the splanchnic circulation (Basu et al. 2001), increase total circulating NEFA (Arner 2002) which may impact on skeletal muscle insulin resistance (Boden 1997;Corcoran, Lamon-Fava, & Fielding 2007) and cause pancreatic beta-cell toxicity (Taylor 2008;Tushuizen et al. 2007) and may be a source of pro-inflammatory adipokines (such as TNF $\alpha$ ) which promote insulin resistance by direct action on insulin signalling and indirectly by inducing endothelial dysfunction (Montecucco, Steffens, & Mach 2008;Salmenniemi et al. 2005;Sydow, Mondon, & Cooke 2005).

Further support for visceral fat as an important metabolic contributor to insulin resistance is observed in patients with lipodystrophy. Lipodystrophy is a condition of disordered adipose tissue distribution and may be localised, partial or generalised. Lipodystrophy may be congenital or acquired and both partial and generalised conditions are associated with insulin resistance and dyslipidaemia (Hegele 2003;Monajemi et al. 2007). Congenital lipodystrophies are a heterogenous group of syndromes which are associated with a variety of defects in insulin signalling, lipid synthesis, connective tissue substrate production and PPARγ function (Monajemi et al. 2007). Acquired lipodystrophies are most commonly observed in Human Immunodeficiency Virus (HIV) sufferers who are treated with protease inhibitors (Carr et al. 2001;Garg 2004). The metabolic defects observed in lipodystrophy are frequently associated with reduced subcutaneous adipose tissue and increased visceral adipose tissue, although in some syndromes reduced peripheral subcutaneous adipose tissue is seen whilst truncal subcutaneous adiposity is preserved (Hegele 2003). Insulin resistance in these conditions is often severe (Beltrand et al. 2007), and is associated with reductions in

both leptin and adiponectin, as well as increased triglycerdides, low HDL-cholesterol and in some syndromes, elevated circulating NEFA (Hegele 2003).

Many of these defects are common to the general insulin resistance state seen in pre-diabetic conditions such as the metabolic syndrome, impaired glucose tolerance, and in high risk groups such as South Asians and offspring of patients with T2D. Some lipodystrophic conditions have shown metabolic response to treatment with leptin (Beltrand et al. 2007;Park et al. 2008) and PPAR $\gamma$  agonists (Maraldi et al. 2007). An improved metabolic phenotype in lipodystrophy is associated with increased subcutaneous adipose tissue mass (Hegele 2003;Monajemi et al. 2007), which emphasises the importance of adipose tissue site and function, rather than total adipose tissue mass, in the development of insulin resistance.

#### 1.10 The effect of insulin resistance on people with a family history of T2D

First-degree relatives of patients with T2D are recognised to be insulin resistant (Humphriss et al. 1997;Perseghin et al. 1997) and also display increased clustering of cardiovascular risk factors (Micossi et al. 1987;Sarlund et al. 1992). In addition, these subjects also display prothrombotic tendencies with elevated levels of D-Dimer and Plasminogen Activator Inhibitor-1 (PAI-1) (Fernandez-Castaner et al. 1996;Mansfield, Stickland, & Grant 1997). Relatives may also exhibit an atherogenic lipoprotein profile, with elevated apolipoprotein B (apoB), and reduced levels of apolipoprotein A-1, with increased apoB in those with greater insulin resistance (Stewart et al. 1998). Further objective evidence of cardiovascular risk in this group was observed with respect to vascular function, where relatives displayed increased carotid intima-media thickness (Pannacciulli et al. 2003) and impaired endothelial function characterised by reduced bio-availability of nitric oxide (Caballero et al. 1999;Cersosimo & DeFronzo 2006;Goldfine et al. 2006).

#### **1.11** Insulin resistance in women

Insulin resistance appears to have a greater impact on cardiovascular risk in women, both prior to and after the development of T2D (Becker et al. 2003; Hu et al. 2001; Hu et al. 2002). Tailored public health strategies have recently been suggested to tackle cardiovascular risk in women (Engberding & Wenger 2008; Evangelista & McLaughlin 2009). The presence of insulin resistance in women may be associated with other metabolic disturbances which are not present to the same extent in men, and which may confer increased risk (Regitz-Zagrosek, Lehmkuhl, & Weickert 2006). In insulin resistant subjects higher inflammatory markers have been reported in women compared to men, and remain significantly higher in women even after adjustment for BMI (Lakoski et al. 2006; Saltevo et al. 2008). This observation may simply reflect a greater amount of adipose tissue in women compared to men of a given BMI, although Saltevo and colleagues have also suggested that with increasing insulin resistance women display relatively greater reductions in adiponectin compared to men and that this may be indicative of increased adipose tissue macrophage infiltration (Saltevo, Kautiainen, & Vanhala 2009). The influence of circulating oestrogen in pre-menopausal women is recognised to have metabolic benefit; promoting fat oxidation, reduced inflammatory cytokines, reduced oxidative stress, reduced hepatic glucose output and subcutaneous, rather than visceral fat accumulation (Geer & Shen 2009). The development of insulin resistance in women is associated with attenuation of these protective mechanisms (Geer & Shen 2009). Polycystic ovarian syndrome (PCOS) is an insulin resistant state characterised by hyperandrogenism, visceral obesity, endothelial dysfunction and low grade inflammation (Giallauria et al. 2008a). It is therefore conceivable that the development of insulin resistance in women is mediated by different mechanisms to those observed in men and that low-grade inflammation may have greater influence on metabolic health in insulin resistant women compared to men. Therefore, chapter 3 of this thesis aims to specifically investigate mediators of insulin sensitivity in pre-menopausal women with, and without a family history of T2D.

#### **1.12** Insulin resistance, cardiovascular risk and low grade inflammation

Adiposity is associated with higher circulating levels of pro-inflammatory cytokines such as c-reactive protein (CRP), interleukin-6 (IL-6), tumour necrosis factor- $\alpha$  (TNF $\alpha$ ) and interleukin-1 (IL-1) (Montecucco et al. 2008). In addition, insulin resistance and obesity are associated with low levels of adiponectin which has anti-inflammatory activity (Folco et al. 2009). Increased circulating inflammatory cytokines may contribute to both the aetiology of insulin resistance, and the end-organ damage observed in the insulin resistant state (Festa et al. 2000;Kowalska et al. 2008). Therefore, low grade inflammation may be both a co-founder in the development of insulin resistance and co-conspirator in the metabolic dysfunction observed in this state. Elevated inflammatory factors are observed in obesity, however increased adipose tissue mass is also associated with increased macrophage recruitment to adipose tissue. Hormones such as leptin and adiponectin are produced by the adipocyte, IL-6 and TNF $\alpha$  are produced by both adipocytes and macrophages (Berg & Scherer 2005) and other inflammatory cytokines are likely to be derived from intra-adipose macrophages (Fain et al. 2004).

There is strong mechanistic evidence that TNF $\alpha$  is a major inflammatory protein and that it is directly involved in the pathogenesis of insulin resistance and atherosclerosis. Elevated plasma expression of TNF $\alpha$  is observed in T2D and atherosclerosis and is likely to be derived from adipose tissue (Berg & Scherer 2005). Interference with Akt/PKB activity has previously been outlined (Section 1.8.3) as a method by which TNF $\alpha$  induces endothelial dysfunction. Additional activation of the intracellular enzyme c-Jun N-terminal kinase (JNK) by TNF $\alpha$  contributes to endothelin-mediated vasoconstriction in the presence of insulin (Eringa et al. 2006). TNF $\alpha$  also activates NADPH oxidase (via PKC activation) resulting in superoxide generation which depletes NO, and through activated NF $\kappa$ B produces cellular adhesion molecules (CAM) which stimulate leucocyte endothelial binding and further promote local inflammation and damage the endothelial barrier (Wagenmakers et al. 2006). These factors result in NO deficiency and endothelin dominance producing both skeletal muscle insulin resistance (through reduced capillary recruitment) and increased peripheral vascular resistance (Serne et al. 2006;Pedersen 2006). This mechanism is complemented by the induction of lipolysis in adipose tissue (Moller 2000), exposing the vascular wall, skeletal muscle and other tissues to circulating NEFA (Pedersen 2006). Additionally, TNF $\alpha$  promotes the 'atherogenic lipid profile' characterised by reduced HDL and increased oxidised LDL, which increase plaque expansion (Popa et al. 2007). Blockade of TNF $\alpha$  using immunomodulators has had mixed results in human subjects, whilst reduction in inflammatory factors was achieved using specific agents, there have been less impressive improvements in insulin sensitivity (Dominguez et al. 2005).

The role of IL-6 in the pathogenesis of insulin resistance is controversial and recent appreciation of IL-6 as a 'myokine' reflects the complexity of the interaction between cytokines and their local and systemic effects (Pedersen 2006). Vettor and colleagues observed that IL-6 interferes with IRS-1 autophosphorylation and that plasma IL-6 concentrations correlate with the degree of insulin resistance (Vettor et al. 2005). However, plasma IL-6 concentrations also correlate with body mass and the relationship between IL-6 and insulin resistance may represent the more robust association between adipose tissue mass and insulin resistance (Jensen 2008). The effect of IL-6 is likely to depend on the target tissue studied. Visceral fat appears to 'overproduce' IL-6 (Jensen 2008), whereas in healthy individuals IL-6 stimulates both lipolysis and fat oxidation without net hypertriglyceridaemia (van et al. 2003). Contraction of skeletal muscle stimulates IL-6 production by the myocyte, which stimulates fatty acid oxidation by activation of adenosine monophosphate kinase (AMPK) (Pedersen 2006).

TNFα, IL-1 and CRP may promote further inflammatory and pro-coagulatory protein production (i.e. ICAM/VCAMs) through stimulation of NFκB (Sarada et al. 2008), and these

effects are enhanced in the presence of increased glucose (Iwasaki et al. 2007;Wilson, Ryan, & Boyle 2006). CRP correlates with insulin sensitivity in non-diabetic women, and may predict the development of T2D (Pradhan et al. 2001;Pradhan et al. 2003), however it is unclear whether CRP has a direct physiological role in insulin resistance, or whether it is merely a marker of inflammatory activity and the underlying pathogenic processes (Sattar & Lowe 2006;Wilson, Ryan, & Boyle 2006).

Normoglycaemic first-degree relatives of patients with T2D display evidence of a proinflammatory state (Tesauro et al. 2007), and evidence of insulin resistance and impaired vascular function (McEleavy et al. 2004;Scuteri et al. 2008). However, McEleavy and colleagues and Scuteri and colleagues used carotid-radial pulse wave velocity and flow mediated brachial artery dilatation respectively. These techniques are considered to be surrogates for endothelial function but prinicipally measure vascular function in conduction vessels. Therefore these techniques demonstrate impaired large vessel reactivity, but are not designed to examine microvascular function, particularly skeletal muscle capillary recruitment. To date, the author is not aware of any studies on skeletal muscle capillary recruitment in normoglycaemic offspring of patients with T2D.

Kriketos and colleagues (2004) did not demonstrate increased inflammation in normoglycaemic first-degree relatives, however this study compared groups who were well matched for adiposity, whereas Scuteri and co-workers (2008) studied relatives who had greater adiposity than the comparator control group (Scuteri et al. 2008;Kriketos et al. 2008). Non-obese women with polycystic ovarian syndrome did not show increased evidence of inflammation compared to controls, despite displaying significantly lower insulin sensitivity (Martinez-Garcia et al. 2009). It seems likely that inflammation contributes to the pathogenesis of both insulin resistance and atherosclerosis (Vettor et al 2005;Jeppesen et al. 2008), but that measurable increases in inflammation may be only be apparent with increased adiposity or detectable change in insulin sensitivity (Ruotsalainen et al. 2006;Ruotsalainen et al. 2008). It is also conceivable that systemic measurements of inflammation are too crude to detect the interaction between low grade inflammation and skeletal muscle capillary endothelial dysfunction which are likely to be present early in the pathogenesis of insulin resistance (Bakker et al. 2009).

# 1.13 Adipose tissue hormones and insulin resistance

Adipose tissue hormones also contribute to insulin sensitivity, however their physiological effects differ between states of metabolic health and disease (Stefanyk & Dyck 2010). Plasma leptin concentrations correlate directly with adipose tissue mass (Fruhbeck, Jebb, & Prentice 1998) and obese individuals with high circulating leptin are insulin resistant (Considine et al. 1996; Weyer et al. 2001). Plasma adiponectin concentration is inversely related to adipose tissue mass (Arita et al. 1999) and circulating adiponectin concentrations are lower in patients with obesity and T2D (Hotta et al. 2000; Weyer et al. 2001). In pre-diabetic states adiponectin is reduced (Lihn et al. 2003;Kowalska et al. 2008), whilst high circulating adiponectin is associated with reduced inflammatory markers and reduced risk of T2D (Duncan et al. 2004). In first-degree relatives of patients with T2D circulating leptin is frequently increased, however these increases are more strongly related to adiposity and insulin resistance than family history and are indicative of the underlying mechanisms which increase risk of T2D in this group (Nyholm et al. 1997; Vauhkonen et al. 1998). Initial speculation on the role of adipokines in insulin resistance was based on these observations and was predicated on the assumption that the development of obesity and insulin resistance was 'caused' by hyperleptinaemia and/or hypoadiponectinaemia. However, reversal of leptin deficiency in both rodents (Larcher et al. 2001) and humans (Beltrand et al. 2007) improves insulin sensitivity. Therefore recent evidence would suggest that the relationship between adiponectin and leptin in metabolic health is one of synergistic promotion of insulin

sensitivity and in states of insulin resistance and obesity, that this synergy is lost (Dyck 2009).

In lean insulin sensitive subjects both leptin and adiponectin stimulate skeletal muscle fatty acid oxidation (Kieffer & Habener 2000;Margetic et al. 2002). Leptin stimulates fatty acid partition towards oxidation and through phosphorylation of ACC2 prevents accumulation of malonyl CoA (Dyck 2009). These actions prevent accumulation of DAG and ceramides, maintaining insulin sensitivity and are mediated by direct activation of AMPK (Minokoshi et al. 2002;Steinberg & Jorgensen 2007)(Havel 2004). Leptin may also reduce expression of fatty acid transporters (CD36) which could contribute to reduced IMCL however this effect has, to date, only been observed in rodents (Dyck 2009). Leptin also has central actions, acting as an energy balance hormone, triggering hypothalamic satiety (Vettor et al. 2005) and stimulating skeletal muscle AMPK by  $\alpha$ -adrenergic activity (Minokoshi et al. 2002).

In healthy individuals adiponectin also stimulates fatty acid oxidation in skeletal muscle through AMPK activation by the adiponectin receptor 1 (AdipoR1) and may directly inactivate ACC2 (Dyck 2009;Steinberg & Jorgensen 2007). Adiponectin also stimulates fatty acid oxidation in the liver, reduces *de <u>novo* lipogenesis</u> and decreases both hepatic glucose output and circulating NEFA and VLDL (Tsochatzis, Papatheodoridis, & Archimandritis 2009). These effects are associated with improved total body insulin sensitivity (Furler et al. 2006).

Elevated concentrations of circulating leptin are found in obesity in humans and in states of high-fat feeding in rodents (Stefanyk & Dyck 2010). In these conditions, leptin is unable to stimulate skeletal muscle fat oxidation (Steinberg et al. 2002) but since administration of the AMP analogue 5-aminoimidazole-4-carboxamide-1-β-D-ribonucleoside (AICAR) fully

activates AMPK in this situation it is suggested that the defect in obesity is an impairment of leptin-mediated AMPK activation (Dyck 2009;Steinberg & Jorgensen 2007). Knockout mice, high-fat fed rodents and obese humans display increased levels of the suppressor of cytokine signalling 3 (SOCS3) protein (in skeletal muscle) which appears to interfere with stimulation of AMPK by inhibiting leptin signalling (Steinberg et al. 2006). In addition to the reduction in adiponectin concentration in obesity and insulin resistance, skeletal muscle also appears to be unable to promote fatty acid oxidation in response to adiponectin exposure (Mullen et al. 2009). Acute saturated fat feeding provokes reductions in skeletal muscle fatty acid oxidation and ACC2 phosphorylation despite adiponectin exposure and is followed by chronic (4 weeks) increased expression of fatty acid transporters and myocyte DAG and ceramide content (Mullen et al. 2009). This is unlikely to be related to changes in adiponectin receptor expression (Steinberg & Jorgensen. 2007; Mullen et al. 2009). However, longer term obesity and insulin resistance may reduce receptor expression. In cultured myotubes derived from obese, diabetic or formerly-obese subjects an inability to increase adiponectin receptor expression after exposure to leptin and adiponectin was observed (McAinch et al. 2006). Finally, adiponectin deficiency may contribute to the low grade inflammatory state in obesity since reduced adiponectin receptor expression in adipose tissue results in increased expression of the macrophage attracting chemokine monocyte chemoattractant protein-1 (MCP1) (Yamauchi & Kadowaki 2008).

In health, leptin and adiponectin appear to exert a complementary effect upon skeletal muscle fat oxidation, which promotes insulin sensitivity. However in obesity and insulin resistance they are unable to maintain these processes despite an intact downstream mechanism (AMPK). In fact the development of obesity would seem to induce skeletal muscle resistance to these hormones and chronic supraphysiological concentrations of leptin may downregulate adiponectin receptor expression resulting in a vicious circle of impaired skeletal muscle fat oxidation, insulin resistance and adipose tissue inflammation (Fang et al. 2009;Stefanyk & Dyck 2010;Yamauchi & Kadowaki 2008). Therefore one of the aims of this thesis is to

examine the interaction between adiposity, adipokines and inflammation with respect to their related and independent effects on insulin sensitivity.

#### **1.14** Genetic predisposition to insulin resistance

The previous sections of this chapter have outlined a number of metabolic abnormalities which contribute to the development of insulin resistance and subsequently increase the risk of T2D and premature cardiovascular disease. Evidence for a gene-environment interaction has been observed in specific ethnic groups (Pimas, Oji-Cree, Torres-Strait Islanders etc.) and this hypothesis has been generalised to more genetically diverse populations such as South Asians and offspring of people with T2D.

The search for the genetic origin of T2D has focused on potential 'diabetogenic' gene polymorphisms, clustering of specific polymorphisms in high-risk populations and mechanisms of inheritance. Paradigms for the single gene predisposition to diabetes exist in patients with Maturity Onset Diabetes of the Young (MODY). These patients display a strong family history of diabetes at an early age and have frequently been misdiagnosed as type 1 diabetes due to the age of onset. MODY patients have a monogenic diabetes which has a variable treatment and natural history depending upon the genetic mutation, in most cases however, insulin resistance is not a key component of the pathophysiology (Olek 2006). The prevalence of MODY is very low, affecting approximately 0.17 per 100,000 children in the UK (Ehtisham et al. 2004).

Other forms of inherited diabetes are observed in maternally transmitted mitochondrial DNA (mtDNA) mutations. These patients suffer from T2D but also from a host of other mitochondrial diseases, such as deafness, myopathy, lactic acidosis and encephalopathy and are therefore distinct from 'common' T2D (McIntyre & Walker 2002). Rare monogenic

forms of insulin resistance are seen in inherited forms of lipodystrophy (Dunnigan-type) (Hegele 2003) and Type A insulin resistance syndrome (Young et al. 2005). These conditions are clinically striking and have a markedly different phenotype when compared with T2D (Jiang et al. 2006).

In view of the insidious onset of T2D, accompanied by a gradual change in phenotype it is clear that significant monogenic abnormalities are not involved in the majority of cases of T2D. Instead T2D has a polygenic basis with more subtle alterations in gene function which appear to require specific environmental conditions to be prevalent to allow the manifestation of the disease. Single-nucleotide polymorphisms (SNPs) for the leptin receptor are associated with increased risk of T2D in people from Northern China (Qu et al. 2008). Abnormalities near a fatty acid binding protein on chromosome 4 are associated with insulin resistance in Pima Indians and Europeans with T2D (Humphreys et al. 1994). Variations in a gene which codes for a calcium sensing protein may have a minor effect on insulin action in Pima Indians (Ma et al. 2008). SNPs for the PPAR $\alpha$  gene are associated with increased risk of progression from IGT to T2D (Andrulionyte et al. 2007). Combinations of SNPs for the Calpain-10 protease (which may have a role in insulin signalling (Paul et al. 2003)) appears to increase risk of T2D, especially when combined with other local SNPs (Malecki 2005;McIntyre & Walker 2002). Multiple genetic abnormalities in insulin signalling, insulin secretion, fatty acid uptake and intracellular substrate utilisation have been determined (McIntyre & Walker 2002). These observations support the hypothesis that type 2 diabetes arises from a complex interaction between fatty acid metabolism, skeletal muscle substrate utilisation, hepatic autoregulation and both insulin resistance and reduced secretion. In addition the roles of adipose tissue hormones and cytokines would appear to modulate the hepatic, skeletal muscle and adipose tissue responses to fatty acids, insulin and glucose and in situations of chronic energy excess, contribute further to the progression towards T2D.

#### **1.15** Environmental impact on genetic predisposition

The comparison between rates of diabetes in genetically identical groups who are exposed to different environments is compelling evidence for the role of environment in the aetiology of diabetes. This is most striking in comparison between American and Mexican Pima Indians (Valencia et al. 1999), but is also apparent in other groups (e.g. Torres Strait Islanders, Oji Cree Indians) who have been exposed to a rapid change in their environment (Hegele et al. 2003;Longstreet et al. 2007). The high prevalence of diabetes in the developed world and expected increase in the prevalence of the disease in the Indian sub-continent and China (Wild et al. 2004) provides evidence that the influence of a sedentary lifestyle and an energy dense diet with high quantities of fat and refined carbohydrate is a powerful environmental influence at a population level. This is likely to be related to increased obesity, but there may also be an independent contribution from diet (Ventura et al. 2009) and physical activity (Balkau et al. 2008). Ethnic groups who have migrated from cultures where lifestyles are traditionally very active and consume diets which are high in fibre and low in fat, to a Westernised lifestyle often display a rapid increase in prevalence of T2D compared with their parent populations (Nakanishi et al. 2004). People with a parent with T2D who are sedentary are insulin resistant and display increased risk of diabetes, but in offspring who are active, the tendency towards metabolic dysfunction appears to be ameliorated (Gill & Malkova 2006).

Environmental influences are likely to begin at an early stage in human development and appear to be able to induce physiological changes which predispose to a variety of chronic diseases. Osmond and Barker (2000) proposed that poor fetal growth predisposed to later development of hypertension, coronary heart disease, insulin resistance and T2D (Osmond & Barker 2000). It is suggested that fetal undernutrition 'programmes' the fetus to protect cerebral development by ensuring adequate perfusion and nutrition, at the expense of the development of other viscera and the fetus as a whole (Osmond & Barker 2000). With respect to the later development of T2D, it is recognised that small babies display insulin resistance

in later life and in some ethnic groups (e.g. South Asians) this is manifest as dysglycaemia in childhood (Yajnik et al. 1995). Protection of the developing fetal brain results in a catabolic state *in utero*, with a metabolism which is glucose-sparing to allow adequate substrate delivery to the brain. This glucose-sparing is achieved by the development of peripheral insulin resistance, evidenced by low rates of glycolysis and glycolytic ATP production during exercise in people who were small babies (Taylor et al. 1995). A catabolic fetal state is proposed to result in reduced circulating insulin, and increased glucocorticoids, this would produce a metabolic shift towards protein and fat metabolism for energy production and may explain low subcutaneous fat and muscle bulk in low birth weight babies (Bjorntorp 1995;Robinson et al. 1991). Visceral 'sacrifice' may lead to lower beta-cell mass and subsequently a tendency towards insulin deficiency (Osmond & Barker 2000).

Small baby fetal programming results in early metabolic changes which adapt to an environment of 'thrift'. The plasticity of the developing fetus allows these influences to affect organogenesis and physiological function to a greater extent than at any other time in life and because this occurs at a critical developmental period these changes are subsequently 'locked in'. The risk of developing T2D is increased in small babies who display rapid 'catch-up' growth, particularly after the age of 2 (Barker 2005). It is suggested that the rapid increase in weight after the age of 2 is characterised by a disproportionate increase in fat mass relative to muscle mass, which would be expected to result in insulin resistance (Eriksson et al. 2002).

Fetal macrosomia is more common in women with gestational diabetes or T2D, and is also associated with subsequent risk of T2D in the offspring (McCance et al. 1994). Birthweight would appear to have a 'U-shaped' relationship to the development of T2D, and appears to be increased by both low and high birthweight, particularly when there is increased adiposity after the age of 2 years (Barker 2005). These findings would appear to be contradictory, however it would seem likely that there are multiple routes towards T2D, as might be expected given the complex nature of the metabolic dysfunction. Those with low birthweight and accelerated catch-up growth can be considered to be programmed towards an environment of thrift by their intrauterine environment and are subsequently unable to alter their insulin resistant 'glucose-sparing' metabolism in the presence of increased nutrient supply. Those with high birthweight are more likely to be born to diabetic mothers, and therefore display genetic predisposition towards diabetes (McCance et al. 1994), however it may be suggested that this could also be an effect of intrauterine programming where the fetus is exposed to high circulating insulin, carbohydrate and NEFA levels and generating an obese insulin resistant baby (Dabelea et al. 2000). Hattersley and Tooke have suggested that the observed 'small baby' phenotype could also be due to a genetically mediated insulin resistance, which prevents adequate growth in utero (Hattersley & Tooke 1999). In support of Hattersley and Tooke's 'fetal insulin hypothesis' Lindsay and colleagues (2000) observed that low fetal birth weight may be mediated by paternal diabetes, suggesting that low birth weight is genetically influenced since paternal physiology has no effect on intrauterine environment (Lindsay et al. 2000). Embryo transfer studies in genetically normoglycaemic and hyperglycaemic rats also suggests that a normal intrauterine environment cannot overcome the genetic tendency towards diabetes, but that exposure of genetically euglycaemic rats to a hyperglycaemic intrauterine environment promotes diabetes in later life (Gill-Randall et al. 2004). Additionally, recent analysis of T2D-promoting SNPs in mothers and offspring has shown that fetal inheritance of specific pro-diabetes SNPs are associated with low birthweight, which supports the fetal insulin hypothesis (Freathy et al. 2009).

In summary, T2D is increased in those with a genetic predisposition and in those who are subjected to intrauterine undernutrition (Barker et al. 2009;Lindsay et al. 2000). The risk may be augmented by increased weight gain after the age of two, in both small and large babies (Barker 2005). However early 'catch-up' growth may be a physiological compensation for poor intra-uterine growth and not pathogenic (Beltrand et al. 2009). Intrauterine programming may occur in small and large babies, predisposing to obesity, insulin resistance

and other chronic diseases and is likely to be mediated by both genetic and environmental influences (Freeman 2009;Lindsay 2008). Individuals with a positive family history of diabetes are more susceptible to the development of metabolic dysfunction related to a sedentary lifestyle and display relative insulin resistance and dyslipidaemia at normal body weights (Gill & Malkova 2006;Guerrero-Romero & Rodriguez-Moran 2006).

### 1.16 Physical activity and diabetes risk

Lifestyle interventions in people with impaired glucose tolerance (IGT) have been proven to reduce or delay progression to T2D (Knowler et al. 2002;Pan et al. 1997;Tuomilehto et al. 2001). In each of these studies lifestyle intervention consisted of intensive dietary modification as well as increased physical activity and goals included weight loss targets, although the Da Qing study did show reduced incidence of diabetes in an 'exercise-only' group who did not reduce weight (Pan et al. 1997). In addition, lifestyle intervention successfully prevents T2D in other groups at increased risk of the condition: specifically, obese men (Davey et al. 2005) and South Asians (Ramachandran et al. 2006).

Early epidemiological studies of environmental influences on diabetes risk observed a reduced risk of diabetes in non-diabetics who participated in regular exercise, in addition this reduction in risk appeared to be greater in those who participated in more frequent exercise (Hu et al. 1999;Manson et al. 1992;Paffenbarger et al. 1997). Although diabetes is associated with obesity and high physical activity levels are associated with lean body morphology, the benefit of exercise in preventing T2D persists even in obese patients (Hu et al. 2004) and may have greater metabolic effects in those at greatest risk of diabetes (Gill & Cooper 2008;Helmrich et al. 1991;Lynch et al. 1996).

Exercise may reduce risks of both diabetes and cardiovascular disease by mediating multiple metabolic changes. However, it should be appreciated that exercise and physical activity do not necessarily mean the same thing. 'Physical activity' may refer to the amount of habitual movement made during a usual day, whilst 'exercise' may be interpreted as a volitional attempt to engage in an undertaking which is specifically designed to raise activity for a period above that of habitual levels. When examining the impact of exercise interventions upon insulin sensitivity and obesity, it is therefore important to appreciate the type, duration, frequency and workload of the intervention and the pre-intervention characteristics of the studied population. With this in mind, the following sections examine the effects of an acute single exercise session, an exercise training intervention and the mechanisms by which these interventions affect metabolic health, as a background to Chapters 4 and 5. In addition, the methods by which exercise induces reduction in body mass and fat mass specifically are examined to provide background to Chapter 6.

#### 1.16.1 Metabolic impact of a single exercise session

Acute effects of exercise can be observed in a number of metabolic pathways, even in response to a single session of exercise. Consistent reductions in triglycerides and increases in HDL-cholesterol have been observed at 12-48 hours after a single session and the degree of benefit is likely to be related to the energy deficit of the session (Thompson et al. 2001). Additionally, reduced blood pressure is present for 12-16 hours after an exercise session (Thompson et al. 2001).

Exercise mediates improves skeletal muscle glucose uptake through a final common pathway of increased GLUT4 expression and it is recognised that this is not achieved through acute activation of the insulin signalling pathway (Holloszy 2005). Instead, skeletal muscle contraction may stimulate insulin-independent translocation of GLUT4 by release of calcium

from the sarcoplasmic reticulum and subsequent activation of calmodulin-dependent protein kinase (CaMK) (Wright et al. 2005). Glucose delivery to skeletal muscle is also enhanced by increased capillary recruitment (Wojtaszewski & Richter 2006). Contraction may stimulate muscle glucose uptake by other mechanisms. AMPK activation can promote insulinindependent glucose uptake, but it is unclear if contraction can independently activate AMPK to a significant degree (Bosselaar, Smits, & Tack 2009;Wojtaszewski & Richter 2006). These changes describe an acute increase in skeletal muscle glucose uptake, rather than in insulin sensitivity. Therefore, these effects are related to increased insulin-independent membrane expression of intramyocellular GLUT4 from established storage pools. These mechanisms may partially account for the acute exercise-mediated increase in glucose uptake by skeletal muscle, however they do not persist beyond approximately three hours after exercise (Young et al. 1987). Subsequent increases in acute post-exercise insulin sensitivity persist beyond the initial period and are thought to be related to separate mechanisms.

After the initial increase in contraction-related glucose uptake, other mechanisms contribute to continued stimulation of GLUT4 translocation. These mechanisms appear to be related to increased activity of components of the insulin signalling cascade. At a period of approximately 4 hours after a single exercise session, increased insulin receptor substrate-2 (IRS-2) activity results in production of phosphatidylinositol-3,4,5-triphosphate (PIP<sub>3</sub>) by stimulation of PI3K (Howlett et al. 2006). PIP<sub>3</sub> promotes atypical PKC function which is associated with increased insulin-mediated translocation of GLUT4 and may be implicated in exercise-mediated translocation and improved insulin sensitivity in the post-exercise state (Wojtaszewski & Richter 2006). Additionally, since AMPK is considered to be sensitive to the energy state of the myocyte it has been suggested that glycogen depletion may stimulate AMPK to directly influence glucose uptake (Borghouts & Keizer 2000;Jensen, Wojtaszewski, & Richter 2009).

Perseghin and colleagues (1996) described increased insulin sensitivity, intramuscular glucose-6-phosphate concentrations and glycogen synthesis in offspring of patients with T2D in response to a single exercise session and further increases after a six-week exercise intervention (Perseghin et al. 1996). AMPK has binding affinity for both glycogen and the enzyme glycogen synthase and McBride and Hardie (2009) have hypothesised that AMPK stimulates glucose uptake and suppresses glycogen synthesis during exercise as glycogen stores deplete (McBride & Hardie 2009). The exact mechanisms by which AMPK activation may increase direct glucose uptake in the acute post-exercise period are currently incompletely understood but may, in part, be mediated through Akt effects on GLUT4 vesicle exocytosis (Geiger et al. 2006; Jensen, Wojtaszewski, & Richter 2009). Myocellular energy deficit may be manifest by elevated levels of AMP or depletion of ATP or glycogen. AMPK has a well recognised role as a 'guardian of cell energy' and responds to relative energy deficit by inhibiting ACC2 which reduces concentrations of malonyl CoA, subsequently stimulating fatty acid oxidation (Jensen, Wojtaszewski, & Richter 2009;Turcotte & Fisher 2008). As previously described, reduced intramyocellular DAG and ceramide improve insulin signalling (Section 1.8.2). Exercise may induce activity of the p38 mitogen-activation protein kinases (MAPK) which can increase the activation of translocated GLUT4, and subsequently act in concert with insulin to promote further glucose uptake (Geiger et al. 2005). However, other authors have suggested that there is little evidence for an effect on the proximal components of the insulin-signalling cascade in increased exercise-mediated insulin sensitivity (Wojtaszewski et al. 2000). Instead, it is suggested that the GLUT4 pools which have recently recycled from the cell membrane after the initial contraction/hypoxia related stimulus remain more sensitive to insulin stimulation than prior to exercise (Geiger et al. 2006;Holloszy 2005). Although the mechanisms are incompletely understood there is a consistent observation that a single acute exercise session can increase insulin sensitivity and a recent review of this subject suggested that even given the wide variety of durations, intensities, subjects and types of exercise previously studied, a single session can induce improvements in insulin sensitivity for up to forty-eight hours (Turcotte & Fisher 2008).

#### **1.16.2 Metabolic impact of repeated exercise sessions**

The acute effects of a single exercise session do not persist and appear to be at least partly dependent upon the energy status of the myocyte, with reversal of exercise-mediated insulin sensitivity by glycogen repletion (Young et al. 1983). However, after an endurance training programme elderly women displayed persisting increases in insulin sensitivity despite a 72-hour interval between testing and the final exercise session (DiPietro et al. 2006). Younger but not older women display prolonged improvements in insulin sensitivity and this may be attributable to chronic adaptations in response to exercise (Goulet et al. 2005). Adaptive changes in skeletal muscle are at least partly responsible for improved insulin sensitivity outwith the 'acute' period. Although the precise changes in insulin signalling in response to exercise remain controversial (Christ-Roberts et al. 2004;Hawley & Lessard 2008), it is likely that chronic increased expression (as well as acute translocation) of GLUT4 is involved (Henriksen 2002). There is some evidence that exercise training can increase insulin receptor, Akt, GLUT4 and glycogen synthase levels and that increased PI3K signalling may occur also however, the significance of these findings in the greater context of muscle insulin sensitivity is unclear (Wojtaszewski & Richter 2006).

Repeated exercise sessions may stimulate other adaptive changes which can contribute to both improved insulin sensitivity and metabolic health. Increased oxidative capacity and capillary density were observed in skeletal muscle in response to aerobic exercise (Dube et al. 2008). In addition, increased adipose tissue insulin sensitivity has been observed 72-hours after completion of a 6-week exercise intervention (Shojaee-Moradie et al. 2006). The latter occurred in the absence of reduced body mass or fat mass but was associated with reduced visceral adipose tissue (Shojaee-Moradie et al. 2006). Other metabolic changes which occur after the 'acute' post-exercise period may also influence insulin sensitivity, for example: reduced circulating leptin has been observed in response to the energy deficit of an exercise intervention, but only after a 48-hour delay (Essig et al. 2000). Improved endothelial function has been observed in response to acute exercise, but this effect is augmented by further bouts of exercise and persists for approximately one week (Haram et al. 2006). Increased AMPK activity may be associated with acute post-exercise increases in insulin sensitivity but repeated exercise sessions induce upregulation of AMPK and this may be partly responsible for skeletal muscle adaptation to exercise (Viollet et al. 2009).

# 1.16.3 Other insulin-sensitising effects of exercise

Exercise also mediates improved insulin sensitivity through a number of different mechanisms. Although total body weight may not change, body morphology may alter with increased lean tissue, reduced total body fat and reduction in specific adipose tissue compartments (Bouchard et al. 1990). Exercise may preferentially reduce central adiposity (Bo et al. 2008) and particularly visceral adipose tissue, producing different metabolic effects in comparison to a global reduction in total body fat (Johnson et al. 2009;Miyatake et al. 2002). Exercise also reduces markers of low grade inflammation in insulin resistant conditions and does so without significant weight or fat loss (Kadoglou et al. 2007;Petersen & Pedersen 2005). Additionally, exercise induces increased fat oxidation without reduction in adiposity (Venables & Jeukendrup 2008). Exercise mediated reduction of fat mass is associated with reduced circulating leptin (O'Leary et al. 2006) and increased fat oxidation (Goodpaster, Katsiaras, & Kelley 2003). In addition, circulating leptin concentrations after an exercise intervention correlated with fat oxidation in obese subjects, suggesting that increased fat oxidation and insulin sensitivity are partially mediated by improved leptin sensitivity in response to exercise (Solomon et al. 2008).

Further evidence that fat loss *per se* is less important than reduction in metabolically active adipose (such as visceral fat, or deep subcutaneous fat) or improvement in adipose tissue function can be observed in studies of obese patients who undergo fat reduction surgery.
Large volume liposuction has been shown to produce impressive cosmetic reduction in central adiposity, but does not alter visceral fat or other risk factors for T2D or coronary heart disease (Klein et al. 2004;Mohammed et al. 2008). Klein and colleagues (2004) report reduced circulating leptin after removal of large volumes of adipose but this is not associated with evidence of improved leptin, or insulin, sensitivity (Klein et al. 2004). Leptin is recognised to correlate with fat mass, however reduction of fat mass by exercise or dietary methods is likely to alter fat composition and function differently from surgical intervention (Mohammed et al. 2008). Ybarra and co-workers observed alterations in NEFA and lipids after liposuction, but not insulin sensitivity (Ybarra et al. 2008).

In summary, exercise improves insulin sensitivity after an acute session by mechanisms of insulin-independent and insulin-dependent stimulation of GLUT4 cycling. These mechanisms of action are multiply redundant but are probably related to both the physical contraction of the myocyte and the acute energy deficit (especially glycogen depletion) of the cell (Eriksen et al. 2007;Holloszy 2005). Further acute metabolic benefits are recognised with respect to blood pressure and lipids (Thompson et al. 2005). Chronic adaptive responses to exercise are also seen and with respect to increased peripheral insulin sensitivity, are likely to be related to increased oxidative capacity, skeletal muscle vascularity and upregulation of the pathways of both GLUT4 expression and insulin signalling (Colberg & Grieco 2009; Dube et al. 2008). These metabolic changes are accompanied by improved endothelial function and reduction in low grade inflammation although the variety in study design and associated results make analysis of the chronic effects of exercise independent of the effects of reduced body mass or adiposity difficult to determine (Esposito et al. 2003; Petersen & Pedersen 2005). However, it is likely that exercise interventions also improve insulin sensitivity in 'at risk' groups through reduction of body fat and in particular, visceral adipose. In addition independent of fat loss, altered adipose tissue function appears to contribute to improved insulin sensitivity by increasing lipid oxidation and restoring leptin sensitivity and is likely to contribute to a reduction in the chronic inflammatory state. Chapter 4 of this thesis will explore the acute

72

metabolic effects of exercise, by studying subjects within twenty four hours of an exercise session with repeat assessment after forty eight hours of sedentary activity, with the aim of detecting persisting metabolic adaptations to repeated exercise and defining those which are a consequence of a single bout of activity.

## 1.16.4 Exercise in persons at risk of type 2 diabetes

Exercise may have amplified metabolic effects in those who are at greater risk of the development of T2D. Post-prandial hyperinsulinaemia was reduced to a greater extent in obese insulin resistant men compared to lean insulin sensitive men in response to a preprandial exercise session (Gill et al. 2004). The magnitude of the exercise-mediated improvement in insulin sensitivity in the HERITAGE family study was related to baseline insulin resistance (An et al. 2005). Studies of physical activity and insulin sensitivity in offspring of people with T2D have observed that the increased insulin resistance associated with a family history of T2D is partially (Ahn et al. 2004) or completely (Higgins et al. 2005) ameliorated by high levels of habitual activity. Similarly in low-birthweight adult males, the tendency to hyperinsulinaemia and the metabolic syndrome were attenuated by regular participation in vigorous physical activity (Laaksonen et al. 2003). By careful matching for age and body mass and by applying recruitment criteria specifying less than one hour of planned physical activity per week, the studies described in chapters 3-5 aim to detect baseline and post-intervention differences in metabolic and physical characteristics which are determined largely by the presence, or absence of a parental history of T2D.

## 1.17 Lifestyle interventions for weight loss

Weight loss is a desired outcome for many individuals (particularly women) who are overweight, whether they are motivated to do so for cosmetic, psychological or medical reasons (Green et al. 1997). Individual dietary counselling is effective in reducing body mass and is associated with reduced incidence of diabetes and improved blood pressure control (Avenell et al. 2004). However dietary weight loss is frequently unsuccessful and there are a number of psychosocial factors which limit an individual's ability to adhere to restriction of energy intake (Thomas et al. 2008). The addition of a physical activity intervention to increase energy expenditure in combination with reduction of energy intake results in greater weight loss and reduced weight regain (Steyn, Lambert, & Tabana 2009). The major determinant of fat loss in response to a lifestyle intervention is the energy deficit incurred (Strasser, Spreitzer, & Haber 2007) however, there is marked inter-individual variation in the extent of weight and fat loss in response to an exercise intervention (Byrne et al. 2006;King et al. 2008; Snyder et al. 1997). Predicted reductions in weight or fat mass are dependent upon the individual achieving the energy deficit calculated by either dietary restriction (Vogels & Westerterp-Plantenga 2007), increased physical activity (Byrne et al. 2006;Colley et al. 2008;McTiernan et al. 2007), or a combination of both. The observed variation between actual and predicted weight loss in response to an exercise intervention has previously been attributed to volitional or subconscious behavioural change such as increased energy intake (King et al. 2007), alteration in dietary composition (Finlayson et al. 2009) and decreased habitual physical activity or 'Non-Exercise Activity Thermogenesis' (NEAT) (Levine et al. 1999; Wang et al. 2008). However, recently identified variations in the FTO gene have suggested that individuals with these genetic variants are predisposed to overweight and obesity (Frayling et al. 2007) and it is possible that similar 'innate' factors could predispose to suboptimal weight loss response to a negative energy balance.

## 1.17.1 Diet and weight loss

The use of a hypocaloric diet to invoke energy deficit has been shown to reduce total body mass, fat mass and fat-free mass (Durrant et al. 1980). Low calorie, low carbohydrate, very low calorie and low fat diets are effective in inducing short-term weight reduction but are limited by patient tolerance and compliance and may not be associated with long-term weight maintenance (Strychar 2006). Very low calorie diets have the greatest reduction in energy

intake and are therefore associated with the greatest weight loss (~ 30% of body weight at 1 year) but have risks related to electrolyte abnormalties and malnutrition (Strychar 2006). For this reason these diets require close medical supervision. Dietary intervention appears to be an effective weight loss strategy in all groups; children (Shalitin et al. 2009), adults (Abete et al. 2009), the elderly (Houston, Nicklas, & Zizza 2009), both sexes (Al et al. 2009;Dengel et al. 1994) and patients with insulin resistance (Dyson 2008). However, restriction of energy intake is associated with reduced muscle mass which may be deleterious to both functional capacity and total body insulin sensitivity, particularly in the elderly (Frimel, Sinacore, & Villareal 2008;Kelley et al. 1999).

In addition to the physical reduction in body mass, dietary induced weight loss has additional metabolic benefits. A 6.1 kg reduction in body mass after a 3-month dietary intervention was associated with reduced liver fat and increased hepatic glucose uptake (Sato et al. 2007). Low carbohydrate diets have been shown to decrease body mass, markers of hepatic steatosis and increase insulin sensitivity (Ryan et al. 2007). Hypocaloric diet induced a 6.7 kg reduction in body mass over 12 weeks and was associated with increased insulin sensitivity, improved lipid profile and both total, and multimeric circulating adiponectin (Polak et al. 2007). Raatz and co-workers (2005) did not observe a metabolic or weight loss benefit of different diet compositions beyond that observed by the energy deficit incurred (Raatz et al. 2005). However, recent results from Volek and colleagues (2009) suggest that carbohydrate restriction may have greater impact upon lipids, insulin sensitivity, weight loss and regional adiposity than fat restriction (Volek et al. 2009).

## 1.17.2 Exercise and weight loss

Establishing progressive weight loss requires a consistent state of negative energy balance, therefore increasing energy expenditure through exercise should also promote weight loss.

However, a recent meta-analysis suggests that exercise alone is less effective than diet, or combined diet and exercise interventions (Franz et al. 2007). Maintenance of reduced body mass requires a state of energy balance and it is suggested that this is most effectively achieved when dietary modification is combined with increased physical activity (Donnelly et al. 2009). The potential metabolic benefits of exercise have already been outlined in this chapter and a recent Cochrane review has supported the positive impact of exercise on 'metabolic fitness' independent of reduced body mass (Macfarlane & Thomas 2009;Shaw et al. 2006). Therefore, Chapter 5 of this thesis aims to address the metabolic impact of a seven-week exercise programme and in particular will examine the mechanisms of improved 'metabolic health' with respect to the energy deficit of the intervention and independent of this factor. In addition, Chapter 6 of this thesis in response to an exercise intervention with particular focus on the role of behavioural compensation for the energy deficit of exercise and associated metabolic processes which may contribute to an exercise-mediated reduction in body fat.

## 1.18 Clinical approaches to treatment of insulin resistance

Despite a growing appreciation of the multiple metabolic defects inherent to the development of insulin resistance, clinicians have had relatively few pharmacological options available to treat those with a pre-diabetic insulin resistant state. Acarbose, an alpha-glucosidase inhibitor has been shown to reduce post-prandial hyperglycaemia and its use in patients with impaired glucose tolerance appears to reduce progression to T2D (Chiasson et al. 2002). Concerns have however been expressed about the validity of the methodology and data analysis in this trial (Sawicki & Kaiser 2004) and in addition, acarbose is generally poorly tolerated by patients due to gastrointestinal side effects (Catalan, Couture, & LeLorier 2001).

Metformin has also been shown to reduce progression to T2D from IGT in the Diabetes Prevention Program (Knowler et al. 2002) and the Diabetes Prevention Study (Tuomilehto et al. 2001). Metformin is considered to have pleotropic effects on insulin resistance including reduction of dietary carbohydrate absorption, reduced hepatic glucose output and increased skeletal muscle insulin sensitivity (Setter et al. 2003). The thiazolidinedione class of drugs are also recognised to have insulin-sensitising effects and have been shown to prevent progression to diabetes (Gerstein et al. 2006), but concerns exist about long-term therapy with respect to possible cardiovascular (Nissen & Wolski 2007) and bone toxicity (Kahn et al. 2006). These concerns may be particularly important in the context of the potential duration of therapy, and with limited knowledge about the persistence of the preventative effect after cessation of drug therapy (Crandall et al. 2008).

Lifestyle intervention trials have frequently combined dietary advice with measures to increase physical activity and in those studies where groups received specific diet or exercise advice both were still efficacious, but not as effective as when combined (Knowler et al. 2002;Pan et al. 1997;Tuomilehto et al. 2001;Ramachandran et al. 2006). These interventions are often associated with weight loss, hence the greater success of a combined diet and exercise approach. However not all exercise interventions produce weight loss, but frequently continue to be associated with improved metabolic health (Ross et al. 2000;Simmons et al. 2008). Indeed, a recent study by Burton and colleagues (2008) highlighted the metabolic benefit of exercise, showing that exercise with replacement of the energy deficit incurred was still beneficial, reducing post-prandial insulinaemia and increasing fat oxidation (Burton et al. 2008).

Obesity surgery is proven to promote large reductions in body mass and is also associated with improved metabolic fitness. Sjöström and co-workers have more than ten years of follow up data for Swedish subjects who had bariatric surgery and have reported sustained reductions in mortality and morbidity from diabetes and cardiovascular risk (Sjostrom et al. 2004). Despite the initial high economic cost of bariatric surgery it is a cost-effective way of treating severe obesity and may be an appropriate modality in the treatment of the non-obese patient with T2D who is judged to be 'metabolically unhealthy' and whose glycaemic control has been poor despite aggressive management with conventional agents (Picot et al. 2009).

## 1.19 Summary

In conclusion, the literature review detailed in this chapter has explored the increasing prevalence of obesity and insulin resistance, with the subsequent risk of T2D and cardiovascular disease, as well as the aetiology of insulin resistance and possible methods of preventing its progression to chronic disease. The aetiology of insulin resistance in many individuals is likely to be a consequence of genetic predisposition such as a family history of diabetes or an ethnic group with characteristic tendencies and exposure to an environment which promotes a sedentary lifestyle with a 'Westernised' diet. In some cases, the risk of diabetes may be attributable to early origin factors although these too may be influenced by parental history and subsequent environmental conditions.

The pathophysiology of insulin resistance is complex and multifactorial but consistent observations have suggested that adipose tissue factors such as NEFA and adipokines have an important role in determining insulin resistance in liver, muscle, adipose tissue and the vascular endothelium. Insulin resistance results in an inability to utilise energy substrates efficiently, leading to a chronic, progressive deterioration in metabolic function. The development of insulin resistance may have relatively poorer health outcomes in women compared to men and given the proven efficacy of lifestyle interventions in insulin resistant states, women may represent an important group to target for intervention. In addition, those with a family history of T2D are a clearly defined population who are at greater risk of

chronic disease and may display early metabolic defects. These defects may be acutely sensitive to exercise intervention and therefore targeting high risk subjects, such as those with a parental history of T2D may represent a cost-effective strategy to prevent this disease.

Exercise is of benefit in the treatment of insulin resistance and overweight and obesity, however the mechanisms by which these benefits occur are incompletely understood, particularly with respect to benefits independent of energy deficit. In addition, exercise may impact upon metabolic health through different mechanisms depending upon the physical and genetic characteristics of the individual as well as alterations in other aspects of lifestyle in response to a change in physical activity. With respect to exercise for the promotion of weight loss, it is unclear whether 'failure' to achieve predicted weight loss with a known energy deficit is a consequence of behavioural compensation or a consequence of acquired or innate factors.

The hypotheses of the studies contained within this thesis are as follows. Firstly, that sedentary, pre-menopausal women with a parental history of T2D are more insulin resistant than matched controls and that despite normoglycaemia, other metabolic defects are present which are involved in the aetiology of insulin resistance. Secondly, that a familial history of diabetes is evidence of a 'thrifty genotype' which is sensitive to increases in physical activity and that exercise will have a greater impact upon those women with a family history of T2D. Thirdly, that any potential augmented response to exercise in women with a family history of T2D is associated with either enhanced metabolic or physical responses to exercise or that in these subjects exercise influences separate metabolic pathways. Finally, that individual variation in exercise-mediated reduction in fat mass is a consequence of behavioural compensation, differences in innate metabolic responses to exercise or a combination of both.

Therefore the aims of this thesis are as follows: In Chapter 3: to determine the physical and metabolic characteristics of sedentary, normoglycaemic pre-menopausal women with a parental history of T2D compared to matched subjects without a familial history of T2D. In addition, to determine the influence of lifestyle and adiposity as potential mediators of metabolic dysfunction in those with evidence of insulin resistance and whether mediators of insulin resistance are dependent on a family history of T2D. In Chapter 4: to determine the metabolic impact of a seven-week aerobic exercise intervention on the same two cohorts and to determine whether accompanying changes in physical, metabolic or lifestyle parameters provide evidence as to the pathways by which exercise might improve metabolic health. In Chapter 5: to examine the contributors to an exercise-mediated increase in metabolic fitness, whether there are different mechanisms of action depending upon a familial predisposition towards T2D and to what extent improved metabolic health is achieved independently of reduced fat mass and energy deficit. In Chapter 6: to examine both the role of possible behavioural compensation and potential innate exercise-responsive pathways.

#### **CHAPTER 2**

#### **GENERAL METHODS**

#### **2.1 Introduction**

This chapter describes the methods employed in recruiting volunteers for the study and the measurement of the anthropometric, metabolic and cardio-respiratory variables required in the study protocol. This chapter also discusses the exercise intervention and the procedures involved in monitoring adherence to the exercise regime. This study was conducted with the ethical approval of the Ethics Committee of North Glasgow Hospitals University NHS Trust and was registered with the National Institute for Health (NIH) in the United States of America (Reference No. NCT00268541). Registration details can be viewed at the following website; <a href="http://www.clinicaltrials.gov/ct">http://www.clinicaltrials.gov/ct</a>.

#### 2.1.1 Inclusion & exclusion criteria

Volunteers were deemed suitable to participate in the trial having met the following criteria; Female, aged between 20 and 45 years old with regular menstrual cycle, non-smoking, systolic blood pressure <160 mmHg, diastolic blood pressure <90mmHg, sedentary lifestyle (less than 1 hour planned physical activity per week, and sedentary job), in good general health without diabetes or a past history of cardiovascular disease. Volunteers were not excluded from the study if they were on prescribed medication, provided that their prescription was stable and did not include medications which were recognised to modulate glucose or lipid homeostasis. Volunteers were allocated into one of two study groups according to family history of type 2 diabetes. Volunteers with a parent who had developed type 2 diabetes prior to the age of 65 years were designated 'Offspring', and those with no first or second degree relative with diabetes prior to the age of 65 years were designated as 'Controls'.

## 2.1.2 Recruitment, Screening & Selection

Volunteers were recruited from the Greater Glasgow area (See Section 2.14.1, Power Calculation). A variety of media were utilised to publicise the study. Initial recruitment was generated from an advertisement in the Glasgow University Newsletter, further exposure occurred through an article in the Evening Times newspaper which coincided with a press release from the British Heart Foundation. Recruitment posters were displayed around the Glasgow University main campus, and associated campuses across Glasgow (e.g. Jordanhill and Garscube campus). A number of general practitioners who provide primary care for patients with type 2 diabetes through the Glasgow Diabetes Service agreed to display recruitment posters in their surgery waiting rooms. Posters aimed specifically at recruiting subjects with a family history of type 2 diabetes were displayed in outpatient clinic areas and diabetic day units of Stobhill Hospital, Gartnavel General Hospital and Glasgow Royal Infirmary. Potential volunteers could access information about the study through a weblink sited on the University Sports and Recreation website. In addition to this, volunteers were able to provide basic contact details and note their interest by submission of a webform which was emailed to the principal investigators. A small number of volunteers was also recruited through an internet football discussion forum. Information on the study was disseminated by word-of-mouth, by both the investigators and volunteers themselves, and was responsible for a small proportion of the total number of subjects recruited.

Interested volunteers were initially contacted by telephone to provide information and to arrange a formal screening visit to the Metabolic Investigation Suite in the Institute of Diet, Exercise and Lifestyle (IDEAL) in the West Medical Building of the University of Glasgow. Volunteers also completed a health questionnaire (Appendix A1) to exclude significant health problems and to determine any important family history. They were provided with an information sheet (Appendix A2), given an oral presentation which explained the study and were then encouraged to give feedback and ask questions. Informed consent was obtained in triplicate. One copy was given to the volunteer for reference, one copy was retained as part of

82

an individualised case record for the volunteer (Appendix A3), the third copy was sent to the volunteer's general practitioner (GP) as part of an information pack. The information pack contained an explanatory letter (Appendix A4), with a copy of the consent form (Appendix A3) and the information sheet (Appendix A2).

Volunteers attended after a 12-hour overnight fast. Blood samples were obtained by routine venepuncture using the Vacutainer<sup>™</sup> (BD Vacutainer Systems Preanalytical Solutions, Plymouth, PL6 7BP) system. Fasting plasma glucose (hexokinase method), lipids - total cholesterol (cholesterol oxidase method), triglycerides (glycerol oxidase method), HDL cholesterol (detergent/cholesterol oxidase method) and LDL cholesterol (Friedewald equation (Friedewald, Levy, & Fredrickson 1972), hepatic enzymes (alanine aminotransferase, aspartate aminotransferase (international federation of clinical chemistry method), bilirubin (diazo spectrophotometry), alkaline phosphatase (AMP colorimetry), albumin (BCP quantative colorimetry), and gamma-glutamyl transferase (carboxy nitroanilide spectrophotometry), renal - sodium, potassium, chloride (all by ion selective electrode method) urea (urease method) and creatinine (Kinetic Jaffe spectrophotometry), and thyroid (thyroid stimulating hormone, free T4 both by immuno-enzymatic reaction) function were analysed using Abbott commercial reagents (Abbott Laboratories, Illinois, USA) on C8000 or Architect analysers (Abbott Laboratories, Illinois, USA) in the NHS biochemistry laboratories at Glasgow Royal Infirmary.

A basic physical examination was performed, including cardiac auscultation to exclude significant murmurs, and auscultation of carotid arteries to exclude carotid bruits. Two volunteers were noted to have previously undiagnosed cardiac murmurs. These two volunteers underwent echocardiography at the Cardiology Department of the Western Infirmary, Glasgow and in both cases no significant valvular abnormality was detected. One volunteer was noted to have a right carotid bruit and had Carotid Doppler Ultrasonography performed in the Acute Stroke Unit at the Western Infirmary, Glasgow, which also excluded abnormal carotid flow. It was necessary to perform carotid imaging as the Complior SP system uses a probe which applies pressure over the carotid vessels and could theoretically precipitate a cerebral thromboembolic event arising from a vessel containing an atherosclerotic plaque.

All subjects performed a Bruce Protocol Exercise Tolerance Test (BPETT) (Bruce, Kusumi, & Hosmer 1973). This test is an incremental exercise test consisting of 3-minute stages which produce a stepwise increase in workload by increasing both treadmill speed and gradient (Figure 2.1). The test was performed with electrocardiograph and blood pressure monitoring, and was performed and interpreted by the author (a registered medical practitioner with Advanced Life Support (ALS) accreditation). All volunteers achieved an adequate workload on BPETT and remained free of symptoms and electrocardiographical signs of coronary artery insufficiency.



Time (mins)

**Figure 2.1** Diagramatic representation of Bruce Protocol Exercise Tolerance Test. Table insert displays changes in gradient and speed of treadmill at given stage of protocol.

Figure 2.2 illustrates the numbers of people involved in the study, including dropouts and exclusions.



**Figure 2.2** Flow chart of volunteer recruitment and randomisation with details of dropouts and exclusions at each stage of the study.

## 2.1.3 Study Design

The two groups (Offspring and Controls) were screened and investigated in parallel and participants were randomly allocated into one of two subgroups. The 'Delayed Exercise' subgroup was designed to provide a study period of 'usual activity', with metabolic, anthropometric and cardiorespiratory testing before and after a 7-week period where subjects were encouraged to maintain their usual lifestyle (Figure 2.3). This provided a control period to determine whether a 'study effect' influenced any of the studied variables. Subjects in the Delayed Exercise group entered a 7-week exercise programme after completing their 7-week period of 'usual activity', and had one further testing period after this. Figure 2.3 displays the timeline of testing and intervention in the Delayed Exercise group.

The 'Exercise' group was tested in the week prior to commencing the 7-week exercise programme. Repeat exercise testing was performed in week 7 of the intervention, with two post-intervention assessments of metabolic parameters in week 8. The first post-intervention metabolic assessment occurred within 15-24 hours of completion of the final exercise session of the intervention. After an interval of 3 days, during which time volunteers were asked to avoid planned exercise, a further metabolic assessment was performed (Figure 2.4) to ascertain the chronic versus the acute effects of exercise training on the measured variables. The volunteers were randomised to one of the two intervention arms using the 'randbetween' function in an Excel spreadsheet (Microsoft Corp. Redmond, WA, USA). Volunteers whose entry into the spreadsheet generated an odd number between 1 and 10 were placed in the Delayed Exercise group and those who generated an even number between 1 and 10 were placed in the Exercise group.

87



Figure 2.3 Study design for Delayed Exercise group.



Figure 2.4 Study design for Exercise group.

Volunteers were asked to attend for a number of separate appointments at baseline, and subsequent re-testing intervals. As exercise acutely influences a number of metabolic variables, such as insulin sensitivity and lipid metabolism (Koopman et al. 2005;Magkos et al. 2007;Magkos et al. 2008;Wooten, Biggerstaff, & Anderson 2008), it was important to structure the timing of the submaximal fitness tests at baseline in the Exercise group, and at both baseline and 'after usual lifestyle' period in the Delayed Exercise group, to prevent an impact on the metabolic testing.

## 2.2 Anthropometry and Body Composition

## 2.2.1 Height

Height was measured using a stadiometer (Invicta Plastics Ltd, Leicester, UK). The volunteer was measured barefoot, with their back positioned against a fixed backboard and their arms relaxed in the lateral position. The head was also positioned against the backboard, with the line of eyesight perpendicular to the backboard. Measurement was performed when the volunteer was positioned and relaxed, and a moveable headboard was lowered on to the top of the head with light pressure allowing hair compression. The investigator applied gentle upwards pressure underneath the angle of the mandible and measurement was made to the nearest 0.01 metre.

## 2.2.2 Body Mass

Body mass was measured using a manual balance weighing scale (Avery, Birmingham, UK). The same scale was used for all volunteers throughout the study. Subjects were weighed wearing lightweight clothing (e.g. shorts and t-shirt) without wearing shoes. Extraneous jewellery and clothing was removed prior to weighing. Body mass was measured with both feet flat on the balance and with arms positioned in the lateral position. Measurement was made to 0.05 kg.

## 2.2.3 Skinfold Thickness

All measurements of skinfold thickness were performed in private, by the same investigator on each occasion. Measurements were made to 0.1 mm using Harpenden skinfold callipers (Cranlea & Company, Birmingham, UK). The same callipers were used throughout the study. The skinfold was held between the investigator's thumb and index finger, the callipers were applied and the measurement made after pressure had been applied for between 5 and 8 seconds. The left hand side of the body was used for measurement, and measurements were made with the volunteer standing up. Biceps skinfold was marked with the elbow flexed at an angle of 90°, at the midpoint between the most lateral point of the acromion and the inferior border of the olecranon. Once marked, the arm was allowed to hang loosely with the palm facing anteriorly, and the measurement made in the midline of the biceps muscle, at the level of the mark. Triceps skinfold was measured at the same level as the biceps measurement, on the posterior aspect of the arm, in the midline. The subscapular skinfold was measured along the line of natural skin cleavage inferior to the scapula, with both arms hanging loosely by the volunteer's side. The suprailiac fold was measured above the iliac crest, on a diagonal fold beginning at the anterior axillary line.

Skinfold measurements were used to illustrate regional adiposity and were not used to determine body density, or body fat. Coefficients of variation were 1.9% for biceps, 0.8% for triceps, and 0.7% for both subscapular and suprailiac skinfolds.

## 2.2.4 Waist & Hip Circumference

Circumferences were measured using a non-elastic tape measure, and measurement was made with the abdominal muscles relaxed, at the end of normal expiration. Both hip and waist measurements were taken three times, and an average of the three readings was noted. Waist circumference was measured horizontally around the waist, at a point midway between the superior border of the ilium and the inferior border of the lateral margin of the ribs (costal margin). Hip circumference was measured around the point of maximal width around the hip region, at approximately the level of the pubic symphysis. The same investigator performed the measurements on each volunteer, on every occasion. Waist to hip ratio was determined by dividing the waist circumference by the hip circumference.

#### 2.2.5 Dual X-Ray Absorptiometry (DEXA) Scanning

Dual X-Ray Absorptiometry (DEXA) scanning is recognised as an accurate method of measuring body composition (Gallagher & Song 2003). DEXA uses X-ray beam radiation of two energies, and is able to interpret the differential attenuation of the two energies to identify bone, lean and fat tissue. Scans were performed by an experienced radiographer (Senior II Grade) using a GE Medical LUNAR Prodigy DEXA scanner (GE Healthcare Diagnostic Imaging, Slough, Berkshire, UK), in the Radiology Department of the Royal Hospital for Sick Children, Yorkhill, Glasgow. Scan data was interpreted using the generic software.

DEXA scanning was carried out at intervals as described in Figures 2.3 - 2.4. The amount of radiation used in a DEXA scan is approximately 0.0007 mSv, equivalent to 10% of one day of background radiation. For further information, volunteers were advised that the amount of radiation delivered during one DEXA scan was fifty times less than one chest x-ray.

## **2.3 Metabolic Testing**

## 2.3.1. Protocol of metabolic testing

Volunteers attended the Metabolic Investigation Suite after a 12-hour overnight fast. Those volunteers randomised to the Exercise group who were attending for the first metabolic testing after the exercise intervention were asked to complete their final exercise session 15-24 hours prior to testing. Following a 10-minute rest lying on a couch, a 20-minute continuous expired air collection was performed using a ventilated hood system (Deltatrac Metabolic Monitor, Datex Engstrom, Kent, UK). Volunteers remained in a supine position in a quiet room after the completion of the expired air sampling and had blood pressure measurement using an automated blood pressure monitor (Omron Healthcare UK Limited, Milton Keynes, UK) (section 2.3.5). Whilst still recumbent, peripheral pulse wave velocity

91

was assessed using the Complior SP system (Artech Medical, Pantin, France). A cannula was introduced into an antecubital vein, and after a 10-minute interval, a fasting blood sample was taken. In order to gain adipose tissue for analysis of gene expression (data not presented in this thesis), volunteers had a 0.5-1 gram sample of subcutaneous abdominal adipose tissue obtained by liposuction technique from an area lateral to the umbilicus. Since the data derived from these biopsies is not presented in this thesis, the method relating to the biopsy procedure is not described in detail, but included to explain the delay between fasting blood sampling and administration of the oral glucose load. After completion of the liposuction biopsy, the volunteers underwent an oral glucose tolerance test (OGTT). A second period of 20-minute continuous expired air collection was made using the ventilated hood following the 120-minute blood sample, and a further assessment of blood pressure and peripheral pulse wave velocity was made (Figure 2.5). Since volunteers had engaged in a prolonged fast, they were offered lunch after concluding the metabolic tests.





Figure 2.5 Timeline illustrating metabolic testing protocol.

#### 2.3.2 Metabolic Rate & Energy Substrate Utilisation

Resting metabolic rate (RMR) was measured in the morning, after a 12-hour fast and 24-hour abstention from exercise, using the Deltatrac Metabolic Monitor (Datex Engstrom, Kent, UK). The Deltatrac was switched on 30 minutes prior to use, and calibrated for ambient atmospheric pressure using a traditional barometer. To ensure accurate assessment of FEO<sub>2</sub> and FECO<sub>2</sub> the gas analyser component of the Deltatrac was calibrated against a known standard gas (95% O<sub>2</sub>, 5% CO<sub>2</sub>) (Quick Cal<sup>™</sup> Calibration Gas, Datex-Ohmeda, Cranlea & Company, Birmingham, UK). Environmental conditions were controlled to ensure a quiet room, with minimal external stimuli and a temperature of 24°C. Volunteers reclined on an examination couch, with the hood in position. No measurements were made for a ten-minute period to allow the volunteer to relax and to acclimatise to both the hood, and the temperature of the room.

Measurements were recorded every 60 seconds for 20 minutes. This system uses a ventilated hood which is placed over the subject's head and neck, with an attached plastic skirt which covers the upper torso. The ventilated hood has a cranial aperture which allows the incursion of ambient air, and a caudal aperture which is attached by tubing to the main body of the Deltatrac (Figure 2.6). The instrument draws air through the hood at a constant flow rate and from this and the fraction of expired CO<sub>2</sub> (FECO<sub>2</sub>) and O<sub>2</sub> (FEO<sub>2</sub>) is able to calculate both rate of oxygen consumption ( $\dot{V}O_2$ ) and rate of carbon dioxide production ( $\dot{V}CO_2$ ). One minute average values of  $\dot{V}O_2$  and  $\dot{V}CO_2$  were recorded and an average of the final 15 minutes of measurement was used to calculate respiratory exchange ratio (RER), and rate of energy substrate utilisation and energy expenditure, using indirect calorimetry equations derived by Frayn (Frayn 1983). (see section **2.3.3**). Measurement of metabolic rate and substrate utilisation was performed in the same manner, 2 hours following glucose ingestion in the oral glucose tolerance test. Continuous measurement of expired air during the whole period of the OGTT may have allowed more thorough analysis of diet-induced thermogenesis, however the ventilated hood is uncomfortable for prolonged periods. In

93

addition, the majority of the volunteers were in full or part-time employment and were keen to use the 2-hour period to perform sedentary work-related tasks (e.g. checking emails). In retrospect, a further 20-minute expired air collection at 1 hour post-glucose ingestion might have provided evidence of subtle differences in diet-induced thermogenesis and metabolic inflexibility.



Figure 2.6 Photograph of Deltatrac Metabolic Monitor with male volunteer.

## 2.3.3 Indirect calorimetry calculations

Mean values for  $\dot{V}O_2$  and  $\dot{V}CO_2$  over the 20-minute measurement period were used in the indirect calorimetry equations derived by Frayn (Frayn 1983). Frayn's equations require a value for nitrogen excretion to provide an estimation of protein oxidation. Urinary nitrogen excretion was not measured during this study, therefore a constant rate of nitrogen excretion of 0.00011 g.kg<sup>-1</sup>.min<sup>-1</sup> was assumed in accord with published values in the literature (Flatt et al. 1985;Melanson et al. 2005).

$$N(g.min^{-1}) = 0.00011 x body mass (Equation 2.1)$$

Frayn's equations allow calculation of  $\dot{V}O_2$  and  $\dot{V}CO_2$  according to equations 2.2 and 2.3 respectively.

$$\dot{V}O_2 (1.min^{-1}) = 0.746 \text{ CHO} + 2.03 \text{ F} + 6.04 \text{ N} (Equation 2.2)$$
  
 $\dot{V}CO_2 (1.min^{-1}) = 0.746 \text{ CHO} + 1.43 \text{ F} + 4.89 \text{ N} (Equation 2.3)$ 

In these equations, CHO is carbohydrate (grams), N is the constant for nitrogen excretion (grams) and F is fat (grams). Combining these equations with a correction for protein oxidation allowed calculation of non-protein oxygen consumption (NPVO<sub>2</sub>) and non-protein carbon dioxide production (NPVCO<sub>2</sub>), and therefore allowed the non-protein respiratory exchange ratio (NPRER) to be determined. Therefore, equations 2.4 and 2.5 determine NPV O<sub>2</sub> and NPVCO<sub>2</sub> respectively, whilst equation 2.6 determines NPRER.

NP $\dot{V}O_2$  (l.min<sup>-1</sup>) = $\dot{V}O_2 - 6.04$  N (**Equation 2.4**) NP $\dot{V}CO_2$  (l.min<sup>-1</sup>) =  $\dot{V}CO_2 - 4.89$  N (**Equation 2.5**) NPRER = NP $\dot{V}CO_2$ /NP $\dot{V}O_2$  (**Equation 2.6**)

Use of protein corrected values allowed calculation of individual energy substrate oxidation (Equations 2.7 - 2.9).

Fat Oxidation  $(g.min^{-1}) = (N\dot{V}O_2 - NP\dot{V}CO_2) / 0.6$  (Equation 2.7)

Carbohydrate Oxidation  $(g.min^{-1}) = (NP\dot{V}O_2 - 2.03 \text{ x F Ox.}) / 0.746$  (Equation 2.8) Estimated Protein Oxidation  $(g.min^{-1}) = N \times 6.25$  (Equation 2.9)

Total energy expenditure was the sum of energy obtained from fat, CHO and protein oxidation. Energy obtained from individual substrate oxidation was determined by the product of the energy density of each substrate and the amount of substrate oxidised (Equation 2.10).

EE (kJ) = (fat x 39.0) + (carbohydrate x 15.5) + (protein x 17.0) (Equation 2.10)

## 2.3.4 Blood Pressure

The controlled environment and supine positioning described in section 2.3.2, conforms to the European Society of Hypertension guidelines on blood pressure measurement (O'Brien et al. 2003). An automated monitor was used (Omron Healthcare UK Limited, Milton Keynes, UK), and the lowest value of three readings was taken as the recorded value.

#### 2.3.5 Pulse Wave Velocity

The Complior SP system (Artech Medical, Pantin, France) was used to assess peripheral pulse wave velocity (PWV). Assessment of peripheral PWV by carotid –radial measurement was a less intimate examination for female volunteers than central PWV by carotid – femoral measurement. The main investigator was male, therefore central measurement would have required a chaperone. Anthropometric data (height, body mass, waist and hip circumference), and blood pressure measurements were entered into the Complior SP generic software prior to each measurement of PWV. The carotid and radial arteries were identified by manual

palpation, and a sensor applied over each. The manufacturers supply clamps which hold the sensors in place with light pressure. The radial clamp was used to allow the operator a free hand, however the carotid clamp was found to be uncomfortable for the volunteer and easily dislodged leading to loss of sensor signal, therefore the carotid sensor was held in place by the operator's left hand. Both sensor positions were manipulated until a clear waveform was identified on the computer screen. After maintenance of good quality signals from both areas for approximately 10 seconds, a foot pedal was depressed which triggered measurement of the time delay of a single pulse wave being detected by the carotid sensor and the radial sensor. The time delay measured by the Complior SP device is the time taken for a single waveform generated by one left ventricular systole to travel from one site in the arterial tree to another more distal site (in this case between the carotid and radial arteries). The distance between the two sites was measured by a non-elastic tape measure, and this distance was also entered into the Complior SP software (Figure 2.7). This allowed calculation of PWV by the following equation (Equation 2.11).



PWV  $(m.s^{-1})$  = distance (m) / sensor – sensor time (s) (Equation 2.11)

**Figure 2.7** Illustration of Pulse Wave Velocity arterial sensors and measurement of arterial distance.

Six PWV measurements were made, and the average value recorded. Measurement of PWV was performed in the fasted state and upon completion of the oral glucose tolerance test (Figure 2.5). Pulse wave velocity is a marker of arterial stiffness (Tomiyama & Yamashina 2010) and correlates with flow-mediated arterial dilatation (FMD) suggesting that it is a surrogate for endothelial function (Jadhav & Kadam 2005). Nitric oxide is derived from vascular endothelium and is a key mediator of vasodilatation throughout the vascular tree, FMD is considered to be a non-invasive marker of local NO bioavailability. Pulse wave velocity is therefore also considered to be indicative of endothelial function and has been shown to be impaired in offspring of patients with T2D (McEleavy et al. 2004). However whilst PWV correlates with FMD, it is a direct measurement of large vessel elasticity and is not a direct measurement of microvascular endothelial function (Tomiyama & Yamashina 2010). Skeletal muscle microvascular function can be measured more directly using invasive techniques such as laser doppler flowmetry, microdialysis or contrast enhanced ultrasound (Clark 2008;Coggins et al. 2001). However, these techniques are considerably more invasive, technically demanding and time-consuming than PWV and have tended to be used in conjunction with insulin infusions which would have been unsuitable for the study protocol. Venous occlusion plethysmography offers a non-invasive and simple technique which has been purported to measure microvascular perfusion (Alexandraki et al. 2006; Woodman et al. 2005). However whilst measurements of reactive hyperaemia in insulin resistance are consistent with measurements made by FMD and are therefore representative of endothelial function, they are actually measurements of total forearm bloodflow and are not isolated measurements of capillary recruitment (Alexandraki et al. 2006;Clark 2008). Unlike PWV, venous occlusion plethysmography does provide an index of vascular function which includes capillary recruitment therefore despite its limitations it may have been a more sensitive method of assessing differences in vascular function in these groups.

### 2.3.6 Fasting Blood Sampling

After PWV measurements an 18G cannula was placed in an antecububital vein. After a 10minute interval, a fasting blood sample was taken using two 10 ml ethylenediamine tetraacetic acid (EDTA) Vacutainer<sup>™</sup> tubes (BD Vacutainer Systems, Plymouth, UK). After the samples were obtained, the cannula was flushed to maintain patency, using 5 mls of 0.9% saline solution.

## 2.3.7 Oral Glucose Tolerance Test

A 75 g oral glucose load was given to the volunteer. This was prepared using 82.5g of Dextrose Monohydrate (Nutrivit Ltd, Corby, UK) in 340 ml of water, with 10 ml of lemon juice (Jif, Colmans of Norwich, Norwich, UK) to give a total volume of 350 ml. Volunteers were instructed to drink the full volume in a 2-minute period, and a stopwatch was started once the volunteer commenced drinking.

At 30, 60, 90, and 120 minutes after ingestion of the glucose drink a 10 ml blood sample was taken through the cannula, into an EDTA tube. Volunteers were supine at all blood collection points. After each sample the cannula was flushed with 5 ml of 0.9% saline.

#### 2.4 Plasma Preparation and Analysis

#### 2.4.1. Plasma preparation and storage

All blood samples were placed on ice prior to refridgerated centrifugation at 4°C. Samples were centrifuged at 3000 rpm for 15 minutes. Plasma was aspirated after centrifugation using a disposable plastic Pasteur pipette. Fasting plasma was dispensed in 0.5 ml aliquots into labelled 2 ml Eppendorf tubes (Alpha Laboratories Ltd, UK), and frozen at -80°C until analysis. Twelve aliquots were stored from fasting plasma, and six aliquots were stored for each subsequent time point during the oral glucose tolerance test.

## 2.4.2. Enzymatic colorimetric methods

Lipid assays were performed in the Department of Vascular Biochemistry at Glasgow Royal Infirmary. Total cholesterol, HDL-cholesterol and triglycerides were analysed by enzymatic colorimetric methods using commercially available kits (Roche Diagnostics GmbH, Mannheim, Germany). LDL-cholesterol concentrations were determined using the Friedewald equation (Friedewald, Levy, & Fredrickson 1972). NEFA concentrations were determined by enzymatic colorimetric methods, using a commercially available kit (Wako Chemicals GmbH, Neuss, Germany). Glucose was assayed using the hexokinase method, utilising a Randox kit (Randox Laboratories, Crumlin, Co. Antrim, UK). High sensitivity Creactive protein (hsCRP) was measured using an immunoturbidimetric assay (Randox Laboratories, Crumlin, Co. Antrim, UK). All assays were performed on an ILAB™ 600 clinical chemistry analyzer (Instrumentation Laboratory, USA). Analysis was performed by colleagues at the Department of Vascular Biochemistry. Alanine aminotransferase (ALT), Gamma Glutamyl Transferase (GGT) and 3-hydroxybutyrate (3-OHB) were assayed in the biochemistry laboratory at IDEAL, University of Glasgow using a Cobas Mira Plus (ABX Diagnostics, France). ALT, GGT and 3-OHB were measured using commercial kits (Randox Laboratories, Crumlin, Co. Antrim, UK) by a research assistant in IDEAL. All samples for

each participant were analysed in a single analyser run. The accuracy and precision of the assays was monitored using quality control sera (Roche Diagnostics GmbH, Mannheim, Germany. Randox Laboratories Ltd, Co. Antrim, Ireland. Wako Chemicals GmbH, Germany). Coefficients of variation were < 3.1% for colorimetric assays.

## 2.4.3. Enzyme-linked immunoassays

All enzyme-linked ammunoassay (ELISA) procedures were based on a 'sandwich' technique. Commercially produced plates were used for all ELISA. The wells of the plates were coated with a monoclonal antibody to the protein of interest. The addition of plasma to the wells bound protein to the antibody, with unbound molecules removed by washing. A second antibody specific to another area of the protein was then added, with unbound molecules removed by a second washing. The second antibody was linked to an enzyme which would catalyse the conversion of a nonfluorescent substrate to a fluorescent product. The intensity of the subsequent fluorescence was directly proportional to the amount of protein in the initial sample (Stryer 1988).

Insulin was measured using a commercially available ELISA with <0.01% cross-reactivity with proinsulin (Mercodia, Uppsala, Sweden). Adiponectin, resistin, leptin, IL-6 and TNF $\alpha$ were analysed using commercially available kits (R & D Systems Europe, Abingdon, UK). High sensitivity kits were used for TNF $\alpha$  and IL-6. All samples for each participant were analysed in a single analyser run. The accuracy and precision of the assays was monitored using quality control sera (Mercodia AB, Uppsala, Sweden. R & D Systems Europe, Oxford, UK.). Coefficients of variation were 3.9% for adiponectin, 8.0% for leptin, 7.1% for resistin, 11.3% for IL-6, 9.3% for TNF $\alpha$ , and <4% for insulin.

# 2.5 Establishment of relationships of heart rate to $\dot{V}O_2$ and $\dot{V}CO_2$ during rest and exercise

Volunteers attended the exercise laboratories at IDEAL as noted in figures 2.3 and 2.4. Volunteers were asked to fast for 2 hours prior to the visit, to prevent elevation of serum lactate as a consequence of intestinal active transport (Frayn 2003). At this visit, volunteers participated in assessment of both cardiorespiratory fitness and measurements of heart rate with expired air collection to establish relationships between  $\dot{V}O_2$ ,  $\dot{V}CO_2$  and heart rate during sedentary activities.

# 2.5.1 Relationship of heart rate to $\dot{V}O_2$ and $\dot{V}CO_2$ during low intensity activities

Volunteers participated in measurements of expired air and heart rate whilst performing low intensity activities. These measurements were employed to provide data on individual relationships (rather than mean values for Offspring and Control groups) between  $\dot{V}O2$  and  $\dot{V}$  CO2 which are required for energy expenditure calculations. The importance of an individual calibration technique to determine energy expenditure at low intensity activities is previously recognised (Ceesay et al. 1989;Spurr et al. 1988). Consistent with previous literature (Ceesay et al. 1989;Spurr et al. 1988) three stages were used to assess  $\dot{V}O_2$ ,  $\dot{V}CO_2$  and heart rate during low intensity activities. Measurements were made with the volunteer: sitting still, standing still and standing swaying with arms swinging. Hyperventilation may occur at rest when Douglas bags are used for expired air analysis, this may alter  $\dot{V}CO_2$  values but does not affect  $\dot{V}O_2$  therefore resting data cannot be used to calculate substrate utilisation but calculation of metabolic rate remains robust. Heart rate and expired air analysis was performed as described in section 2.6.1 – 2.6.2.

Volunteers were seated on a stool and were instructed to apply the nose-clips and breathe through the mouthpiece and two-way valve. To allow the volunteer to acclimatise to the environment, mouthpiece and nose-clips, a 5-minute period without expired air collection or heart rate monitoring was performed. A 5-minute expired air collection was then performed. During the last minute of the collection, heart rate was measured. Upon completion of this stage, the volunteer was asked to remove the nose clips and mouthpiece, and offered a drink of water.

Volunteers were then asked to stand up. The mouthpiece and nose-clips were introduced. A one minute run-in period was allowed to clear the mixed expired air and ambient air from the collection apparatus. A further 5-minute expired air collection was performed. Heart rate was recorded in the final minute.

Volunteers were asked to remain standing, transferring weight from one foot to another, with their arms swinging loosely by their side. The mouthpiece and nose-clips were introduced. A further one minute period was allowed to clear the collection tubing of mixed air, before a further 5-minute expired air collection with heart rate recording as described in section 2.6.1 -2.6.2.

## 2.6 Assessment of Cardiorespiratory Fitness

Upon completion of the low intensity assessments, volunteers had a 10 minute break at the end of which baseline capillary lactate was measured. The volunteers then performed an incremental submaximal treadmill walking test to assess cardiorespiratory fitness. Capillary lactate sampling was also performed to determine the lactate threshold and to help guide the

duration of the test. Measurements of heart rate and expired air were performed as described in sections 2.6.1 - 2.6.2.

Volunteers had been familiarised with the treadmill during their screening Bruce protocol exercise tolerance test. The same treadmill (Woodway PPS 55Med–I, Woodway GmbH, Weil am Rhein, Germany) was used for all submaximal exercise tests. Each test began with a 5 minute stage with the walking speed set at 3 km.h<sup>-1</sup>, with a 0% gradient. This stage allowed the volunteer to become familiar with the pattern of mouthpiece and nose-clip application, rate of perceived exertion (RPE) assessment and capillary lactate sample whilst walking on the treadmill.

A stopwatch was started once the volunteer began walking at 3 km.h<sup>-1</sup>. For the first two minutes of the stage, no expired air was collected. At the start of the third minute, the mouthpiece and nose-clips were introduced, and a one minute expired air collection into Douglas bags was started at the beginning of the 4<sup>th</sup> minute. A minimum of 15-20 seconds is recommended for expired air collection during exercise at a steady state (Macfarlane 2001). In the 5<sup>th</sup> minute of the stage, RPE and heart rate were recorded. RPE was recorded using the Borg Scale of 6 to 20, the subject was asked to point at the appropriate point on the scale, and this was confirmed verbally by an investigator (Borg 1973). At 5 minutes, the expired air collection was completed for that stage. The volunteer was assisted in removing nose-clips and mouthpiece and simultaneously, a second investigator took the capillary lactate sample.

Once the lactate sample had been obtained the next stage was commenced, and the treadmill was increased in speed to 5 km.h<sup>-1</sup>. Real-time capillary lactate analysis was performed during the test to allow detection of the lactate threshold (section 2.6.4). The volunteer continued to walk at 5 km.h<sup>-1</sup> for the duration of the test. Subsequent increases in workload were achieved

by a 2% increase in treadmill gradient with each 5 minute stage (Figure 2.8). Concentration of capillary lactate vs. workload was plotted on a graph, and the volunteer was judged to have breached their lactate threshold once an obvious deviation (> 1 mmol.l<sup>-1</sup>) had occurred from baseline concentrations.



Figure 2.8 Submaximal exercise test protocol.

The submaximal test was terminated once the investigators were convinced that the lactate threshold had been breached, the volunteer asked to stop, or an RPE of greater than 18 was reported.

## 2.6.1 Measurement of Heart Rate

Heart rate was measured during the establishment of the relationship between  $\dot{V}O_2$ ,  $\dot{V}CO_2$ and heart rate at rest, during low intensity activity and the submaximal exercise test using a short-range telemetry system, which utilised a chest wall sensor with transmitter and a watch device which received the signal, and displayed the heart rate (Polar s610i Heart Rate Monitor, Polar Electro Oy, Kempele, Finland). The Polar heart rate monitors were also used by volunteers during all exercise sessions in the intervention, and for monitoring of heart rate during periods of assessment of habitual physical activity. In the submaximal exercise test, heart rates were recorded every 15 seconds during the 5<sup>th</sup> minute of each stage of the test. An average of the 4 heart rate recordings was given as the heart rate for that stage. The predicted maximal heart rate was determined by the formula (Fox, III & Haskell 1968):

 $HR_{max} = 220 - Age$  (equation 2.12)

## 2.6.2 Expired Air Collection & Analysis

During the low intensity activity measurements and the submaximal exercise test, expired air was collected in Douglas bags, using standard Douglas bag technique (Consolazio CF 1963).

Analysis of expired air was performed in two stages:

Stage 1: 0.5 litres of air was extracted through the sampling port of the Douglas bag at a constant flow rate, controlled by a flow meter. This air was passed into a gas analyser (Servomex 4000 series, Servomex Group Limited, East Sussex, UK) and the percentage fraction of oxygen and carbon dioxide was measured.

Stage 2: The remaining air was extracted by a vacuum at a constant flow rate, through a dry gas meter (Harvard Apparatus Ltd, Kent, UK) which provided a measurement of the volume of expired gas, and through the dry gas meter's thermometer, which provided a measurement of the temperature of the expired air.

Prior to all tests, the gas analyser was calibrated against known reference gases (BOC Gases, BOC Limited, Surrey, UK) and barometric pressure was measured. All gas measurements were corrected to standard room temperature and pressure (STPD) for a dry gas. Values of  $\dot{V}$  O<sub>2</sub> and  $\dot{V}$ CO<sub>2</sub> were calculated from expired air using the Haldane transformation of the Fick equation as shown in equations 2.13 – 2.15 (Wilmore & Costill 1973).

 $\dot{V}I = \dot{V}E_{STPD} x (100 - Exp. Fraction O_2 - Exp. Fraction CO_2) / 79.04 Equation 2.13$ 

$$\dot{V}O_2$$
 (1.min<sup>-1</sup>) =  $\dot{V}I \ge 0.2093 - (\dot{V}E_{STPD} \ge Exp. Fraction O_2 / 100)$  Equation 2.14

 $\dot{V}CO_2 (l.min^{-1}) = (\dot{V}E_{STPD} x Exp. Fraction CO_2 / 100) - \dot{V}I x 0.0003$  Equation 2.15

## 2.6.3 Estimation of $\dot{VO}_{2max}$

Mean heart rate was plotted against calculated  $\dot{V}O_2$  for each stage of the submaximal exercise test. In all cases a linear relationship was observed between heart rate and  $\dot{V}O_2$  (equation 2.16). The equation of the line describing the relationship between heart rate and  $\dot{V}O_2$  was used with the predicted maximal heart rate (HR<sub>max</sub>) to calculate predicted  $\dot{V}O_{2max}$  (American College of Sports Medicine 1995).

$$HR = m(\dot{V}O_2) + c$$

$$VO_{2max} = (HR_{max} - c) / m$$
 (equation 2.16)

Measurement of maximal heart rate and  $\dot{VO}_{2max}$  was not performed because this study required continued commitment to the exercise programme and it was felt that an initial maximal test might have dissuaded some subjects from continued participation in the study. In addition, the validity of comparison of pre- and post-intervention maximal testing is more questionable when the group is sedentary at baseline since maximal workload may be falsely perceived (Church et al. 2001;WHO Consultation 1968).
#### 2.6.4 Capillary Lactate Sampling

Capillary lactate was measured in the resting state, prior to the test, and in the last minute of each exercise stage. The initial resting sample was taken after the volunteer's left hand had been immersed for ten minutes in a bath of water heated to 42°C, to ensure increased blood flow and arteriolisation of the sample. The thumb of the left hand was swabbed with an alcohol wipe (Uni-Wipe Alcotip Swabs, Sani System, Enfield, UK). Sampling was performed by digital puncture, using Accu-Chek Softclix lancets (Roche Diagnostics, Welwyn Garden City, UK). The initial blood droplet was wiped away, and a 30-50µL capillary blood sample was drawn into a pre-treated (Heparin/Fluoride/Nitrite) capillary tube (Analox Instruments Ltd, London, UK) by capillary attraction. Lactate was assayed using an Analox GM7 lactate analyser (Analox Instruments Ltd, London, UK).

Capillary lactate samples were taken during the submaximal exercise test. Sampling did not require repeated digital immersion in a water bath, due to increased peripheral blood flow induced by exercise. Samples were taken within one minute of the completion of expired air collection at each stage of the submaximal test. Volunteers were asked to rest their left hand on the edge of the treadmill rail and allow repeat alcohol swabbing, digital puncture and sample collection. Once an adequate sample had been obtained, the volunteer was given a tissue to prevent excessive bleeding.

#### **2.7 Exercise Intervention**

#### **2.7.1 Exercise Intensity and Frequency**

Volunteers used their heart rate monitors to guide the intensity of their exercise sessions. Volunteers were asked to complete all prescribed sessions exercising within a twenty beat heart rate range which was within 65-80% of their predicted maximum heart rate. This heart rate range is considered to represent a moderate intensity exercise prescription which promotes increased cardiorespiratory fitness (Pollock et al. 1998). Volunteers were asked to exercise for seven weeks, increasing the number and duration of sessions on a weekly basis. Week 1 of the regime consisted of 3 x 30 minute sessions, and the volunteers completed 5 x 60 minute sessions in weeks 6 and 7 (Table 2.1).

Week	<b>Duration</b> 1	No. Sessions	<b>Duration 2</b>	No. Sessions
1	30	3	-	-
2	30	2	45	2
3	30	3	45	2
4	45	5	-	-
5	45	3	60	2
6	60	5	-	-
7	60	5	-	-

**Table 2.1** Duration (minutes) and number of sessions for each week in the exercise intervention.

Both the duration and the heart rate achieved during each session were stored in the memory of the Polar s610i heart rate monitor. Volunteers met a study investigator on a weekly basis, this allowed infrared data transfer from the heart rate monitors to a laptop computer using Polar software (Polar Electro Oy, Kempele, Finland), providing a complete record of all sessions performed in the intervention. In addition, the weekly meeting provided the volunteer with the opportunity to ask questions about the study or exercise regime, identify any problems and arrange future visits. The investigator participated in an exercise session with the volunteer at the weekly meetings. This meeting allowed the investigator to supervise an exercise session and verify the volunteer's ability to participate in an exercise session for the given duration and at the given workload. The weekly meetings also maintained the focus and motivation of the volunteer. Each session was analysed for duration and mean heart rate achieved.

#### 2.7.2 Exercise Activities and Facilities

Volunteers were given free access to the Sports and Recreation Service facilities at Glasgow University for the duration of their involvement with the study. These facilities provided a focal point for exercise sessions and weekly meetings, and allowed volunteers access to a wide variety of exercise equipment, training areas, swimming pool and the opportunity to participate in organised classes. Volunteers were allowed to use whichever cardiovascular exercise equipment they preferred (e.g stepper, treadmill, cycle ergometer, rowing ergometer) or attend scheduled aerobic exercise classes. Alternatively, they were allowed to run, cycle or perform other modes of exercise at other locations, provided they completed the required duration and intensity of exercise.

To assist volunteers in planning their next week of exercise, an exercise planner (Appendix B2) was provided at each weekly meeting. The planner contained a reminder of their target heart rate, number and duration of sessions, and timing of the next weekly meeting. The planner also provided a diary facility to allow the volunteer to record the time of each session, and the exercise performed.

#### 2.8 Dietary Monitoring and Analysis

Participants were instructed to keep a food diary (Appendix B3) for seven consecutive days which involved weighing all food and drink on electronic scales (Salter Housewares, Kent, United Kingdom) and recording the weight and time of consumption in the diary. Diaries were completed prior to each metabolic test. Instructions were provided in addition to a visual demonstration by the researcher to show how to use the scales and the diary. The participants were advised to maintain their normal dietary intake. Volunteers were encouraged to weigh constituents of each meal and record the details of the food type and weight of food used. If a meal was not completely consumed, they were asked to weigh and record the remaining food constituents to allow investigators to calculate accurate consumption. In circumstances where it was not possible to weigh constituent components of a meal (e.g. a pre-packed sandwich), the volunteers were asked to record the weight of the total meal and provide a description of the meal. CompEat 5.0 (Nutrition Systems, Banbury, United Kingdom) was then used to analyse the food diaries and determine macronutrient and micronutrient intake for each subject.

# 2.9 Evaluation of habitual physical activity

For the seven day period prior to each metabolic test, volunteers participated in physical activity recording. Volunteers completed an activity diary (Appendix B4) during this period, and were asked to record all activities performed during waking hours. Detailed description of activities was not required, however the type of activity and duration was requested. Activity diaries are frequently used as the criterion measure in the validation of physical activity questionnaires (Pols et al. 1996) and appear to be a reliable method of performing repeated measurements of physical activity (Bouchard et al. 1983). Additionally, for waking hours, volunteers were asked to wear their heart rate monitors. The memory storage facility of the monitors was used to record a mean heart rate for each minute worn.

#### 2.9.1 Heart Rate Recording

Continuous daily heart rate recording provided a measurement of habitual physical activity. Activity diaries served as a validation tool for the accuracy of the heart rates recorded, and allowed the use of an estimated heart rate (based on the recorded activity) for periods where heart rate data was not successfully recorded. For example, close proximity to devices producing electromagnetic fields (e.g. laptop fans, induction loops) caused interference which served to cause a recorded heart rate of 230 beat.min<sup>-1</sup>. Activity diaries were useful in these situations to allow the use of an appropriate estimated heart rate for these periods.

#### 2.9.2 Evaluation of energy expenditure of exercise sessions

For each exercise training session the mean heart rate of the session was converted into an equivalent  $\dot{V}O_2$  based on the individualised heart rate versus  $\dot{V}O_2$  relationship. The individual's  $\dot{V}O_2$  associated with sedentary activity was subtracted from this value to determine the net exercise  $\dot{V}O_2$  (i.e. the oxygen cost of exercise over and above that of sedentary daily activities). The net exercise  $\dot{V}O_2$  (1.min<sup>-1</sup>) was then multiplied by the duration of exercise to determine the total net oxygen cost of the exercise session, and the net energy cost of the session (in kJ) was obtained by multiplying the net oxygen cost by 20.3 (Frayn & Macdonald I.A. 1997). For exercise sessions performed during the first 3.5 weeks of the programme, energy expenditure values were determined using the heart rate versus  $\dot{V}O_2$  relationship from the baseline fitness test; for sessions during the second 3.5 weeks of the programme, values obtained during the post-intervention fitness test were used in the energy expenditure calculations. The net energy expenditure of all exercise sessions was summed to determine the total net energy expenditure of the exercise training programme.

#### 2.9.3 FLEX Pilot Study

The 'FLEX' method of energy expenditure calculation utilises free-living measurements of minute-by-minute heart rates to estimate total daily energy expenditure (Ceesay et al. 1989;Spurr et al 1988;Spurr 1990). This method compares well (Spurr et al. 1988;Livingstone et al. 1990;Leonard 2003) with the 'gold-standard' method of energy expenditure calculation using doubly-labelled water (Ainslie, Reilly, & Westerterp 2003;Klein et al. 1984). The FLEX method is less expensive and less technically complicated than the doubly-labelled water method (Schoeller & Webb 1984;Spurr et al. 1988) and takes into account the loss of the linear relationship between heart rate and  $\dot{VO}_2$  (and thus energy expenditure) at low intensity heart rates (Montoye HJ 1996) (Figure 2.9).



**Figure 2.9**. **A** Oxygen consumption determined by heart rate<sup>3</sup> showing non-linear relationship. **B** Separation of 'active' and 'non-active' heart rates by observed activity, and corresponding determination of threshold above which linear relationship between HR and  $\dot{V}$  O<sub>2</sub> applies, and below which a non-linear relationship exists. Dotted red line indicates inflexion point (FLEX) between active and inactive heart rates.

Determination of the FLEX point requires accurate measurement of heart rates at rest and during sedentary activities. It was observed that the difference in resting heart rate before and after the exercise intervention was considerably greater when determined during the FLEX measurements (i.e. immediately prior to exercise, as described in section 2.5.1), than when measured using the Complior method (COMP), or from the lowest heart rate values recorded during the Continuous Heart Rate Monitoring weeks (CHRM) (Figure 2.10). This discrepancy between pre- and post-intervention FLEX measurements was greater than would have been expected as a results of improved cardiorespiratory fitness, and therefore questioned the accuracy of the heart rates measured during determination of the FLEX point.



**Figure 2.10**. Difference between pre- and post-intervention resting heart rate measured by: FLEX measurements, Complior and Continuous Heart Rate Monitoring (CHRM). \*p < 0.001 compared with change by FLEX measurement.

# 2.9.3.1 Methodology for the evaluation of the FLEX Method

To determine the impact of a possible exaggerated difference in pre- and post-intervention FLEX heart rate values on calculated energy expenditure, a pilot study was performed using the data from the first fifteen volunteers to complete pre- and post-intervention testing. The FLEX method defined a heart rate threshold (the 'FLEX' point) above which was categorised as 'active', and below which was categorised as 'inactive'. For 'active' heart rates, the

relationship between heart rate and oxygen consumption was assumed to be valid, and therefore active energy expenditure (Active EE) was calculated based on an individual's oxygen consumption and heart-rate relationship derived from an exercise test (Section 2.6). For sleep periods, energy expenditure was given as equivalent to resting metabolic rate (RMR) (Section 2.3.2). For waking periods at  $\leq$  FLEX, so-called 'sedentary energy expenditure' was a mean value calculated as the average energy expenditure for the sedentary stages (lying down, sitting, standing) during an active-inactive calibration (FLEX EE) (Ceesay et al. 1989;Spurr et al. 1988). The FLEX point was an individually determined heart rate which was defined as the average of the highest resting heart rate, and lowest exercise heart rate recorded during the assessment of heart rate and  $\dot{VO}_2$  during rest and exercise (Section 2.8 – 2.9) (Ceesay et al. 1989;Spurr et al. 1988;Spurr 1990).

Energy expenditure was calculated using the equations of indirect calorimetry (section 2.3.3), where total daily energy expenditure was composed of:

- 1. Sleeping energy expenditure ((1440 number of minutes recorded) x RMR)
- 2. Sedentary energy expenditure (number of minutes  $\leq$  FLEX) x FLEX EE

3. Active energy expenditure (number of minutes > FLEX) x Active EE To adjust for the observed difference in resting heart rates, energy expenditure was recalculated using a variety of adjusted FLEX heart rates. Pre-intervention values of daily energy expenditure were compared to post-intervention values calculated using:

- 1. Post-intervention FLEX heart rate derived at post-intervention re-assessment.
- 2. Pre-intervention FLEX heart rate (FLEX 1).
- Pre-intervention FLEX heart rate adjusted by the difference between pre- and postintervention resting heart rate as calculated by continuous heart rate monitoring (FLEX CHRM).

 Pre-intervention FLEX heart rate adjusted by the difference between pre- and postintervention resting heart rate as calculated by Complior measurement. (FLEX COMP).

# 2.9.3.2 Results

By using the post-intervention values derived by the FLEX method, and after adjustment for the energy cost of the exercise sessions in the monitoring week, a significant increase in energy expenditure of  $321.9 \pm 107.6$  kcal.day<sup>-1</sup> was noted (p < 0.01). However, using adjusted values of FLEX did not reveal any difference in habitual energy expenditure (Table 2.2).

Method of EE assessment Value (kcal.day<sup>-1</sup>) p-value Pre-Intervention FLEX EE 2396 Post-Intervention FLEX EE 2718 < 0.01 Post-Intervention FLEX 1 NS 2409 Post-Intervention FLEX COMP 2501 NS Post-Intervention FLEX CHRM NS 2462

**Table 2.2** Estimated total daily energy expenditure at baseline and post-intervention.

Post-intervention values adjusted using baseline FLEX point (FLEX 1), pre-intervention FLEX heart rates adjusted as described. P-values calculated comparing pre-intervention FLEX EE with post-intervention values using repeated measures ANOVA with Tukey HSD *post-hoc* analysis.

#### 2.9.3.3 Discussion

In summary, discrepancies between those resting heart rate measurements made during the active-inactive calibration and those made during both the Complior measurements and CHRM recordings suggested that the calculated FLEX heart rates might be incorrect. Resting heart rates measured using the CHRM method are representative of free-living values and Complior resting heart rate is measured after a rest period of 30 minutes in an environment with minimal external stimuli, these are likely to be more accurate measurements of true resting heart rate. This was borne out by the significant difference observed between calculated energy expenditure pre- and post-intervention using the FLEX heart rates, which was not observed using heart rates which were adjusted for more realistic and physiological changes in resting heart rates.

It is unclear why there were such differences in resting heart rates measured during the determination of the relationships between heart rate and  $\dot{V}O_2$  during rest and exercise. The FLEX measurements were amongst the first tests in which volunteers participated. The initial exposure to a new laboratory environment, and to unfamiliar technical staff and procedures could have been a more stressful experience than anticipated, leading to falsely elevated resting heart rates at this time. This pilot study suggests that in this case the FLEX heart rates calculated prior to, or after the exercise intervention (or both) were not accurate representations of the 'true' FLEX point(s) and subsequently estimated energy expenditure values were not deemed to be valid. It was therefore decided not to use the FLEX method to determine energy expenditure in this study. Further attempts to use the FLEX method to determine energy expenditure may require volunteers to be 'familiarised' with the laboratory environment and for resting heart rate measurements to be validated against established methods (Voors, Webber, & Berenson 1982).

#### 2.10 Indices of physical activity

In view of the difficulties with the FLEX method, an alternative index of habitual activity was used. The problems with the FLEX technique were apparent only after the majority of the study had been conducted and there was no opportunity to design an alternative method of calculating habitual energy expenditure. Instead, indices of physical activity based on heart rate observed during the continuous seven day monitoring period were used. Average daily heart rate was calculated from heart rate data obtained during the waking hours. Additionally, resting heart rate was determined as the lowest heart rate recorded during waking hours. Indices of physical activity were determined by calculating the number of minutes spent at multiples of resting heart rate. Time spent at heart rates greater than resting multiplied by 1.5 and 2 were used as markers of light and moderate physical activity, respectively. Thus, given a typical resting heart rate of 70 beat.min<sup>-1</sup>, light activity would be equivalent to a heart rate of around 105 beat.min<sup>-1</sup>, and moderate activity at a heart rate of around 140 beat.min<sup>-1</sup>. Average daily heart rate minus resting heart rate (i.e. the time averaged area under the heart rate versus time curve, using resting heart rate as baseline) was used as a surrogate measure of total activity.

#### **2.11 Power Calculation**

Based on reports in the literature, the within-subject reproducibility coefficient of variation for OGTT-assessed insulin resistance is approximately 8% (Mari et al. 2001); the standard deviation for between-subject variation in insulin resistance in normoglycaemic individuals is approximately 25-40% (Lihn et al. 2003;Perseghin et al. 1997) ; offspring of patients with type 2 diabetes are approximately 40-50% more insulin resistant than matched control subjects (Higgins et al. 2005;Humphriss et al. 1997;Lihn et al 2003) and the planned exercise intervention would be expected to improve insulin sensitivity by approximately 20-30% (Borghouts & Keizer 2000). Assuming the between-subject standard deviation for insulin resistance is 40%, 80 volunteers (40 control, 40 offspring) was expected to enable detection of a 25% difference in insulin resistance between the control and offspring groups at  $\alpha = 0.05$ 

118

and 80% power. Previous studies have shown that this number of volunteers is sufficient to show a difference in insulin resistance between control and offspring subjects at the p = 0.001level (Higgins et al. 2005). Twenty subjects in each group (Offspring Exercise, Offspring Delayed Exercise, Control Exercise, Control Delayed Exercise) was expected to allow detection of a 5% change in insulin resistance with exercise and a 5% difference in the change in insulin resistance with exercise between the control and offspring groups with the same power.

#### 2.11.1 Statistical analysis

All statistical analysis were performed using Statistica (version 6.0, StatSoft Inc., Tulsa, Oklahoma) and Minitab (version 13.1, Minitab Inc., State College, Pennsylvania). Differences between the Offspring and Control groups at baseline were assessed by unpaired t-tests. Exercise-induced effects were assessed by comparison of the change in variables (i.e. post-intervention minus pre-intervention) among the groups by 2-way ANOVA (family history x intervention) with *post-hoc* Tukey tests. Determinants of change in insulin sensitivity were assessed by univariate linear correlations. Multiple regression analyses were performed to establish the independence of relationships. Anderson-Darling normality testing was performed on all data sets, and logarithmic transformation was performed (where necessary), prior to analysis.

#### 2.11.2 Index of Insulin Sensitivity calculation

In all experimental chapters insulin sensitivity was measured using the Insulin Sensitivity Index (ISI) devised by Matsuda and DeFronzo (1999). This utilises the glucose and insulin values for each time point of an oral glucose tolerance test and is highly correlated with the rate of whole-body glucose disposal during a clamp (Matsuda & DeFronzo 1999). The ISI was calculated using equation 2.17.

# ISI = 10000 / $\sqrt{$ [(Fasting Glucose x Fasting Insulin) x (Mean Glucose during OGTT x Mean Insulin during OGTT)] **Equation 2.17**

Mean glucose and insulin concentrations during OGTT were calculated as the average of the 30, 60, 90 and 120 minute values. For the purposes of equation 2.17, glucose was measured in mg.dl<sup>-1</sup> and insulin in  $\mu$ U.ml<sup>-1</sup>. Based on data from the two baseline measurements made in the volunteers randomised to the 'Delayed Exercise' arm, the within-participant test-retest coefficient of variation for ISI was 12.2%, with mean values for ISI differing by 3.2% between the two baseline measurements.

#### **CHAPTER 3**

# PHYSICAL AND METABOLIC CHARACTERISTICS OF SEDENTARY WOMEN WITH AND WITHOUT A FAMILY HISTORY OF TYPE 2 DIABETES

# **3.1 Introduction**

First degree relatives of patients with type 2 diabetes have a greater risk of developing diabetes than their counterparts with no family history of diabetes (Kobberling J & Tillil H 1982;Ohlson et al. 1988). Insulin resistance is the best risk predictor for the development of diabetes (Ferrannini 1998;Laws & Reaven 1993), which is consistent with the observation that healthy first degree relatives of type 2 diabetic patients (relatives) are often more insulin resistant than individuals without a family history of diabetes (controls) (Nyholm et al. 2004; Perseghin et al. 1997; Tesauro et al. 2007; Higgins et al. 2005; Ahn et al. 2004; Humphriss et al. 1997;Nyholm et al. 1996;Ostergard et al. 2006;Johanson et al. 2003). In some of the studies (Humphriss et al. 1997; Higgins et al. 2005) impaired insulin sensitivity observed in relatives could be explained by higher measures of adiposity, a central contributing factor to insulin sensitivity (Chang et al. 2004;Salmenniemi et al. 2005;Kriketos et al. 2004). However in other studies the differences in insulin sensitivity were evident despite no difference in body mass or total body fat between relative and control groups (Nyholm et al. 1997; Johanson et al. 2003; Nyholm et al. 2004; Ostergard et al. 2006). Similar observations have been made in other groups (e.g. South Asians, American Pima Indians) who display an increased predisposition to type 2 diabetes in comparison to Caucasians with comparable adiposity (McCance et al. 1994; Razak et al. 2007; Chandalia et al. 2007). This suggests that, similar to these other groups, insulin sensitivity in relatives of patients with type 2 diabetes may be influenced by adiposity to a greater degree than in individuals without family history of diabetes. One aim of the current study was therefore to explore the influences of adiposity on insulin sensitivity in this group, using a robust assessment of adiposity.

In addition to differences in adipose tissue mass and distribution, relatives of patients with type 2 diabetes have been reported to have altered adipose tissue function, characterised by higher circulating NEFA (Perseghin et al. 1997;Lattuada et al. 2005) and pro-inflammatory cytokines such as IL-6, TNF $\alpha$  (Maltezos et al. 2002;Straczkowski et al. 2002) and reduced concentrations of insulin-sensitising cytokines such as adiponectin (Lattuada et al. 2005;Tesauro et al. 2007). In the presence of insulin resistance, altered vascular endothelial function has been observed (Scuteri et al. 2008;Tesauro et al. 2007;McEleavy et al. 2004), which may predispose to premature cardiovascular disease. A further aim of this study was therefore to explore the interaction between adipose tissue function, vascular function and insulin resistance using measurements of circulating adipokines as a proxy measure of adipose tissue function and Pulse Wave Velocity (PWV) as a marker of vascular function.

There is some evidence to suggest that both physical activity and diet play an important role in determining differences in insulin sensitivity seen between those with and without a family history of T2D. For example, it has been shown that early metabolic abnormalities of offspring of type 2 diabetic patients are partially ameliorated by participation in regular aerobic exercise (Ahn et al. 2004). In addition, Higgins and colleagues (2005) reported lower levels of habitual physical activity in relative subjects compared to control subjects and observed a stronger relationship between physical activity and insulin sensitivity in the relative group (Higgins et al. 2005). However, as Higgins and colleagues used self-report measures for the physical activity assessment, which have limited reliability (Klesges et al. 1990), and assessed insulin resistance by HOMA, using fasting glucose and insulin concentrations only (Matthews et al. 1985), the data may not be entirely robust and require confirmation. Using more robust measurements of both insulin sensitivity and habitual activity, the current study aimed to investigate the notion that insulin sensitivity in relatives may be modulated to a greater degree by physical activity, than in control volunteers.

122

To date only three studies have made an attempt to investigate differences in dietary intake between relatives and control individuals (Adamson et al. 2001; Johanson et al. 2003; Higgins et al. 2005), and only two studies have investigated the role of diet in determining differences in insulin sensitivity in these groups (Johanson et al. 2003;Higgins et al. 2005). Two existing studies reported that there were no significant differences in daily energy and macronutrient intake or the proportion of energy provided by the main nutrients between relative and control groups (Higgins et al. 2005; Johanson et al. 2003). Furthermore, Higgins and colleagues (2005) showed that neither the control nor the relative group's dietary factors correlated with insulin sensitivity (Higgins et al. 2005). Contrary to this, Adamson and coworkers (2001) observed that non-diabetic relatives of type 2 diabetic patients consumed diets which are expected to promote the development of diabetes (Adamson et al. 2001). Indeed, in this study relatives were reported to have higher absolute intakes of total fat, saturated fat and cholesterol and lower intake of non-starch polysaccharides (Adamson et al. 2001). In both studies, relatives were heavier than controls, therefore the current study aimed to examine the role of dietary factors in determining insulin sensitivity in offspring of people with type 2 diabetes, and matched controls.

Therefore, aims of this chapter were: firstly, to determine differences in the metabolic phenotype between sedentary females with and without a family history of type 2 diabetes; secondly, to explore the interaction between adipose tissue function, inflammation and both insulin resistance and vascular function in this group; additionally, to determine the extent by which adiposity and lifestyle factors, such as physical activity, contribute to the insulin resistant phenotype observed in those with a family history of diabetes; finally, to ascertain whether effects of adiposity and lifestyle factors on insulin sensitivity differ between those with, and without a family history of the disease.

#### 3.2 Methods

#### 3.2.1 Volunteers

Volunteers were recruited and screened as detailed in section 2.1. Specific inclusion criteria are defined in section 2.1.3. Further information detailing recruitment response and excluded volunteers can be found in Figure 2.2. The Offspring group was composed of women with a first-degree relative with type 2 diabetes. The Control group was composed of women with no first or second degree relative with type 2 diabetes. This study was conducted on thirty four Offspring and thirty six Control volunteers. Control and Offspring groups were well matched for age (Controls  $33.6 \pm 6.1$  years, Offspring  $35.6 \pm 7.0$  years, p = 0.20) and BMI (Controls  $27.3 \pm 4.7$  kg.m<sup>-2</sup>, Offspring  $28.1 \pm 5.1$  kg.m<sup>-2</sup>, p = 0.50).

# 3.2.2. Study Design

All participants underwent assessment of physical characteristics, body composition, cardiorespiratory fitness and participated in metabolic testing as detailed in Sections 2.2-2.3 and 2.6 respectively. They were also asked to complete a diary of habitual physical activity and food intake. In addition they were asked to participate in continuous waking hours heart rate monitoring prior to which the individual relationships of heart rate to  $\dot{V}O_2$ , and  $\dot{V}CO_2$  under resting and exercise conditions were established.

#### 3.2.3 Energy and macronutrient intake

For a seven day period prior to metabolic investigation, volunteers were asked to keep a record of a food diary for seven consecutive days as described in Section 2.8.

# 3.2.4 Establishment of relationships of heart rate to $\dot{V}O_2$ , and $\dot{V}CO_2$

Individual relationships between heart rate,  $\dot{V}O_2$  and  $\dot{V}CO_2$  were established during both rest and exercise as described in Sections 2.5 and 2.6 respectively.

# 3.2.5 Evaluation of habitual physical activity

Heart rate measurement throughout the day (as described in section 2.6.1) was used to assess indices of habitual physical activity (Section 2.10). Assessment of time spent at heart rates which were 1.5 and 2 times greater than resting heart rate were considered to be represent approximate thresholds for light and moderate activity respectively (Section 2.10). Average daily heart rate minus resting heart rate was used as a surrogate measure of total activity (Section 2.10).

#### 3.2.6 Evaluation of cardiorespiratory fitness

Cardiorespiratory fitness was measured by predicted  $\dot{V}O_{2max}$  and lactate threshold, as described in section 2.6.

#### **3.2.7** Anthropometric assessment

Measurements of body mass, height, skinfold thickness, waist and hip circumference were made as described in Section 2.2. Additional assessment of body composition was made using DEXA scanning, as described in Section 2.2.5.

# **3.2.8 Metabolic testing**

As described in Section 2.3, volunteers attended the Metabolic Investigation Suites at the Institute of Diet, Exercise and Lifestyle (IDEAL) after a 12-hour overnight fast. Volunteers had fasting measurements of resting metabolic rate, substrate utilisation and peripheral PWV (see Sections 2.3.2, 2.3.5 respectively) which was followed by fasting blood sampling (Section 2.3.6). After fasting samples were taken, the volunteers proceeded to subcutaneous abdominal adipose tissue liposuction biopsy (Appendix B1, data not presented in this thesis). On completion of the biopsy process, volunteers commenced a 75 g oral glucose tolerance test (section 2.3.7) with further blood samples were taken at 30, 60, 90 and 120 minutes after glucose ingestion. Samples were processed and analysed as described in Section 2.4. Upon completion of the glucose tolerance test, the volunteers had repeat PWV assessment and measurements of substrate utilisation and metabolic rate (Figure 2.5).

#### 3.2.9 Blood analysis

After centrifugation all plasma samples were stored at -80°C in labelled 2 ml Eppendorf tubes (Alpha Laboratories, Hampshire, UK). Glucose, NEFA, 3-OHB, ALT, GGT, total cholesterol, HDL-cholesterol, TG and CRP were all measured by spectrophometric techniques as described in Section 2.4.2. LDL-cholesterol was calculated by the Friedewald equation (Friedewald, Levy, & Fredrickson 1972). Insulin, adiponectin, resistin, leptin, TNFα, IL-6 were measured by ELISA technique as described in section 2.4.3.

# **3.2.9.1** Calculation of glucose, insulin, NEFA, and 3-OHB concentrations and insulin sensitivity during oral glucose tolerance test

The mean concentration of glucose, insulin, NEFA and 3-OHB during the oral glucose tolerance test (OGTT) was calculated by dividing the area under variable *vs.* time curve (total AUC) by the duration of the OGTT (mean during OGTT). The rise (or suppression) of glucose, insulin, NEFA and 3-OHB was calculated by subtracting the area under the fasting concentration vs. time curve from the total AUC and dividing the obtained value by the duration of the OGTT (rise during OGTT). For variables where values decreased from baseline after glucose ingestion (e.g. NEFA), the rise during OGTT was negative and represented the area above the curve extending up to the baseline value, providing an index of post-glucose suppression. Insulin sensitivity was calculated using the Insulin Sensitivity Index (ISI) as described by Matsuda and DeFronzo (Matsuda & DeFronzo 1999). This equation is outlined in section 2.11.2.

## 3.2.9.2 Statistical analysis

Data were analysed using Statistica (version 6.0, StatSoft Inc., Tulsa, Oklahoma) and Minitab (version 13.1, Minitab Inc., State College, Pennsylvania). Prior to analysis, all data were tested for normality using the Anderson-Darling normality test and, if necessary, logarithmically transformed. All data between Offspring and Control groups was compared using unpaired t-tests. Univariate linear regressions were performed, using logarithmically transformed data where required. Correlations between ISI and other variables were performed for the Offspring and Control groups. To determine the extent to which variables were related to ISI independently of fat mass, univariate linear regressions were then performed between the residuals for ISI of the regression between ISI and total fat mass. This classic statistical approach has been used previously to determine the independent effects of individual variables within multi-factorial systems on biological outcomes (Speakman et al. 2002;Westerterp & Speakman 2008) and, in effect, provides the correlations between ISI and other variables, adjusted for the effect of total fat mass. Significance was accepted at p  $\leq$  0.05.

# 3.3 Results

# 3.3.1 Concentrations of glucose, insulin, non-esterified fatty acid and 3-hydroxybuytrate

Fasting concentrations of plasma glucose, insulin, non-esterified fatty acid (NEFA) and 3hydroxybutyrate (3-OHB) are presented in Table 3.1 and Figure 3.1. No significant differences between Controls and Offspring were found for glucose, NEFA or 3-OHB but Offspring had significantly higher fasting insulin concentrations compared to Controls (Table 3.1, Figure 3.1). Mean concentrations of glucose, insulin, NEFA and 3-OHB during the OGTT are presented in Table 3.1 and Figure 3.1. Mean concentration of glucose, insulin and NEFA were not significantly different between groups whilst mean concentration of 3-OHB was significantly higher (p < 0.01) in Offspring compared to Controls. Mean change of glucose, insulin, NEFA and 3-OHB concentration during OGTT did not differ between Control and Offspring groups.

	Controls	Offspring	P - value
Glucose			
Fasting (mmol.l <sup>-1</sup> )	$4.8\pm0.5$	$4.9\pm0.7$	0.52
Mean during OGTT (mmol.1 <sup>-1</sup> )	$6.4 \pm 1.3$	$7.0 \pm 1.5$	0.09
Change from fasting during OGTT (mmol.1 <sup>-1</sup> )	$1.6 \pm 0.2$	$2.1\pm0.2$	0.11
Insulin			
Fasting <sup>a</sup> (mU. $l^{-1}$ )	$6.2 \pm 3.8$	$7.9\pm4.5$	< 0.05
Mean during OGTT <sup>a</sup> (mU. 1 <sup>-1</sup> )	$50.4\pm35.5$	$61.8\pm38.9$	0.11
Change from fasting during OGTT <sup>a</sup> (mU.l <sup>-1</sup> )	$44.2\pm5.4$	$53.9\pm6.0$	0.12
NEFA			
Fasting (mmol.l <sup>-1</sup> )	$0.52\pm0.18$	$0.55\pm0.17$	0.39
Mean during OGTT <sup>a</sup> (mmol.l <sup>-1</sup> )	$0.25\pm0.11$	$0.26\pm0.07$	0.28
Change from fasting during OGTT (mmol.1 <sup>-1</sup> )	$\textbf{-}0.27\pm0.02$	$\textbf{-}0.29\pm0.02$	0.41
3-ОНВ			
Fasting <sup>a</sup> (mmol.1 <sup>-1</sup> )	$0.06\pm0.05$	$0.09\pm0.07$	0.16
Mean during OGTT (mmol.1 <sup>-1</sup> )	$0.04\pm0.004$	$0.07\pm0.05$	< 0.01
Change from fasting during OGTT (mmol.1 <sup>-1</sup> )	$-0.02 \pm 0.01$	$-0.01 \pm 0.01$	0.11

**Table 3.1** Glucose, insulin, non-esterified fatty acid (NEFA) and 3-hydroxybutyrate (3-OHB) concentrations in the fasted state and during OGTT. Values are mean ± SEM.

P-values are for unpaired *t*-tests between groups. <sup>a</sup>statistical analysis performed on logarithmically transformed data.



**Figure 3.1** Plasma glucose (**a**), insulin (**b**), NEFA (**c**) and 3-OHB (**d**) concentrations for Offspring and Control in the fasting state and during the OGTT. Values are mean ± SEM.

#### 3.3.2 Insulin sensitivity

The Insulin Sensitivity Index (ISI) (Matsuda & DeFronzo 1999) of Offspring was 22% lower (p = 0.047) than Controls (Figure 3.2).



**Figure 3.2** ISI in Control and Offspring. Values are Mean  $\pm$  SEM. P-value is for unpaired *t*-tests between groups. \*significantly different (p < 0.05) from Control group.

Fasting concentrations of non-esterified fatty acids (NEFA) and 3-hydroxybutyrate (3-OHB) were not correlated with ISI in either group, either before or after adjustment for fat mass. In Offspring, mean concentration of NEFA during the OGTT was significantly negatively associated with ISI (r = -0.42, p < 0.05), but only after adjustment for the effects of fat mass.

# 3.3.3 Physical characteristics and body composition

There were no differences between groups for any index of body composition (Table 3.2).

	Controls	Offspring	P - value
	( <b>n=36</b> )	(n=34)	
Body mass (kg)	$73.0\pm14.8$	$76.6 \pm 15.1$	0.33
Total fat mass (kg)	$28.9 \pm 10.1$	$31.3\pm10.2$	0.33
Trunk fat mass (kg) <sup>a</sup>	$14.2 \pm 5.3$	$16.3 \pm 5.6$	0.13
Leg fat mass (kg) <sup>a</sup>	$10.0\pm3.2$	$10.3 \pm 3.2$	0.76
Arm fat mass (kg) <sup>a</sup>	$3.6 \pm 1.6$	$3.9 \pm 1.9$	0.58
Android fat mass (kg)	$2.3\pm1.0$	$2.7 \pm 1.2$	0.12
Gynoid fat mass (kg)	$5.2 \pm 1.4$	$5.5 \pm 1.3$	0.91
Total lean mass (kg)	$42.1 \pm 5.6$	$42.6\pm6.0$	0.69
Trunk lean mass (kg)	$20.9\pm3.0$	$21.3 \pm 3.2$	0.59
Leg lean mass (kg)	$13.8 \pm 2.1$	$14.0 \pm 2.3$	0.64
Arm lean mass (kg) <sup>a</sup>	$4.2\pm0.8$	$4.4\pm0.9$	0.39
Upper Body Fat (kg) <sup>a</sup>	$17.7\pm6.8$	$20.2\pm7.3$	0.18
Lower Body Fat (kg) <sup>a</sup>	$10.0\pm3.2$	$10.3 \pm 3.2$	0.76
Waist Circ (cm) <sup>a</sup>	$84.4 \pm 11.6$	89.1 ± 13.1	0.12
Hip Circ (cm) <sup>a</sup>	$106.7\pm9.7$	$109.0\pm10.1$	0.33
Biceps Skinfold (mm) <sup>a</sup>	$14.7 \pm 5.7$	$16.8 \pm 8.3$	0.33
Triceps Skinfold (mm) <sup>a</sup>	$24.0\pm 6.6$	$23.5\pm7.3$	0.55
Subscapular Skinfold (mm)	$23.2\pm8.9$	$23.7 \pm 7.1$	0.78
Suprailiac Skinfold (mm) <sup>a</sup>	$23.9\pm8.0$	$27.0 \pm 9.4$	0.12

Table 3.2 Physical characteristics and body composition of Control and Offspring.

Values are Mean  $\pm$  SD. P-values are for unpaired *t*-tests between groups. <sup>a</sup>statistical analysis performed on logarithmically transformed data.

Total fat mass was significantly negatively correlated with ISI in both Controls and Offspring and correlated more strongly with ISI in Offspring than Controls (Figure 3.3). In Offspring alone lean mass was also negatively correlated with ISI, (r = -0.614, p < 0.001) with consistent negative association in all regional measurements of lean mass (trunk: r = -0.601, p < 0.001; leg: r = -0.534, p < 0.01; arm: r = -0.650, p < 0.001). In the Control group, most indices of adiposity were negatively associated with ISI (Table 3.3). Offspring showed stronger and more consistent association between adiposity and ISI than Controls, exhibiting significant negative correlation with all indices of adiposity. After adjustment for the influence of fat mass, neither Controls nor Offspring showed any significant correlation between indices of adiposity and ISI.

**Figure 3.3** Scattergrams illustrating the correlation of fat mass with ln ISI in Control and Offspring groups.



		Controls		Offspi	ring
		Unadjusted	Adjusted	Unadjusted	Adjusted
Biceps skinfold <sup>a</sup>	r	-0.31	0.12	-0.54**	-0.14
Triceps skinfold <sup>a</sup>	r	-0.26	0.23	-0.45**	-0.09
Subscapular skinfold	r	-0.54***	-0.14	-0.45**	-0.02
Suprailiac skinfold <sup>a</sup>	r	-0.45**	-0.03	-0.61***	-0.28
Waist – hip ratio	r	-0.51**	-0.23	-0.58***	-0.27
BMI	r	-0.37*	0.23	-0.72***	-0.26
Waist circumference <sup>a</sup>	r	-0.50**	0.05	-0.65***	-0.23
Hip circumference <sup>a</sup>	r	-0.32	0.26	-0.62***	-0.17
Trunk fat mass <sup>a</sup>	r	-0.45**	0.15	-0.72***	-0.27
Leg fat mass <sup>a</sup>	r	-0.30	0.30	-0.58***	-0.12
Arm fat mass <sup>a</sup>	r	-0.39*	0.20	-0.62***	-0.14
Upper body fat <sup>a</sup>	r	-0.44**	0.16	-0.72***	-0.24
Lower body fat <sup>a</sup>	r	-0.30	0.30	-0.58***	-0.12

**Table 3.3** Correlations between ISI and indices of adiposity for Control (n = 36) and Offspring (n = 34) groups.

r = Univariate correlations of variable vs. ISI. Unadjusted = variable correlated against ISI. Adjusted = variable correlated against residual of ISI and fat mass. <sup>a</sup>statistical analysis performed on logarithmically transformed data. Statistically significant correlation shown in bold. <sup>\*</sup> p < 0.05, <sup>\*\*</sup> p < 0.01, <sup>\*\*\*</sup> p < 0.001.

#### 3.3.4 Metabolic rate & substrate utilisation

No differences were observed between Offspring and Controls in resting metabolic rate (RMR), respiratory exchange ratio (RER), or rates of fat and carbohydrate oxidation measured in the fasting state (Table 3.4). Furthermore, no significant differences between Controls and Offspring were found in metabolic rate, RER, rate of fat oxidation or carbohydrate oxidation measured 2 hours after the administration of a 75 g glucose load (Table 3.4). The magnitude of the change (value at 2 hours after glucose load minus fasting value) in metabolic rate, RER, fat and carbohydrate oxidation rates was not significant different between the groups (data not shown).

**Table 3.4** Metabolic rate, respiratory exchange ratio (RER), rates of fat and carbohydrate(CHO) oxidation in Controls and Offspring in the fasted state and 2 hours after glucose load.

	Controls	Offspring	P - value
	( <b>n=36</b> )	( <b>n=34</b> )	
Resting metabolic rate (kJ.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.06 \pm 0.01$	$0.06 \pm 0.01$	0.43
Post-glucose metabolic rate (kJ.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.06\pm0.007$	$0.06\pm0.008$	0.29
Fasting RER	$0.86\pm0.05$	$0.85\pm0.05$	0.66
Post-glucose RER	$0.95\pm0.04$	$0.95\pm0.05$	0.67
Fasting fat oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.47\pm0.33$	$0.50 \pm 0.26$	0.68
Post-glucose fat oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.07\pm0.18$	$0.04 \pm 0.23$	0.56
Fasting CHO oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$1.85 \pm 0.91$	$1.69 \pm 0.68$	0.42
Post-glucose CHO oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$3.01 \pm 0.69$	$2.95 \pm 0.80$	0.72

Values are mean  $\pm$  SD. P-values are for unpaired *t*-tests between groups.

# 3.3.5 Blood pressure measurement and peripheral pulse wave velocity

No significant differences in fasting systolic or diastolic blood pressure were observed (Table 3.5). In addition, no differences in pulse wave velocity were observed between Controls and Offspring in either the fasted or post-glucose state (Table 3.5).

**Table 3.5** Fasting systolic and diastolic blood pressure and peripheral pulse wave velocity(PWV) in fasting and at 2 hours after glucose load.

	Controls	Offspring	P - value
	(n=36)	(n=34)	
Systolic blood pressure (mmHg)	$117.4 \pm 12.7$	$120.3 \pm 14.3$	0.37
Diastolic blood pressure (mmHg)	72.1 ± 8.9	$76.4 \pm 11.1$	0.07
Fasting PWV (m.s <sup>-1</sup> ) <sup>a</sup>	9.5 ± 1.5	$9.7 \pm 1.6$	0.79
Post-glucose PWV (m.s <sup>-1</sup> ) <sup>a</sup>	9.4 ± 1.3	9.6 ± 1.6	0.57

Values are mean  $\pm$  SD. P-values are for unpaired *t*-tests between groups. <sup>a</sup>Statistical analysis performed on logarithmically transformed data.

# 3.3.6 Plasma lipids

No significant differences were found in total cholesterol, LDL-cholesterol, HDL cholesterol or triglycerides between Offspring and Controls (Table 3.6).

	Controls	Offspring	p-value
	( <b>n=36</b> )	(n=34)	
Total cholesterol (mmol.l <sup>-1</sup> )	$4.23 \pm 1.04$	$4.17 \pm 1.06$	0.80
Fasting triglycerides (mmol.l <sup>-1</sup> ) <sup>a</sup>	$0.86 \pm 0.30$	$0.98 \pm 0.36$	0.14
HDL cholesterol (mmol.l <sup>-1</sup> )	$1.22\pm0.31$	$1.27\pm0.28$	0.55
LDL cholesterol (mmol.l <sup>-1</sup> )	$2.62 \pm 0.94$	$2.45\pm0.94$	0.48

**Table 3.6** Fasting concentrations of total cholesterol, triglycerides, HDL-cholesterol and LDL-cholesterol in Offspring and Control groups.

Values are mean  $\pm$  SD. P-values are for unpaired *t*-tests between groups. <sup>a</sup>statistical analysis performed on logarithmically transformed data.

# 3.3.7 Liver enzymes

No significant differences in fasting values for GGT (Control,  $13.6 \pm 8.6 \text{ mmol.l}^{-1}$ , Offspring  $20.1 \pm 21.4 \text{ mmol.l}^{-1}$ , p = 0.35) or ALT (Control  $16.9 \pm 6.5 \text{ mmol.l}^{-1}$ , Offspring  $17.5 \pm 7.6 \text{ mmol.l}^{-1}$ , p = 0.92) were found between Controls and Offspring.

# 3.3.8 Adipokines & C-reactive protein

Fasting plasma concentrations of adiponectin, leptin, TNFα, IL-6, resistin and CRP are presented in Table 3.7 and Figure 3.4. No significant differences were found between Controls and Offspring for any of the adipokines measured but there was a trend towards higher resistin concentrations in the Offspring group. No significant difference was observed for CRP concentration between groups.

	Controls	Offspring	P -
	( <b>n=36</b> )	(n=34)	value
Adiponectin (µg.ml <sup>-1</sup> ) <sup>a</sup>	$7.4 \pm 2.8$	$7.2 \pm 3.2$	0.66
Leptin (ng.ml <sup>-1</sup> ) <sup>a</sup>	$17.8 \pm 10.8$	$22.2 \pm 14.7$	0.33
TNF- $\alpha$ (pg.ml <sup>-1</sup> ) <sup>a</sup>	$1.3 \pm 0.8$	$1.1 \pm 0.5$	0.47
IL-6 (pg.ml <sup>-1</sup> ) <sup>a</sup>	$1.7 \pm 1.4$	$1.4 \pm 0.9$	0.58
Resistin (ng.ml <sup>-1</sup> )	$14.5 \pm 4.2$	$16.9 \pm 5.8$	0.06
CRP (mmol.l <sup>-1</sup> ) <sup>a</sup>	$2.5 \pm 3.0$	$2.3 \pm 3.4$	0.74

**Table 3.7** Fasting plasma adipokine concentrations in Control and Offspring groups.

Values are mean  $\pm$  SD. P-values are for unpaired *t*-tests between groups. <sup>a</sup>statistical analysis performed on logarithmically transformed data.

-



**Figure 3.4** Individual value plots of baseline adipokines in Controls and Offspring. p-values are for unpaired t-tests between groups. Statistical analysis performed on logarithmically transformed data for all adipokines except resistin.

In both Control and Offspring groups, fasting adiponectin concentrations correlated positively with ISI (Table 3.8). After adjustment for fat mass, Offspring alone continued to display a significant positive association between fasting adiponectin concentration and ISI (Table 3.8, Figure 3.5). Leptin concentration was negatively correlated with ISI in both groups, but was no longer statistically significant in either group after adjustment for fat mass (Table 3.8). IL-6 concentration was negatively correlated with ISI in Offspring alone, but this observation was no longer statistically significant after adjustment for fat mass (Table 3.8). CRP was negatively correlated with ISI in both Controls and Offspring, but remained significant in Offspring alone after adjustment for fat mass (Table 3.8, Figure 3.6).

<b>Table 3.8</b> Correlations between adiponectin, leptin, TNF-α, IL-6, resistin and CRP
concentrations and ISI in Controls ( $n = 36$ ) and Offspring ( $n = 34$ ), before and after
adjustment for fat mass.

		Controls		Offs	pring
		Unadjusted	Adjusted	Unadjusted	Adjusted
Adiponectin	r	0.38*	0.07	0.51**	0.56***
Leptin	r	-0.57***	-0.17	-0.52**	-0.05
TNF-α	r	0.16	0.19	-0.22	-0.20
IL-6	r	-0.22	0.06	<b>-0.37</b> *	-0.07
Resistin	r	-0.20	0.18	-0.12	-0.15
CRP	r	-0.36*	0.05	-0.66***	<b>-0.41</b> <sup>*</sup>

r = Univariate correlations of variable vs. ISI. Unadjusted = variable correlated against ISI. Adjusted = variable correlated against residual of ISI and fat mass. <sup>a</sup>statistical analysis performed on logarithmically transformed data. Statistically significant correlation shown in bold. <sup>\*</sup> p < 0.05, <sup>\*\*</sup> p < 0.01, <sup>\*\*\*</sup> p < 0.001.



**Figure 3.5** Scattergram showing correlation between ln adiponectin and ln ISI residual after adjustment for fat mass in Controls and Offspring.



**Figure 3.6** Scattergram showing correlation between ln CRP and ln ISI residual after adjustment for fat mass in Controls and Offspring.

#### 3.3.9 Cardiorespiratory Fitness & Habitual Physical Activity

There was no significant difference between  $\dot{V}O_{2max}$  in Controls compared to Offspring (Table 3.9). Lactate threshold occurred at a similar absolute  $\dot{V}O_2$  in both Controls and Offspring (Table 3.9). There were no observed differences in resting heart rate or average daily heart rate. Controls displayed a trend towards a higher average heart rate minus resting heart rate (a surrogate measure of total activity) (Table 3.9). In addition, Controls spent significantly greater amounts of time per day at heart rates which were twice that of resting (p < 0.05) (Table 3.9).

**Table 3.9** Indices of cardiorespiratory fitness and physical activity in Control and Offspring groups.

	Controls	Offspring	P - value
	( <b>n=36</b> )	(n=34)	
$\dot{V}O_{2max}^{b}$ (ml.kg <sup>-1</sup> .min <sup>-1</sup> )	32.7 ± 6.7	$30.4 \pm 4.8$	0.10
$\dot{V}O_2$ at lactate threshold $(ml.kg^{-1}.min^{-1})^{b,a}$	$21.0 \pm 4.1$	$21.0 \pm 3.8$	0.97
Resting HR <sup>c</sup> (b.min <sup>-1</sup> )	$66.1 \pm 10.0$	$69.6 \pm 7.8$	0.15
Average daily HR (b.min <sup>-1</sup> )	$85.7\pm8.6$	$86.0 \pm 6.3$	0.85
Average daily HR - resting HR	$19.6 \pm 1.11$	$16.5 \pm 1.10$	0.05
Time spent above 1.5 times resting HR (min.day <sup>-1</sup> ) <sup>a</sup>	$133.0 \pm 130.9$	86.7 ± 65.7	0.15
Time spent above 2 times resting HR (min.day <sup>-1</sup> )	$18.4 \pm 28.8$	$6.3 \pm 6.3$	< 0.05

Values are mean  $\pm$  SD. P-values are for unpaired *t*-tests between groups. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>b</sup>Estimated from submaximal incremental treadmill test. <sup>c</sup>From pulse wave velocity determination.  $\dot{V}O_2$ : Oxygen uptake;  $\dot{V}O_{2max}$ : Maximal oxygen uptake; HR: heart rate.
The Control group displayed significant negative associations between resting heart rate, average daily heart rate and ISI. After adjustment for fat mass, only resting heart rate remained significantly negatively correlated with ISI (Table 3.10). Time spent at 1.5 times resting heart rate was significantly positively associated with ISI in the Control group, and this association remained significant after adjustment for fat mass (Table 3.10). In the Offspring group, ISI correlated negatively with resting heart rate and positively with time spent at 1.5 times resting heart rate but these associations became non significant after adjustment for fat mass (Table 3.10). However, these predicted  $\dot{V}O_{2max}$  and ISI, and  $\dot{V}O_2$  at lactate threshold and ISI (Table 3.10). The Offspring group did not display any significant associations between indices of cardiorespiratory fitness and ISI (Table 3.10).

		Controls		Offspi	ring
		Unadjusted	Adjusted	Unadjusted	Adjusted
$\dot{VO}_{2max}$ (ml.kg <sup>-1</sup> .min <sup>-1</sup> )	r	<b>0.40</b> *	0.06	0.20	-0.15
$\dot{V}O_2$ at lactate threshold (ml kg <sup>-1</sup> min <sup>-1</sup> ) <sup>a</sup>	r	0.46**	0.11	0.29	-0.04
Resting HR	r	-0.59***	-0.46*	<b>-0.47</b> <sup>*</sup>	-0.37
Average daily HR	r	-0.50**	-0.30	-0.32	-0.23
Average daily HR - resting HR	r	0.27	0.33	0.29	0.25
Time spent above 1.5 times resting HR <sup>a</sup>	r	0.53**	0.46*	0.39*	0.27
Time spent above 2 times resting HR	r	0.33	0.23	0.23	0.08

**Table 3.10** Correlations between indices of physical activity and ISI, and cardiorespiratory

 fitness and ISI in Controls and Offspring, before and after adjustment for fat mass.

r = Univariate correlations of variable vs. ISI, Adjusted R = correlation of variable vs. ISI adjusted for influence of fat mass. <sup>a</sup>statistical analysis performed on logarithmically transformed data. Statistically significant correlation shown in bold.\* p < 0.05, \*\* p < 0.01, \*\*\*p < 0.001.

# 3.3.10 Energy intake and dietary composition

Data on energy and macronutrient intake are presented in Table 3.11. No significant differences were observed in energy or macronutrient intake between Control and Offspring groups (Table 3.11). No dietary factors were associated with ISI in the Control or Offspring groups.

	Controls	Offspring	P - value
	(n=32)	(n=32)	
Energy (kJ.kg <sup>-1</sup> .day <sup>-1</sup> )	$108 \pm 27$	$114 \pm 34$	0.39
Total fat (g. kg <sup>-1</sup> .day <sup>-1</sup> )	$0.90\pm0.27$	$1.00 \pm 0.32$	0.17
SFA (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$0.32\pm0.12$	$0.35\pm0.12$	0.19
MUFA (g. kg <sup>-1</sup> .day <sup>-1</sup> )	$0.29\pm0.09$	$0.34\pm0.12$	0.10
PUFA (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$0.13 \pm 0.06$	$0.20 \pm 0.10$	0.15
Total CHO (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$3.36 \pm 1.04$	$3.57 \pm 1.41$	0.68
Sugar (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$1.52 \pm 0.78$	$1.62 \pm 0.86$	0.66
Starch (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$2.50 \pm 4.01$	$1.95 \pm 0.72$	0.76
Protein (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$1.00 \pm 0.25$	$1.00 \pm 0.28$	0.62

**Table 3.11** Energy and macronutrient intake in Control and Offspring groups.

Values are mean ± SD. P-values are for unpaired *t*-tests between groups. <sup>a</sup>statistical analysis performed on logarithmically transformed data. SFA: Saturated fatty acids; MUFA: Monounsaturated fatty acids; PUFA: Polyunsaturated fatty acids, CHO: Carbohydrate.

#### **3.4 Discussion**

As expected (Higgins et al. 2005;Perseghin et al. 1997;Humphriss et al. 1997;Straczkowski et al. 2003;Nyholm et al. 2004;Ahn et al. 2004) it was observed that sedentary female offspring of type 2 diabetic patients were more insulin resistant than matched females without a family history of type 2 diabetes and that adiposity is an important contributing factor to insulin sensitivity in both Offspring and Control groups. Most importantly, by using accurate and detailed measurements of body composition it was demonstrated that the effect of adiposity on insulin sensitivity is influenced by a family history of diabetes. Indeed, the Offspring group was found to have stronger and more consistent negative associations between adiposity indices and ISI than Controls. In contrast to reports from others (Higgins et al. 2005;Adamson et al. 2001), there were no differences in dietary intake between Controls and Offspring and insulin sensitivity was not related to diet in either group. Consistent with previous evidence, Offspring were less habitually active but contrary to previous evidence from Higgins and colleagues (2005), Controls showed a stronger association than Offspring between physical activity and insulin sensitivity (Higgins et al. 2005).

Although there is some evidence that in offspring of type 2 diabetic patients, measures of adiposity (Humphriss et al. 1997;Perseghin et al. 1997;Nyholm et al. 2004;Higgins et al. 2005) or body mass index (Adamson et al. 2001;Higgins et al. 2005) are increased in comparison with control subjects, in this study no differences in body mass, body mass index or measurements of adiposity were observed between Offspring and Control groups. Although measurements of adiposity such as BMI, total body fat, regional body fat, skinfold measurements and waist and hip circumferences were not different between the groups, the associations between adiposity markers and insulin sensitivity were stronger in the Offspring group. Cross-sectional analysis of heterogenous groups of relatives and controls has previously suggested greater susceptibility to the effect of adiposity on insulin sensitivity in relatives (van Dam et al. 2001). However, subsequent studies have not provided evidence of

this in matched groups of relatives and controls (Nyholm et al. 2004;Johanson et al. 2003;Higgins et al. 2005). The results of the current study are important since contrary to the previous research, it would appear that insulin sensitivity is more susceptible to the influence of adipose tissue in Offspring than in Controls.

Other groups who are predisposed to the development of type 2 diabetes (e.g. South Asians, Pima Indians) are recognised to display insulin resistance at lower levels of adiposity than matched subjects who are thought to have a lower predisposition for the condition (Gautier et al. 1999;Razak et al. 2007). South Asians also manifest both insulin resistance and diabetes at lower BMI values than Caucasians (Razak et al. 2007). Indeed, comparison between BMI matched South Asian and Caucasian males revealed similar visceral fat mass, but larger subcutaneous adipose mass in South Asians (Chandalia et al. 2007). Sniderman and colleagues (2007) suggest that this group has elevated deep subcutaneous adipose mass, which has greater contribution to insulin resistance (Sniderman et al. 2007). Thus, the current study findings support the notion that relatives, similar to other high-risk groups, have greater susceptibility to the adverse effects of obesity on glycaemia and insulin sensitivity than their counterparts with no family history of diabetes (Johanson et al. 2003;van Dam et al. 2001).

Although no differences were observed in cardiorespiratory fitness between the two groups, Offspring displayed a trend towards a lower total activity, calculated as average daily heart rate minus resting heart rate, and spent significantly less time per day at heart rates which were two times higher than resting heart rates. Thus, in this study Offspring were significantly less active than Control subjects. This is consistent with the work of Higgins and colleagues (2005) which reported that offspring display lower daily energy expenditure estimated from 7 day activity diaries (Higgins et al. 2005). However, correlations between physical activity measures and ISI found in the current study were different from those reported by Higgins and colleagues (2005). In the current study, only Controls displayed significant positive correlations between indices of physical activity (such as time spent at 1.5 times resting heart rate) and ISI, after adjustment for fat mass, whilst Higgins and co-workers observed that HOMA<sub>IR</sub> was significantly negatively correlated with habitual levels of daily activity only in non-diabetic daughters of patients with type 2 diabetes (Higgins et al. 2005). Therefore, the notion that in Offspring, insulin sensitivity is influenced by physical activity to a greater degree than in individuals without family history of diabetes was not confirmed in this study.

Reasons for these differences may be related to the differing methods used to evaluate physical activity. The heart rate analysis utilised in this study is a more accurate and objective description of physical activity than the estimated energy expenditure method used by Higgins and colleagues (Higgins et al. 2005). In addition, the subject group in the current study were different to those described by Higgins and co-workers. The current study specified that all volunteers were sedentary, and results suggest that Offspring in the current study were more sedentary than the Control groups. The absence of significant correlations between physical activity and ISI in Offspring might suggest that habitual physical activity does not play a major role in the determination of insulin sensitivity. However, it should be appreciated that the range of physical activity in the Offspring volunteers showed less variability than Controls, and this homogeneity of physical activity may preclude detection of a direct interaction. Additionally, despite the absence of a direct relationship between physical activity and ISI it should be appreciated that physical activity may have an indirect effect through other mechanisms. One such mechanism is a potential alteration in partitioning of dietary saturated fats, which in the context of extreme physical inactivity, promotes palmitate storage in muscle and contributes to increased intramuscular lipid content (Bergouignan et al. 2009).

In agreement with previous studies (Johanson et al. 2003; Higgins et al. 2005), there were no significant differences in daily energy and macronutrient intake or proportion of energy provided by the main energy providing nutrients between Offspring and Control groups. Furthermore, in both Control and Offspring groups dietary factors did not correlate with insulin sensitivity. Taken together these data suggest that impaired insulin sensitivity seen in Offspring can not be explained by the difference in dietary factors. It should be noted however that other studies (Adamson et al. 2001) found that non-diabetic relatives of type 2 diabetic patients consumed diets which are expected to promote the development of diabetes (Adamson et al. 2001). Indeed, in the above study (Adamson et al. 2001) relatives were reported to have higher absolute intakes of total fat, saturated fat and cholesterol and lower intake of non-starch polysaccharides and suggested that there is scope for decreasing the disease risk through improvements in the diet (Adamson et al. 2001). However, relatives studied by Adamson and colleagues (2001) were significantly heavier, with higher waist circumferences than controls and may therefore be considered to be of a different metabolic phenotype than those described in this study (Adamson et al. 2001). Indeed, consideration of dietary habits may be of a particular importance since the effects of dietary fatty acids may differ between offspring of type 2 diabetic patients and individuals without family history of diabetes. Negative associations between saturated fat and insulin sensitivity, and a positive association between unsaturated fats and insulin sensitivity have been observed in relatives alone (Ntali et al. 2009).

This study also aimed to find whether early metabolic abnormalities seen in Offspring can be explained by impairment in metabolic flexibility. Metabolic inflexibility has been defined as an increase in muscle glucose oxidation in fasting conditions, and decreased glucose oxidation under insulin-stimulated circumstances (Kelley & Mandarino 2000) and has been shown to contribute to the accumulation of intramuscular triglyceride and thus insulin resistance (Felber et al. 1987;Boden 1997;Delarue & Magnan 2007). In this study however, there were no differences in the rate of fat and carbohydrate oxidation in fasted or post-

glucose states between the Offspring and Control groups. To some extent this finding is not surprising, since precise investigation of metabolic inflexibility has been performed most successfully in isolated tissue groups (Kelley & Mandarino 2000) and there is evidence that 'whole body' respiratory quotient may be a poor index of a metabolic abnormality which is predominantly skeletal-muscle based (Galgani et al. 2008;Galgani, Moro, & Ravussin 2008). In addition, metabolic inflexibility is typically observed in obese and insulin resistant subjects who display evidence of elevated circulating NEFA concentrations (Delarue & Magnan 2007;Corpeleijn, Saris & Blaak 2009). The Offspring group in this study were observed to have neither greater adiposity, nor higher NEFA concentrations.

Although NEFA concentration was not different between the two groups, this is not surprising, since there were no obvious differences in subcutaneous abdominal adipose tissue mass which is the major determinant of systemic NEFA concentrations (Bickerton et al. 2008;Frayn et al. 1993). Additionally, Bickerton and colleagues (2008) have demonstrated relatively suppressed NEFA concentrations due to hyperinsulinaemia in insulin resistant subjects (Bickerton et al. 2008). Therefore, in the current Offspring group a tendency towards increased circulating NEFA may have been suppressed by hyperinsulinaemia. After adjustment for fat mass, NEFA was significantly negatively correlated with ISI in Offspring alone, suggesting that women with a family history of type 2 diabetes may be more sensitive to the effects of fatty acids on insulin sensitivity (Roden et al. 2000;Boden 1997;Boden & Shulman 2002).

Single measurements of NEFA concentrations over an OGTT may be misleading in terms of the metabolic interaction between lipolysis in adipose tissue and NEFA uptake in skeletal muscle. *In vivo* NEFA flux between adipose tissue and skeletal muscle could be greater in Offspring than in Control volunteers and may represent insulin resistance in adipose tissue and greater NEFA uptake by insulin resistant skeletal muscle. Confirmation of this

supposition would require a study of NEFA efflux from adipose tissue under OGTT conditions. However this would have required superficial abdominal vein cannulation (Frayn, Coppack, & Humphreys 1993) and forearm cannulation to determine arteriovenous differences (Evans, Clark, & Frayn 1999). This would have provided information about lipolysis in adipose tissue in response to an oral glucose load and subsequent increase in circulating insulin concentration and the skeletal muscle uptake of increased NEFA efflux. Measurement of the VLDL-triglyceride arteriovenous difference in skeletal muscle would have given evidence of increased local delivery of NEFA from circulating VLDL-triglyceride, since skeletal muscle LPL activity is dependent on plasma lipid concentrations (Frayn, Arner, & Yki-Jarvinen 2006). In this study fasting triglyceride concentration was not different between Controls and Offspring but no assessment was made of plasma triglyceride concentration over the OGTT. Future exploration of the early mediators of insulin resistance in Offspring could include both analysis of lipid kinetics in both adipose tissue and skeletal muscle and lower doses of oral glucose in order to assess more subtle changes in glucose-insulin-lipid dynamics (Bickerton et al. 2007).

No differences were observed in concentrations of adipokines, however in Offspring alone, adiponectin was significantly positively correlated with ISI, independent of fat mass. In Offspring alone, these findings may indicate an important role for adipose tissue function, independent of total adipose mass, in determining ISI. In particular, Offspring appear to be more sensitive to the influences of these factors on insulin sensitivity. Fasting adiponectin is an independent predictor of both liver fat and insulin sensitivity and in addition to promotion of both fatty acid uptake and oxidation in muscle cells, is thought to increase hepatic insulin sensitivity through stimulation of peroxisome proliferators-activated receptor alpha (PPARα) (Furler et al. 2006;Yamauchi & Kadowaki 2008;Koska et al. 2008). Similar to the potential relationship between NEFA and ISI observed in this study, the positive correlation between ISI and adiponectin (independent of fat mass) in Offspring may reflect a compensatory response to the effects of NEFA on insulin sensitivity in muscle and liver, perhaps reflecting

subjects who are able to counteract increased NEFA by adiponectin mediated stimulation of hepatic fatty acid oxidation (Bonet et al. 2007). In this study, supportive evidence for this suggestion may be considered by the absence of elevated biomarkers of hepatic steatosis (ALT, GGT) (Fraser et al. 2009) in Offspring, which is consistent with previously observed normal hepatic lipid content in male offspring subjects despite the presence of insulin resistance (Johanson et al. 2003).

The significance of CRP in the pathogenesis of insulin resistance remains controversial. Elevated concentrations of CRP are associated with both cardiovascular disease and insulin resistant states (Festa et al. 2000;Pradhan et al. 2003;Tesauro et al. 2007). However, it is possible that CRP is simply a marker of the chronic inflammatory state common to both cardiovascular disease and insulin resistance, and as yet, no causal link has been identified (Sattar & Lowe 2006;Kriketos et al. 2004). The current study reports a negative correlation between CRP and ISI in Offspring alone. This is thought to indicate a low grade chronic inflammatory state which is implicated in the pathogenesis of insulin resistance and vascular dysfunction, and appears to be associated with increased visceral fat mass (Kowalska et al. 2008;Salmenniemi et al. 2005). The current study did not determine corroborative evidence of a pro-inflammatory state, increased visceral fat mass or altered vascular function in Offspring.

This study has advantages over previous studies using objective assessment of body composition by using DEXA scanning. In addition, objective measurements of both cardiorespiratory fitness and habitual physical activity were performed, allowing a more complete description of both physical fitness and daily activity. Furthermore, use of accurate values of total fat mass allowed correction for this factor and an understanding of the influences upon insulin sensitivity which cannot be directly attributed to fat mass.

There are areas which would strengthen this study further. Direct imaging using computed tomography or magnetic resonance imaging would provide a direct assessment of visceral fat and partitioned subcutaneous fat, which would be of great value when further considering the influences upon hepatic insulin sensitivity and adjusting for the influence of fat mass. Dietary data is likely to have been under-reported, however more robust evaluation of dietary intake is difficult in free-living subjects (Hill & Davies 2001; Weber et al. 2001). This study was unable to assess habitual energy expenditure, due to the limitations in the application of the FLEX technique. It would also be reasonable to consider the use of venous occlusion plethysmography as a tool for vascular assessment since it may be a more useful tool in assessing skeletal muscle microvascular function, although more invasive methods such as contrast enhanced ultrasound would be more robust (Clark 2008; Wilkinson & Webb 2001). The gold standard technique, the euglycemic hyperinsulinaemic clamp was not used for the assessment of insulin sensitivity. However, the insulin sensitivity index that was used has been shown to correlate well with the rate of whole-body glucose disposal during a euglycemic hyperinsulinaemic clamp (r = 0.73, p < 0.0001), and can be considered to be a more physiological index of insulin sensitivity than HOMA<sub>IR</sub> which is primarily an index of hepatic insulin sensitivity (Matsuda & DeFronzo 1999; Chang et al. 2006). Use of the OGTT was a more practical method of assessing insulin sensitivity and performing 70 clamp studies was not feasible in the available timescale.

In conclusion, this study observed differences in the metabolic phenotype of Offspring characterised by lower insulin sensitivity. In addition, adiposity more strongly influenced insulin sensitivity in Offspring, and circulating fatty acids were also associated with insulin sensitivity in Offspring alone, regardless of fat mass. Adiponectin was positively correlated with ISI in Offspring alone, and CRP negatively correlated in this group only. These differences were apparent despite similar dietary intake, but were associated with lower habitual activity levels in Offspring, whose insulin sensitivity was not influenced by physical activity. These observations suggest an important role for adipose tissue and lipotoxicity in the early pathogenesis of insulin resistance, with possible early compensatory mechanisms, in sedentary premenopausal daughters of patients with type 2 diabetes.

#### CHAPTER 4

# PHYSICAL AND METABOLIC RESPONSES TO A STRUCTURED EXERCISE PROGRAMME IN WOMEN WITH AND WITHOUT A FAMILY HISTORY OF TYPE 2 DIABETES

# 4.1 Introduction

The rapid increase in the prevalence of type 2 diabetes over recent years (Wild et al. 2004) must be attributed to changes in environmental factors, such as body fatness and exercise habits, as the human genome has not changed over this short period (Booth et al. 2000). Epidemiological data suggests that rates of both overweight/obesity and increasing urbanisation are closely related to prevalence of diabetes (King, Aubert, & Herman 1998). Prospective observational data utilising proportional hazard models (Colditz et al. 1995) and multiple logistic regression modelling (Manson et al. 1992) suggests that being overweight with an abdominal fat distribution probably accounts for 80-90% (Astrup & Finer 2000), and that a sedentary lifestyle accounts for at least 25% (Manson et al. 1992), of all type 2 diabetes incidence. However, as not all overweight and sedentary individuals develop the disease, it is clear that other non-environmental factors also play an important role in determining risk for type 2 diabetes. In sedentary populations, first-degree relatives of patients with type 2 diabetes have about three times the risk of developing diabetes than their counterparts with no family history of diabetes (Kobberling & Tillil 1982;Ohlson et al. 1988) and even when matched for BMI, are often more insulin resistant than control participants (Kriketos et al. 2004;Nyholm et al 2004;Perseghin et al 1997), suggesting that relatives have an innate susceptibility (probably due to some combination of genetic, epigenetic and/or 'early-origins' factors) to insulin resistance and diabetes. However, in epidemiological observations, the excess risk of developing diabetes that is associated with a diabetes family history is much greater in sedentary population groups than in physically active population groups, suggesting that the increased risk associated with a diabetes family history may be diminished (although not eliminated) at high levels of physical activity (Hu et al.

1999;Sargeant et al. 2000). In addition, recent cross-sectional reports suggest that modulation of insulin resistance in offspring of patients with type 2 diabetes may be particularly amenable to habitual physical activity or cardiorespiratory fitness level, such that inactive or unfit offspring are insulin-resistant compared with control participants with no diabetes family history, whereas active or fit offspring exhibit levels of insulin sensitivity that are much closer to persons with no diabetes family history (Ahn et al. 2004;Gill & Malkova 2006;Higgins et al. 2005). Taken together, these observations suggest that healthy offspring of patients with type 2 diabetes possess an innate predisposition to insulin resistance, with consequent increased diabetes risk, which manifests when sedentary. However, this appears to be highly modulated by level of physical activity. It is therefore hypothesised that, in response to the same exercise training intervention, offspring of type 2 diabetes patients would improve insulin sensitivity to a greater extent than control participants with no diabetes family history.

#### 4.2 Methods

#### 4.2.1 Volunteers

Volunteers were recruited and screened as detailed in section 2.1. Specific inclusion criteria are defined in section 2.1.3. Further information detailing recruitment response and excluded volunteers can be found in Figure 2.2. The Offspring group was composed of women with a first-degree relative with type 2 diabetes (T2D). The Control group was composed of women with no first or second degree relative with T2D. This study was conducted on twenty eight Offspring and thirty four Control volunteers. Control and Offspring groups were well matched for age (Controls  $33.5 \pm 6.3$  years, Offspring  $35.5 \pm 6.7$  years, p = 0.22) and BMI (Controls  $27.0 \pm 4.3$  kg.m<sup>-2</sup>, Offspring  $28.1 \pm 4.9$  kg.m<sup>-2</sup>, p = 0.36).

#### 4.2.2 Study Design

Volunteers were randomised into 'Exercise' or 'Delayed-Exercise' intervention arms using a computer programme. The study design is shown in Figures 2.3-2.4. The testing regime for each intervention arm is described in Section 2.1.3. All post-intervention OGTTs were performed 15-24 hours after the volunteers' final exercise session. The Exercise group participated in further metabolic testing 3 days after the final exercise session (Figure 2.4), during which period the volunteers were asked to avoid any planned exercise, to assess the acute and chronic effects of the exercise intervention. Analysis of the exercise-mediated change in variables was made by comparison of baseline and acute post-intervention data in the Exercise group, and by comparison of the data collected after the period of 'usual lifestyle' (prior to inclusion in the exercise intervention) and the post-intervention data in the Delayed-Exercise group. Therefore irrespective of the intervention arm, volunteers were studied at eight-week intervals to allow, as far as possible, for all metabolic, fitness and body composition assessments for each individual to be made in the same menstrual cycle phase

(Creinin, Keverline, & Meyn 2004). Volunteers were asked to maintain their usual dietary and habitual physical activity habits thoughout the studied periods.

#### 4.2.3 Fitness testing

Cardiorespiratory fitness was measured by predicted  $\dot{V}O_{2max}$  and lactate threshold, as described in section 2.6.

#### **4.2.4 Metabolic testing**

Volunteers attended for metabolic testing as described in section 3.2.8. Assessment of metabolic rate and substrate utilisation was performed in the fasted state and 2 hours after a 75 g oral glucose load (section 2.3.2). Additionally, pulse wave velocity and blood pressure were measured in the fasted and post-glucose state as described in section 2.3.4 - 2.3.5. Fasting plasma samples were analysed as described in sections 2.4.2 and 2.4.3. An OGTT was performed using a 75 g oral glucose load with sampling from an intravenous cannula sited in the antecubital fossa, as described in section 2.3.7. Samples were analysed at 30, 60, 90 and 120 minutes for insulin, glucose, NEFA and 3-OHB (Section 2.4).

# 4.2.5 Body composition assessment

Dual X-Ray Absorptiometry (DEXA) scans (LUNAR Prodigy DEXA scanner, GE Healthcare Diagnostic Imaging, Slough, Berkshire, UK) were used to determine body composition and fat distribution (section 2.2.5). Height, body mass, waist and hip circumferences, and biceps, triceps, suprailiac and subscapular skinfolds were also determined using standard protocols (section 2.2) (Marfell-Jones M et al. 2006).

#### 4.2.6 Assessment of diet and habitual physical activity

Self-reported dietary intake and activity diaries were completed for a seven-day period prior to each episode of metabolic, anthropometric and fitness testing (Section 2.8-2.9). Seven day waking-hours heart rate monitoring was also performed prior to each testing episode (Section 2.9). Indices of habitual physical activity were calculated as detailed in section 2.10.

# **4.2.7 Exercise intervention**

Volunteers participated in a 7-week endurance-type exercise training programme with weekly supervised sessions as described in section 2.7.

#### 4.3 Statistical analysis

Data were analysed using Statistica (version 6.0, StatSoft Inc., Tulsa, Oklahoma) and Minitab (version 13.1, Minitab Inc., State College, Pennsylvania). Prior to analysis, all data were tested for normality using the Anderson-Darling normality test and, if necessary, logarithmically transformed. Based on data from the two baseline measurements made in the volunteers randomised to the Delayed-exercise intervention arm, the within-subject test-retest coefficient of variation for ISI was 12.2%, with mean values for ISI differing by 3.2% between the two baseline measurements. Changes from the immediately pre-intervention assessment to the post-intervention assessment in all volunteers who completed the intervention (i.e. between the first and second sets of tests in volunteers randomised to the 'Exercise' intervention and between the second and third sets of tests in volunteers randomised to the 'Exercise' intervention) were compared by 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. The group x trial interaction term was used to determine whether the Offspring and Control subjects responded differently to the intervention and *post-hoc* Tukey tests were used to identify in which group(s) changes occurred. In separate analyses, similar comparisons were made between the first and second

sets of tests in volunteers randomised to the 'Delayed-Exercise' intervention, to determine whether a 'study effect' could have contributed to the observed changes. Additionally, comparisons were made between the second and third sets of tests in volunteers randomised to the 'Exercise' intervention, to determine whether a three day period without planned exercise would result in further metabolic changes. Associations between variables were assessed using univariate linear correlations, with adjustments for the influence of change in other variables as described in Section 3.2.9.2. Multiple regression analyses were performed to establish the independence of relationships.

# 4.3.1 Calculation of glucose, insulin, NEFA, and 3-OHB concentrations and insulin sensitivity during oral glucose tolerance test

Concentrations of glucose, insulin, NEFA and 3-OHB were measured as described in section 2.4. Mean concentrations of glucose, insulin, NEFA and 3-OHB during the OGTT were calculated as described in section 3.2.9.1. Insulin sensitivity was expressed as the Insulin Sensitivity Index (Matsuda & DeFronzo 1999) and calculated as described in section 2.11.2.

# 4.4 Results

# 4.4.1 Changes during the usual lifestyle period for the Delayed Exercise group

There were no significant changes in energy intake or dietary composition, or in indices of habitual physical activity over the course of the 7-week 'usual lifestyle' period in either the Control (n=16) or the Offspring (n=13) (data not shown). The only significant changes in metabolic variables during the usual lifestyle period were observed in the Offspring group. Offspring showed an increase in fasting triglycerides (Pre:  $0.91 \pm 0.07 \text{ mmol.1}^{-1}$ ; Post  $1.03 \pm 0.09 \text{ mmol.1}^{-1}$ , NS) whilst Controls displayed a slight decrease in fasting triglycerides (Pre:  $0.82 \pm 0.08 \text{ mmol.1}^{-1}$ ; Post  $0.73 \pm 0.06 \text{ mmol.1}^{-1}$ , NS). Although the changes were not significant within the groups, these small changes resulted in a significant difference between

fasting triglycerides (p < 0.05) after the period of usual lifestyle. Offspring also displayed a significant increase in fasting GGT (Pre:  $16.6 \pm 2.82 \text{ mmol.1}^{-1}$ ; Post  $22.8 \pm 4.66 \text{ mmol.1}^{-1}$ , p < 0.05) and although Controls did not display a significant change in fasting GGT (Pre:  $12.28 \pm 1.90 \text{ mmol.1}^{-1}$ ; Post:  $12.3 \pm 1.39 \text{ mmol.1}^{-1}$ , NS) there was no significant difference between the groups after the period of usual lifestyle (p = 0.06). There were no significant changes in any indices of cardiorespiratory fitness over the course of the 7-week usual lifestyle period in either the Control or the Offspring volunteers (data not shown). However a small, but significant, increase in lean body mass in the Offspring (Pre:  $44.7\pm1.7 \text{ kg}$ ; Post:  $45.6\pm1.8 \text{ kg}$ , p<0.05) but not the Control (Pre:  $42.8\pm1.7 \text{ kg}$ ; Post:  $42.5\pm1.6 \text{ kg}$ , NS) group was observed.

# 4.4.2 Compliance to exercise intervention

Six Offspring subjects and two Control subjects dropped out of the study before completion of the exercise intervention. Reasons for their withdrawal are shown in Figure 2.2. For subjects who completed the intervention and attended for post-intervention testing, Control subjects (n = 34) completed 28 ± 6 exercise sessions (out of 32 possible sessions), undertaking 1353 ± 58 minutes of exercise at an average heart rate of  $141 \pm 2$  beat.min<sup>-1</sup> (73 ± 1% of maximum heart rate), over the 7-week intervention period. The Offspring subjects (n = 28) completed 29 ± 7 exercise sessions, undertaking  $1451 \pm 107$  minutes of exercise at an average heart rate of  $144 \pm 2$  beat.min<sup>-1</sup> (74 ± 1% of maximum heart rate). None of these values differed significantly between the Control and Offspring groups.

## 4.4.3 Post-intervention data

# 4.4.3.1 Physical characteristics and body composition

The Offspring group had small but statistically significant reductions in body mass (-1.6%, p < 0.05), total fat mass (-3.7%, p < 0.01), trunk fat mass (-3.0%, p < 0.05), leg fat mass (-2.8%, p < 0.05), arm fat mass (-6.9%, p < 0.05), waist circumference (-1.8%, p < 0.01), hip

circumference (-1.1%, p < 0.05), upper body (-4.5%, p < 0.01) and lower body fat (-2.8%, p < 0.05) following the exercise intervention, whereas in the Control group, the only statistically significant anthropometric changes were a small reduction in hip circumference (-0.9%, p<0.05), upper body fat (3.1%, p < 0.05) and triceps skinfold (6.6%, p < 0.01). There was also a significant interaction for the change in body mass with the intervention between the Control and Offspring groups (p<0.05), indicating a significantly greater reduction in body mass in response to the intervention the latter group (Table 4.1).

	Controls	Offspring	P - value
	(n = 34)	( <b>n</b> = <b>28</b> )	
Body mass (kg)	$-0.15 \pm 0.28$	$-1.20 \pm 0.44^{*}$	< 0.05
Total fat mass (kg)	$-0.58 \pm 0.23$	$-1.16 \pm 0.33^{**}$	0.14
Trunk fat mass (kg) <sup>a</sup>	$-0.26 \pm 0.17$	$-0.63 \pm 0.25^{*}$	0.79
Leg fat mass (kg) <sup>a</sup>	$-0.22 \pm 0.08$	$-0.28 \pm 0.13^{*}$	0.59
Arm fat mass (kg) <sup>a</sup>	$-0.10 \pm 0.06$	$-0.33 \pm 0.13^{*}$	0.23
Android fat mass (kg)	$\textbf{-}0.02\pm0.04$	$\textbf{-}0.04\pm0.05$	0.74
Gynoid fat mass (kg)	$-0.15 \pm 0.05$	$\textbf{-}0.04\pm0.10$	0.27
Total lean mass (kg)	$0.15\pm0.43$	$\textbf{-}0.04\pm0.28$	0.73
Trunk lean mass (kg)	$0.19\pm0.20$	$0.12\pm0.20$	0.79
Leg lean mass (kg)	$0.19\pm0.07$	$0.05\pm0.11$	0.28
Arm lean mass (kg) <sup>a</sup>	$0.04\pm0.05$	$\textbf{-}0.05\pm0.06$	0.44
Upper Body Fat (kg) <sup>a</sup>	$-0.37 \pm 0.21^{*}$	$-0.96 \pm 0.25^{**}$	0.46
Lower Body Fat (kg) <sup>a</sup>	$-0.22 \pm 0.08$	$-0.28 \pm 0.13^{*}$	0.59
Waist Circ (cm) <sup>a</sup>	$-0.84 \pm 0.37$	$-1.61 \pm 0.45^{**}$	0.28
Hip Circ (cm) <sup>a</sup>	$-0.94 \pm 0.32^{*}$	$-1.15 \pm 0.41^{*}$	0.83
Biceps Skinfold (mm) <sup>a</sup>	$0.74\pm0.65$	$-1.69 \pm 0.68$	< 0.05
Triceps Skinfold (mm) <sup>a</sup>	$-1.69 \pm 0.46^{**}$	$-1.00 \pm 0.50$	0.31
Subscapular Skinfold (mm)	$-0.73 \pm 0.85$	$\textbf{-}0.79\pm0.60$	0.96
Suprailiac Skinfold (mm) <sup>a</sup>	$-1.26 \pm 0.64$	$1.68 \pm 1.62$	0.24

**Table 4.1** Absolute changes in physical characteristics and body composition of Control andOffspring. Values are Mean ± SEM.

Column p-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>\*</sup>p < 0.05, <sup>\*\*</sup>p < 0.01 for difference from baseline within group.

# 4.4.3.2 Concentrations of glucose, insulin, non-esterified fatty acid and 3hydroxybuytrate

During the oral glucose tolerance test (OGTT) the Offspring group had significant reductions in fasting insulin (-17%, p<0.05), mean insulin concentration during OGTT (-14%, p < 0.01), and change in concentration from fasting during the OGTT (-12%, p < 0.01). There was also a significant interaction for the change in insulin concentration during the OGTT (p<0.05), and change from fasting (p<0.05) with the intervention between the Control and Offspring groups, indicating a significantly greater reduction in circulating insulin in response to the intervention in the latter group (Table 4.2). The Offspring group also showed significant reductions in 3-OHB concentrations during the OGTT as measured by the mean concentration during OGTT (-49%, p < 0.001) and change in concentration from fasting (p < 0.001), whilst Controls showed a significant change in mean 3-OHB concentration during the OGTT (-34%, p < 0.01). There was also a significant interaction for the change in both mean concentration during OGTT (p < 0.05), and change from fasting during OGTT (p < 0.01) with the intervention between the Control and Offspring groups, indicating a significantly greater reduction in 3-OHB in response to the intervention in the Offspring group (Table 4.2).

	Controls	Offspring	P - value
	(n = 34)	( <b>n=28</b> )	
Glucose			
Fasting (mmol.l <sup>-1</sup> )	$\textbf{-}0.08\pm0.08$	$-0.11 \pm 0.08$	0.82
Mean during OGTT (mmol.l <sup>-1</sup> )	$-0.08 \pm 0.18$	$-0.30 \pm 0.19$	0.39
Change from fasting during OGTT (mmol.l <sup>-1</sup> )	$-0.003 \pm 0.18$	$-0.20 \pm 0.17$	0.46
Insulin			
Fasting <sup>a</sup> (mU.1 <sup>-1</sup> )	$-0.27 \pm 0.44$	$-1.81 \pm 0.62^{*}$	0.11
Mean during OGTT <sup>a</sup> (mU.l <sup>-1</sup> )	$-3.78 \pm 2.86$	$-12.91 \pm 5.62^{**}$	< 0.05
Change from fasting during OGTT <sup>a</sup> (mU.l <sup>-1</sup> )	$-3.51 \pm 2.79$	-11.11 ± 5.14 <sup>**</sup>	< 0.05
NEFA			
Fasting (mmol.l <sup>-1</sup> )	$-0.01 \pm 0.03$	$-0.01 \pm 0.03$	0.93
Mean during OGTT <sup>a</sup> (mmol.l <sup>-1</sup> )	$0.002\pm0.018$	$-0.003 \pm 0.012$	0.52
Change from fasting during OGTT (mmol.l <sup>-1</sup> )	$0.009\pm0.023$	$0.006\pm0.027$	0.92
3-ОНВ			
Fasting <sup>a</sup> (mmol.l <sup>-1</sup> )	$0.01\pm0.01$	$0.04\pm0.02$	0.62
Mean during OGTT (mmol.l <sup>-1</sup> )	$-0.02 \pm 0.01^{**}$	$-0.05 \pm 0.01^{***}$	< 0.05
Change from fasting during OGTT (mmol.l <sup>-1</sup> )	$-0.03 \pm 0.01$	$-0.09 \pm 0.02^{***}$	< 0.01

**Table 4.2** Absolute changes in glucose, insulin, NEFA and 3-OHB in the fasted state and during OGTT. Values are Mean ± SEM.

P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>\*</sup>p < 0.05, <sup>\*\*</sup>p < 0.01, <sup>\*\*\*</sup>p < 0.001 for difference from baseline within group.

# 4.4.3.3 Insulin Sensitivity Index (ISI)

Offspring showed an increase of 23% (p < 0.01) in insulin sensitivity, whilst Controls increased ISI by 7% (NS). The interaction for the change in ISI was significant (p < 0.05)

suggesting a greater increase in ISI in Offspring in response to the exercise intervention (Figures 4.1-4.2).



**Figure 4.1** ISI in Controls (white bars) and Offspring (black bars) prior to, and after exercise intervention.  $*^{*}p < 0.01$  for within-group change from baseline.



**Figure 4.2** Magnitude of change in ISI in Controls (white bar) and Offspring (black bar) in response to exercise intervention. p < 0.05 for between-group change.

# 4.4.3.4 Plasma lipids

No significant differences were observed in total cholesterol, LDL-cholesterol, HDLcholesterol or triglycerides in either Controls or Offspring in response to the exercise intervention (data not shown).

# 4.4.3.5 Blood pressure and peripheral pulse wave velocity

Offspring displayed a trend towards reduction in diastolic blood pressure (p = 0.07) in response to the exercise intervention (Table 4.3). The interaction for the change in diastolic blood pressure was significant, suggesting that the exercise intervention produced different results in Controls compared to Offspring (Table 4.3). No significant changes were observed in fasting or post-glucose PWV in either group (Table 4.3).

**Table 4.3** Absolute changes in fasting systolic and diastolic blood pressure and peripheralPWV in fasted, and 2-hour post-glucose states. Values are Mean ± SEM.

	Controls	Offspring	P - value
	(n=34)	( <b>n=28</b> )	
Systolic blood pressure (mmHg)	$-1.59 \pm 1.78$	$-3.36 \pm 2.27$	0.54
Diastolic blood pressure (mmHg)	$1.12 \pm 1.16$	$-3.43 \pm 1.50$	< 0.05
Fasting PWV (m.s <sup>-1</sup> ) <sup>a</sup>	$-0.46 \pm 0.23$	$0.08\pm0.20$	0.86
Post-glucose PWV $(m.s^{-1})^a$	$-0.21 \pm 0.20$	$-0.28 \pm 0.26$	0.86

P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data.

#### 4.4.3.6 Liver enzymes

No significant differences in ALT or GGT were observed in either group after the intervention (data not shown).

# 4.4.3.7 Adipokines & C-reactive protein

Controls and Offspring had significant reductions in fasting adiponectin in response to exercise (Table 4.4). No significant difference was noted in the magnitude of the change in either group (Table 4.4). Controls displayed a significant reduction in IL-6 (-2.7%, p < 0.05). Offspring alone showed a significant reduction in fasting leptin in response to exercise, the interaction for the change between groups was also significant, suggesting a greater response due to exercise in Offspring (Table 4.4, Figures 4.3-4.4).

	Controls	Offspring	P -
	(n=34)	( <b>n=28</b> )	value
Adiponectin $(\mu g.ml^{-1})^{a}$	$-1.09 \pm 0.22^{***}$	$-0.94 \pm 0.34^{**}$	0.48
Leptin (ng.ml <sup>-1</sup> ) <sup>a</sup>	$-0.04 \pm 0.90$	$-5.3 \pm 1.4^{***}$	< 0.01
TNF- $\alpha$ (pg.ml <sup>-1</sup> ) <sup>a</sup>	$-0.02 \pm 0.06$	$0.01 \pm 0.04$	0.92
IL-6 $(pg.ml^{-1})^a$	$-0.43 \pm 0.16^{*}$	$\textbf{-0.08} \pm 0.08$	0.20
Resistin (ng.ml <sup>-1</sup> )	$0.33 \pm 0.34$	$-0.68 \pm 0.56$	0.11
$CRP (mmol.l^{-1})^{a}$	$-0.31 \pm 0.22$	$-0.01 \pm 0.33$	0.10

**Table 4.4** Absolute change in fasting adipokines. Values are Mean ± SEM.

P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>\*</sup>p < 0.05, <sup>\*\*</sup>p < 0.01, <sup>\*\*\*</sup>p < 0.001 for difference from baseline within group.



**Figure 4.3** Leptin in Controls (white bars) and Offspring (black bars) prior to, and after exercise intervention.  $*^{*}p < 0.001$  for within-group change from baseline.



**Figure 4.4** Magnitude of change in leptin in Controls (white bar) and Offspring (black bar) in response to exercise intervention.  $^{\ddagger}p < 0.01$  for between-group change.

## 4.4.3.8 Metabolic rate & substrate utilisation

Offspring displayed a significant reduction in respiratory exchange ratio (RER) in both fasted and post-glucose states, significant reductions in carbohydrate oxidation in both fasted and post-glucose conditions, and significantly increased fat oxidation in both states (Table 4.5). The interaction for the change between groups was also significant for these measurements, suggesting a greater response in the Offspring group. Overall metabolic rate did not change in either group, in either fasting or post-glucose conditions, after the exercise intervention.

**Table 4.5** Absolute change in metabolic rate, respiratory exchange ratio (RER), rates of fat and carbohydrate (CHO) oxidation in Controls and Offspring in the fasted state and 2 hours after glucose load. Values are Mean ± SEM.

	Controls	Offspring	P - value
	( <b>n=34</b> )	( <b>n=28</b> )	
Resting metabolic rate (kJ.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.0002 \pm 0.0005$	$0.0006 \pm 0.001$	0.93
Post-glucose metabolic rate (kJ.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.001 \pm 0.0009$	$0.0006 \pm 0.0008$	0.55
Fasting RER	$-0.01 \pm 0.01$	$-0.05 \pm 0.01^{***}$	< 0.01
Post-glucose RER	$-0.02 \pm 0.01$	$-0.05 \pm 0.01^{***}$	< 0.05
Fasting fat oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.09\pm0.07$	$0.41 \pm 0.08^{***}$	< 0.01
Post-glucose fat oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.12\pm0.07$	$0.37 \pm 0.081^{***}$	< 0.05
Fasting CHO oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$-0.17 \pm 0.16$	$-0.87 \pm 0.18^{***}$	< 0.01
Post-glucose CHO oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$-0.14 \pm 0.19$	$-0.78 \pm 0.20^{***}$	< 0.05

P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>\*</sup>p < 0.05, <sup>\*\*</sup>p < 0.01, <sup>\*\*\*</sup>p < 0.001 for difference from baseline within group.

#### 4.4.3.9 Cardiorespiratory fitness and habitual physical activity

Both Offspring and Controls significantly improved cardiorespiratory fitness as assessed by predicted  $\dot{V}O_{2max}$  (12% in the Controls and 15% in the Offspring, p<0.001 for both) and by  $\dot{V}$  O<sub>2</sub> at lactate threshold (Table 4.6). The improvement in cardiorespiratory fitness in both groups with the intervention is also illustrated by the right-shift in the  $\dot{V}O_2$  vs. heart rate relationship during sub-maximal exercise (Figure 4.5). The magnitude of improvement was similar in both groups, illustrated by a non-significant p-value for the interaction of group x trial.



**Figure 4.5**.  $\dot{VO}_2$  vs. Heart rate relationship pre- (black circles) and post-intervention (white circles) in Controls and Offspring.

The Control volunteers had a significant reduction in resting heart rate compared to their baseline values (6%, p < 0.001), whilst the Offspring group did not show a significant reduction (3%, NS) (Table 4.6). Neither Controls nor Offspring had a significant change in average daily heart rate compared to baseline however, although these changes were not

significant within the groups, the interaction for the change between groups was significant suggesting that the average daily heart rate had responded differently in Offspring compared to Controls (Table 4.6). Offspring alone showed a significant increase in habitual physical activity measured by the index of average heart rate – resting heart rate (Table 4.6). No significant differences were observed in time spent at heart rates indicative of light (1.5 times resting) and moderate (2 times resting) activity in either group, after adjustment for participation in the exercise intervention (Table 4.6).

**Table 4.6** Absolute change in cardiorespiratory fitness and indices of habitual physicalactivity. Values are Mean  $\pm$  SEM.

	Controls	Offspring	P - value
	( <b>n=34</b> )	( <b>n=28</b> )	
$\dot{V}O_{2max}^{b}$ (ml.kg <sup>-1</sup> .min <sup>-1</sup> )	3.89 ± 0.59 <sup>***</sup>	4.66 ± 0.82 <sup>****</sup>	0.46
$\dot{V}O_2$ at lactate threshold $(ml.kg^{-1}.min^{-1})^{b,a}$	$2.84 \pm 0.49^{***}$	$2.37 \pm 0.71^{**}$	0.59
Resting HR <sup>c</sup> (b.min <sup>-1</sup> )	$-4.2 \pm 1.0^{***}$	$-2.5 \pm 1.1$	0.27
Average daily HR (b.min <sup>-1</sup> )	$-1.2 \pm 0.9$	$2.6 \pm 1.3$	< 0.05
Average daily HR - resting HR	$3.3 \pm 1.3$	$5.1 \pm 1.4^{**}$	0.36
Time spent above 1.5 times resting HR (min.day <sup>-1</sup> ) <sup>a</sup>	$12.83\pm20.83$	$6.22\pm9.14$	0.84
Time spent above 2 times resting HR (min.day <sup>-1</sup> )	$2.08\pm4.10$	$4.86\pm2.06$	0.55

P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001 for difference from baseline within group. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>b</sup>Estimated from submaximal incremental treadmill test. <sup>c</sup>From pulse wave velocity determination.  $\dot{VO}_2$ : Oxygen uptake;  $\dot{VO}_{2max}$ : Maximal oxygen uptake; HR: heart rate.

# 4.4.3.10 Energy intake and dietary composition

The only significant change in energy or macronutrient intake after the exercise intervention was a reduction (9%, p < 0.05) in protein intake in the Control group.

	Controls	Offspring	P - value
	(n=34)	( <b>n=28</b> )	
Energy (kJ.kg <sup>-1</sup> .day <sup>-1</sup> )	$-5.41 \pm 3.27$	$-5.70 \pm 4.60$	0.96
Total fat (g. kg <sup>-1</sup> .day <sup>-1</sup> )	$-0.01 \pm 0.04$	$-0.05 \pm 0.05$	0.56
SFA $(g. kg^{-1}.day^{-1})^a$	$-0.01 \pm 0.02$	$-0.01 \pm 0.02$	0.94
$MUFA (g. kg^{-1}.day^{-1})$	$-0.01 \pm 0.02$	$-0.02 \pm 0.02$	0.53
$PUFA (g. kg^{-1}.day^{-1})^{a}$	$-0.01 \pm 0.01$	$-0.03 \pm 0.02$	0.59
Total CHO (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$-0.18 \pm 0.15$	$-0.29 \pm 0.15$	0.60
Sugar (g. $kg^{-1}$ .day $^{-1}$ ) <sup>a</sup>	$-0.17 \pm 0.11$	$-0.21 \pm 0.10$	0.74
Starch (g. $kg^{-1}$ .day <sup>-1</sup> ) <sup>a</sup>	$-0.10 \pm 0.07$	$0.0001 \pm 0.10$	0.23
Protein (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$-0.10 \pm 0.03^{*}$	$\textbf{-0.07} \pm 0.05$	0.58

Table 4.7 Absolute change in energy and macronutrient intake. Values are Mean ± SEM.

P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>\*</sup>p < 0.05, <sup>\*\*</sup>p < 0.01, <sup>\*\*\*</sup>p < 0.001 for difference from baseline within group.

#### **4.4.4** Chronic effects of exercise intervention

Volunteers randomised to the Exercise group had repeated metabolic assessment 3 days after the final exercise session (section 4.2.2). Volunteers had been asked to avoid any further planned exercise during this period to determine which metabolic variables might be acutely influenced by exercise, and which of the metabolic changes observed might persist in response to the cumulative effects of the intervention.

# 4.4.4.1 Three-day 'detraining' results

Insulin sensitivity did not change significantly in Offspring or Controls after 3 days without planned exercise (Table 4.8). Offspring did show a significant reduction in fasting 3-OHB concentration (-43%, p < 0.001) and the interaction for this change between the groups was also significant (Table 4.8). This would suggest a reduction in fasting hepatic fat oxidation, and this suggestion is supported by the observed significant reductions in fat oxidation in both fasting, and post-glucose conditions, with concurrent increases in respiratory exchange ratio (RER) in both states in Offspring alone (Table 4.8). The interaction for the change in fasting fat oxidation between the groups is also significant, suggesting that 3 days of detraining has a different effect on fasting fat oxidation in Offspring compared to Controls.

The only other significant metabolic variable to change with detraining was the fasting leptin concentration, which was significantly increased in Offspring (33%, p < 0.001). Fasting leptin concentration in the Control group did not alter significantly (7%, NS) (Table 4.8). The interaction for the change in leptin between groups was significant, suggesting a differing response to detraining in Offspring compared to Controls (Table 4.8).

	Controls	Offspring	P - value
	( <b>n=19</b> )	( <b>n=17</b> )	
ISI <sup>a</sup>	$-0.16 \pm 0.67$	$0.14 \pm 0.94$	0.64
Fasting 3-OHB (mmol.l <sup>-1</sup> ) <sup>a</sup>	$-0.021 \pm 0.017$	$-0.068 \pm 0.020^{***}$	< 0.05
Fasting RER	$0.02 \pm 0.01$	$0.05 \pm 0.01^{**}$	0.08
Post-glucose RER	$0.01 \pm 0.01$	$\boldsymbol{0.03 \pm 0.01}^{*}$	0.08
Fasting fat oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$-0.12 \pm 0.10$	$-0.42 \pm 0.09^{***}$	< 0.05
Post-glucose fat oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$-0.086 \pm 0.048$	$\textbf{-0.245} \pm \textbf{0.092}^{**}$	0.10
Fasting CHO oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.29 \pm 0.22$	$0.87 \pm 0.25^{**}$	0.11
Post-glucose CHO oxidation (mg.kg <sup>-1</sup> .min <sup>-1</sup> )	$0.086\pm0.13$	$0.514 \pm 0.23^{*}$	0.82
Leptin (ng.ml <sup>-1</sup> )	$0.19\pm0.80$	$5.80 \pm 2.15^{***}$	< 0.01

**Table 4.8** Absolute change in metabolic variables in Controls and Offspring after 3 days of 'detraining'. Values are Mean ± SEM.

P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data. <sup>\*</sup>p < 0.05, <sup>\*\*</sup>p < 0.01, <sup>\*\*\*</sup>p < 0.001 for difference from immediate post-intervention testing within group.

#### **4.5 Discussion**

The main finding of this study was that women with a family history of type 2 diabetes significantly improved insulin sensitivity in response to an exercise training programme, whereas no significant change in insulin sensitivity was observed in women matched for age and BMI but with no diabetes family history. ISI was 22% lower (p < 0.05) in the Offspring than the Controls group at baseline, but only 8% (NS) lower following exercise training. This augmented response to exercise training in the Offspring supports the primary hypothesis and is consistent with the literature from cross-sectional studies (Ahn et al. 2004;Gill & Malkova 2006;Higgins et al. 2005). Changes in body composition over the intervention were small in both groups, although the Offspring lost slightly more body fat than the Controls (by ~0.6 kg). The larger increase in ISI in the Offspring in response to the exercise intervention could not be explained by differences in compliance to the exercise intervention, or differences in changes in energy intake, or fitness between the Control and Offspring groups.

Although compliance with the intervention was not significantly different between the groups there remains a possibility that volunteers could have altered their habitual activity levels and that this could have influenced insulin sensitivity. The improvement in cardiorespiratory fitness in both groups was similar. Average daily heart rate did not change significantly in either group, however because the Controls decreased their average heart rate (-1%, NS) and Offspring increased their average heart rate (3%, NS) these changes were found to be statistically significant between the groups. This might suggest an increase in habitual activity in the Offspring group, however other indices of physical activity, such as the difference between average heart rate and resting heart rate, and the time spent at 1.5 times and 2 times resting heart rates (crude markers of time spent at heart rates consistent with 'light' and 'moderate' activity) were not different between Controls and Offspring. Thus, no evidence exists for a significant change in habitual activity levels outwith participation in the intervention, although it is accepted that the indices of physical activity are relatively crude.

Despite having similar levels of adiposity and fitness, and similar energy and macronutrient intakes, the Offspring group were more insulin resistant than the Control group at baseline, consistent with other data in the literature (Kriketos et al. 2004;Nyholm et al. 2004;Perseghin et al. 1997). It is possible that this increased baseline insulin resistance contributed to the Offspring group's greater change in insulin sensitivity in response to the intervention, since other studies have reported greater exercise-induced changes in insulin sensitivity in those who are more insulin resistant at baseline (An et al. 2005;Gill & Malkova 2006;Magkos et al. 2007). As Offspring have a genetic predisposition to increased insulin resistance and thus increased baseline insulin resistance is a well-established component of the Offspring 'phenotype' (Kriketos et al. 2004;Nyholm et al. 2004;Perseghin et al. 1997), it is difficult to determine the extent to which diabetes family history versus increased baseline insulin resistance itself contributed to the increased change in ISI with exercise training in the present study. However, irrespective of whether this is due to increased insulin resistance at baseline or other underlying factors, the present data indicate that Offspring form an easily identifiable group who will particularly benefit from increasing their level of physical activity.

Circulating leptin concentrations decreased by 24% in response to the intervention in the Offspring but were unchanged in the Control group. Leptin concentrations correlate with fat mass (Correia & Rahmouni 2006), and a change in the Offspring group might have been expected since fat mass was reduced. However, a 'rebound' effect on fasting leptin in Offspring alone was observed after 3 days of 'detraining' and therefore, the metabolic effect of exercise on leptin cannot solely be attributed to a change in fat mass, since body mass did not change over these 3 days. The reduction in leptin concentrations in the Offspring group may reflect an increase in 'leptin sensitivity'. Normally, leptin promotes lipid oxidation by increasing PPAR $\alpha$  expression (Lee et al. 2002). Thus, it has been proposed that leptin

resistance leads to insulin resistance by causing ectopic accumulation of lipids in non-adipose tissues, such as skeletal muscle, the liver and the pancreas (Correia & Rahmouni 2006). Indeed, a recent study reported a significant correlation between changes in circulating leptin concentrations and changes in intramuscular lipid content following an exercise intervention in older obese adults with impaired glucose tolerance (Solomon et al. 2008) and significant correlations between changes in insulin concentrations and changes in leptin concentrations following weight loss interventions have been reported (Carantoni et al. 1999;Doucet et al. 2000). In this study, Offspring alone displayed significant exercise mediated increases in rates of fat oxidation, and after 'detraining' significant reduction in fasting fat oxidation. Therefore, in the Offspring group alone, reduced leptin concentration, increased fat oxidation and increased insulin sensitivity were found after the exercise intervention. Also, consistent with a possible causal relationship between leptin, fat oxidation and insulin sensitivity, in this group alone the improvement in both fasting leptin concentration and fat oxidation were rapidly reduced after a period without exercise, and these changes were not accompanied by a significant change in insulin sensitivity. Thus, improved leptin sensitivity in the Offspring could have contributed to their increased insulin sensitivity following the intervention, and may signify, or even mediate, a subsequent reduction in insulin sensitivity which has been previously observed with cessation of exercise (Liu et al. 2008). It is however unclear from the present data why the exercise intervention only influenced leptin concentrations in the Offspring. Differential effects of exercise-training on adipose tissue expression of leptin and its receptor between the Offspring and Controls may have contributed, and this warrants further investigation.

These results differ slightly from the findings of earlier smaller scale studies which found similar improvements in insulin sensitivity in response to an exercise training intervention in adults with and without a family history of type 2 diabetes (Ostergard et al. 2006;Perseghin et al. 1996). One possible explanation for this discrepancy relates to the volume of exercise undertaken. In the progressive exercise intervention in the present study, subjects performed
300 minutes of exercise per week during the final two weeks of the intervention, compared to 135-180 minutes of exercise per week in the previously published reports (Ostergard et al. 2006;Perseghin et al. 1996). In an evolutionary context, even three hours of exercise per week represents a relatively low volume (Eaton & Eaton 2003), and it is possible that improvements in insulin sensitivity with exercise would plateau at higher exercise volumes in those with a family history of diabetes compared with those with no diabetes family history (Gill & Malkova 2006). Thus, in the earlier reports (Ostergard et al. 2006;Perseghin et al. 1996), the training stimulus, while sufficient to induce improvements in insulin sensitivity, may not have been great enough to induce optimal benefits in the group with a diabetes family history. Further study is needed to establish the dose-response relationship between physical activity and insulin sensitivity in those with a diabetes family history, which has potential implications for physical activity guidelines.

An additional explanation for the influence of exercise in this study is the sex of the participants. The earlier reports (Ostergard et al. 2006;Perseghin et al. 1996) studied mixed groups of men and women, whereas the current study involved only women. In the HERITAGE family study, the improvement in insulin sensitivity in response to exercise training was three times as great in men as in women (16% vs. 5%) (Boule et al. 2005), suggesting that changes in insulin sensitivity differ between the sexes. Thus, it is possible that the augmented response to exercise training in the Offspring group in the present study is sex-specific and further investigation is required to determine whether these findings extend to men.

The present investigation has a number of strengths. It is the largest study to date investigating the effects of exercise training on insulin sensitivity in Offspring of patients with type 2 diabetes, and thus well-powered to detect changes in the main outcome measures, and the exercise intervention was tightly controlled and quantified. The Control and Offspring groups were well-matched for age and adiposity. Diet was not strictly controlled during the intervention, however subjects were asked not to change their dietary habits during the study and subjects undertook 'gold-standard' 7-day weighed food records during the weeks preceding each OGTT: these data indicate that dietary intakes did not differ between the groups or change over the course of the intervention. It is always difficult to undertake intervention studies in pre-menopausal women because of the potentially confounding effects of menstrual cycle. However, it is important to examine this group because they are relatively understudied in the field of cardiovascular and diabetes risk and because the development of diabetes in women has a more profound effect on cardiovascular risk than in men (Mak & Haffner 2003). Attempts to control for the effects of menstrual cycle were made by performing pre- and post-intervention assessments at an interval of 8 weeks, so that, as far as possible, all measurements for each individual would be made in the same menstrual cycle phase (Creinin, Keverline, & Meyn 2004). Adopting this approach does mean that women with very long or short menstrual cycles may not have been studied in the same phase of menstrual cycle in their pre- and post-intervention tests, but the number of women to which this would apply would have been small (Creinin, Keverline, & Meyn 2004) and this is unlikely to have had a major influence on the study findings.

This study is not without limitations. Insulin sensitivity was assessed using OGTTs, rather than 'gold standard' euglycaemic hyperinsulinaemic clamps. The reason for this was pragmatic; over 160 OGTTs were performed over the course of this study and it would not have been feasible to undertake a study of this size using euglycaemic clamps. However, the ISI index used in this study correlates well with clamp-derived measures of insulin sensitivity (20) and has been widely used in the literature (915 citations from April 2000 to November 2009, ISI Web of Knowledge, <a href="http://apps.isiknowledge.com">http://apps.isiknowledge.com</a>, accessed 11 November 2009).

In conclusion, the data from the present study indicate that women with a family history of type 2 diabetes experience greater improvements in insulin sensitivity following an exercise intervention than women with no diabetes family history undertaking the same programme. Further study is needed to elucidate the mechanisms underpinning why Offspring exhibited this augmented response to exercise training and to determine the dose-response relationship for exercise-induced changes to insulin sensitivity in those with and without a family history of type 2 diabetes, which could potentially lead to refinement of physical activity guidelines for the improvement of insulin sensitivity and prevention of diabetes.

#### **CHAPTER 5**

# DETERMINANTS OF THE EXERCISE-MEDIATED CHANGE IN INSULIN SENSITIVITY

#### **5.1 Introduction**

Exercise has been shown to improve insulin sensitivity in the obese (DeFronzo, Sherwin, & Kraemer 1987), the overweight (Nassis et al. 2005), normal weight subjects (Schafer et al. 2007) and in patients with type 2 diabetes (Christ-Roberts et al. 2004; Winnick et al. 2008). Additionally, exercise has been of benefit in preventing diabetes in patients with impaired glucose tolerance (Knowler et al. 2002; Pan et al. 1997; Tuomilehto et al. 2001; Eriksson et al. 1991;Ramachandran et al. 2006) and normal glucose tolerance (Davey et al. 2005) and has been shown to be of greater benefit to insulin sensitivity in pre-menopausal women with a diabetes family history (Offspring) than age and BMI matched controls (Controls) (see Chapter 4). The latter improvement was seen despite a similar exercise 'dose' and similar improvement in cardiorespiratory fitness. However, changes in adiposity and adipose tissue hormone concentrations, particularly leptin, were observed in the Offspring group. Additionally, exercise appeared to influence fat oxidation to a greater extent in Offspring compared to Controls, and the Offspring alone displayed increased circulating leptin, and decreased fat oxidation (measured by both fasting 3-OHB concentration and by measurements of indirect calorimetry) after three days of exercise avoidance. Since leptin is strongly correlated with fat mass (Considine et al. 1996), and decreased fat oxidation is also associated with overweight and obesity (Lillioja et al. 1986), it could be argued that the acute changes in these factors occurred due to the change in body fat with exercise and that this simply accompanies, rather than influences the change in insulin sensitivity. However, this does not explain the changes in fat oxidation, 3-OHB and leptin observed after 3 days of detraining, which occurred in the absence of changes in body mass.

Reduction in fat mass is recognised to correlate with improved insulin sensitivity (Ross et al. 2000). The mechanisms underpinning this improvement are multifactorial and the mechanism by which fat loss is achieved is also likely to be important. To illustrate this point, Giugliano and colleagues (2004) found improved insulin sensitivity and reduced inflammatory markers in obese women who had lost an average of 2.7 kg of fat after liposuction (Giugliano et al. 2004). Klein and co-workers (2004) did not find improvements in metabolic health represented by increased insulin sensitivity or reduced inflammation after liposuction fat loss of around 10.5 kg (Klein et al. 2004).

However, reduction in fat mass by dietary restriction, exercise or bariatric surgery is consistently associated with reduced markers of inflammation (Zahorska-Markiewicz et al. 2008; Ryan & Nicklas. 2004; Murri et al. 2010). Reduced fat mass is also associated with increased circulating adiponectin (Polak et al. 2006) which is considered to have systemic anti-inflammatory effects, promoting insulin signalling and lipid oxidation, reduction in ectopic fat deposition and increasing NO bioavailability (Sowers 2008). Therefore through a reduction in inflammatory proteins such as TNFa and increase in adiponectin, endothelial dysfunction in skeletal muscle capillary beds and impaired insulin signalling might be ameliorated (Ziccardi et al. 2002). Reductions in circulating leptin are observed with reduced fat mass (Considine et al. 1996) and some groups have suggested that after weight loss, the reduction in leptinaemia and the associated increase in fat oxidation is evidence of improved 'leptin sensitivity' (Solomon et al. 2008). However, although mechanisms have been described which link the action of leptin to skeletal muscle fat oxidation in health and the development of apparent 'leptin resistance' in states of obesity and insulin resistance (Dyck 2009), none have yet been described to delineate cause and effect in the circumstance of improved metabolic health after weight loss. Reduced fat mass is also associated with alterations in lipid metabolism which would be expected to improve insulin sensitivity. Reduced lipolysis and increased skeletal muscle fat oxidation are reported after weight loss due to caloric restriction alone and caloric restriction plus exercise (Savage, Petersen, &

Shulman 2007;Schenk et al. 2009). Thus, reduced fat mass is also likely to improve insulin sensitivity by reducing exposure of skeletal muscle to lipid intermediates as a result of decreased fatty acid uptake and enhanced oxidation of intramyocellular lipid.

Aerobic exercise interventions are frequently associated with both improvements in insulin sensitivity, and reductions in adiposity (Boudou et al. 2003;Kim et al. 2007;Polak et al.2006;Solomon et al. 2008). However, not all exercise interventions which induce improved insulin sensitivity are associated with decreased adiposity (Nassis et al. 2005;Ostergard 2006). Therefore fat loss *per se* may not be the key determinant of exercise mediated increases in insulin sensitivity. One aim of this study was therefore to determine the role of body composition changes in mediating exercise-associated changes in insulin sensitivity.

Insulin sensitivity may be altered in adipose tissue, liver and/or skeletal muscle. However in the context of an exercise mediated increase in insulin sensitivity, the predominant target tissue is likely to be skeletal muscle (Corcoran et al. 2007;DeFronzo et al. 1981;Shulman et al. 1990), resulting in both increased insulin-dependent, and independent glucose uptake (Hawley & Lessard 2008). The final common pathway in this process appears to be increased translocation of the GLUT4 transporter (although it is unlikely that exercise mediates improved insulin sensitivity entirely via this pathway) (O'Gorman et al. 2006). As described in Section 1.16.1, a single exercise session produces acute effects on skeletal muscle which persist for up to 4 hours. These include contraction-induced GLUT4 translocation, mediated by activation of CaMK, increased capillary recruitment producing enhanced glucose delivery to muscle and may activate AMPK to promote GLUT4 cycling (Wojtaszewski & Richter 2006). Beyond 4 hours IRS-2 activity is increased and downstream atypical PKC activity mediates further GLUT4 translocation (Sakamoto & Goodyear 2002;Zierath et al. 2000;Howlett et al. 2006) and glycogen depletion is likely to stimulate AMPK further (Jensen, Wojtaszewski, & Richter 2009; McBride & Hardie. 2009). Although the proximal

parts of the insulin signalling cascade do not appear to be as acutely sensitive to exercise as the downstream components, there is some evidence from *in vitro* studies that after exercise, GLUT4 cycles to areas of the myocyte which are more insulin sensitive, leading to more GLUT4 translocation in response to insulin (Geiger et al. 2006).

With respect to aerobic training, both seven day, short duration intervention (O'Gorman et al. 2006) and chronic endurance regimes (Yu et al. 2001) have shown alteration in the expression of components of the insulin signalling cascade which are not thought to be directly responsible for improved insulin sensitivity (Yu et al. 2001;O'Gorman et al. 2006). Short term training has not shown consistent increases in expression of key components of the insulin signalling cascade although insulin receptor activity may be increased (Wojtaszewski & Richter 2006; Youngren et al. 2001) however, chronic training does increase expression of the insulin receptor, Akt and GLUT4 and glycogen synthase (Wojtaszewski & Richter 2006). Chronic exercise training may induce increased insulin sensitivity indirectly, by ameliorating pathological processes such as intramyocellular lipid accumulation and chronic inflammation. Increased lipid oxidation in exercised myocytes leads to reduced IMCL and intermediates (Goodpaster et al. 2003) (ceramides, diacylglycerol (DAG) and long-chain fatty acids) leading to improved insulin signalling by increased Akt and reduced PKC activity (Bonen, Dohm, & van Loon 2006)(Delarue & Magnan 2007;Turcotte & Fisher 2008;Wolf 2008). Alterations in muscle structure and function are likely to improve insulin sensitivity as fibres 'switch' to more oxidative type and increase capillary density (Daugaard & Richter 2001). Exercise-mediated increases in circulating adiponectin and local expression of adiponectin receptors may have direct effects on muscle lipid oxidation and indirect effects on inflammation and endothelial function (Kim et al 2007; Vu et al 2007; Sowers 2008). Therefore, whilst a sedentary lifestyle may predispose to insulin resistance partly through disrupted insulin signalling, exercise may induce improved insulin sensitivity by methods which circumvent, rather than reverse, these changes (Turcotte & Fisher 2008;O'Gorman et al. 2006).

It is difficult to generalise the role of adipose tissue associated factors (such as circulating fatty acid concentrations, rates of substrate utilisation, and adipokine concentrations) in the exercise-mediated change in insulin sensitivity, given the heterogeneity of the research methods used thus far. Variations in study subjects and training regimes in particular often provide apparently conflicting information. Polak and colleagues (2006) described an aerobic training related improvement in insulin sensitivity associated with reduced serum leptin in obese females (Polak et al 2006). However the same group reported improved insulin sensitivity without any change in adipokines in obese men participating in a resistance-type training programme (Klimcakova et al. 2006). Similarly, adiponectin has not been consistently associated with exercise mediated improvements in insulin sensitivity (Boudou et al. 2003;Nassis et al. 2005;Hulver et al. 2002;Yatagai et al. 2003) in groups ranging from lean subjects to middle aged men with type 2 diabetes. A further aim of the current study was therefore to examine the contribution of adipose tissue derived hormones to an exercisemediated change in insulin sensitivity. Additionally, since insulin resistance is characterised by reduced skeletal muscle fatty acid oxidation this study also aimed to assess the associations between whole body fat oxidation and the change in insulin sensitivity after the exercise intervention.

The aim of this study was therefore to examine the interactions between the changes in metabolic, cardiorespiratory, lifestyle and anthropometric factors and the change in insulin sensitivity, in an attempt to determine which processes contribute to the change in insulin sensitivity seen following a seven-week aerobic training programme in sedentary premenopausal women. Additionally, by identification of the factors which contribute to an exercise-mediated change in insulin sensitivity, this study aimed to describe possible metabolic pathways which might explain these changes. In view of the alterations in adiposity and circulating adipose tissue hormone concentrations in response to exercise, the

interaction between the change in these variables and the change in insulin sensitivity was of particular interest. Finally, this study aimed to determine whether exercise improves insulin sensitivity by different mechanisms in Offspring compared to Controls, or whether the fundamental difference in exercise response is mediated by the same metabolic pathway which is more responsive to exercise in the Offspring group.

#### 5.2 Methods

#### 5.2.1 Volunteers

Volunteers were recruited and screened as detailed in section 2.1. Specific inclusion criteria are defined in section 2.1.3. Further information detailing recruitment response and excluded volunteers can be found in Figure 2.2. The Offspring group was composed of women with a first-degree relative with T2D. The Control group was composed of women with no first or second degree relative with T2D. This study was conducted on twenty eight Offspring and thirty four Control volunteers. Baseline volunteer characteristics are described in Section 4.2.1.

## 5.2.2 Study Design

Volunteers were randomised into 'Exercise' or 'Delayed-Exercise' intervention arms using a computer programme. The study design is shown in Figures 2.3-2.4 (Section 2.1.3). Metabolic, anthropometric and cardiorespiratory testing (Section 2.2-2.3, 2.6) was performed at baseline and after a progressive 7-week exercise intervention (Section 2.7). Volunteers were studied at eight-week intervals to allow, as far as possible, for all metabolic, fitness and body composition assessments for each individual to be made in the same menstrual cycle phase (Creinin, Keverline, & Meyn 2004). Volunteers were asked not to alter their dietary habits during their participation in the study. Evaluation of both diet and habitual physical activity was performed at baseline and during the last week of the intervention, as described in sections 2.8 - 2.9.

#### 5.2.3 Fitness testing

Cardiorespiratory fitness was determined by calculation of predicted  $VO_{2max}$  and determination of the lactate threshold, as described in section 2.6

# **5.2.4 Metabolic testing**

Volunteers attended for metabolic testing as described in Section 2.3. Assessment of metabolic rate, substrate utilisation, blood pressure and PWV (Section 2.3.2 - 2.3.5) was made in the fasted state and 2 hours after a 75 g oral glucose tolerance test (Section 2.3.7). Fasting plasma samples were analysed for insulin, glucose, NEFA, 3-OHB, leptin, adiponectin, resistin, CRP, IL-6, TNF $\alpha$ , ALT, GGT, total cholesterol, HDL-cholesterol, and triglycerides as described in Section 2.4. During the OGTT plasma samples were obtained at 30, 60, 90 and 120 minutes and analysed for insulin, glucose, NEFA and 3-OHB (Section 2.4). Mean concentrations of glucose, insulin, NEFA and 3-OHB during the OGTT were calculated as described in section 3.2.9.1. Insulin sensitivity was expressed as the Insulin Sensitivity Index (Matsuda & DeFronzo 1999) and calculated as described in section 2.11.3.

# 5.2.5 Body composition assessment

DEXA scans were used to determine body composition and fat distribution (section 2.2.5). Height, body mass, waist and hip circumferences, and biceps, triceps, suprailiac and subscapular skinfolds were also determined using standard protocols (section 2.2) (Marfell-Jones et al. 2006).

## **5.2.6 Exercise intervention**

Volunteers underwent a 7-week endurance-type exercise training programme as described in Section 2.7.

# **5.3 Statistical analysis**

Data were analysed using Statistica (version 6.0, StatSoft Inc., Tulsa, Oklahoma) and Minitab (version 13.1, Minitab Inc., State College, Pennsylvania). Prior to analysis, all data were tested for normality using the Anderson-Darling normality test and if necessary, logarithmically transformed. Associations between variables were assessed using univariate linear correlations. To determine the extent to which variables were related to change in insulin sensitivity index independently of other variables, univariate linear regressions were then performed between the residuals for change in insulin sensitivity index of the regression between change in insulin sensitivity index (ISI) and the other variables as previously described in Section 3.2.9.2. Finally, to determine which variables were independent predictors of the change in ISI, multiple-regression analysis was performed. Statistical significance was accepted at the  $p \le 0.05$  level.

# **5.4 Results**

#### 5.4.1 Major influences on exercise mediated change in insulin sensitivity

Exercise-mediated insulin sensitivity improved to a greater extent in the Offspring group compared to Controls, although there was inter-individual variability within each group (Figure 5.1). This improvement was accompanied by significant increases in fat oxidation (as measured by 3-hydroxybutyrate concentrations (Williamson & Whitelaw 1978) in response to oral glucose, and by evaluation of the rate of fat oxidation by indirect calorimetry), and significant reductions in body mass, fat mass and circulating leptin (Section 4). These changes were observed in Offspring alone, despite similar adherence to the exercise regime, energy expended during exercise and similar improvements in cardiorespiratory fitness.



**Figure 5.1** Individual changes in insulin sensitivity index (ISI) in whole group (n = 62) after exercise intervention. Volunteers ranked in order of change in unadjusted ISI. Horizontal lines indicate mean change in ISI in Controls (solid line) and Offspring (dotted line).

## 5.4.2 Baseline influences on exercise-mediated change in insulin sensitivity

The change in insulin sensitivity in response to the exercise intervention did not correlate with baseline anthropometric or behavioural variables, but was significantly negatively associated with baseline insulin sensitivity (Figure 5.2) and with diabetes family history (with 0 and 1 included as dummy variables for negative and positive diabetes family history, respectively) in the combined group analysis. In addition, resting and post-glucose metabolic rate were positively correlated with the exercise-mediated increase in insulin sensitivity (Table 5.1).

When analysed separately, Controls displayed a positive correlation between baseline postglucose metabolic rate and change in insulin sensitivity and a negative correlation between baseline CRP and change in insulin sensitivity (Table 5.1). In Offspring, baseline insulin sensitivity alone was negatively associated with the change in insulin sensitivity after exercise (Figure 5.2).



Figure 5.2 Scattergram showing relationship between baseline ln ISI and change in ln ISI.

		Correlations with change in insulin sensitiv						
	Variable	Combined	Controls	Offspring				
Family history variables	Diabetes family history	0.27	-	-				
	mstory	(< 0.05)						
Body composition variables	Body Mass	0.10	0.17	-0.03				
		(0.45)	(0.36)	(0.87)				
	Total fat mass	0.13	0.25	-0.07				
		(0.36)	(0.19)	(0.75)				
Physiological/metabolic variables	VO <sub>2max</sub>	-0.01	0.01	0.12				
		(0.94)	(0.94)	(0.55)				
	RMR	0.29	0.17	-0.02				
		(< 0.05)	(0.34)	(0.91)				
	Post-glucose MR	0.27	0.37	-0.16				
		(< 0.05)	(< 0.05)	(0.42)				
Behavioural variables	Energy intake	0.01	0.23	-0.23				
		(0.96)	(0.22)	(0.25)				
	Average HR – Resting	-0.25	-0.17	-0.20				
	IIIX	(0.06)	(0.40)	(0.32)				
	Resting HR	0.13	-0.02	0.20				
		(0.33)	0.94	(0.31)				
Biochemical variables	ISI <sup>a</sup>	-0.47	-0.33	-0.53				
		(< 0.0005)	(0.06)	(< 0.005)				
	Leptin <sup>a</sup>	0.07	-0.16	0.18				
		(0.61)	(0.37)	(0.35)				
	CRP <sup>a</sup>	-0.01	-0.35	0.35				
		(0.95)	(< 0.05)	(0.07)				
	$TNF\alpha^{a}$	0.01	-0.09	0.18				
		(0.96)	(0.63)	(0.36)				
	adiponectin <sup>a</sup>	-0.15	0.02	-0.33				
		(0.26)	(0.92)	(0.09)				

**Table 5.1** Correlations between baseline variables and change in ISI in response to exercise

training in Combined, Control and Offspring groups.

n = 62, values are correlation coefficients with p-values in brackets. <sup>a</sup>Statistical analysis performed on logarithmically transformed data. Significant correlations in bold.

## 5.4.3 Exercise mediated changes influencing insulin sensitivity

When examined as a combined group, the exercise-mediated improvement in insulin sensitivity was associated with a reduction in circulating leptin (Figure 5.3). The change in insulin sensitivity in response to exercise also correlated significantly with a reduction in resting heart rate and an increase in the surrogate of habitual physical activity, average daily heart rate minus resting heart rate (Table 5.2). Additionally, improved exercise-mediated insulin sensitivity was associated with reduced lower body fat mass and hip circumference and was positively correlated with the net energy cost of the total exercise intervention and the net energy cost of the exercise of the final week of the intervention (Table 5.2).

Separate analysis of the Control group did not show significant correlations between the exercise-mediated change in insulin sensitivity and energy expended during the exercise intervention (Table 5.2). However, a reduction in both leptin and TNF $\alpha$  was significantly correlated with improved insulin sensitivity whilst increased fasting 3-OHB concentration was associated with improved insulin sensitivity (Table 5.2, Figure 5.3). A reduction in energy intake was also associated with improved insulin sensitivity in Controls (Table 5.2).

In Offspring, improved insulin sensitivity was positively correlated with net energy expended during the complete intervention and during the final week of exercise, but was not associated with the net energy cost of the final session (Table 5.2). In addition, reduced lower body fat, circulating CRP and leptin were also associated with improved insulin sensitivity in those with a family history of T2D (Table 5.2, Figure 5.3). Reduction in resting heart rate and increased habitual physical activity after the intervention was associated with improved insulin sensitivity (Table 5.2).

**Figure 5.3** Scattergram showing relationship between the change in ln leptin and the change in ln ISI.



Insulin sensitivity at baseline was strongly associated with adiposity in both groups. Therefore, in view of an apparent relationship between circulating leptin and both baseline insulin sensitivity, and exercise-mediated change in insulin sensitivity, correlations were performed between all measured variables and residuals of the regression of the change in ln ISI against the change in total fat mass (Figure 5.4). After correcting for fat mass in this manner, reductions in circulating leptin concentration, lower body fat mass, resting heart rate and an increase in habitual physical activity remained correlated with the change in insulin sensitivity in the combined group (Table 5.2). When analysed separately, the Controls exhibited a significant positive correlation between fasting 3-OHB concentration and change in insulin sensitivity (Table 5.2). The exercise-mediated change in insulin sensitivity in the Offspring group was associated with a reduction in lower body fat mass, hip circumference, resting heart rate and both circulating leptin and CRP concentrations (Table 5.2). Offspring also showed a significant positive correlation between fat mass adjusted change in insulin

sensitivity and the surrogate marker of habitual physical activity, average heart rate minus resting heart rate (Table 5.2).



**Figure 5.4** Individual changes in ln insulin sensitivity index (ISI) residuals after adjustment for change in fat mass in whole group (n = 61) after exercise intervention. Volunteers ranked in order of change in unadjusted ISI.

After adjustment for the change in fat mass, baseline insulin sensitivity continued to influence the change in insulin sensitivity after exercise in the combined group (r = -0.43, p < 0.01), Control group (r = -0.41, p < 0.05) and Offspring group (r = -0.49, p < 0.05). Due to the apparent influence of baseline insulin sensitivity on the exercise mediated change in insulin sensitivity, correlations were performed between all measured variables and residuals for the regression of the change in ln ISI against baseline ln ISI (Figure 5.5). After correction for baseline insulin sensitivity, only the reductions in circulating leptin concentration and lower body fat mass and increase in habitual physical activity were still significantly associated with the change in insulin sensitivity after the exercise intervention in the combined group (Table 5.2). In the Control group, after adjustment for baseline insulin sensitivity; energy intake, circulating leptin and TNF $\alpha$  concentrations were all negatively correlated with postintervention change in insulin sensitivity and change in fasting 3-OHB was positively correlated with improved insulin sensitivity (Table 5.2). In the Offspring group only reduced circulating leptin concentration remained significantly correlated with the improved insulin sensitivity after adjustment for baseline ISI (Table 5.2).



**Figure 5.5** Individual changes in ln insulin sensitivity index (ISI) residuals after adjustment for baseline insulin sensitivity in whole group (n = 62) after exercise intervention. Volunteers ranked in order of change in unadjusted ISI.

		Correlations with change in insulin sensitivity								
		Combined			Controls			Offspring		
		(a)	(b)	(c)	(a)	<b>(b</b> )	(c)	(a)	<b>(b</b> )	(c)
Intervention	Total net energy	0.33	-	-	-0.32	-	-	0.74	-	-
variables	expenditure of exercise	(< 0.05)			(0.12)			(< 0.0001)		
	Final week net exercise energy expenditure	0.33	-	-	-0.38	-	-	0.65	-	-
	energy enpendicate	(< 0.05)			(0.07)			(< 0.001)		
	Last session net energy	0.19	-	-	-0.14	-	-	0.37	-	-
	expenditure	(0.20)			(0.50)			(0.08)		
Body composition	Change in fat mass	-0.11	-	0.06	-0.04	-	-0.16	-0.05	-	0.25
vallables		(0.42)		(0.65)	(0.67)		(0.41)	(0.80)		(0.21)
	Change in lower body fat mass <sup>a</sup>	-0.35	-0.36	-0.29	-0.16	-0.14	-0.24	-0.46	-0.49	-0.31
		(< 0.01)	(< 0.01)	(< 0.05)	(0.39)	(0.45)	(0.20)	(< 0.05)	(< 0.05)	(0.11)
	Change in hip	-0.29	-0.24	-0.23	-0.09	0.08	-0.17	-0.48	-0.45	-0.29
	circuimerence	(< 0.05)	(< 0.08)	(0.08)	(0.63)	(0.66)	(0.35)	(< 0.05)	(< 0.05)	(0.14)
Physiological/ metabolic variables	Change in $\dot{VO}_{2max}$	-0.10	-0.10	-0.05	-0.24	-0.19	-0.26	-0.06	-0.04	0.13
metabolic variables		(0.42)	(0.47)	(0.72)	(0.17)	(0.32)	(0.14)	(0.78)	(0.85)	(0.51)
	Change in RMR	0.08	0.11	-0.03	-0.06	-0.12	-0.15	0.15	0.23	0.05
		(0.52)	(0.45)	(0.81)	(0.76)	(0.53)	(0.40)	(0.46)	(0.28)	(0.82)
	Change in Post-glucose	-0.12	-0.15	-0.15	-0.26	-0.34	-0.25	0.11	0.09	-0.01
	IVIN	(0.37)	(0.28)	(0.24)	(0.13)	(0.07)	(0.15)	(0.60)	(0.71)	(0.97)

Table 5.2 Correlations between post-exercise variable change and change in ln ISI unadjusted (a) and adjusted for: (b) fat mass, (c) baseline ISI.

Table 5.2 continued		Correlations with change in insulin sensitivity										
			Combined		Control			Offspring				
		(a)	<b>(b</b> )	(c)	(a)	<b>(b</b> )	(c)	(a)	<b>(b)</b>	(c)		
Behavioural variables	Change in Energy intake	-0.09	0.00	-0.10	-0.43	-0.30	-0.41	0.11	0.14	0.11		
variables	intuke	(0.53)	(0.99)	(0.48)	(< 0.05)	(0.15)	(< 0.05)	(0.60)	(0.50)	(0.59)		
	Change in Resting HR	-0.27	-0.34	-0.20	-0.23	-0.20	-0.22	-0.41	-0.48	-0.19		
		(< 0.05)	(< 0.05)	(0.13)	(0.22)	(0.33)	(0.24)	(< 0.05)	(< 0.05)	(0.33)		
	Change in Average HR	0.41	0.35	0.31	0.21	0.20	0.25	0.50	0.48	0.35		
	- Resting HK	(< 0.01)	(< 0.01)	(< 0.05)	(0.30)	(0.33)	(0.22)	(< 0.01)	(< 0.05)	(0.07)		
Biochemical	Change in leptin <sup>a</sup>	-0.43	-0.32	-0.40	-0.35	-0.31	-0.44	-0.40	-0.39	-0.42		
variables		(< 0.0005)	(< 0.05)	(< 0.01)	(< 0.05)	(0.10)	(< 0.05)	(< 0.05)	(< 0.05)	(< 0.05)		
	Change in CRP <sup>a</sup>	-0.05	-0.16	-0.01	0.18	0.10	0.18	-0.53	-0.51	-0.30		
		(0.70)	(0.25)	(0.94)	(0.31)	(0.61)	(0.32)	(< 0.01)	(< 0.01)	(0.12)		
	Change in TNF $\alpha^a$	-0.09	-0.09	-0.13	-0.35	-0.34	-0.38	0.14	0.13	0.13		
		(0.47)	(0.50)	(0.30)	(< 0.05)	(0.06)	(< 0.05)	(0.47)	(0.52)	(0.50)		
	Change in Fasting 3-	0.13	0.20	0.24	0.41	0.40	0.38	-0.06	0.02	0.09		
	OHR	(0.92)	(0.13)	(0.06)	(< 0.05)	(< 0.05)	(< 0.05)	(0.94)	(0.91)	(0.67)		

n = 62, values are correlation coefficients with p-values in brackets. (a): unadjusted data; (b): variables correlated against ln ISI adjusted for change in fat mass; (c): variables correlated against ln ISI adjusted for baseline ln ISI. <sup>a</sup>Statistical analysis performed on logarithmically transformed data. Significant correlations in bold.

The exercise-mediated change in insulin sensitivity was significantly correlated with the net energy cost of the total exercise intervention and the net energy cost of the exercise performed in the final week of the study. In order to determine which variables might influence exercise-mediated alterations in insulin sensitivity, correlations were performed between all measured variables and residuals for the regression of the change in ln ISI against firstly: the net energy cost of the total exercise intervention and secondly: the net energy cost of the study of the intervention (Figure 5.6). The energy cost of the final session of the intervention was not associated with the change in insulin sensitivity in either the combined, Control or Offspring groups.

In the combined group, changes in lower body fat, resting heart rate and circulating leptin were negatively correlated, and changes in habitual physical activity positively correlated, with the exercise-mediated change in insulin sensitivity after adjustment for net total exercise energy expenditure (Table 5.3). After adjustment for the energy cost of the final week the previously observed association with change in resting heart rate was no longer statistically significant (Table 5.3). After adjustment for the net energy cost of the total exercise intervention, the Control group displayed significant negative correlations between the changes in energy intake, circulating leptin and TNF $\alpha$  and the change in insulin sensitivity (Table 5.3). A positive correlation between the change in fasting 3-OHB concentration and the change in insulin sensitivity was observed after adjustment for the net energy expended during the final week of the intervention (Table 5.3). In the Offspring group adjustment for both the net total energy expenditure and the net energy expenditure of the final week of the intervention did not alter the significant correlations observed in unadjusted regressions, except for the association between the change in leptin concentrations and the change in insulin sensitivity, which was no longer significant (Table 5.2, 5.3).



**Figure 5.6** Individual changes in ln ISI residuals after adjustment for (a) net total energy expenditure of exercise, (b) net total energy expenditure of final week of exercise in whole group (n = 60). Volunteers ranked in order of change in unadjusted ISI.

		Correlations with change in insulin sensitivity								
		Combined			Controls			Offspring		
		( <b>d</b> )	(e)	( <b>f</b> )	( <b>d</b> )	<b>(e)</b>	<b>(f)</b>	( <b>d</b> )	(e)	( <b>f</b> )
Body composition variables	Change in fat mass	-0.11	-0.10	-0.05	-0.01	0.04	-0.03	-0.08	-0.06	-0.05
vulluoites		(0.42)	(0.48)	(0.73)	(0.95)	(0.98)	(0.88)	(0.69)	(0.78)	(0.80)
	Change in lower body fat mass <sup>a</sup>	-0.28	-0.32	-0.35	-0.09	-0.14	-0.16	-0.38	-0.44	-0.46
		(< 0.05)	(< 0.05)	(< 0.01)	(0.67)	(0.46)	(0.40)	(< 0.05)	(< 0.05)	(< 0.05)
	Change in hip	-0.23	-0.21	-0.25	-0.05	0.15	0.12	-0.39	-0.45	-0.47
	eneumerenee	(0.08)	(0.11)	(0.06)	(0.79)	(0.43)	(0.53)	(< 0.05)	(< 0.05)	(< 0.05)
Physiological/ metabolic variables	Change in VO <sub>2max</sub>	-0.17	-0.09	-0.07	-0.32	-0.16	-0.16	-0.11	-0.09	-0.06
		(0.20)	(0.49)	(0.58)	(0.08)	(0.40)	(0.41)	(0.58)	(0.65)	(0.76)
	Change in RMR	0.06	0.09	0.09	-0.06	-0.02	-0.03	0.12	0.14	0.15
		(0.65)	(0.50)	(0.49)	(0.74)	(0.90)	(0.87)	(0.54)	(0.47)	(0.47)
	Change in Post-glucose MR	-0.14	-0.11	-0.12	-0.27	-0.21	-0.25	0.07	0.08	0.10
		(0.30)	(0.42)	(0.39)	(0.14)	(0.25)	(0.17)	(0.73)	(0.68)	(0.62)

**Table 5.3** Correlations between post-exercise variable change and ln ISI adjusted for (d) net total energy expenditure of the exercise

intervention, (e) net total energy expenditure of final week of exercise, (f) net energy expenditure of final exercise session.

Table 5.3 continued	l	Correlations with change in insulin sensitivity										
			Combined Control						Offspring			
		( <b>d</b> )	<b>(e)</b>	( <b>f</b> )	( <b>d</b> )	(e)	<b>(f)</b>	( <b>d</b> )	<b>(e)</b>	( <b>f</b> )		
Behavioural variables	Change in Energy intake	-0.06	-0.05	-0.05	-0.49	-0.33	-0.33	-0.20	0.12	0.11		
variableb	intuite	(0.70)	(0.73)	(0.73)	(< 0.05)	(0.11)	(0.10)	(0.30)	(0.56)	(0.60)		
	Change in Resting HR	-0.27	-0.26	-0.27	-0.20	-0.22	-0.22	-0.42	-0.40	-0.42		
		(< 0.05)	(0.06)	(< 0.05)	(0.29)	(0.27)	(0.27)	(< 0.05)	(< 0.05)	(< 0.05)		
	Change in Average HR	0.37	0.39	0.41	0.25	0.21	0.21	0.40	0.46	0.49		
	- Resting HR	(< 0.01)	(< 0.01)	(< 0.01)	(0.24)	(0.30)	(0.30)	(< 0.05)	(< 0.05)	(< 0.01)		
Biochemical	Change in leptin <sup>a</sup>	-0.38	-0.41	-0.44	-0.36	-0.27	-0.30	-0.30	-0.36	-0.39		
variables		(< 0.01)	(< 0.01)	(< 0.01)	(< 0.05)	(0.15)	(0.10)	(0.12)	(0.06)	(< 0.05)		
	Change in CRP <sup>a</sup>	-0.07	-0.07	-0.08	0.16	0.11	0.10	-0.53	-0.53	-0.53		
		(0.62)	(0.59)	(0.54)	(0.39)	(0.56)	(0.58)	(< 0.01)	(< 0.01)	(< 0.01)		
	Change in $TNF\alpha^a$	-0.16	-0.08	-0.08	-0.52	-0.32	-0.32	0.16	0.15	0.14		
		(0.23)	(0.54)	(0.57)	(< 0.01)	(0.08)	(0.08)	(0.43)	(0.46)	(0.47)		
	Change in Fasting 3-	0.06	0.21	0.20	0.22	0.42	0.41	-0.05	0.01	0.02		
	ОПВ	(0.63)	(0.12)	(0.13)	(0.23)	(< 0.05)	(< 0.05)	(0.79)	(0.98)	(0.94)		

n = 62, values are correlation coefficients with p-values in brackets. (d): ISI adjusted for total net energy expenditure of intervention; (e): ISI adjusted for total net energy expenditure of final week of intervention; (f): ISI adjusted for total net energy expenditure of final session of intervention. <sup>a</sup>Statistical analysis performed on logarithmically transformed data. Significant correlations in bold.

#### **5.4.4 Multivariate analysis**

To determine whether the greater improvement in insulin sensitivity in the Offspring in response to the exercise intervention could be explained by differences in changes in fitness, energy intake, body fat, compliance to the exercise intervention or energy expended during the intervention between the Control and Offspring groups, multivariate regression analyses were undertaken. Change in insulin sensitivity was the response variable and group (Control or Offspring), together with number of minutes of exercise completed, energy expended during the intervention, change in body fat, change in energy intake, change in  $\dot{V}O_{2max}$  or change in  $\dot{V}O_2$  at lactate threshold, as the predictor variables. In all of these analyses, group remained a significant predictor of the change in insulin sensitivity in response to the intervention (all p < 0.05), whereas none of the other factors included in the models were significant predictors of the change in insulin sensitivity. Thus, the finding that insulin sensitivity increased to a greater extent in the Offspring than the Control group was not confounded by differences in compliance to the exercise intervention, energy expended during exercise, differences in fat loss or differences in changes in fitness between the groups.

When change in leptin concentration and group were included in a multivariate regression model, change in leptin was a significant predictor of change in insulin sensitivity (p < 0.01), whereas the statistical significance of group was lost (p = 0.29), indicating that the greater change in insulin sensitivity in the Offspring group was associated with their greater change in circulating leptin. When change in fat mass was added to this model, change in leptin remained a significant predictor of change in insulin sensitivity (p < 0.0005). Additionally, in multivariate regression analysis including baseline insulin sensitivity and group as the predictor variables, baseline insulin sensitivity was a stronger significant predictor of the change in insulin sensitivity (p < 0.0005) than group (p = 0.13) (Figure 5.3). When baseline insulin sensitivity, change in leptin and group were included in the same model, baseline insulin sensitivity (p < 0.0005) and change in leptin (p < 0.0005) were both independent

predictors of change in insulin sensitivity, whereas the significant effect of group was lost (p = 0.81). In view of the correlations observed in section 5.3, the change in CRP, adiponectin, TNF $\alpha$  and 3-OHB were added to the model, however only the change in leptin (p < 0.0005) and baseline insulin sensitivity (p < 0.0005) remained independent predictors of the change in insulin sensitivity. The addition of behavioural variables such as the change in energy intake, resting heart rate and habitual physical activity to the model determined similar results for the change in leptin and baseline insulin sensitivity but added the change in energy intake (p < 0.05) as a significant independent predictor of exercise-mediated change in insulin sensitivity. This model, which included all factors with significant univariate correlations, explained 47.9% of the variance in the change in insulin sensitivity in response to exercise (adjusted R<sup>2</sup>).

#### **5.5 Discussion**

The main findings of this study were that improved exercise-mediated insulin sensitivity was observed in those who were most insulin resistant at baseline. This differs slightly from the results in Chapter 4 where Offspring displayed the greatest improvement in insulin sensitivity therefore baseline insulin sensitivity is a stronger predictor of exercise response than a family history of diabetes. Additionally, the insulin sensitivity response was greatest in those who reduced circulating leptin concentrations, resting heart rate and lower limb fat mass and those who increased habitual physical activity and expended the most energy during the intervention. Despite the strong associations between baseline adiposity and insulin sensitivity, the associations between the exercise-mediated increase in insulin sensitivity and leptin concentration, baseline insulin sensitivity, lower limb fat mass and habitual physical activity were independent of changes in fat mass. Reductions in circulating leptin, lower body fat mass and increased habitual physical activity continued to positively influence insulin sensitivity after adjustment for the influence of baseline insulin sensitivity. After adjustment for the net energy cost of the exercise intervention, reduction in lower body fat mass, circulating leptin and an increase in habitual physical activity remained associated with improved insulin sensitivity after exercise.

Exercise-mediated changes in anthropometric variables were associated with improved insulin sensitivity in response to the exercise programme. A reduction in lower body fat mass was associated with improved post-exercise insulin sensitivity in the combined and Offspring groups and was independent of reduced total fat mass, energy expended during exercise and in the combined group, baseline insulin sensitivity. The observation of a reduction in lower limb fat mass appearing to influence insulin sensitivity is surprising, since reductions in visceral and total abdominal adipose tissue are more frequently associated with this outcome (Boudou et al. 2003;O'Leary et al. 2006). However, this association has previously been reported in a smaller study of obese subjects (Goodpaster et al. 2003), and it has been

suggested that thigh fat may be a marker of intramuscular or deep subcutaneous fat accumulation, which is recognised to correlate with insulin sensitivity (Goodpaster, Thaete, & Kelley 2000;Ryan, Nicklas, & Berman 2002;Yim et al. 2007). In the Offspring group, reduced lower body fat did not correlate with insulin sensitivity after adjustment for baseline insulin sensitivity and was not a significant independent predictor of exercise-mediated increased insulin sensitivity in multivariate analysis. The participants in this study engaged in a variety of exercise activities, however although the majority of these were lower limb dominant (walking, running, cycling) there is no evidence that this preferentially favours lower limb fat reduction (McTiernan et al. 2007;Nindl et al. 2000;Wilmore et al. 1999). It is therefore unclear why reduced lower body fat was correlated with improved insulin sensitivity, and why this observation was not significant in Controls. More detailed imaging of adipose tissue compartments, direct assessment of intramuscular lipid content or greater knowledge of the specific metabolic role of deep subcutaneous adipose tissue (Sniderman et al. 2007) would be required to provide evidence which would agree with the observations of Goodpaster and colleagues (Goodpaster, Thaete, & Kelley 2000;Goodpaster et al. 2003).

Lifestyle factors were associated with increased insulin sensitivity in response to exercise. The surrogate index of habitual physical activity, average heart rate minus resting heart rate, was associated with increased insulin sensitivity in the combined and Offspring, but not the Control groups. Increased habitual physical activity was associated with improved insulin sensitivity in the combined group independent of change in fat mass, baseline insulin sensitivity and energy expended during exercise. Increased habitual physical activity was correlated with improved insulin sensitivity in Offspring independent of change in fat mass and energy expenditure of exercise but not baseline insulin sensitivity. Offspring showed a significant increase in habitual physical activity during the intervention and the observation that this factor was not independent of baseline insulin sensitivity may suggest that the more insulin resistant volunteers were those who increased physical activity most.

A reduction in resting heart rate was also correlated with increased insulin sensitivity in the combined and Offspring groups, and was independent of change in fat mass and the energy expenditure of exercise but not of baseline insulin sensitivity. Reduced resting heart rate can be considered to be a crude marker of improved cardiorespiratory fitness (Jurca et al. 2005), however a more accurate description of the change in cardiorespiratory fitness, the change in predicted VO<sub>2max</sub>, was not associated with increased insulin sensitivity in any of the groups. High resting heart rate may be indicative of dysfunction of the autonomic nervous system which in turn is recognised to influence inflammatory responses and glucose metabolism (Carnethon & Craft 2008;Tracey 2002). Chronic sympathetic over-stimulation may promote an inflammatory and pro-glycaemic state which contributes to insulin resistance and endothelial dysfunction (Carnethon et al. 2003; Piwonska et al. 2008; Thayer & Fischer 2009). Sympathetic hyperactivity has been observed in offspring of subjects with T2D and correlates with insulin resistance (Huggett et al. 2006). It is unclear whether sympathetic hyperactivity is a cause, or an effect, of insulin resistance (Huggett et al. 2006); however sympathetic hyperactivity could theoretically augment insulin resistance by stimulation of chronic inflammation (Egan 2003). Therefore, reduction in resting heart rate in response to exercise may represent reversal of early autonomic dysfunction which is associated with insulin resistance and a mechanism by which exercise might improve insulin sensitivity (Giallauria et al. 2008b). In this study, reduced resting heart rate was associated with improved insulin sensitivity in Offspring and combined groups but was not independent of baseline insulin sensitivity. This may suggest that those with greater baseline insulin resistance also have a degree of early autonomic dysfunction which contributes to the aetiology of the insulin resistance but which is sensitive to the effects of exercise (Frontoni et al. 2003; Giallauria et al. 2008).

Baseline resting and post-glucose metabolic rates were correlated with an exercise-mediated increase in insulin sensitivity. Higher resting metabolic rates have been recorded in obese compared to lean women (Weststrate et al. 1990), therefore it might have been expected to

observe a correlation between reductions in metabolic rate and improved insulin sensitivity which was dependent upon change in fat mass. However, no change in metabolic rate or substrate utilisation in response to the exercise intervention was observed to significantly correlate with the change in insulin sensitivity. This is a surprising finding, particularly in the Offspring group who manifest significantly increased fat oxidation and reduced carbohydrate oxidation after the intervention (Section 4.4.3.8). Furthermore, there is consistent evidence to support an association between increased fat oxidation and insulin sensitivity in other studies (Gan et al. 2003;Goodpaster et al. 2003). However, these associations were reported in obese subjects and have not been observed in relatives of patients with type 2 diabetes (Ostergard et al. 2006). Although increases in fat oxidation are recognised in response to an exercise intervention (Blaak & Saris 2002), these changes may not influence insulin sensitivity directly, given that a weight loss intervention has been shown to increase insulin sensitivity to the same degree as a weight loss plus exercise intervention, but with increased oxidative capacity observed in the exercise group alone (Schenk et al. 2009). Additionally, Schenk and colleagues (2009) have shown that despite increased fat oxidation in the exercise plus weight loss group, lipid infusion can reverse the improved insulin sensitivity to the same extent in both groups (Schenk et al. 2009). It has also been previously suggested that whole body indirect calorimetry may be a relatively crude method of assessing alterations in skeletal muscle fat oxidation (Galgani et al. 2008;Galgani, Moro & Ravussin 2008). Although exercise has been shown to increase fat oxidation in association with reduced circulating leptin and reduced intramuscular lipid content it has not been associated with the observed increase in insulin sensitivity (Solomon et al. 2008).

Improved insulin sensitivity in Controls alone was correlated with an increase in fasting hepatic fatty acid oxidation (as assessed by 3-OHB concentrations), and was independent of both baseline insulin sensitivity and change in fat mass. However, this association was not independent of the net energy cost of the exercise intervention. Exercise is recognised to increase hepatic fat oxidation (Gill et al. 2007), reduce hepatic steatosis and increase total

body insulin sensitivity (Shah et al. 2009b). However, a six-week aerobic exercise intervention has been associated with improved hepatic insulin sensitivity (characterised by reduced endogenous glucose production in response to insulin infusion) without evidence of reduced steatosis (Shojaee-Moradie et al. 2007). Therefore a mechanism in which exercise stimulates oxidation of intrahepatic lipid directly resulting in increased insulin sensitivity would appear to be an oversimplification (Johanson et al. 2003). Black and colleagues (2005) observed improved hepatic insulin sensitivity after exercise only in the context of energy deficit however measurements of total body and hepatic fatty acid metabolism were not performed (Black et al. 2005). Neither Controls nor Offspring displayed a significant increase in fasting hepatic fat oxidation after the exercise intervention (Section 4.4.3.2) and the absence of a significant independent contribution to the exercise mediated change in insulin sensitivity in multivariate analysis suggests that increased fasting hepatic fatty acid oxidation is not a key determinant of the exercise-mediated change in insulin sensitivity.

Reductions in circulating TNF $\alpha$  were correlated with increased insulin sensitivity in the Control group, however these associations were not observed after adjustment for the change in fat mass or the net energy cost of the final week and final session of the intervention. Exercise intervention in obese women has been shown to reduce circulating TNF $\alpha$ concentrations (Kondo, Kobayashi, & Murakami 2006), and a combined diet and physical activity intervention in obese women has shown a significant association between the amount of weight lost and the reduction in circulating TNF $\alpha$  (Zahorska-Markiewicz et al. 2008). The results from the current study would be consistent with the previous observations that exercise-mediated reductions in circulating TNF $\alpha$  are more strongly associated with fat mass than with insulin sensitivity (Polak et al. 2006). Additionally, it has been suggested that TNF $\alpha$  influences insulin sensitivity in a paracrine fashion and that circulating TNF $\alpha$  may not be the best marker of the influence of changes in the TNF system on insulin sensitivity (Straczkowski et al. 2001). Thus, the observed association between reduced TNF $\alpha$  and insulin

sensitivity in Controls is likely to be related to reduced fat mass, rather than a direct effect on insulin sensitivity and this suggestion is supported by the multivariate regression model.

Reduced CRP was significantly correlated with improved insulin sensitivity in the Offspring group but not the combined or Control groups. The association was independent of the net energy cost of the intervention and the influence of reduced fat mass, but was not significant after adjustment for baseline insulin resistance. Dekker and colleagues (2007) observed greater CRP concentrations in obese subjects, compared to lean subjects and subjects with T2D (Dekker et al. 2007). Changes in insulin sensitivity in response to exercise were not associated with change in CRP in this study, but it must be appreciated that Dekker and coworkers analysed the effects of an exercise intervention with preserved body mass and that this study had only 8 subjects in each group and was under-powered to detect correlations between insulin sensitivity and other variables which did not change over the intervention (Dekker et al. 2007). Balducci and colleagues (2009) described reductions in CRP after a twelve month exercise intervention in patients with type 2 diabetes (Balducci et al. 2009). Balducci and co-workers observed greater reduction in CRP and other indices of inflammation and increased insulin sensitivity in those who participated in high intensity aerobic activity, and a maximal response in those who engaged in a mixed aerobic and resistance regime (Balducci et al. 2009). However, there was no reported correlation between the reduction in CRP and the increase in insulin sensitivity and the methods of assessing insulin sensitivity (HOMA-IR) in subjects with confirmed T2D who are treated with a range of pharmaceuticals can be questioned (Anderson et al. 1995). A number of other authors have found that exercise-mediated improvements in insulin sensitivity are unrelated to changes in CRP (Marcell et al. 2005; Oberbach et al. 2006), whilst Kadoglou and colleagues (2007) reported a significant reduction in CRP after exercise in patients with T2D, but did not report whether this was associated with improved insulin sensitivity (Kadoglou et al. 2007). The results from the current study would suggest that CRP is a marker, rather than a mediator, of insulin resistance and that it is indicative of a potential beneficial effect of exercise only in

that it identifies those who are more insulin resistant. This supposition would be supported by the absence of change in CRP as a significant independent predictor of the exercise-mediated change in insulin sensitivity in multivariate analysis.

Reduced circulating leptin was significantly correlated with improved insulin sensitivity in the combined, Control and Offspring groups. However the association was dependent upon adiposity and the net energy expended during the final week and final session of the intervention in the Control group. In Control subjects the influence of reduced leptin appears to be inter-related to reduced fat mass, it is therefore difficult to be certain whether reduction in leptin exerts an insulin sensitising effect or whether it is simply a marker of adiposity. The relationship between leptin and insulin sensitivity was independent of the energy cost of the intervention, but not the energy cost of either the final week or the final session. This would suggest that in Controls, the effect of leptin on insulin sensitivity is related to reduced fat mass and an acute effect of exercise (Olive & Miller 2001). For Controls, one unifying pathway which links the associations between changes in energy intake, fasting hepatic fatty acid oxidation, leptin concentration and the exercise-mediated change in insulin sensitivity is through stimulation of AMP-kinase (AMPK). Leptin is recognised to act centrally (via AMPK) on the hypothalamus to reduce appetite and to stimulate fat oxidation in hepatic and skeletal muscle cells (Hutchinson, Summers, & Bengtsson 2008). However, no change in total body fat oxidation was observed in Controls therefore the suggestion that improved leptin sensitivity might mediate improved insulin sensitivity through increased fat oxidation cannot be substantiated by these results.

In Offspring, the influence of reduced leptin on insulin sensitivity was independent of the change in fat mass and the degree of baseline insulin resistance, but not the net energy cost of the total or final week of the intervention. This may reflect a mechanism by which energy deficit, independent of fat loss leads to reduction in circulating leptin. The relationship

between exercise-mediated energy deficit and reduced leptin concentration has previously been described (McMurray & Hackney 2005). Minokoshi and colleagues (2002) have identified increased skeletal muscle fat oxidation triggered by leptin-stimulated, AMPKmediated suppression of acetyl CoA carboxylase (Minokoshi et al. 2002). In addition, previously observed increases in insulin sensitivity and reductions in leptin are thought to be related to the impact of exercise on fat oxidation, rather than the change in fat mass (Goodpaster et al. 2003;Solomon et al. 2008). However despite significantly increased fat oxidation in Offspring, the exercise-mediated improvement in insulin sensitivity in this group was not correlated with fat oxidation. Therefore, despite a plausible metabolic pathway by which reduced leptin concentrations reflect improved leptin sensitivity which in turn mediates improved fat oxidation and subsequently increased insulin sensitivity, the results of this chapter do not support this supposition.

Despite a positive correlation between adiponectin and insulin sensitivity at baseline, there were no associations between the change in adiponectin and the change in insulin sensitivity after the exercise intervention. Positive correlations between improvement in insulin resistance and adiponectin have previously been observed (Bluher et al. 2006;Oberbach 2006). However, Oberbach and colleagues only observed this association in subjects with impaired glucose tolerance (IGT) or type 2 diabetes (Oberbach et al. 2006) and although Bluher and co-workers did observe a change in subjects with normal glucose tolerance, this was in an older population than the current study. Monzillo and colleagues also illustrated improved insulin sensitivity correlated with increased adiponectin, however this was only found in patients with type 2 diabetes, and in response to a combined physical activity and hypocaloric diet regime (Monzillo et al. 2003). This is consistent with the work of Hotta and colleagues (2000), who observed increased adiponectin after dietary induced weight loss (Hotta et al. 2000). Other recent studies have observed exercise mediated changes in insulin sensitivity in subjects with type 2 diabetes (Boudou et al. 2003), obese women (Polak et al. 2006), and subjects with IGT which are not associated with changes in circulating

adiponectin. Additionally, Polak and colleagues (2007) reported an increase in total and multimeric adiponectin but no relationship between adiponectin and insulin sensitivity at baseline or after dietary-induced weight loss (Polak et al. 2007). There may be a number of reasons why adiponectin is not significantly associated with exercise mediated change in insulin sensitivity in the current study. Firstly, the volunteers in the current study were normally glucose tolerant, and therefore compared to subject groups who did show an association between exercise mediated change in insulin sensitivity and an increase in plasma adiponectin, the current volunteers may be too insulin sensitive to trigger an exercise mediated response in adiponectin (Oberbach et al. 2006). Additionally, if adiponectin acts to stimulate hepatic fat oxidation as a compensatory response to increased circulating NEFA (Bonet et al. 2007), then increased lipid oxidation in muscle in response to improved leptin sensitivity (Solomon et al. 2008) may downregulate adiponectin, particularly since adiponectin may increase skeletal muscle lipid oxidation in a similar AMPK dependent pathway to both leptin and exercise (Boudou et al. 2003; Yamauchi et al. 2002). Finally, adiponectin has recently been discovered to have more than one active form, and therefore the assay used in this (and other) studies may not have detected changes in different types of adiponectin, which may have differing metabolic properties (Yamauchi & Kadowaki 2008).

Multivariate regression analysis produced a model in which parental history of diabetes was a significant predictor of exercise mediated improvement in insulin sensitivity. However, any individual, regardless of family history, who displays insulin resistance prior to exercise, or who displays an ability to reduce circulating leptin (independent of fat mass) would be expected to improve insulin sensitivity after an exercise intervention to a greater degree than a matched individual who is more insulin sensitive or who does not modulate leptin in response to exercise. This observation is consistent with previous evidence which found serum leptin to be a stronger predictor of insulin resistance than diabetes family history (Nyholm et al. 1997). Reduction in circulating leptin also predicted improvements in insulin resistance in individuals with IGT who participated in a diet and exercise intervention
(Corpeleijn et al. 2007). To date, this study is the first to describe improved insulin sensitivity, mediated by reduced circulating leptin in response to exercise, in relatives of patients with type 2 diabetes.

This study has a number of limitations, the imaging modality utilised to assess body composition was unable to delineate separate subcutaneous adipose compartments, or determine intramuscular lipid content, both of which would have been of value in examining the anthropometric mediators of exercise-mediated change in insulin sensitivity. In addition, investigation of fat oxidation was limited to whole body assessment by indirect calorimetry and may be not be sensitive enough to determine specific alterations in skeletal muscle oxidative function, however this method was sensitive enough to determine a different postintervention effect in Controls and Offspring. This study did not assay high molecular weight adiponectin, or circulating TNFa receptor concentrations and therefore lacks the ability to more completely determine adipose hormone function. Further exploration of adipose tissue hormones and exercise is warranted to determine the interaction between change in fat mass, energy deficit and insulin sensitivity. The analysis contained within this chapter was an attempt to determine whether exercise-meditated alterations in other variables were significantly associated with the change in insulin sensitivity. For this reason, the results and conclusions contained within this chapter cannot be considered to be definitive evidence of a mechanistic relationship. Instead, these associations suggest possible mechanisms allowing the generation of hypotheses which relate exercise to alterations in metabolic health and are therefore intended to provide a basis for future hypothesis driven research. Although it may be argued that a more focused approach to the mechanisms which mediate the insulinsensitising response might have been appropriate, the complexity of studying a large group of free living individuals should be appreciated therefore inclusion of variables such as diet and habitual activity were felt to be important to prevent confounding. In addition, whilst the inclusion of further variables in the analysis may reduce the statistical power of the conclusions it was preferable to minimise the statistical significance of any interaction than to

ignore a potential confounding factor which would obscure associations which might guide further study.

In summary, exercise-mediated increase in insulin sensitivity can be predicted by a positive family history of type 2 diabetes, although baseline insulin resistance and an ability to reduce circulating leptin are stronger predictors of an exercise related response. In terms of the exercise-mediated change in insulin sensitivity, greater responses were observed in volunteers who were most insulin resistant at baseline, who improved leptin sensitivity, and who increased habitual physical activity levels. Only lower body adiposity was associated with an exercise-mediated increase in insulin sensitivity, whilst total body fat oxidation did not influence the change in insulin sensitivity.

Women without a family history of diabetes stimulate improved insulin sensitivity after exercise in concert with reductions in TNF $\alpha$ , leptin, energy intake and increased fasting hepatic fat oxidation however, these changes are likely to be dependent on reduced fat mass and an acute effect of energy deficit. Women with a family history of diabetes increase insulin sensitivity in association with reduced leptin, CRP, lower body fat, resting heart rate and increased physical activity, independent of fat mass. These factors are also independent of the energy deficit of the intervention but not baseline insulin resistance. A putative mechanism which is currently understudied in Offspring subjects is the interaction between exercise and sympathetic hyperactivity. This study is unable to determine the pathways by which these factors might affect baseline insulin resistance, although this information provides more evidence that planned aerobic exercise and increased habitual activity has greater metabolic effects on those with greater metabolic dysfunction related to insulin resistance. Reduced circulating leptin was an independent predictor of improved insulin sensitivity, and was independent of reduced fat mass and baseline insulin sensitivity but not the energy deficit of exercise. Reduced circulating leptin would appear to mediate an insulin-

sensitising effect through a different, but currently incompletely described pathway in Offspring compared to Controls.

Future studies are clearly required to identify the mechanisms by which exercise improves insulin sensitivity, particularly in Offspring where the mechanism may be less dependent on reduced fat mass than in Controls. Detailed imaging to identify changes in adipose tissue compartments and intramuscular fat storage would be of interest, particularly to determine whether the observed association with reduced lower body fat is a marker of intramuscular lipid content. In addition, more detailed assessment of the inflammatory cascade (including TNF $\alpha$  receptor status, CAM assay, adipocyte size and adipose tissue macrophage content and examination of the dynamic function of adipose with respect to NEFA and adipokine efflux) would be of value. This should be paired with measurement of microvascular function to determine the anti-inflammatory effect of exercise. Further exploration of the role of autonomic hyperactivity is a further area of study which theoretically links low grade inflammation and microvascular disease. However, prior to an exercise intervention trial exploration of the cause-and-effect relationship between autonomic activity and inflammation would be required since mechanisms have been suggested which consider both factors as causal (Thayer 2009).

If the primary cause of skeletal muscle insulin resistance is impaired microvascular function (Serne et al. 2006) then improvement in microvascular function in response to exercise would be expected to be closely associated with the magnitude of increased insulin sensitivity (Pedersen 2006). If in addition to this, reduction in both adipokine release from adipose tissue and inflammatory activity in skeletal muscle were observed in concert with improved skeletal muscle capillary recruitment and insulin sensitivity, then this would be indicative of a metabolic process of local and systemic inflammation which is prone to modulation by exercise and which mediates improved insulin sensitivity by increased endothelial function. However whilst this finding would suggest a metabolic pathway which is sensitive to

exercise, it does not describe specific mechanisms which effect the increase in insulin sensitivity.

Description of putative metabolic pathways may allow specific mechanistic studies of the final determinants of change in insulin sensitivity in response to exercise. With respect to the exercise-mediated change in insulin sensitivity in Offspring, the association with reduced circulating leptin represents an important target for further study. Reduction in circulating leptin has consistently been associated with increased insulin sensitivity and increased fat oxidation (Barwell et al. 2008;Corpeleijn et al. 2007;Polak et al. 2006;Solomon et al. 2008). The data in this chapter does not suggest that reduced circulating leptin mediates a direct insulin sensitivity' requires investigation of other 'leptin-centric' mechanisms (such as insulin-stimulated phosphorylation of the AS160 substrate of Akt (Alkhateeb et al. 2009) or AMPK activation (Uotani, Abe & Yamaguchi. 2006;Dyck 2009)) in skeletal muscle after an exercise intervention.

#### CHAPTER 6

# DETERMINANTS OF FAT LOSS IN RESPONSE TO EXERCISE IN PREVIOUSLY SEDENTARY WOMEN

# **6.1 Introduction**

It is well-established that extent of weight or fat loss in response to exercise training varies between individuals (Byrne et al. 2006;King et al. 2008;Snyder et al. 1997). The extent of compliance to the exercise intervention clearly contributes to this variation (Byrne et al. 2006;Colley et al. 2008;McTiernan et al. 2007) but even when this is accounted for, differences in weight and fat loss between individuals are observed (King et al. 2008;Snyder et al. 1997). Failure to achieve predicted weight loss due to alterations in energy expenditure has been termed 'adaptive thermogenesis' and its origin is multifactorial, suggestive of both behavioural and metabolic effects (Major et al. 2007).

It has been suggested that individual differences in compensatory adjustments to the increased exercise energy expenditure are responsible for this variability (King et al. 2007). Indeed, it has recently been reported that those who lost less weight than predicted in response to an exercise intervention increased their energy intake over the course of the intervention, whereas those who lost more weight than predicted decreased energy intake, although there was no overall change in energy intake before and after the intervention for the group as a whole (King et al. 2008). Additionally, those who compensate for the energy deficit of exercise by increased energy intake may show greater desire for food after exercise, and may preferentially select energy dense foods after an exercise session (Finlayson et al. 2009). There may also be a difference between compensatory energy intake behaviour in males compared to females in response to a similar exercise programme (McLaughlin et al. 2006).

Furthermore, compensatory decreases in energy expended in spontaneous activity have been shown to lead to smaller than expected increases in total energy expenditure in response to exercise interventions (Goran & Poehlman 1992;Kempen, Saris, & Westerterp 1995) and a wide variation in the extent of change in non-intervention energy expenditure when individuals undergo an exercise-training intervention has been reported (Goran & Poehlman 1992;King et al. 2007). The extent to which this contributes to individual differences in weight or fat loss is unclear, but it has been shown that individuals with the greatest increases in non-exercise activity thermogenesis (NEAT) in response to overfeeding are protected from weight gain (Levine, Eberhardt, & Jensen 1999), which suggests that individual differences in activity compensation could conceivably play a role in determining responsiveness to exercise-induced weight loss.

However, differences in behavioural compensation probably cannot explain all of the variability in responsiveness to exercise-induced weight and fat loss. For example, Bouchard and colleagues reported that in men residing at an isolated experimental station in a highly controlled environment, imposition of an exercise-induced energy deficit of 4.2 MJ per day for 84 days, with constant energy intake, led to reductions in body weight ranging from 3 to 12 kg (Bouchard et al. 1990), a range which is unlikely to be fully explained by differences in compensatory activity between subjects. Relatively homogenous metabolic responses to an exercise intervention were observed in identical twins, compared to other matched pairs, suggesting a constitutive influence on the metabolic response to exercise (Bouchard et al. 1994). Thus non-behavioural metabolic factors are also likely to contribute to individual responsiveness to exercise-induced weight loss. One metabolic factor which could contribute is resting metabolic rate, however, a recent report found no significant difference in the change in resting metabolic rate between the start and end of an exercise intervention between subjects who lost more weight than expected and who lost less weight than expected, suggesting that this does not play a major role (King et al. 2008). A further metabolic factor which might play a role is the magnitude of the exercise-induced change in

resting fat oxidation. Both fasting and postprandial fat oxidation have been shown to increase for at least 24 hours following an exercise session (Burton et al. 2008; Hansen, Shriver, & Schoeller 2005; Votruba et al. 2002), even in the absence of an exercise-induced energy deficit (Burton et al. 2008) and studies have reported greater fat oxidation at rest in endurance-trained versus untrained individuals (Romijn et al. 1993). The exercise-induced increase in resting fat oxidation varies between individuals (Blaak & Saris 2002;Burton et al. 2008;Goodpaster et al. 2003), and indeed the extent of this increase has been demonstrated to be a strong predictor of other exercise-induced metabolic changes, such as the magnitude of change to postprandial lipid metabolism (Burton et al. 2008) and insulin sensitivity (Goodpaster et al. 2003). Furthermore, a number of studies have shown that a high respiratory quotient (RER) (indicating high carbohydrate and low fat oxidation) measured fasting (Marra et al. 1998; Seidell et al. 1992) or over 24 hours (Zurlo et al. 1990) is a significant predictor of long-term weight gain, independent of metabolic rate (Marra et al. 1998;Seidell et al. 1992;Zurlo et al. 1990). Thus effects of exercise on resting fat oxidation may contribute to fat loss in response to an exercise intervention. It is therefore hypothesised that variability in the upregulation of resting fat oxidation following an exercise training intervention would contribute to the inter-individual variability in exercise-induced fat loss and that this effect would be independent of energy expended during exercise training sessions and the effects of behavioural compensatory responses. In addition, this study aimed to determine whether any baseline physiological, metabolic, anthropometric or behavioural characteristics could predict responsiveness to exercise-induced fat loss.

## 6.2 Methods

## 6.2.1 Volunteers

Volunteers for this study participated in an intervention trial to determine the effects of exercise training on insulin sensitivity in women with and without a family history of diabetes as described in Chapter 4 (Barwell et al. 2008). They were recruited and screened as described in Section 2.1. Sixty-two women completed the intervention and complete data sets for the variables of interest were available in 55 volunteers. These 55 women were included in the data analysis and their characteristics are presented in Table 6.1. Twenty seven of the women were offspring of patients with type 2 diabetes and 28 had no family history of the disease.

# 6.2.2 Study design

The overall study design has been described previously in section 2.1.3 but, in brief, subjects all underwent a metabolic assessment, body composition measurements (by Dual X-ray Absorptiometry, DEXA) and a cardio-respiratory fitness test, including individualised calibration of the heart rate versus oxygen uptake relationship, at baseline and after a controlled 7-week endurance-type exercise training programme. Dietary intake and physical activity were monitored during the week preceding baseline metabolic assessment and during the final week of the exercise training programme.

## 6.2.3 Metabolic assessment

Volunteers reported to the metabolic suite after 12-hour overnight fast and a 20-minute expired air sample was collected using a ventilated hood system (Deltatrac Metabolic Monitor, Datex Engstrom, Kent, UK) to determine resting oxygen uptake ( $\dot{V}O_2$ ), carbon dioxide production ( $\dot{V}CO_2$ ), RER (i.e.  $\dot{V}CO_2/\dot{V}O_2$ ) and metabolic rate (Sections 2.3.2-2.3.3). Resting heart rate was recorded immediately following the ventilated hood measurement using an automated device (Complior, Artech Medical, Pantin, France) as described in Section 2.3.5. Volunteers completed their final exercise session 15-24 hours prior to the post-intervention measurements.

# 6.2.4 Body composition assessment

Dual X-Ray Absorptiometry (DEXA) scans (LUNAR Prodigy DEXA scanner, GE Healthcare Diagnostic Imaging, Slough, Berkshire, UK) were used to determine body composition and fat distribution (Section 2.2.5). Height, body mass and waist circumferences were also determined using standard protocols (Sections 2.2.1 – 2.2.4) (Marfell-Jones et al. 2006).

# 6.2.5 Fitness test and calibration of the heart rate versus VO<sub>2</sub> relationship

Volunteers attended for this test at least two hours after eating. After a 10-minute rest, 5minute expired air samples were collected via a mouthpiece into Douglas bags for the determination of  $\dot{V}O_2$  and  $\dot{V}CO_2$  during three sedentary activities – i.e. sitting, standing, and standing with arms swaying. The mean  $\dot{V}O_2$  of these three activities was taken as representative of  $\dot{V}O_2$  during sedentary activities (Section 2.5.1). Following this, subjects performed an incremental, submaximal, treadmill walking test to determine their heart rate versus  $\dot{V}O_2$  relationship and estimate maximal oxygen uptake ( $\dot{V}O_{2max}$ ) (Section 2.6).

## 6.2.6 Physical activity and dietary assessment

For seven days preceding each metabolic assessment day, subjects wore a heart rate monitor for all waking hours (Polar 610i; Polar Electro, Kempele, Finland) as described in section 2.9. Average daily heart rate minus resting heart rate was used as a surrogate measure of total activity and time spent above 1.5 times and 2 times resting heart rate were used as surrogate indices of time spent engaging in activities of at least light and at least moderate intensity, respectively (Section 2.10). During this time subjects also completed 7-day weighed food diaries which were analysed using a computerised version of food composition tables (CompEat Pro; Nutrition Systems, Banbury, UK) (Section 2.8).

# 6.2.7 Exercise intervention

Volunteers participated in a 7-week endurance type exercise programme, as described in section 2.7. A total of 32 exercise sessions were prescribed over the course of the intervention. Other than participating in the exercise intervention, subjects were requested to make no changes to their lifestyle for the duration of the study.

The energy expended during each exercise session was calculated from the mean exercise heart rate for the session and the individualised heart rate versus  $\dot{V}O_2$  relationship as described in section 2.9.2. The net energy expenditure of all exercise sessions was summed to determine the total net energy expenditure of the exercise training programme.

# 6.2.8 Reproducibility and statistical power

To determine the reproducibility of the resting RER measurements in a free-living situation, 28 of the women in this study underwent two ventilated hood measurements following a 12hour fast, at an interval of eight weeks, with no specific guidelines other than to maintain their usual diet and physical activity habits throughout the interval between measurements. Fasting RER was  $0.84 \pm 0.04$  on the first visit and  $0.85 \pm 0.05$  on the second visit (NS) and the SD for the difference in RER between visits was 0.05. Based on these data, the present study, with 55 participants, had sufficient statistical power to detect a difference in RER with exercise training of 0.02 with 90% power. This number of volunteers also enables detection of a correlation between variables of 0.26 at the  $p \le 0.05$  level.

#### **6.2.9 Statistical analysis**

Data were analysed using Statistica (version 6.0; StatSoft, Tulsa, OK, USA) and Minitab (version 13.1, Minitab Inc., State College, Pennsylvania). Univariate linear regression analyses were performed to determine the relationship between change in total and regional (trunk, upper body (i.e. trunk plus arm), and leg) fat mass over the course of the intervention and: total net energy expenditure of the exercise intervention; behavioural compensation variables (i.e. change in energy, fat, carbohydrate and protein intake, change in average daily heart rate minus resting heart rate, and change in time spent above 1.5 and 2 times resting heart rate excluding exercise training); changes in physiological variables (i.e. change in V O<sub>2max</sub>, change in resting metabolic rate, change in fasting RER); and baseline physiological and behavioural characteristics (age, BMI, total and regional fat mass, VO<sub>2max</sub>, resting metabolic rate, fasting RER, energy, fat, carbohydrate and protein intake, average heart rate minus resting heart rate, time spent above 1.5 and 2 times resting heart rate). Using the same method as described in Section 3.2.9.2, the extent to which variables were related to change in fat mass independently of exercise energy expenditure was determined. Univariate linear regressions were performed between the residuals for change in fat mass of the regression between change in fat mass and net total energy expenditure of exercise and the other variables. To further adjust for the effect of change in energy intake on change in fat mass, the residual for change in fat mass after regression with net total energy expenditure of exercise and change in energy intake was used in univariate linear regressions with the other variables. Finally, to determine which variables were independent predictors of the change in fat mass, multiple-regression analysis was performed. Statistical significance was accepted at the p  $\leq$  0.05 level.

# **6.3 Results**

Baseline characteristics and group changes in response to the exercise intervention are shown in Table 6.1. In response to the exercise training intervention, the group as a whole significantly reduced BMI (by 0.9%), total fat mass (by 3.3%), trunk fat mass (by 3.0%), upper body fat mass (by 3.8%), leg fat mass (by 2.5%), percentage body fat (by 2.6%) and waist circumference (by 1.4%), but lean body mass was not significantly changed.  $\dot{VO}_{2max}$ (by 13.6%) was significantly increased. Resting metabolic rate was not changed but fasting RER was significantly reduced, indicating a shift in substrate utilisation towards fat oxidation. Although volunteers were asked not to alter their diets over the course of the intervention, energy (by 6.3%), carbohydrate (by 7.6%) and protein (by 8.8%) intakes were significantly lower at the end of the intervention than at baseline, but there were no significant differences in fat intake between the start and end of the intervention. Average daily heart rate minus resting heart rate and total time spent above 1.5 times and 2 times resting heart rate (including time spent in the exercise sessions) were all significantly higher at the end of the intervention than at baseline (p < 0.01), but the latter two factors did not differ between baseline and the end of the intervention when the time spent during the intervention exercise sessions was excluded.

	Baseline	Change with
		exercise training
Age (years)	$34.7 \pm 6.4$	-
BMI (kg.m <sup>-2</sup> )	$27.5 \pm 4.7$	$-0.24 \pm 0.65^{**}$
Lean mass (kg)	$41.8 \pm 5.8$	$0.34 \pm 1.37$
Fat mass (kg)	$29.9\pm9.6$	$-0.97 \pm 1.48^{**}$
% body fat	$39.3\pm6.0$	$-1.02 \pm 1.47^{**}$
Waist circumference (cm)	$85.9 \pm 11.9$	$-1.20 \pm 2.26^{**}$
$VO_2max (ml.kg^{-1}.min^{-1})$	$31.3 \pm 5.1$	$4.2 \pm 4.0^{**}$
Insulin sensitivity index	$6.79 \pm 3.82$	0.75 ± 2.33*
Resting metabolic rate (kJ.d <sup>-1</sup> )	$6250\pm918$	$-12 \pm 410$
Fasting RER	$0.853 \pm 0.051$	$-0.030 \pm 0.055^{**}$
Post-glucose RER	$0.948\pm0.042$	$-0.035 \pm 0.052^{**}$
Energy intake (kJ.d <sup>-1</sup> )	$7904 \pm 1787$	$-500 \pm 1530^{*}$
Fat intake (g.d <sup>-1</sup> )	$66.9\pm20.4$	$-1.8 \pm 19$
Carbohydrate intake (g.d <sup>-1</sup> )	$247.3\pm72.0$	$-18.9 \pm 53.4^{**}$
Protein intake (g.d <sup>-1</sup> )	$71.6 \pm 17.5$	-6.3 ± 14.1**
Resting heart rate (beat.min <sup>-1</sup> )	$68 \pm 8$	$-3.4 \pm 5.7$
Time above 1.5 times resting heart rate inc.exercise training (min.d <sup>-1</sup> )	-	$40 \pm 11$
Time above 2 times resting heart rate inc.exercise training (min.d <sup>-1</sup> ) -		$23 \pm 3$
Time above 1.5 times resting heart rate exc.exercise training (min.d <sup>-1</sup> )	$106 \pm 91$	$9 \pm 11$
Time above 2 times resting heart rate exc.exercise training (min.d <sup>-1</sup> )	$11 \pm 15$	$4\pm 2$
Number of exercise training sessions completed	-	$28.3 \pm 6.3$
Total duration of exercise training completed (min)	-	$1402\pm469$
Mean heart rate during exercise (beat.min <sup>-1</sup> )	-	$143 \pm 9$
Total net exercise energy expenditure (MJ)	-	$36.9 \pm 17.0$

**Table 6.1.** Volunteer characteristics at baseline and in response to exercise training.

N = 55, values are mean  $\pm$  SD. \*p < 0.05 for change with exercise training (\*\*p < 0.01)

Across the group, there was a wide individual variation in change in total fat mass over the intervention, with individual changes in total fat mass ranging from a 2.1 kg gain to a 5.3 kg loss (Figure 6.1).



**Figure 6.1** Individual changes in fat mass in whole group (n = 55) after exercise intervention. Volunteers ranked in order of change in unadjusted fat mass.

Correlates of the change in fat mass over the intervention are shown in Table 6.2. The strongest correlate of change in total fat mass during simple univariate regression analysis was total net energy expenditure of exercise (r = 0.60, p < 0.0005), explaining 36% of the variance in this factor. Change in average daily heart rate minus resting heart rate (which incorporates heart rate elevations from the exercise intervention) also significantly correlated with change in total fat mass (r = -0.31, p < 0.05) but change in time spent over 1.5 and 2 times resting heart rate (excluding exercise training) did not. Change in fat intake (r = -0.25, p

= 0.06) and change in fasting RER (r = -0.25, p = 0.05), had borderline significant correlations with change in total fat mass in simple univariate regression analysis.

		Correlation with change in fat mass		
		No adjustment	After adjustment for total net exercise energy expenditure	After adjustment for total net exercise energy expenditure and change in energy intake
Intervention variables	Total net energy expenditure of exercise	-0.60	-	-
		(<0.0005)		
Behavioural compensation variables	Change in energy intake	0.20	0.20	-
		(0.14)	(0.14)	
	Change in fat intake	0.25	0.24	0.10
		(0.06)	(0.07)	(0.48)
	Change in carbohydrate intake	0.11	0.17	0.00
		(0.42)	(0.22)	(0.99)
	Change in protein intake	0.05	0.05	-0.06
		(0.73)	(0.71)	(0.65)
	Change in time spent above 1.5 times resting heart rate Change in time spent above 2 times resting heart rate	0.06	-0.05	-0.03
		(0.64)	(0.73)	(0.85)
		-0.04	-0.10	-0.07
		(0.75)	(0.48)	(0.60)
Physiological variables	Change in VO <sub>2</sub> max	0.03	0.20	0.15
		(0.82)	(0.14)	(0.26)
	Change in resting metabolic rate	0.13	0.15	0.15
		(0.33)	(0.28)	(0.28)
	Change in Fasting RER	0.25	0.31	0.26
		(0.06)	(< 0.05)	(0.05)

**Table 6.2.** Correlations between interventions, behavioural compensation and physiological

 variables and change in fat mass in response to exercise training.

N = 55, values are correlation coefficients with p-values in brackets.

Figure 6.2 shows the individual residual changes in fat mass after adjustment for the net total energy expenditure of exercise, and Figure 6.3 shows the individual residual changes in total fat mass after adjustment for the net total energy expenditure of exercise and changes in energy intake. In other words, figures 6.2-6.3 show the variation in change in total fat mass which cannot be explained by changes in exercise energy expenditure and energy intake over the intervention.



**Figure 6.2** Individual residual changes in fat mass for whole group (n = 55) after adjustment for net total energy expenditure of exercise intervention. Volunteers ranked in order of change in unadjusted fat mass.



**Figure 6.3** Individual residual changes in fat mass for whole group (n = 55) after adjustment for net total energy expenditure of exercise intervention and change in energy intake. Volunteers ranked in order of change in unadjusted fat mass.

The relationship between change in total fat mass and change in fasting RER was strengthened after adjustment for the total net energy expenditure of the exercise (r = -0.31, p < 0.05) and remained after further adjustment for change in energy intake (r = -0.26, p = 0.05) (Figure 6.4). Thus, change in fasting RER explained ~7% of the variance in the change in total fat mass over the course of the intervention after adjustment for effects of the total net energy expenditure of exercise and change in energy intake. Adjusting for exercise energy expenditure and change in energy intake both weakened the relationship between change in total fat mass and change in fat intake. None of the included baseline factors correlated significantly with change in total fat mass in any of these analyses (data not shown). In multiple-regression analysis including all of the variables described in the statistical analysis section in the model, net total energy expenditure of exercise (p < 0.0005) and change in fasting RER (p = 0.02) were the only statistically significant independent predictors of

change in fat mass. Together, these two variables explained 40.2% of the variance in change in fat mass (adjusted  $R^2$ ).



**Figure 6.4** Scattergram showing the relationship between change in fasting RER and residual change in fat mass, adjusted for the effects of net energy expenditure of exercise and change in energy intake (N = 55, r and p values for univariate linear correlations).

In univariate regression, changes in trunk (r = 0.87, p < 0.001), upper body (r = 0.92, p < 0.001) and leg (r = 0.52, p < 0.001) fat were strongly associated with changes in total fat mass. In addition, changes in trunk (r = -0.43, p < 0.01), upper body (r = -0.49, p < 0.001) and leg (r = -0.45, p < 0.01) fat all significantly correlated with total net energy expenditure of exercise. Change in trunk fat also significantly correlated with change in average daily heart rate minus resting heart rate (r = -0.29, p < 0.05) and change in upper body fat correlated significantly with fat intake at baseline (r = -0.28, p < 0.05) (i.e. those with highest baseline fat intake lost the greatest amount of upper body fat). Changes in trunk, upper body or leg fat were not significantly associated with change in fasting RER.

Because the exercise training-induced change in fasting RER was a significant predictor of change in total fat mass, univariate and multivariate regression analyses were performed to determine whether any of the other measured variables could predict the exercise-induced change in RER. Change in fasting RER was significantly associated with diabetes family history (with 0 and 1 included as dummy variables for negative and positive diabetes family history, respectively) r = -0.35, p < 0.01) (i.e. those with a positive diabetes family history had a bigger exercise-induced reduction in fasting RER); baseline BMI (r = 0.29, p < 0.05) and total (r = 0.29, p < 0.05) and leg (r = 0.36, p < 0.01) fat mass (i.e. those with lower baseline BMI and fat mass had bigger reductions in RER); and changes in energy (r = 0.29, p < 0.05), carbohydrate (r = 0.28, NS) and fat (r = 0.27, p < 0.05) intake (i.e. those with bigger decreases in energy, carbohydrate and fat intake had bigger reductions in RER). In multiple regression analysis, baseline fasting RER (p < 0.001), diabetes family history (p < 0.001) and change in energy intake (p < 0.05) were independent significant predictors of the change in fasting RER, together explaining 55.9% of the variance. Baseline fasting RER alone explained 44.2% of the variance in the change in fasting RER.

#### 6.4 Discussion

In this study, 55 women underwent a 7-week exercise training programme, which induced a mean total fat loss for the group of 0.97 kg. The mean net total energy expenditure of the exercise programme was 36.9 MJ, so, assuming that fat loss requires a negative energy balance of 39.4 MJ/kg (Elia, Stratton, & Stubbs 2003), fat loss in the group as a whole was broadly at the expected level. However, considering the change in fat mass at group level obscures the wide inter-individual variability in change in fat mass with the intervention (see Figure 6.1). The most important correlate of change in fat mass was net total energy expenditure of exercise, which explained 36% of the variance, but even after adjusting for this and for changes in energy intake over the intervention, a wide variation for the residual change in fat mass was evident, ranging from +2.5 kg to -2.9 kg. The main novel finding of the study was that the residual change in total fat mass in response to an exercise intervention, after adjustment for the energy expended during the exercise intervention and for changes in energy intake was related to the change in fasting RER between the start and end of the intervention. Change in fasting RER explained  $\sim 7\%$  of the variance of the residual change in fat mass. Indeed, the relationship between change in fasting RER and change in total fat mass was independent of all other physiological and behavioural variables included in a multivariate analysis model and in this model change in RER and net total energy expenditure of exercise were the only significant independent predictors of change in total fat mass.

It is not possible from the design of this study to determine the direction of causality between change in fasting RER and change in fat mass conclusively. However, the case that exercise-induced reduction in RER was a significant predictor of exercise-induced fat loss fits well with the body of data indicating that a high RER predicts long-term weight gain (Marra et al. 1998;Seidell et al. 1992;Zurlo et al. 1990) – in other words, good fat oxidisers were protected from future weight gain. This data builds on this work, indicating that individuals who had

the largest shifts in resting substrate utilisation towards fat oxidation in response to exercise training experienced the greatest losses in fat mass, independent of exercise energy expenditure and change in energy intake. The studies of RER and long-term weight gain found this effect to be independent of metabolic rate (Marra et al. 1998;Seidell et al. 1992;Zurlo et al. 1990) and the results of this study are consistent with this. Indeed, in agreement with the study of King and co-workers (King et al. 2008), it was found that changes in resting metabolic rate between the start and end of the intervention were not associated with change in fat mass.

On the other hand, it is well established that negative energy balance leads to increased fat oxidation (Pagliassotti et al. 1997) and the women with the greatest fat losses in this study would, by definition, have incurred the greatest negative energy balances over the course of the intervention. However, this shift towards fat oxidation appears to occur largely in response to acute negative energy balance over the relatively short-term, and over the longer term, a number of studies have reported that weight loss (i.e. incurring a large cumulative negative energy balance over a number of weeks or months, with consequent changes in body composition) is often associated with no change (Valtuena et al. 1997) or even an increase (Luscombe et al. 2003; Pasman, Westerterp, & Saris 1999; Weyer et al. 2000) in fasting and 24-hour RER. Indeed, an increase in RER in response to weight loss is thought be one of the factors predisposing to weight regain following weight loss: a number of reports have indicated that the individuals with the highest RERs following weight loss are those who are most susceptible to weight regain (Froidevaux et al. 1993; Valtuena et al. 1997). Furthermore, the change in fat oxidation in response to the same degree of short-term negative energy balance (Burton et al. 2008), and long-term change in body weight (Weyer et al. 2000), is highly variable between individuals. For example, previously reported individual changes in whole-body postprandial fat oxidation ranged from a decrease of 4 grams to an increase of 16 grams over an 8.5-hour observation on the day following an exercise session inducing an identical energy deficit of 27 kJ.kg<sup>-1</sup> in all subjects (Burton et al. 2008). It has been

demonstrated that this variability has metabolic consequences beyond the regulation of body weight: individuals to upregulate post-exercise fat oxidation to the greatest extent in response to a given energy deficit also experience the largest changes to postprandial lipid metabolism (Burton et al. 2008) and insulin sensitivity (Goodpaster et al. 2003). In addition, it has been shown that exercise increases subsequent fat oxidation for at least 24 hours even in the absence of an associated energy deficit and that the extent of this increase also varies markedly between individuals (Burton et al. 2008). As the post-training metabolic assessment was undertaken 15-24 hours after an exercise session, it is likely that the variation in change in RER in the present study at least partly reflects inter-individual differences in the acute post-exercise increase in fat oxidation. Thus, the evidence from the literature indicates that exercise-induced changes in RER exhibit a large degree of inter-individual variability and are evident in the absence of energy deficit, and that RER is not consistently reduced following weight loss, so does not support the rationale that larger reductions in RER are simply the consequence of larger incurred negative energy balances and greater weight losses.

As the change in fasting RER was significantly, independently, associated with change in total fat mass, this study aimed to determine what factors influenced change in RER. By far, the most important predictor of the change in fasting RER was its baseline value – those with the highest baseline values for RER experienced the greatest reductions. Thus, it appears that individuals who are 'carbohydrate oxidisers' when sedentary experience the greatest shift towards fat oxidation in response to exercise training. This is an exciting possibility which suggests that those who are most susceptible to weight gain due to their high RER (Marra et al. 1998;Seidell et al. 1992; Zurlo et al. 1990), experience the greatest benefits in terms of increase in resting fat oxidation from exercise. However, baseline fasting RER was not significantly associated with change in total fat mass, a degree of caution is advised when interpreting this observation. Change in fasting RER was also significantly associated with other 'innate' baseline factors, namely, BMI, total and leg fat mass and diabetes family

history; however, change in components of dietary intake were also implicated, albeit to a much lesser degree than baseline RER, suggesting that 'behavioural' and 'metabolic' factors influencing change in fat mass are not completely independent of each other. Further investigation in this area could confirm that inter-individual ability to lose weight in response to exercise is predicated on ability to upregulate fasting fat oxidation. Alteration in fasting RER in response to a single exercise session would allow stratification of subjects into potential 'responders' and 'non-responders' prior to an exercise intervention designed to induce weight loss. Such a study would require blinding of the volunteers to their potential exercise response and close control of diet and levels of spontaneous physical activity over the period of the exercise intervention. Actual reduction in fat mass after the exercise intervention would provide evidence that ability to reduce weight in response to exercise is predictable and determined, in part, by innate metabolic response to exercise.

The present study has a number of strengths. With 55 participants, it is the largest study to date attempting to address the issue of factors influencing individual responsiveness to exercise-induced fat loss, which provides statistical high power to detect associations between variables, making the statistical findings of this study robust. In addition, both metabolic and behavioural variables were included in the analyses enabling the relative importance of each to be determined, and energy expenditure of the exercise intervention was objectively quantified on an individual-by-individual basis. Furthermore, baseline and post-intervention testing was performed at an interval of eight weeks to ensure that, as far as possible, women were in the same phase of menstrual cycle for baseline and post-intervention testing and thus limiting the confounding effects of cyclical hormonal changes on the results.

The main limitations to this study, which are common to the majority of reports in this field, relate to the measurement of behavioural compensation variables. Firstly, it is well established that underreporting is a common problem incurred in measurement of dietary

intake, and that this problem is greater in obese than lean individuals (Black & Cole 2001;Westerterp & Goris 2002). However, it appears that the extent of underreporting is relatively consistent within an individual (Black & Cole 2001), implying that differences in dietary intake between two observation points (e.g. baseline and post-intervention) are likely to be determined with greater accuracy than absolute dietary intakes at a single time-point. Thus, the repeated-measures design in the present study may attenuate this potential error. Furthermore, the effect of inaccuracy in measurement of dietary intake would act to diminish any association with exercise-induced change in fat mass (a regression dilution bias effect (Frost C 2000)): thus the finding that there was a borderline significant relationship between change in fat intake and exercise-induced change in fat mass in the present study, despite potential errors in the assessment on dietary intake, suggests that this association was likely to be real. This would be in agreement with the recent work of King and colleagues who reported that dietary compensation influenced the extent of weight loss in response to exercise training (King et al. 2008). In contrast, this study found no evidence to suggest that individual differences in compensation by reducing non-exercise activities influenced the extent of fat loss with the heart rate monitoring methodology employed in the present study. However, because factors other than activity (e.g. excitement and stress) can influence heart rate, particularly when heart rates are relatively low (as they are over most of the day), this approach may have missed subtle changes in activity which could have contributed to responsiveness to exercise-induced fat loss. Thus, it is conceivable that individual differences in compensatory changes in spontaneous physical activity in response to an exercise intervention could have an effect on the extent of exercise-induced changes in fat mass which it was not possible to detect in the present study.

In conclusion, this study found that exercise-induced change in fat loss is associated not only with exercise energy expenditure but also with changes in RER at rest. Thus, development of strategies to maximise the shift in resting substrate utilisation towards fat oxidation in

response to an exercise training programme may help individuals to maximise exerciseinduced fat loss.

#### CHAPTER 7

## **GENERAL DISCUSSION**

## 7.1 Experimental chapter summaries

The aim of Chapter 3 was to describe the differences between sedentary, premenopausal women with and without a parental history of type 2 diabetes. The findings confirm that familial history of T2D is associated with greater insulin resistance and that this may be mediated by a greater sensitivity to the deleterious metabolic effects of adiposity. Further support for this suggestion may be derived from the finding that although circulating NEFA concentrations were not different between the groups, after adjustment for fat mass Offspring displayed a negative correlation between NEFA and insulin sensitivity. The correlation between adiponectin and insulin sensitivity in Offspring alone may represent an adaptive response towards the effects of NEFA on insulin sensitivity, and the relationship between CRP and insulin sensitivity in Offspring suggests a role for a low-grade inflammation in the development of insulin resistance. Although Offspring subjects were less physically active than Controls, habitual physical activity levels influenced baseline insulin sensitivity in Controls alone. However, it should be appreciated that both the methods used to assess habitual activity and the range of activity observed in the Offspring group may prevent detection of a relationship between physical activity and insulin sensitivity in this group.

Chapter 4 examined the effect of a 7-week aerobic exercise programme on physical and metabolic characteristics of women with, and without a parental history of diabetes. Despite similar compliance to the exercise intervention and similar improvements in cardiorespiratory fitness, women with a parental history of T2D exhibited an augmented metabolic response to exercise, showing increased insulin sensitivity, increased fat oxidation and reduced circulating leptin. Three-day detraining results in Offspring showed increased circulating

leptin and reduced fat oxidation without a change in insulin sensitivity. The persistence of improved insulin sensitivity is consistent with the literature comparing exercise responses in younger and older women (Goulet et al. 2005). Taken together, these results suggest that women with a parental history of diabetes are display a more favourable metabolic response to exercise, with relatively improved insulin sensitivity which may be mediated by a metabolic pathway which involves reduction in circulating plasma leptin and increased fat oxidation.

Chapter 5 explored the contribution of changes in body composition, adipose tissue hormones and whole body fat oxidation to improved insulin sensitivity after an exercise intervention. When considered as a single combined group the greatest improvement in insulin sensitivity was seen in those who were most insulin resistant prior to the intervention, those who reduced circulating leptin concentrations, who expended the most energy during the intervention and those who increased habitual physical activity. When considered as a separate group, improved insulin sensitivity in Controls was associated with an increase in hepatic fat oxidation. However although reductions in circulating leptin,  $TNF\alpha$  and energy intake were also associated with improved insulin sensitivity, these associations were not independent of reduced fat mass. This may suggest that in Controls, exercise improves insulin sensitivity primarily through reduction in fat mass. Exercise appears to influence insulin sensitivity by different mechanisms in Offspring. These subjects appeared to be more sensitive to the net energy cost of the total intervention which might suggest a more direct dose-response relationship between exercise and insulin sensitivity in this group. Alternatively, it may suggest that Offspring have a lower threshold for exercise-mediated insulin sensitisation. An interesting association between reduced lower body fat and insulin sensitivity was observed in this group, this association was independent of reduced total fat mass. This could suggest that lower body fat is a marker of intramuscular fat, but this would be contrary to the observation that exercise does not necessarily reduce IMCL but does alter partitioning (Dube et al. 2008). Measured by DEXA scan, thigh fat may be a better marker of

deep subcutaneous fat than in other areas of the body (Ryan et al. 2002), particularly the abdomen where it is not possible for the scanner to delineate subcutaneous from visceral fat. If Offspring, like South Asians (Sniderman et al. 2007), are more sensitive to the metabolic influence of deep subcutaneous fat, exercise may preferentially reduce the more metabolically deleterious adipose pools leading to increased insulin sensitivity. Both CRP and resting heart rate were negatively correlated with the post-intervention increase in insulin sensitivity. These factors were independent of reduced fat mass and energy expended during exercise, but not baseline insulin resistance. A link between these two seemingly disparate observations could be made by considering the effects of autonomic hyperactivity. Autonomic hyperactivity is associated with both hyperinsulinaemia and low grade inflammation (Egan 2003) and exercise has been shown to improve insulin sensitivity in concert with improved autonomic responsiveness (Giallauria et al. 2008). However, a similar pattern was also observed with the index of habitual activity (average heart rate minus resting heart rate) and it is possible that a reduction in resting heart rate is also a marker of increased habitual physical activity.

Increased insulin sensitivity in Offspring was associated with reduced circulating leptin and this relationship was independent of alterations in fat mass but not independent of the energy deficit of the intervention. Other authors have suggested that the reduction in circulating leptin mediates improved insulin sensitivity by increased skeletal muscle fat oxidation (Goodpaster et al. 2003;Solomon et al. 2008). Surprisingly, fat oxidation was not associated with the exercise-mediated increase in insulin sensitivity in any of the groups therefore the results of Chapter 5 cannot support this hypothesis. However, if reduced leptin influences insulin sensitivity, but is dependent upon energy deficit then leptin could be acting as a 'system-wide' signal for energy deficit and may interact with AMPK to stimulate peripheral insulin sensitivity by mechanisms other than increased skeletal muscle fat oxidation (Kahn et al. 2005;Hutchinson et al. 2008).

The exercise-mediated change in circulating leptin and fat oxidation, combined with the rebound leptinaemia and reduction in fat oxidation in response to detraining, suggests that exercise increases skeletal muscle insulin sensitivity by a mechanism of leptin-sensitive fat oxidation. Increased fat oxidation in skeletal muscle would be expected to reduce lipid intermediates (Dube et al. 2008) and result in improved insulin signalling (Bonen, Dohm, & van Loon 2006). Leptin is recognised to promote fat oxidation in healthy individuals (Minokoshi et al. 2002; Steinberg et al. 2002). If this was the causative mechanism, increased fat oxidation would have been expected to have been significantly associated with the exercise-mediated increase in insulin sensitivity. In Chapter 5 this was not found to be the case. Therefore, increased fat oxidation may be a marker of both improved skeletal muscle insulin and leptin sensitivity, rather than a direct causative factor in the observed increase in insulin sensitivity. Additionally, although exercise may directly influence fat oxidation, the effects on insulin sensitivity may be indirectly mediated through the action of intermediates on insulin signalling (Schenk & Horowitz. 2006; Schenk et al. 2009) which is consistent with the observation that intramyocellular DAG and ceramides are more closely associated with insulin sensitivity than IMCL (Delarue & Magnan. 2007). Leptin-stimulated fat oxidation may reduce intramyocellular intermediates which directly influence peripheral insulin sensitivity. Alternatively, leptin may mediate improved skeletal muscle insulin sensitivity by other mechanisms (Stefanyk & Dyck 2010). Rodent studies have shown that reduced skeletal muscle leptin receptor expression in high-fat fed animals and leptin 'resistance' in human myotubes can be overcome by direct AMPK stimulation using AICAR (Dyck 2009). Leptin has also been shown to activate AMPK in hepatic cells (Uotani, Abe, & Yamaguchi 2006). Leptin has been considered to have direct effects on GLUT4 translocation in myotubes in vitro (Berti & Gammeltoft 1999) and in palmitate-induced insulin resistance, leptin has recently been shown to stimulate palmitate oxidation and separately promote GLUT4 expression via increased insulin-stimulated Akt activity (Alkhateeb et al. 2009). Therefore if, in addition to the effects of leptin on fat oxidation, leptin activation of AMPK and Akt can

also promote activation of downstream insulin signalling, then the effects of increased leptin sensitivity on insulin sensitivity may be at least two-fold. Adipose tissue expression of the long isoform of the leptin receptor is increased in response to exercise in Offspring compared to Controls and in combination with improved insulin sensitivity and decreased circulating leptin suggests improved leptin sensitivity (Moran et al. 2010). Leptin receptor expression is increased in trained muscle in humans (Olmedillas et al. 2010). If exercise also increases leptin receptor expression in skeletal muscle in Offspring then it could be considered that improved leptin sensitivity results in increased fat oxidation, reduced lipid intermediates and enhanced leptin-mediated stimulation of insulin signalling.

In adipose tissue, leptin is typically considered to be a pro-inflammatory hormone which potentiates release of TNF $\alpha$  and other cytokines which mediate endothelial dysfunction and skeletal muscle insulin resistance (Gil et al. 2007;Lago et al. 2007). In addition, leptin may directly stimulate adipose tissue macrophage infiltration (Gruen et al. 2007;Gutierrez, Puglisi, & Hasty 2009). Therefore if leptin's role in adipose tissue inflammation is solely stimulatory, improved adipose tissue leptin sensitivity might be expected to promote both inflammation and insulin resistance. However, murine studies of adipose tissue inflammation show that exercise is effective in reducing pro-inflammatory cytokine expression in adipose tissue and that this is paralleled by reduced plasma leptin concentrations (Bradley et al. 2008). In addition, Moran and co-workers describe a state of increased 'leptin sensitivity' which is associated with increased insulin sensitivity (Moran et al. 2010). It would seem likely that an improved state of adipose tissue leptin sensitivity is anti-, rather than pro-inflammatory, however the mechanism(s) which underpin this in adipose tissue are not currently known.

Leptin resistance has been described as a failure to increase metabolic rate and energy expenditure and to reduce fat mass despite rising leptin concentrations (Galic, Oakhill, & Steinberg 2010). Many of the models and mechanisms of increased inflammation which are

attributed to hyperleptinaemia are described in obesity and relatively little work has been done in examining the relationship between leptin and adipose tissue inflammation in lean, insulin sensitive subjects. Therefore, similar to initial assumptions about leptin inducing insulin resistance (due to the observed associations between obesity, high leptin and insulin resistance) the suggestion that in all states, leptin promotes inflammation may be incorrect (Stofkova 2009). Since both free and bound leptin appear to be associated with differing metabolic effects (Magni et al. 2005) and leptin is recognised to inhibit AMPK activity centrally, but stimulate it in peripheral tissues (Galic, Oakhill, & Steinberg 2010) it is perhaps not surprising that leptin action within adipose tissue may alter depending on the metabolic state of the tissue. The observation that Offspring have improved exercise-mediated adipose tissue 'leptin sensitivity' would suggest that at least one leptin-related metabolic process is differently exercise-responsive in Offspring compared to Controls. Potential mechanisms include; improved adipose tissue insulin sensitivity leading to lower rates of lipolysis, restoration of the synergistic relationship between leptin and adiponectin on insulin signalling (Stefanyk & Dyck 2010) and a currently unexplained mechanism by which leptin sensitive adipose tissue reduces inflammatory protein production. Further examination of the adipose tissue samples obtained (Appendix B1) would provide further information about adipose tissue inflammation with respect to alterations in adipocyte size and macrophage infiltration in response to exercise. Objective assessment of alteration in adipose tissue inflammatory cell content would contribute to the understanding of a potential differential anti-inflammatory effect of exercise in Offspring compared to Controls. Analysis of components of the insulin signalling pathway, in particular the activity of AMPK and Akt, would also provide further evidence linking exercise, leptin activity and adipose tissue insulin sensitivity.

Chapter 6 examined the determinants of exercise mediated fat loss, particularly with respect to possible mechanisms which compensate for the energy deficit of exercise and lead to a reduction in predicted fat loss. These results suggest that the most important determinant of fat loss in response to exercise remains the energy deficit of the exercise intervention.

Additionally, the novel finding from this study was that the change in fat mass (after adjustment for changes in energy intake and the energy deficit of exercise) was related to the change in fasting RER, suggesting that fat loss is associated with a shift towards fasting fat oxidation. This observation was independent of other variables, suggesting an innate response to exercise which contributes to an individual's ability to reduce fat mass. Those with the highest pre-exercise RER showed greatest reduction in post-exercise RER, providing further evidence that those who are predisposed to weight gain may benefit most from a change from a sedentary lifestyle. However, baseline RER was not significantly associated with fat loss and change in fasting RER was also associated with baseline factors such as diabetes family history. Therefore an 'innate' predisposition to fasting carbohydrate oxidation in a sedentary state and an exercise stimulated shift towards fasting fat oxidation is associated with greater fat loss, independent of the energy deficit of the intervention.

The remainder of this chapter will focus upon areas of interest generated by these studies and potential future avenues of research which may further expand the understanding of the interaction of diabetes family history, exercise and metabolic health.

# 7.2 Adipose tissue – distribution, function and insulin sensitivity

Results from Chapters 3, 4 and 5 reflect the importance of adiposity in contributing to insulin sensitivity. Increased adiposity is strongly associated with insulin resistance, and the relationship between adiposity and baseline insulin resistance in Offspring suggests an increased sensitivity to adipose tissue in this group. Parallels can be drawn with other groups at high risk of T2D, in particular South Asians (Razak et al. 2007). It is suggested that maximal adipose tissue compartment size may be genetically determined (Chandalia et al. 2007) and that since different adipose tissue pools are recognised to have different metabolic properties (Walker et al. 2007) increased fat mass may have a distinct metabolic impact

which is dependent on the expansion of a specific adipose pool. Central adiposity is recognised to be an important co-founder in the development of T2D however, it is unclear whether visceral fat (Banerji et al. 1999), deep subcutaneous fat or superficial subcutaneous fat (Chandalia et al. 2007; Sniderman et al. 2007) is the most important metabolically active pool, particularly with respect to circulating NEFA and adipose tissue derived hormones. Sniderman and colleagues (2007) suggest that in South Asians a reduction in the size of the subcutaneous adipose compartment leads to relative expansion of deep subcutaneous and visceral pools compared to Caucasian subjects exposed to a similar environment (Sniderman et al. 2007). In Pima Indians, visceral fat mass was not found to be independently associated with insulin resistance (Gautier et al. 1999). Further investigation of the metabolic activity of visceral, deep subcutaneous and superficial subcutaneous fat in Pimas suggests that expanded deep subcutaneous fat is more sensitive to lipolysis than superficial subcutaneous fat and may have greater influence on skeletal muscle insulin sensitivity (Koska et al. 2008b). Additionally, changes in both hyperinsulinaemia and circulating leptin independent of fat mass in response to weight loss suggests that adipose tissue function may be more important than total adiposity (Doucet et al. 2000). It is recognised that basal and stimulated rates of lipolysis differ between superficial and deep adipose tissue pools and that adipokine production (leptin and  $TNF\alpha$ ) may also be dependent on specific pool volume and local effects of systemic signals (such as catecholamines) (van Harmelen et al. 2002). Detailed comparison of the functional aspects of deep and superficial adipose tissue has not yet been performed (Koska et al. 2008), analysis of rates of lipolysis in basal and stimulated conditions, evidence of local inflammation and adipokine (and receptor) expression in these pools would be of value. In addition, alterations in specific tissue pools in response to exercise would be of interest, particularly when comparing Offspring and Control responses.

Access to visceral fat is limited in a research environment however both visceral and subcutaneous adipose tissue could be obtained in elective surgical patients. Laparoscopic cholecystectomy is a frequently performed, minimally invasive surgical technique which

allows access to the abdominal viscera. This operation is usually an elective procedure and therefore would allow recruitment of subjects to a study. Gallstone disease in the United Kingdom has a high prevalence in young, menstruating women (Shaffer 2006) and it would be possible to recruit age and BMI-matched women planned for elective surgery based on their family history of diabetes and obtain samples of visceral and subcutaneous adipose tissue during the process of their surgery. Whilst this would not add to understanding of exercise-mediated change in adipose tissue, it may provide more information about specific characteristics of adipose tissue morphology and inflammatory protein expression in separate tissue pools and differences between Offspring and Controls.

The supraphysiological dose of glucose required in the OGTT completely suppressed circulating NEFA in both Offspring and Control volunteers. Further investigation is required to understand the influence of adipose tissue on insulin resistance in these groups. In future, in vivo subcutaneous fatty acid metabolism in matched Offspring and Control subjects should be studied using the arteriovenous difference technique described by Frayn and colleagues (Frayn, Coppack, & Humphreys 1993). The NEFA and TG response to either a reduced dose (20 grams) of oral glucose or a mixed meal with this technique would provide a more physiological assessment of lipid metabolism in adipose tissue and skeletal muscle (Tan et al. 2008). Increased NEFA efflux from adipose tissue in the post-prandial state promotes hepatic glucose production and hepatic and skeletal muscle steatosis and insulin resistance. Use of these techniques would further describe the role of adipose tissue and NEFA in the aetiology of insulin resistance in Offspring. The use of the hyperinsulinaemic euglycaemic clamp with radiolabelled glucose would be useful to clarify the hepatic and skeletal muscle contributions to total body insulin resistance (DeFronzo, Simonson, & Ferrannini 1982). These techniques are expensive, labour-intensive and more unpleasant for the volunteer than those used in the studies in this thesis but would have the advantage of providing potentially unifying data on the interaction between adipose tissue, liver and skeletal muscle in Offspring and Controls.

Adipokines have autocrine and paracrine activity and this may not be reflected in peripheral concentrations, therefore use of the microdialysis technique with peripheral venous sampling would allow detection of differences in adipokine activity (and possibly the site of activity) between Offspring and Controls (Dostalova et al. 2009;Nielsen et al. 2009). In addition, combining knowledge of adipokine production with direct microscopy of adipose tissue from biopsies would provide a more thorough understanding of the relationship between adipocytes, adipose-bound inflammatory cells and adipokines, particularly if such studies were performed before and after an exercise intervention.

Further exploration of adiposity in sedentary females with a parental history of T2D would be valid. The methods used in the studies contained in this thesis were unable to accurately describe visceral fat or distinguish deep and subcutaneous adipose tissue. Computed tomography (CT), magnetic resonance imaging (MRI) and ultrasonography (US) are able to distinguish between visceral and subcutaneous adipose tissue and by identification of the *fascia superficialis* (Figure 7.1) quantify deep and superficial subcutaneous compartments (Smith et al. 2001). Ultrasound guided placement of deep and superficial subcutaneous microdialysis catheters would allow comparison of adipokine efflux from these two distinct adipose pools (Siklova-Vitkova et al. 2009). This may be of use in determining if the 'overflow hypothesis' suggested by Sniderman and colleagues (Sniderman et al. 2007) has metabolic evidence to support the suggestion that deep subcutaneous adipose tissue is more metabolically active than superficial tissue. To date, there does not appear to have been any attempt to utilise ultrasound guidance to sample these compartments.

Kelley and colleagues (2000) observed associations between deep subcutaneous and visceral adipose tissue and both dyslipidaemia and increased insulin and glucose concentrations
which were not apparent for superficial subcutaneous adipose tissue (Kelley et al. 2000). In addition, Smith and co-workers describe sexual dimorphism in distribution of deep and superficial subcutaneous and visceral fat (Smith et al. 2001) and within males and females, ethnic differences in adipose tissue distribution are well recognised (Chandalia et al. 2007;Lear et al. 2007).



**Figure 7.1** Division of deep and superficial subcutaneous adipose by *fascia superficialis*. Adapted from Smith et al. (2001).

The results described in chapter 5 include an association between improved insulin sensitivity and reduced lower limb adiposity after the exercise intervention. An independent association between lower limb adiposity and insulin resistance has previously been reported (Goodpaster, Thaete, & Kelley 2000;Livingston 2006) and an association between reduced lower limb adiposity and improved insulin sensitivity in response to exercise has been described in only one other study (Goodpaster et al. 2003). Detailed imaging of the lower limb subcutaneous adipose tissue compartments, using modalities described earlier, may be of benefit in investigating these associations. Magnetic resonance imaging of the torso and limbs before and after exercise could determine alterations in subcutaneous and visceral adipose tissue pools. With respect to the lower limb, MRI would give adequate image resolution to assess the degree of subcutaneous and intramuscular adipose deposition. If lower limb fat measured by DEXA is in fact a surrogate for intramuscular adipose content then more detailed imaging should be able to determine this. Goodpaster and colleagues observed that only the subfascial fat compartment in the thigh was associated with improved insulin resistance, but that CT did not suggest any significant change in intramuscular lipid content (Goodpaster et al. 2003). However the 'athlete's paradox' would highlight that IMCL content is less important than lipid intermediate burden in determining insulin sensitivity (Dube et al. 2008). Therefore direct assessment of the lipid content of lower limb muscle would be required to determine whether exercise is associated with a more 'pro-metabolic' profile of intramuscular lipid. Vastus lateralis muscle biopsy allows access to skeletal muscle and has the benefit of being the specific tissue of interest in this case. In addition to determining the overall lipid content of the muscle, thigh biopsy would also allow analysis of lipid droplets and their physical relationship to mitochondria which is thought to be closely associated with peripheral insulin sensitivity (Shaw et al. 2009). Thigh muscle biopsy before and after an exercise intervention in matched Offspring and Controls would provide information on fibre type, lipid and intermediate content and distribution and could also be analysed for the activity of components of the insulin signalling cascade (Cochran et al. 2010). If it were possible to combine these studies with subcutaneous adipose biopsies then the analysis of adipokine receptor expression in both tissues would be of great value in describing the relationship between adipose tissue and skeletal muscle. The author notes the challenge of maintaining adherence to a study which proposes adipose and muscle biopsy, assessment of adipose tissue function by arteriovenous assessment or microdialysis (± ultrasound guided sampling of deep tissue), use of the euglycaemic clamp and a seven week exercise intervention. In addition the additional cost of more complex imaging and the technical burden of the studies would require both an extravagant grant and a very sympathetic ethics committee.

The importance of adipose tissue function in determining insulin sensitivity and exercise mediated changes in insulin sensitivity was discussed in Chapter 3, 4 and 5. In each study adipose tissue activity influenced insulin sensitivity independently of fat mass. At baseline,

circulating NEFA concentrations were negatively associated with insulin sensitivity, independent of fat mass and adiponectin was correlated with insulin sensitivity in Offspring alone. In Chapters 4 and 5, the contribution of fat oxidation and reduced circulating leptin to the exercise mediated increase in insulin sensitivity in Offspring supports the hypothesis that exercise affects insulin sensitivity through altered adipose tissue function. The multivariate analysis in Chapter 5 did not find fat oxidation to be a significant factor in the exercisemediated improvement in insulin sensivitity. This observation is surprising in view of the significant role of reduced leptin in improved insulin sensitivity. Arteriovenous differences in substrate utilisation can be measured across skeletal muscle (Kelley et al. 1999) and this may be a more specific method of examining the relationship between insulin sensitivity, leptin and fat oxidation. The association between adiponectin and insulin sensitivity was only observed at baseline. Adiponectin is now recognised to have more than one active form, and at least two distinct receptors (Yamauchi & Kadowaki 2008) therefore the assay used in these studies may not have been specific enough to detect alterations in specific isoforms of adiponectin. Interestingly, Polak and colleagues (2007) observed increases in adiponectin multimers after dietary induced weight loss, but these were not associated with the observed change in insulin sensitivity (Polak et al. 2007). Abdominal subcutaneous adipose samples were obtained by liposuction biopsy before and after the exercise intervention. Some analysis of these samples has been performed and the increase in adipose tissue 'leptin sensitivity' has been discussed in Section 7.1 (Moran et al. 2010). As previously described, microscopy of the adipose tissue samples with respect to adipocyte size and inflammatory cell infiltration would provide important information about adipose tissue inflammation and the effects of exercise.

Minimally invasive imaging techniques such as magnetic resonance spectroscopy (MRS) offer real-time description of metabolic function in both muscle (Carey et al. 2003) and liver (Ravikumar et al. 2008). Functional skeletal muscle imaging using techniques such as proton MRS offer the opportunity to examine changes in IMCL in response to exercise (Machann,

Stefan, & Schick 2008), whilst <sup>31</sup>P MRS is a marker of skeletal muscle mitochondrial function (Meex et al. 2010). These methods allow minimally invasive assessment of a specific aspect of skeletal muscle function but alone, cannot provide enough information to determine the mechanism of improved insulin sensitivity in response to exercise. Instead, MRS is frequently utilised in concert with such methods as muscle biopsy and indirect calorimetry in an attempt to define the metabolic pathway of interest (Meex et al. 2010).

### 7.3 Lifestyle factors and implications for weight loss and insulin sensitivity

Habitual physical activity was considered to be important in all experimental chapters. At baseline, contrary to the findings of Higgins and co-workers (2005), Controls displayed a relationship between physical activity and insulin sensitivity which was not observed in Offspring despite a lower level of habitual activity. Increased habitual physical activity was also positively associated with change in insulin sensitivity after exercise in analysis of the combined Offspring and Control group. It is appreciated that the method of assessment of habitual physical activity utilised in these studies is crude. Unfortunately, attempts to assess energy expenditure using the FLEX method were unsuccessful and prevented detailed examination of energy expenditure. Higher levels of habitual physical activity are thought to be protective against chronic diseases such as T2D (Helmrich et al. 1991) and cardiovascular disease (Franco et al. 2005), and may be able to compensate for inherited or acquired tendencies to these diseases (Laaksonen et al. 2003;Gill & Malkova 2006;Gill & Cooper 2008).

In Chapter 6 increased fat loss was observed in subjects who had increased habitual physical activity. Therefore, more robust assessment of physical activity is warranted both at baseline and during an exercise intervention to determine the dose-response relationship between the degree of inactivity and deleterious metabolic perturbations and to examine the exercise

associated alteration in lifestyle which may contribute to, or mitigate against both reversal of these metabolic defects and weight loss.

Despite the problems encountered with the FLEX method on this occasion, it remains a well validated (Livingstone et al. 1990) and relatively simple technique. Prior to further use of the FLEX process, further investigation is warranted into the methodology utilised in the pilot study described in Chapter 2. Comparison with the doubly-labelled water technique would be technically complicated and expensive, but would allow definitive assessment of the validity of the technique used in Chapter 2 and validate any adjustments to the methodology.

In all studies participants were asked to maintain their usual dietary habits. However, at baseline (Chapter 3) and after the exercise intervention (Chapter 4), no differences were observed between Offspring and Controls with respect to either energy intake or dietary composition. In those who participated in the analysis of fat loss (Chapter 6), a significant reduction in energy intake was observed ( $500 \pm 1530 \text{ kj.d}^{-1}$ , p < 0.05) after exercise. It is recognised that self-reported dietary intake is prone to misreporting, but that these errors are reasonably consistent within individuals (Black & Cole 2001). It is therefore likely that individuals who under- or over-report dietary data will have done so before and during, the exercise intervention. It would be difficult to achieve more accurate data on energy intake in free living individuals over a seven week period. It is however important for future exercise studies to continue to examine energy balance to ensure that results are not confounded by non-study alterations in energy balance. Understanding of the role of macronutrient intake in the development of insulin resistance in people at risk of diabetes is likely to require not only accurate recording of dietary intake, but also greater understanding of the impact of innate and environmental influences on subsequent fat and carbohydrate metabolism (Bergouignan et al. 2009).

In Chapter 6 the ability to reduce fasting RER in response to exercise was a significant determinant of fat loss over the course of the intervention. The propensity to switch from fasting carbohydrate oxidation to fat oxidation in response to exercise, independent of the energy deficit of a seven week intervention suggests a metabolic mechanism which is exercise-sensitive. However, the study described in Chapter 6 raises further questions. Firstly: What is the exercise-sensitive process which is stimulated in these individuals? Secondly: Can the baseline fasting RER response to a single session in sedentary women predict fat loss response to an exercise intervention and furthermore, can this information allow practical individualised advice on exercise prescription for weight loss? Studies to answer the first question would require more detailed assessment of the in vivo oxidative function of skeletal muscle. Application of the arteriovenous difference method of substrate utilisation (Kelley et al. 1999) in a skeletal muscle group could provide specific data on the metabolic response in muscle, rather than in whole body. Magnetic resonance spectroscopy using <sup>31</sup>P would be a less invasive method of examining the muscle oxidative capacity in subjects who reduce fasting RER. Additionally, since substrate utilisation is dependent on substrate delivery to tissues, an assessment of skeletal muscle microvascular function is warranted. After review of the literature, the author's preference would be to employ contrast enhanced ultrasound, however if less invasive and expensive techniques were more appropriate venous occlusion plethysmography would be an alternative (Clark 2008).

Finally, the relationship between reduced resting heart rate and increased insulin sensitivity after the exercise intervention warrants further exploration. Resting heart rate may simply be a crude indicator of increased fitness or increased habitual activity and therefore a more robust assessment of habitual activity could be helpful in determining if this is the case. The alternative explanation, that resting heart rate is a surrogate for autonomic dysfunction, could be explored by examining heart rate recovery after exercise and it is likely that the current fitness testing protocol could be adapted to allow this to be included (Yeckel et al. 2009).

#### 7.4 Results in women – implications for future studies

#### 7.4.1 General considerations

The findings described in this thesis are exclusively applicable to women, furthermore the majority of the participants in the studies were European Caucasians. These factors have important implications for both future research and any potential public health message. With respect to both of these points it would clearly be beneficial to perform similar studies with sedentary male participants. The mechanisms and mediators of the development of insulin resistance and atherosclerosis, as well as the outcomes from the diseases themselves would appear to differ between the sexes (Engberding & Wenger 2008;Evangelista & McLaughlin 2009;Kilhovd et al. 2005). Observational evidence also suggests that exercise may induce different metabolic changes in women compared to men (Perreault et al. 2004;Soto et al. 2008). In addition, exercise responses in women may be dependent upon the menstrual phase (Northoff et al. 2008) which supports the methods used in this study of test-retest intervals which would be expected to align with menstrual cycle in the majority of women or an alternative method which utilises hormone analysis to dictate testing.

Larger numbers of participants and a longer period of exercise intervention may be desirable for future studies however from the author's perspective the practical considerations of maintaining volunteer involvement and compliance in such a labour intensive process would pose further problems. The remaining sections of this thesis will therefore consider specific studies which would seek to advance understanding of the mechanisms of firstly; increased insulin resistance in Offspring subjects, secondly; the metabolic processes which underpin the observed 'hyper-response' in those who are most insulin resistant and finally; the reasons why a switch from fasting carbohydrate oxidation to fat oxidation in response to exercise might promote fat loss. The process of performing these studies, the data analysis and preparation of the thesis has led the author to conclude that there is no such thing as a 'perfect study' and therefore the suggestions in the three final sections of this thesis are made with an appreciation of financial cost, the practicality of proposed methods and the burden on both the investigator and most importantly, the volunteers.

#### 7.4.2 Insulin resistance in Offspring

The hypothesis that insulin resistance in Offspring is in part determined by adipose tissue derived factors would appear to be correct. However, to consider adipose tissue in isolation is clearly inadequate and with respect to the aetiology of insulin resistance the interaction with the liver, skeletal muscle and vascular endothelium are of particular importance in exploring the mechanisms which confer insulin resistance in Offspring. In retrospect, PWV may have been an inadequate surrogate for endothelial function, particularly in insulin-sensitive subjects. Future baseline assessment of matched Offspring should consider alternative techniques such as contrast enhanced ultrasound to provide minimally invasive assessment of the skeletal muscle microcirculation. It is likely that microvascular dysfunction is an early abnormality in insulin resistance and that it may be a key cofounder in the process (Rattigan et al. 2006). It would be reasonable to consider using a lower dose of oral glucose to prevent the degree of hyperinsulinaemia and NEFA suppression that was found in Chapter 3, accompanied by repetition of the Deltatrac measurement at 1-hour post glucose in an attempt to detect subtle differences in diet-induced thermogenesis between Offspring and Controls. Whilst the 'gold standard' euglycaemic clamp would potentially provide distinct data on muscle and liver insulin sensitivity, it is questionable whether the expense, complexity and tolerability of the procedure is worth the additional data in the context of the other investigations proposed. Since skeletal muscle is the major determinant of total body insulin sensitivity more extensive investigation of skeletal muscle is warranted. Whilst non invasive techniques such as CT/MRI and MRS can provide detailed anatomical and limited functional data, muscle biopsy would be the preferred investigation. This would allow comparison of

muscle fibre type, IMCL content and droplet location, measurement of lipid intermediates, adipokine (particularly leptin) receptor expression and perhaps most importantly analysis of the activity of the components of the insulin signalling cascade. The continued use of abdominal adipose biopsies would allow a unification of the role of the adipokine locally and in skeletal muscle. Direct examination of inflammatory cell infiltration and cytokine (and receptor) expression in adipose tissue and muscle, with evidence of cytokine-mediated insulin signalling dysregulation (e.g. Akt inactivation) in muscle and impaired capillary recruitment would be strongly supportive of a link between adipose tissue-derived low grade inflammation and skeletal muscle insulin resistance.

# 7.4.3 Augmented metabolic response to exercise

The primary hypothesis of Chapters 4 and 5 was that Offspring would show an augmented metabolic response to exercise. This was considered to have been correct. Improved insulin sensitivity was accompanied by increased fat oxidation and a reduction in circulating leptin. Adipose tissue studies before and after an exercise intervention suggest that Offspring display increased 'leptin sensitivity' compared to matched Controls (Moran et al. 2010). Again, consideration of adipose tissue in isolation prevents exploration of a mechanism which is likely to have resulted in increased skeletal muscle insulin sensitivity. Muscle and adipose tissue biopsy before and after the exercise intervention would be of value. Evidence of improved muscle leptin sensitivity allied to increased activity of components of the insulin signalling cascade which are considered to be leptin responsive would be a strong indication of a mechanism by which leptin mediates increased insulin sensitivity without stimulating fat oxidation. In practice it would be very difficult to separate the effects of exercise on fat oxidation and insulin signalling from any effects of leptin, particularly as leptin may also upregulate skeletal muscle fat oxidation and and reduction in ceramides might also stimulate activation of downstream insulin signalling. Ideally, the administration of a short-acting potent malonyl CoA mimic prior to the muscle biopsy would be expected to 'switch off' fat

oxidation and clarify the role of increased leptin expression on insulin signalling. However, the author is not aware of any such compound and the likely toxicity of such a substance would probably preclude its use in human subjects.

For the same rationale as at baseline, post-exercise assessment of adipose tissue inflammation and skeletal muscle capillary function would be of value. Description of adipose tissue compartment response to exercise would be achieved using MRI, this would minimise radiation exposure to a group of relatively young women, whilst providing accurate data on the visceral, superficial subcutaneous and deep subcutaneous adipose pools. With respect to Sniderman and colleagues' (2007) 'overflow' hypothesis, exercise would be expected to preferentially reduce visceral or deep subcutaneous tissue, whilst leaving superficial tissue intact (Sniderman et al. 2007). If MRI were too expensive or unavailable, a single operator could be trained to use ultrasound to assess deep and superficial subcutaneous pools. In addition, identification of the fascia superficialis may allow selective sampling of adipokines by microdialysis although it is unlikely that many volunteers would agree to this in addition to muscle and adipose biopsies.

Assessment of the impact of an exercise intervention in free-living individuals is potentially confounded by alterations of diet and habitual activity. The limitations of dietary reporting have been discussed in Section 7.3. The assessment of habitual activity was limited by the problems with the FLEX method (Chapter 2). Prior to another intervention, review of the protocol for assessment of resting heart rate measurement at baseline should be considered in an attempt to ensure that the initial FLEX heart rate is a true representation of the inflexion point between sedentary and active energy expenditures. In addition, it would be appropriate to incorporate assessment of heart rate variability and/or heart rate recovery as markers of autonomic function (Carnethon et al. 2003). Associations between autonomic dysfunction and low-grade inflammation have been described (Seematter et al. 2004), but not in Offspring

261

subjects. A causal relationship is unlikely to be determined by these simple measurements but as in patients with polycystic ovarian syndrome (Giallauria et al. 2008), evidence of an exercise interaction with these factors would be worthy of further study.

# 7.4.4 Determinants of exercise-mediated fat loss

In Section 7.3 it was suggested that there were two potential studies which should expand on the work described in Chapter 6. Firstly, an attempt should be made to explore the association between exercise-mediated reduction in RER and fat loss. Robust assessment of total and regional adiposity should be performed and specific assessment of skeletal muscle, as well as total body, substrate oxidation would determine whether the change in fasting RER is evidence of metabolic flexibility (Kelley et al. 1999). Assessment of carbohydrate oxidation using <sup>13</sup>C or fatty acid oxidation using <sup>31</sup>P could be performed using MRS (Machann, Stefan, & Schick 2008). Alternatively, cannulation of leg vessels could also be combined with cannulation of the superficial epigastric vein to examine NEFA efflux and muscle substrate utilisation (Frayn et al. 1993;Kelley et al. 1999). The impact of exercise on endothelial function is of interest, since the ability to alter substrate oxidation requires adequate delivery of substrate to the tissue. Therefore contrast enhanced ultrasound would allow assessment of skeletal muscle capillary recruitment (Coggins et al. 2001). If there is evidence of microvascular dysfunction further investigation of inflammatory indices and NO bioavailability would be required since inflammatory cytokines such as TNFa reduce NO activity (Rattigan et al. 2006). Reduced NO action has endothelial effects but also interferes with insulin signalling, this may be a mechanism which promotes over-reliance on carbohydrate metabolism but the effects of exercise can bypass this insulin signalling interference and IL-6 release by skeletal muscle may be able to reverse the effects of  $TNF\alpha$ (Pedersen 2006).

Secondly, a more clinical application could be a prospective study to determine whether reduction in fasting RER in response to a single exercise session can predict fat loss over a subsequent exercise intervention. In this study, both the principle investigator and the volunteers would need to be blinded to their predetermined status as either 'responders' or 'non-responders', to prevent bias affecting the subsequent exercise intervention. If the primary hypothesis of this study was found to be correct, it would have important implications for the treatment of overweight and obesity. The cardiovascular benefits of increased physical activity have been recognised since the ground breaking epidemiological studies by Paffenbarger and Morris (Morris et al. 1980; Paffenbarger, Jr. et al. 1986). However, from a weight-loss perspective it may be of benefit to 'non-responders' to avoid becoming demoralised by lack of weight loss in response to moderate exercise and conversely, it may stimulate greater compliance with exercise advice if 'responders' are aware that they are metabolically predisposed to exercise-mediated weight loss. In the current economic climate health care resources are under increasing pressure to provide 'value for money'. Targetting exercise advice and intervention towards people with proven insulin resistance, women with a first-degree relative with T2D or overweight people who show metabolic response to exercise may, in all three situations, represent an effective use of resource.

### REFERENCES

Abete, I., Parra, D., De Morentin, B. M., & Alfredo, M. J. 2009, "Effects of two energyrestricted diets differing in the carbohydrate/protein ratio on weight loss and oxidative changes of obese men", *Int.J.Food Sci.Nutr.*, vol. 60 Suppl 3, pp. 1-13.

Adamson, A. J., Foster, E., Butler, T. J., Bennet, S., & Walker, M. 2001, "Non-diabetic relatives of Type 2 diabetic families: dietary intake contributes to the increased risk of diabetes", *Diabet.Med.*, vol. 18, no. 12, pp. 984-990.

Ahn, C. W., Song, Y. D., Nam, J. H., Kim, D. M., Woo, S. O., Park, S. W., Cha, B. S., Lim, S. K., Kim, K. R., Lee, J. H., Lee, H. C., & Huh, K. B. 2004, "Insulin sensitivity in physically fit and unfit children of parents with Type 2 diabetes", *Diabet.Med*, vol. 21, no. 1, pp. 59-63.

Ainslie, P., Reilly, T., & Westerterp, K. 2003, "Estimating human energy expenditure: a review of techniques with particular reference to doubly labelled water", *Sports Med.*, vol. 33, no. 9, pp. 683-698.

Al, H. H., Atoum, M., Alboqai, O., Ismail, L. C., Al, D. A., & Hijjawi, N. 2009, "Effectiveness of dietary intervention for obese women in Jordan", *Int.J.Food Sci.Nutr.*, vol. 60 Suppl 3, pp. 76-82.

Alexander, C. M., Landsman, P. B., Teutsch, S. M., & Haffner, S. M. 2003, "NCEP-defined metabolic syndrome, diabetes, and prevalence of coronary heart disease among NHANES III participants age 50 years and older", *Diabetes*, vol. 52, no. 5, pp. 1210-1214.

Alexandraki, K., Protogerou, A. D., Papaioannou, T. G., Piperi, C., Mastorakos, G., Lekakis, J., Panidis, D., & amanti-Kandarakis, E. 2006, "Early microvascular and macrovascular dysfunction is not accompanied by structural arterial injury in polycystic ovary syndrome", *Hormones.*(*Athens.*), vol. 5, no. 2, pp. 126-136.

Alkhateeb, H., Chabowski, A., Glatz, J. F., Gurd, B., Luiken, J. J., & Bonen, A. 2009, "RESTORING AS160 PHOSPHORYLATION RESCUES SKELETAL MUSCLE INSULIN RESISTANCE AND FATTY ACID OXIDATION WHILE NOT REDUCING INTRAMUSCULAR LIPIDS", *Am.J.Physiol Endocrinol.Metab*.

American College of Sports Medicine. Guidelines for exercise testing and prescription. 1995. Williams and Wilkins, Baltimore.

American Diabetes Association 2008, "Standards of medical care in diabetes--2008", *Diabetes Care*, vol. 31 Suppl 1, p. S12-S54.

An, P., Teran-Garcia, M., Rice, T., Rankinen, T., Weisnagel, S. J., Bergman, R. N., Boston, R. C., Mandel, S., Stefanovski, D., Leon, A. S., Skinner, J. S., Rao, D. C., & Bouchard, C. 2005, "Genome-wide linkage scans for prediabetes phenotypes in response to 20 weeks of endurance exercise training in non-diabetic whites and blacks: the HERITAGE Family Study", *Diabetologia*, vol. 48, no. 6, pp. 1142-1149.

Anderson, R. L., Hamman, R. F., Savage, P. J., Saad, M. F., Laws, A., Kades, W. W., Sands, R. E., & Cefalu, W. 1995, "Exploration of simple insulin sensitivity measures derived from

frequently sampled intravenous glucose tolerance (FSIGT) tests. The Insulin Resistance Atherosclerosis Study", *American Journal of Epidemiology*, vol. 142, no. 7, pp. 724-732.

Andersson, C. X., Gustafson, B., Hammarstedt, A., Hedjazifar, S., & Smith, U. 2008, "Inflamed adipose tissue, insulin resistance and vascular injury", *Diabetes Metab Res.Rev.*, vol. 24, no. 8, pp. 595-603.

Andrulionyte, L., Kuulasmaa, T., Chiasson, J. L., & Laakso, M. 2007, "Single nucleotide polymorphisms of the peroxisome proliferator-activated receptor-alpha gene (PPARA) influence the conversion from impaired glucose tolerance to type 2 diabetes: the STOP-NIDDM trial", *Diabetes*, vol. 56, no. 4, pp. 1181-1186.

Arita, Y., Kihara, S., Ouchi, N., Takahashi, M., Maeda, K., Miyagawa, J., Hotta, K., Shimomura, I., Nakamura, T., Miyaoka, K., Kuriyama, H., Nishida, M., Yamashita, S., Okubo, K., Matsubara, K., Muraguchi, M., Ohmoto, Y., Funahashi, T., & Matsuzawa, Y. 1999, "Paradoxical decrease of an adipose-specific protein, adiponectin, in obesity", *Biochem.Biophys.Res.Commun.*, vol. 257, no. 1, pp. 79-83.

Arner, P. 2002, "Insulin resistance in type 2 diabetes: role of fatty acids", *Diabetes Metab Res.Rev.*, vol. 18 Suppl 2, p. S5-S9.

Arner, P. 2005, "Human fat cell lipolysis: biochemistry, regulation and clinical role", *Best.Pract.Res.Clin.Endocrinol.Metab*, vol. 19, no. 4, pp. 471-482.

Astrup, A. & Finer, N. 2000, "Redefining type 2 diabetes: 'diabesity' or 'obesity dependent diabetes mellitus'?", *Obes.Rev.*, vol. 1, no. 2, pp. 57-59.

Avenell, A., Broom, J., Brown, T. J., Poobalan, A., Aucott, L., Stearns, S. C., Smith, W. C., Jung, R. T., Campbell, M. K., & Grant, A. M. 2004, "Systematic review of the long-term effects and economic consequences of treatments for obesity and implications for health improvement", *Health Technol.Assess.*, vol. 8, no. 21, p. iii-182.

Bakker, W., Eringa, E. C., Sipkema, P., & van, H., V 2009, "Endothelial dysfunction and diabetes: roles of hyperglycemia, impaired insulin signaling and obesity", *Cell Tissue Res.*, vol. 335, no. 1, pp. 165-189.

Balducci, S., Zanuso, S., Nicolucci, A., Fernando, F., Cavallo, S., Cardelli, P., Fallucca, S., Alessi, E., Letizia, C., Jimenez, A., Fallucca, F., & Pugliese, G. 2009, "Anti-inflammatory effect of exercise training in subjects with type 2 diabetes and the metabolic syndrome is dependent on exercise modalities and independent of weight loss", *Nutr.Metab Cardiovasc.Dis.* 

Balkau, B., Mhamdi, L., Oppert, J. M., Nolan, J., Golay, A., Porcellati, F., Laakso, M., & Ferrannini, E. 2008, "Physical activity and insulin sensitivity: the RISC study", *Diabetes*, vol. 57, no. 10, pp. 2613-2618.

Bandyopadhyay, G. K., Yu, J. G., Ofrecio, J., & Olefsky, J. M. 2006, "Increased malonyl-CoA levels in muscle from obese and type 2 diabetic subjects lead to decreased fatty acid oxidation and increased lipogenesis; thiazolidinedione treatment reverses these defects", *Diabetes*, vol. 55, no. 8, pp. 2277-2285.

Banerji, M. A., Faridi, N., Atluri, R., Chaiken, R. L., & Lebovitz, H. E. 1999, "Body composition, visceral fat, leptin, and insulin resistance in Asian Indian men", *J.Clin.Endocrinol.Metab*, vol. 84, no. 1, pp. 137-144.

Barker, D. J. 2005, "The developmental origins of insulin resistance", *Horm.Res.*, vol. 64 Suppl 3, pp. 2-7.

Barker, D. J., Osmond, C., Kajantie, E., & Eriksson, J. G. 2009, "Growth and chronic disease: findings in the Helsinki Birth Cohort", *Ann.Hum.Biol.* pp. 1-14.

Baron, A. D. 1994, "Hemodynamic actions of insulin", *Am.J.Physiol*, vol. 267, no. 2 Pt 1, p. E187-E202.

Barrett, E. J., Eggleston, E. M., Inyard, A. C., Wang, H., Li, G., Chai, W., & Liu, Z. 2009, "The vascular actions of insulin control its delivery to muscle and regulate the rate-limiting step in skeletal muscle insulin action", *Diabetologia*, vol. 52, no. 5, pp. 752-764.

Barrett-Connor, E. 1989, "Epidemiology, obesity, and non-insulin-dependent diabetes mellitus", *Epidemiol.Rev.*, vol. 11, pp. 172-181.

Barwell, N. D., Malkova, D., Moran, C. N., Cleland, S. J., Packard, C. J., Zammit, V. A., & Gill, J. M. 2008, "Exercise training has greater effects on insulin sensitivity in daughters of patients with type 2 diabetes than in women with no family history of diabetes", *Diabetologia*, vol. 51, no. 10, pp. 1912-1919.

Basu, A., Basu, R., Shah, P., Vella, A., Rizza, R. A., & Jensen, M. D. 2001, "Systemic and regional free fatty acid metabolism in type 2 diabetes", *Am.J.Physiol Endocrinol.Metab*, vol. 280, no. 6, p. E1000-E1006.

Becker, A., Bos, G., de Vegt, F., Kostense, P. J., Dekker, J. M., Nijpels, G., Heine, R. J., Bouter, L. M., & Stehouwer, C. D. 2003, "Cardiovascular events in type 2 diabetes: comparison with nondiabetic individuals without and with prior cardiovascular disease. 10year follow-up of the Hoorn Study", *Eur.Heart J*, vol. 24, no. 15, pp. 1406-1413.

Beltrand, J., Beregszaszi, M., Chevenne, D., Sebag, G., De, K. M., Huet, F., Polak, M., Tubiana-Rufi, N., Lacombe, D., De Paoli, A. M., & Levy-Marchal, C. 2007, "Metabolic correction induced by leptin replacement treatment in young children with Berardinelli-Seip congenital lipoatrophy", *Pediatrics*, vol. 120, no. 2, p. e291-e296.

Beltrand, J., Nicolescu, R., Kaguelidou, F., Verkauskiene, R., Sibony, O., Chevenne, D., Claris, O., & Levy-Marchal, C. 2009, "Catch-up growth following fetal growth restriction promotes rapid restoration of fat mass but without metabolic consequences at one year of age", *PLoS.ONE.*, vol. 4, no. 4, p. e5343.

Berg, A. H. & Scherer, P. E. 2005, "Adipose tissue, inflammation, and cardiovascular disease", *Circ.Res.*, vol. 96, no. 9, pp. 939-949.

Bergouignan, A., Trudel, G., Simon, C., Chopard, A., Schoeller, D. A., Momken, I., Votruba, S. B., Desage, M., Burdge, G. C., Gauquelin-Koch, G., Normand, S., & Blanc, S. 2009, "Physical inactivity differentially alters dietary oleate and palmitate trafficking", *Diabetes*, vol. 58, no. 2, pp. 367-376.

Berti, L. & Gammeltoft, S. 1999, "Leptin stimulates glucose uptake in C2C12 muscle cells by activation of ERK2", *Mol.Cell Endocrinol.*, vol. 157, no. 1-2, pp. 121-130.

Bi, Y., Zhu, D., Jing, Y., Hu, Y., Feng, W., Shen, S., Tong, G., Shen, X., Yu, T., Song, D., & Yang, D. 2010, "Decreased beta cell function and insulin sensitivity contributed to increasing fasting glucose in Chinese", *Acta Diabetol.* 

Bickerton, A. S., Roberts, R., Fielding, B. A., Hodson, L., Blaak, E. E., Wagenmakers, A. J., Gilbert, M., Karpe, F., & Frayn, K. N. 2007, "Preferential uptake of dietary Fatty acids in adipose tissue and muscle in the postprandial period", *Diabetes*, vol. 56, no. 1, pp. 168-176.

Bickerton, A. S., Roberts, R., Fielding, B. A., Tornqvist, H., Blaak, E. E., Wagenmakers, A. J., Gilbert, M., Humphreys, S. M., Karpe, F., & Frayn, K. N. 2008, "Adipose tissue fatty acid metabolism in insulin-resistant men", *Diabetologia*, vol. 51, no. 8, pp. 1466-1474.

Bjorntorp, P. 1995, "Insulin resistance: the consequence of a neuroendocrine disturbance?", *Int.J.Obes.Relat Metab Disord.*, vol. 19 Suppl 1, pp. S6-10.

Blaak, E. E. & Saris, W. H. 2002, "Substrate oxidation, obesity and exercise training", *Best.Pract.Res.Clin.Endocrinol.Metab*, vol. 16, no. 4, pp. 667-678.

Black, A. E. & Cole, T. J. 2001, "Biased over- or under-reporting is characteristic of individuals whether over time or by different assessment methods", *J.Am.Diet.Assoc.*, vol. 101, no. 1, pp. 70-80.

Black, S. E., Mitchell, E., Freedson, P. S., Chipkin, S. R., & Braun, B. 2005, "Improved insulin action following short-term exercise training: role of energy and carbohydrate balance", *J.Appl.Physiol*, vol. 99, no. 6, pp. 2285-2293.

Blake, D. R., Meigs, J. B., Muller, D. C., Najjar, S. S., Andres, R., & Nathan, D. M. 2004, "Impaired glucose tolerance, but not impaired fasting glucose, is associated with increased levels of coronary heart disease risk factors: results from the Baltimore Longitudinal Study on Aging", *Diabetes*, vol. 53, no. 8, pp. 2095-2100.

Bluher, M., Bullen, J. W., Jr., Lee, J. H., Kralisch, S., Fasshauer, M., Kloting, N., Niebauer, J., Schon, M. R., Williams, C. J., & Mantzoros, C. S. 2006, "Circulating adiponectin and expression of adiponectin receptors in human skeletal muscle: associations with metabolic parameters and insulin resistance and regulation by physical training", *J.Clin.Endocrinol.Metab*, vol. 91, no. 6, pp. 2310-2316.

Bo, S., Ciccone, G., Guidi, S., Gambino, R., Durazzo, M., Gentile, L., Cassader, M., Cavallo-Perin, P., & Pagano, G. 2008, "Diet or exercise: what is more effective in preventing or reducing metabolic alterations?", *Eur.J.Endocrinol.*, vol. 159, no. 6, pp. 685-691.

Boden, G. 1997, "Role of fatty acids in the pathogenesis of insulin resistance and NIDDM", *Diabetes*, vol. 46, no. 1, pp. 3-10.

Boden, G. & Shulman, G. I. 2002, "Free fatty acids in obesity and type 2 diabetes: defining their role in the development of insulin resistance and beta-cell dysfunction", *Eur.J.Clin.Invest*, vol. 32 Suppl 3, pp. 14-23.

Bonen, A., Dohm, G. L., & van Loon, L. J. 2006, "Lipid metabolism, exercise and insulin action", *Essays Biochem.*, vol. 42, pp. 47-59.

Bonet, B., Viana, M., Sanchez-Vera, I., Quintanar, A., Martinez, J., & Espino, M. 2007, "Adipose tissue and liver lipid metabolism in obese children: role of the body mass index and the presence of acanthosis nigricans", *Diabet.Med.*, vol. 24, no. 11, pp. 1192-1198.

Bonora, E., Formentini, G., Calcaterra, F., Lombardi, S., Marini, F., Zenari, L., Saggiani, F., Poli, M., Perbellini, S., Raffaelli, A., Cacciatori, V., Santi, L., Targher, G., Bonadonna, R., & Muggeo, M. 2002, "HOMA-estimated insulin resistance is an independent predictor of cardiovascular disease in type 2 diabetic subjects: prospective data from the Verona Diabetes Complications Study", *Diabetes Care*, vol. 25, no. 7, pp. 1135-1141.

Booth, F. W., Chakravarthy, M. V., & Spangenburg, E. E. 2002, "Exercise and gene expression: physiological regulation of the human genome through physical activity", *J.Physiol*, vol. 543, no. Pt 2, pp. 399-411.

Booth, F. W., Gordon, S. E., Carlson, C. J., & Hamilton, M. T. 2000, "Waging war on modern chronic diseases: primary prevention through exercise biology", *J.Appl.Physiol*, vol. 88, no. 2, pp. 774-787.

Booth, G. L., Kapral, M. K., Fung, K., & Tu, J. V. 2006, "Relation between age and cardiovascular disease in men and women with diabetes compared with non-diabetic people: a population-based retrospective cohort study", *Lancet*, vol. 368, no. 9529, pp. 29-36.

Borg, G. A. 1973, "Perceived exertion: a note on "history" and methods", *Med.Sci.Sports*, vol. 5, no. 2, pp. 90-93.

Borghouts, L. B. & Keizer, H. A. 2000, "Exercise and insulin sensitivity: a review", *Int.J.Sports Med.*, vol. 21, no. 1, pp. 1-12.

Bosselaar, M., Smits, P., & Tack, C. J. 2009, "Ischaemia and insulin, but not ischaemia and contraction, act synergistically in stimulating muscle glucose uptake in vivo in humans", *Clin.Sci.(Lond)*, vol. 116, no. 2, pp. 157-164.

Bouchard, C., Tremblay, A., Despres, J. P., Theriault, G., Nadeau, A., Lupien, P. J., Moorjani, S., Prudhomme, D., & Fournier, G. 1994, "The response to exercise with constant energy intake in identical twins", *Obes.Res.*, vol. 2, no. 5, pp. 400-410.

Bouchard, C., Tremblay, A., Leblanc, C., Lortie, G., Savard, R., & Theriault, G. 1983, "A method to assess energy expenditure in children and adults", *Am.J.Clin.Nutr.*, vol. 37, no. 3, pp. 461-467.

Bouchard, C., Tremblay, A., Nadeau, A., Dussault, J., Despres, J. P., Theriault, G., Lupien, P. J., Serresse, O., Boulay, M. R., & Fournier, G. 1990, "Long-term exercise training with constant energy intake. 1: Effect on body composition and selected metabolic variables", *Int.J.Obes.*, vol. 14, no. 1, pp. 57-73.

Boudou, P., Sobngwi, E., Mauvais-Jarvis, F., Vexiau, P., & Gautier, J. F. 2003, "Absence of exercise-induced variations in adiponectin levels despite decreased abdominal adiposity and improved insulin sensitivity in type 2 diabetic men", *Eur.J.Endocrinol.*, vol. 149, no. 5, pp. 421-424.

Boule, N. G., Weisnagel, S. J., Lakka, T. A., Tremblay, A., Bergman, R. N., Rankinen, T., Leon, A. S., Skinner, J. S., Wilmore, J. H., Rao, D. C., & Bouchard, C. 2005, "Effects of exercise training on glucose homeostasis: the HERITAGE Family Study", *Diabetes Care*, vol. 28, no. 1, pp. 108-114.

Bradley, R. L., Jeon, J. Y., Liu, F. F., & Maratos-Flier, E. 2008, "Voluntary exercise improves insulin sensitivity and adipose tissue inflammation in diet-induced obese mice", *Am.J.Physiol Endocrinol.Metab*, vol. 295, no. 3, p. E586-E594.

Bruce, R. A., Kusumi, F., & Hosmer, D. 1973, "Maximal oxygen intake and nomographic assessment of functional aerobic impairment in cardiovascular disease", *Am.Heart J.*, vol. 85, no. 4, pp. 546-562.

Brunner, E. J., Shipley, M. J., Witte, D. R., Fuller, J. H., & Marmot, M. G. 2006, "Relation between blood glucose and coronary mortality over 33 years in the Whitehall Study", *Diabetes Care*, vol. 29, no. 1, pp. 26-31.

Bulugahapitiya, U., Siyambalapitiya, S., Sithole, J., & Idris, I. 2009, "Is diabetes a coronary risk equivalent? Systematic review and meta-analysis", *Diabet.Med.*, vol. 26, no. 2, pp. 142-148.

Burton, F. L., Malkova, D., Caslake, M. J., & Gill, J. M. 2008, "Energy replacement attenuates the effects of prior moderate exercise on postprandial metabolism in overweight/obese men", *Int.J.Obes.(Lond)*, vol. 32, no. 3, pp. 481-489.

Byrne, N. M., Meerkin, J. D., Laukkanen, R., Ross, R., Fogelholm, M., & Hills, A. P. 2006, "Weight loss strategies for obese adults: personalized weight management program vs. standard care", *Obesity.(Silver.Spring)*, vol. 14, no. 10, pp. 1777-1788.

Caballero, A. E., Arora, S., Saouaf, R., Lim, S. C., Smakowski, P., Park, J. Y., King, G. L., LoGerfo, F. W., Horton, E. S., & Veves, A. 1999, "Microvascular and macrovascular reactivity is reduced in subjects at risk for type 2 diabetes", *Diabetes*, vol. 48, no. 9, pp. 1856-1862.

Carantoni, M., Abbasi, F., Azhar, S., Schaaf, P., & Reaven, G. M. 1999, "Can changes in plasma insulin concentration explain the variability in leptin response to weight loss in obese women with normal glucose tolerance?", *J.Clin.Endocrinol.Metab*, vol. 84, no. 3, pp. 869-872.

Carey, P. E., Halliday, J., Snaar, J. E., Morris, P. G., & Taylor, R. 2003, "Direct assessment of muscle glycogen storage after mixed meals in normal and type 2 diabetic subjects", *Am.J.Physiol Endocrinol.Metab*, vol. 284, no. 4, p. E688-E694.

Carnethon, M. R. & Craft, L. L. 2008, "Autonomic regulation of the association between exercise and diabetes", *Exerc.Sport Sci.Rev.*, vol. 36, no. 1, pp. 12-18.

Carnethon, M. R., Jacobs, D. R., Jr., Sidney, S., & Liu, K. 2003, "Influence of autonomic nervous system dysfunction on the development of type 2 diabetes: the CARDIA study", *Diabetes Care*, vol. 26, no. 11, pp. 3035-3041.

Carr, A., Hudson, J., Chuah, J., Mallal, S., Law, M., Hoy, J., Doong, N., French, M., Smith, D., & Cooper, D. A. 2001, "HIV protease inhibitor substitution in patients with

lipodystrophy: a randomized, controlled, open-label, multicentre study", *AIDS*, vol. 15, no. 14, pp. 1811-1822.

Castillo, C., Bogardus, C., Bergman, R., Thuillez, P., & Lillioja, S. 1994, "Interstitial insulin concentrations determine glucose uptake rates but not insulin resistance in lean and obese men", *J.Clin.Invest*, vol. 93, no. 1, pp. 10-16.

Catalan, V. S., Couture, J. A., & LeLorier, J. 2001, "Predictors of persistence of use of the novel antidiabetic agent acarbose", *Arch.Intern.Med.*, vol. 161, no. 8, pp. 1106-1112.

Ceesay, S. M., Prentice, A. M., Day, K. C., Murgatroyd, P. R., Goldberg, G. R., Scott, W., & Spurr, G. B. 1989, "The use of heart rate monitoring in the estimation of energy expenditure: a validation study using indirect whole-body calorimetry", *Br.J.Nutr.*, vol. 61, no. 2, pp. 175-186.

Centers for Disease Control and Prevention (CDC) 2009, "Differences in prevalence of obesity among black, white, and Hispanic adults - United States, 2006-2008", *MMWR Morb.Mortal.Wkly.Rep.*, vol. 58, no. 27, pp. 740-744.

Cerasi, E. 1995, "Insulin deficiency and insulin resistance in the pathogenesis of NIDDM: is a divorce possible?", *Diabetologia*, vol. 38, no. 8, pp. 992-997.

Cersosimo, E. & DeFronzo, R. A. 2006, "Insulin resistance and endothelial dysfunction: the road map to cardiovascular diseases", *Diabetes Metab Res.Rev.*, vol. 22, no. 6, pp. 423-436.

Chandalia, M., Lin, P., Seenivasan, T., Livingston, E. H., Snell, P. G., Grundy, S. M., & Abate, N. 2007, "Insulin resistance and body fat distribution in South Asian men compared to Caucasian men", *PLoS.ONE.*, vol. 2, no. 8, p. e812.

Chang, A. M., Smith, M. J., Bloem, C. J., Galecki, A. T., Halter, J. B., & Supiano, M. A. 2006, "Limitation of the homeostasis model assessment to predict insulin resistance and betacell dysfunction in older people", *J.Clin.Endocrinol.Metab*, vol. 91, no. 2, pp. 629-634.

Chang, S. A., Kim, H. S., Yoon, K. H., Ko, S. H., Kwon, H. S., Kim, S. R., Lee, W. C., Yoo, S. J., Son, H. S., Cha, B. Y., Lee, K. W., Son, H. Y., & Kang, S. K. 2004, "Body mass index is the most important determining factor for the degree of insulin resistance in non-obese type 2 diabetic patients in Korea", *Metabolism*, vol. 53, no. 2, pp. 142-146.

Chen, X., Iqbal, N., & Boden, G. 1999, "The effects of free fatty acids on gluconeogenesis and glycogenolysis in normal subjects", *J.Clin.Invest*, vol. 103, no. 3, pp. 365-372.

Chiasson, J. L., Josse, R. G., Gomis, R., Hanefeld, M., Karasik, A., & Laakso, M. 2002, "Acarbose for prevention of type 2 diabetes mellitus: the STOP-NIDDM randomised trial", *Lancet*, vol. 359, no. 9323, pp. 2072-2077.

Christ-Roberts, C. Y., Pratipanawatr, T., Pratipanawatr, W., Berria, R., Belfort, R., Kashyap, S., & Mandarino, L. J. 2004, "Exercise training increases glycogen synthase activity and GLUT4 expression but not insulin signaling in overweight nondiabetic and type 2 diabetic subjects", *Metabolism*, vol. 53, no. 9, pp. 1233-1242.

Church, T. S., Kampert, J. B., Wilkinson, W. J., Dunn, A. L., & Blair, S. N. 2001, "Evaluating the reproducibility and validity of the Aerobic Adaptation Test", *Med.Sci.Sports Exerc.*, vol. 33, no. 10, pp. 1770-1773.

Clark, M. G. 2008, "Impaired microvascular perfusion: a consequence of vascular dysfunction and a potential cause of insulin resistance in muscle", *Am.J.Physiol Endocrinol.Metab*, vol. 295, no. 4, p. E732-E750.

Clark, M. G., Wallis, M. G., Barrett, E. J., Vincent, M. A., Richards, S. M., Clerk, L. H., & Rattigan, S. 2003, "Blood flow and muscle metabolism: a focus on insulin action", *Am.J.Physiol Endocrinol.Metab*, vol. 284, no. 2, p. E241-E258.

Cleland, S. J., Petrie, J. R., Ueda, S., Elliott, H. L., & Connell, J. M. 1999, "Insulin-mediated vasodilation and glucose uptake are functionally linked in humans", *Hypertension*, vol. 33, no. 1 Pt 2, pp. 554-558.

Clerk, L. H., Vincent, M. A., Lindner, J. R., Clark, M. G., Rattigan, S., & Barrett, E. J. 2004, "The vasodilatory actions of insulin on resistance and terminal arterioles and their impact on muscle glucose uptake", *Diabetes Metab Res.Rev.*, vol. 20, no. 1, pp. 3-12.

Cochran, A. J., Little, J. P., Tarnopolsky, M. A., & Gibala, M. J. 2010, "Carbohydrate feeding during recovery alters the skeletal muscle metabolic response to repeated sessions of high-intensity interval exercise in humans", *J.Appl.Physiol*, vol. 108, no. 3, pp. 628-636.

Coggins, M., Lindner, J., Rattigan, S., Jahn, L., Fasy, E., Kaul, S., & Barrett, E. 2001, "Physiologic hyperinsulinemia enhances human skeletal muscle perfusion by capillary recruitment", *Diabetes*, vol. 50, no. 12, pp. 2682-2690.

Colberg, S. R. & Grieco, C. R. 2009, "Exercise in the treatment and prevention of diabetes", *Curr.Sports Med.Rep.*, vol. 8, no. 4, pp. 169-175.

Colditz, G. A., Willett, W. C., Rotnitzky, A., & Manson, J. E. 1995, "Weight gain as a risk factor for clinical diabetes mellitus in women", *Ann.Intern.Med.*, vol. 122, no. 7, pp. 481-486.

Colley, R. C., Hills, A. P., O'Moore-Sullivan, T. M., Hickman, I. J., Prins, J. B., & Byrne, N. M. 2008, "Variability in adherence to an unsupervised exercise prescription in obese women", *Int.J.Obes.(Lond)*, vol. 32, no. 5, pp. 837-844.

Considine, R. V., Sinha, M. K., Heiman, M. L., Kriauciunas, A., Stephens, T. W., Nyce, M. R., Ohannesian, J. P., Marco, C. C., McKee, L. J., Bauer, T. L., & . 1996, "Serum immunoreactive-leptin concentrations in normal-weight and obese humans", *N.Engl.J.Med.*, vol. 334, no. 5, pp. 292-295.

Consolazio CF, J. R. P. L. Physiological Measurements of Metabolic Functions in Man. 1-60. 1963. McGraw-Hill, New York.

Corcoran, M. P., Lamon-Fava, S., & Fielding, R. A. 2007, "Skeletal muscle lipid deposition and insulin resistance: effect of dietary fatty acids and exercise", *Am.J.Clin.Nutr.*, vol. 85, no. 3, pp. 662-677.

Corpeleijn, E., Feskens, E. J., Jansen, E. H., Mensink, M., Saris, W. H., & Blaak, E. E. 2007, "Lifestyle intervention and adipokine levels in subjects at high risk for type 2 diabetes: the

Study on Lifestyle intervention and Impaired glucose tolerance Maastricht (SLIM)", *Diabetes Care*, vol. 30, no. 12, pp. 3125-3127.

Corpeleijn, E., Saris, W. H., & Blaak, E. E. 2009, "Metabolic flexibility in the development of insulin resistance and type 2 diabetes: effects of lifestyle", *Obes.Rev.*, vol. 10, no. 2, pp. 178-193.

Correia, M. L. & Rahmouni, K. 2006, "Role of leptin in the cardiovascular and endocrine complications of metabolic syndrome", *Diabetes Obes.Metab*, vol. 8, no. 6, pp. 603-610.

Crandall, J. P., Knowler, W. C., Kahn, S. E., Marrero, D., Florez, J. C., Bray, G. A., Haffner, S. M., Hoskin, M., & Nathan, D. M. 2008, "The prevention of type 2 diabetes", *Nat.Clin.Pract.Endocrinol.Metab*, vol. 4, no. 7, pp. 382-393.

Creinin, M. D., Keverline, S., & Meyn, L. A. 2004, "How regular is regular? An analysis of menstrual cycle regularity", *Contraception*, vol. 70, no. 4, pp. 289-292.

Dabelea, D., Hanson, R. L., Lindsay, R. S., Pettitt, D. J., Imperatore, G., Gabir, M. M., Roumain, J., Bennett, P. H., & Knowler, W. C. 2000, "Intrauterine exposure to diabetes conveys risks for type 2 diabetes and obesity: a study of discordant sibships", *Diabetes*, vol. 49, no. 12, pp. 2208-2211.

Daugaard, J. R. & Richter, E. A. 2001, "Relationship between muscle fibre composition, glucose transporter protein 4 and exercise training: possible consequences in non-insulindependent diabetes mellitus", *Acta Physiol Scand.*, vol. 171, no. 3, pp. 267-276.

Davey, S. G., Bracha, Y., Svendsen, K. H., Neaton, J. D., Haffner, S. M., & Kuller, L. H. 2005, "Incidence of type 2 diabetes in the randomized multiple risk factor intervention trial", *Ann.Intern.Med.*, vol. 142, no. 5, pp. 313-322.

DeFronzo, R. A. 2010, "Insulin resistance, lipotoxicity, type 2 diabetes and atherosclerosis: the missing links. The Claude Bernard Lecture 2009", *Diabetologia*, vol. 53, no. 7, pp. 1270-1287.

DeFronzo, R. A., Gunnarsson, R., Bjorkman, O., Olsson, M., & Wahren, J. 1985, "Effects of insulin on peripheral and splanchnic glucose metabolism in noninsulin-dependent (type II) diabetes mellitus", *J.Clin.Invest*, vol. 76, no. 1, pp. 149-155.

DeFronzo, R. A., Jacot, E., Jequier, E., Maeder, E., Wahren, J., & Felber, J. P. 1981, "The effect of insulin on the disposal of intravenous glucose. Results from indirect calorimetry and hepatic and femoral venous catheterization", *Diabetes*, vol. 30, no. 12, pp. 1000-1007.

DeFronzo, R. A., Sherwin, R. S., & Kraemer, N. 1987, "Effect of physical training on insulin action in obesity", *Diabetes*, vol. 36, no. 12, pp. 1379-1385.

DeFronzo, R. A., Simonson, D., & Ferrannini, E. 1982, "Hepatic and peripheral insulin resistance: a common feature of type 2 (non-insulin-dependent) and type 1 (insulin-dependent) diabetes mellitus", *Diabetologia*, vol. 23, no. 4, pp. 313-319.

Dekker, M. J., Lee, S., Hudson, R., Kilpatrick, K., Graham, T. E., Ross, R., & Robinson, L. E. 2007, "An exercise intervention without weight loss decreases circulating interleukin-6 in

lean and obese men with and without type 2 diabetes mellitus", *Metabolism*, vol. 56, no. 3, pp. 332-338.

Delarue, J. & Magnan, C. 2007, "Free fatty acids and insulin resistance", *Curr.Opin.Clin.Nutr.Metab Care*, vol. 10, no. 2, pp. 142-148.

Dengel, D. R., Hagberg, J. M., Coon, P. J., Drinkwater, D. T., & Goldberg, A. P. 1994, "Effects of weight loss by diet alone or combined with aerobic exercise on body composition in older obese men", *Metabolism*, vol. 43, no. 7, pp. 867-871.

DiPietro, L., Dziura, J., Yeckel, C. W., & Neufer, P. D. 2006, "Exercise and improved insulin sensitivity in older women: evidence of the enduring benefits of higher intensity training", *J.Appl.Physiol*, vol. 100, no. 1, pp. 142-149.

Dominguez, H., Storgaard, H., Rask-Madsen, C., Steffen, H. T., Ihlemann, N., Baunbjerg, N. D., Spohr, C., Kober, L., Vaag, A., & Torp-Pedersen, C. 2005, "Metabolic and vascular effects of tumor necrosis factor-alpha blockade with etanercept in obese patients with type 2 diabetes", *J.Vasc.Res.*, vol. 42, no. 6, pp. 517-525.

Donnelly, J. E., Blair, S. N., Jakicic, J. M., Manore, M. M., Rankin, J. W., & Smith, B. K. 2009, "American College of Sports Medicine Position Stand. Appropriate physical activity intervention strategies for weight loss and prevention of weight regain for adults", *Med.Sci.Sports Exerc.*, vol. 41, no. 2, pp. 459-471.

Dostalova, I., Kavalkova, P., Haluzikova, D., Housova, J., Matoulek, M., & Haluzik, M. 2009, "The use of microdialysis to characterize the endocrine production of human subcutaneous adipose tissue in vivo", *Regul.Pept.*, vol. 155, no. 1-3, pp. 156-162.

Doucet, E., St-Pierre, S., Almeras, N., Mauriege, P., Despres, J. P., Richard, D., Bouchard, C., & Tremblay, A. 2000, "Fasting insulin levels influence plasma leptin levels independently from the contribution of adiposity: evidence from both a cross-sectional and an intervention study", *J.Clin.Endocrinol.Metab*, vol. 85, no. 11, pp. 4231-4237.

Dube, J. J., Amati, F., Stefanovic-Racic, M., Toledo, F. G., Sauers, S. E., & Goodpaster, B. H. 2008, "Exercise-induced alterations in intramyocellular lipids and insulin resistance: the athlete's paradox revisited", *Am.J.Physiol Endocrinol.Metab*, vol. 294, no. 5, p. E882-E888.

Duncan, B. B., Schmidt, M. I., Pankow, J. S., Bang, H., Couper, D., Ballantyne, C. M., Hoogeveen, R. C., & Heiss, G. 2004, "Adiponectin and the development of type 2 diabetes: the atherosclerosis risk in communities study", *Diabetes*, vol. 53, no. 9, pp. 2473-2478.

Durrant, M. L., Garrow, J. S., Royston, P., Stalley, S. F., Sunkin, S., & Warwick, P. M. 1980, "Factors influencing the composition of the weight lost by obese patients on a reducing diet", *Br.J.Nutr.*, vol. 44, no. 3, pp. 275-285.

Dyck, D. J. 2009, "Adipokines as regulators of muscle metabolism and insulin sensitivity", *Appl.Physiol Nutr.Metab*, vol. 34, no. 3, pp. 396-402.

Dyson, P. A. 2008, "A review of low and reduced carbohydrate diets and weight loss in type 2 diabetes", *J.Hum.Nutr.Diet.*, vol. 21, no. 6, pp. 530-538.

Eaton, S. B. & Eaton, S. B. 2003, "An evolutionary perspective on human physical activity: implications for health", *Comp Biochem.Physiol A Mol.Integr.Physiol*, vol. 136, no. 1, pp. 153-159.

Eddy, D. M., Schlessinger, L., & Kahn, R. 2005, "Clinical outcomes and cost-effectiveness of strategies for managing people at high risk for diabetes", *Ann.Intern.Med.*, vol. 143, no. 4, pp. 251-264.

Edelstein, S. L., Knowler, W. C., Bain, R. P., Andres, R., Barrett-Connor, E. L., Dowse, G. K., Haffner, S. M., Pettitt, D. J., Sorkin, J. D., Muller, D. C., Collins, V. R., & Hamman, R. F. 1997, "Predictors of progression from impaired glucose tolerance to NIDDM: an analysis of six prospective studies", *Diabetes*, vol. 46, no. 4, pp. 701-710.

Egan, B. M. 2003, "Insulin resistance and the sympathetic nervous system", *Curr.Hypertens.Rep.*, vol. 5, no. 3, pp. 247-254.

Ehtisham, S., Hattersley, A. T., Dunger, D. B., & Barrett, T. G. 2004, "First UK survey of paediatric type 2 diabetes and MODY", *Arch.Dis.Child*, vol. 89, no. 6, pp. 526-529.

Ekelund, U., Brage, S., Griffin, S. J., & Wareham, N. J. 2009, "Objectively measured moderate- and vigorous-intensity physical activity but not sedentary time predicts insulin resistance in high-risk individuals", *Diabetes Care*, vol. 32, no. 6, pp. 1081-1086.

Ekelund, U., Griffin, S. J., & Wareham, N. J. 2007, "Physical activity and metabolic risk in individuals with a family history of type 2 diabetes", *Diabetes Care*, vol. 30, no. 2, pp. 337-342.

Elia, M., Stratton, R., & Stubbs, J. 2003, "Techniques for the study of energy balance in man", *Proc.Nutr.Soc.*, vol. 62, no. 2, pp. 529-537.

Engberding, N. & Wenger, N. K. 2008, "Cardiovascular disease prevention tailored for women", *Expert.Rev.Cardiovasc.Ther.*, vol. 6, no. 8, pp. 1123-1134.

Eriksen, L., hl-Petersen, I., Haugaard, S. B., & Dela, F. 2007, "Comparison of the effect of multiple short-duration with single long-duration exercise sessions on glucose homeostasis in type 2 diabetes mellitus", *Diabetologia*, vol. 50, no. 11, pp. 2245-2253.

Eriksson, J. G., Forsen, T., Tuomilehto, J., Jaddoe, V. W., Osmond, C., & Barker, D. J. 2002, "Effects of size at birth and childhood growth on the insulin resistance syndrome in elderly individuals", *Diabetologia*, vol. 45, no. 3, pp. 342-348.

Eriksson, K. F. & Lindgarde, F. 1991, "Prevention of type 2 (non-insulin-dependent) diabetes mellitus by diet and physical exercise. The 6-year Malmo feasibility study", *Diabetologia*, vol. 34, no. 12, pp. 891-898.

Eringa, E. C., Stehouwer, C. D., Merlijn, T., Westerhof, N., & Sipkema, P. 2002, "Physiological concentrations of insulin induce endothelin-mediated vasoconstriction during inhibition of NOS or PI3-kinase in skeletal muscle arterioles", *Cardiovasc.Res.*, vol. 56, no. 3, pp. 464-471.

Eringa, E. C., Stehouwer, C. D., Walburg, K., Clark, A. D., van Nieuw Amerongen, G. P., Westerhof, N., & Sipkema, P. 2006, "Physiological concentrations of insulin induce

endothelin-dependent vasoconstriction of skeletal muscle resistance arteries in the presence of tumor necrosis factor-alpha dependence on c-Jun N-terminal kinase", *Arterioscler.Thromb.Vasc.Biol.*, vol. 26, no. 2, pp. 274-280.

Esposito, K., Pontillo, A., Di, P. C., Giugliano, G., Masella, M., Marfella, R., & Giugliano, D. 2003, "Effect of weight loss and lifestyle changes on vascular inflammatory markers in obese women: a randomized trial", *JAMA*, vol. 289, no. 14, pp. 1799-1804.

Essig, D. A., Alderson, N. L., Ferguson, M. A., Bartoli, W. P., & Durstine, J. L. 2000, "Delayed effects of exercise on the plasma leptin concentration", *Metabolism*, vol. 49, no. 3, pp. 395-399.

Evangelista, O. & McLaughlin, M. A. 2009, "Review of cardiovascular risk factors in women", *Gend.Med.*, vol. 6 Suppl 1, pp. 17-36.

Evans, K., Clark, M. L., & Frayn, K. N. 1999, "Effects of an oral and intravenous fat load on adipose tissue and forearm lipid metabolism", *Am.J.Physiol*, vol. 276, no. 2 Pt 1, p. E241-E248.

Ezenwaka, C. E., Davis, G., & Offiah, N. V. 2001, "Evaluation of features of syndrome X in offspring of Caribbean patients with Type 2 diabetes", *Scand.J.Clin.Lab Invest*, vol. 61, no. 1, pp. 19-26.

Fain, J. N., Madan, A. K., Hiler, M. L., Cheema, P., & Bahouth, S. W. 2004, "Comparison of the release of adipokines by adipose tissue, adipose tissue matrix, and adipocytes from visceral and subcutaneous abdominal adipose tissues of obese humans", *Endocrinology*, vol. 145, no. 5, pp. 2273-2282.

Fang, X., Fetros, J., Dadson, K. E., Xu, A., & Sweeney, G. 2009, "Leptin prevents the metabolic effects of adiponectin in L6 myotubes", *Diabetologia*.

Felber, J. P., Ferrannini, E., Golay, A., Meyer, H. U., Theibaud, D., Curchod, B., Maeder, E., Jequier, E., & DeFronzo, R. A. 1987, "Role of lipid oxidation in pathogenesis of insulin resistance of obesity and type II diabetes", *Diabetes*, vol. 36, no. 11, pp. 1341-1350.

Fernandez-Castaner, M., Camps, I., Fernandez-Real, J. M., Domenech, P., & Martinez-Brotons, F. 1996, "Increased prothrombin fragment 1 + 2 and D-dimer in first-degree relatives of type 2 diabetic patients. Prethrombotic state in relatives of type 2 diabetic patients", *Acta Diabetol.*, vol. 33, no. 2, pp. 118-121.

Ferrannini, E. 1998, "Insulin resistance versus insulin deficiency in non-insulin-dependent diabetes mellitus: problems and prospects", *Endocr.Rev.*, vol. 19, no. 4, pp. 477-490.

Ferrannini, E., Haffner, S. M., Mitchell, B. D., & Stern, M. P. 1991, "Hyperinsulinaemia: the key feature of a cardiovascular and metabolic syndrome", *Diabetologia*, vol. 34, no. 6, pp. 416-422.

Ferrier, K. E., Nestel, P., Taylor, A., Drew, B. G., & Kingwell, B. A. 2004, "Diet but not aerobic exercise training reduces skeletal muscle TNF-alpha in overweight humans", *Diabetologia*, vol. 47, no. 4, pp. 630-637.

Festa, A., D'Agostino, R., Jr., Howard, G., Mykkanen, L., Tracy, R. P., & Haffner, S. M. 2000, "Chronic subclinical inflammation as part of the insulin resistance syndrome: the Insulin Resistance Atherosclerosis Study (IRAS)", *Circulation*, vol. 102, no. 1, pp. 42-47.

Finlayson, G., Bryant, E., Blundell, J. E., & King, N. A. 2009, "Acute compensatory eating following exercise is associated with implicit hedonic wanting for food", *Physiol Behav.*, vol. 97, no. 1, pp. 62-67.

Flatt, J. P., Ravussin, E., Acheson, K. J., & Jequier, E. 1985, "Effects of dietary fat on postprandial substrate oxidation and on carbohydrate and fat balances", *J. Clin. Invest*, vol. 76, no. 3, pp. 1019-1024.

Folco, E. J., Rocha, V. Z., Lopez-Ilasaca, M., & Libby, P. 2009, "Adiponectin inhibits proinflammatory signaling in human macrophages independently of interleukin-10", *J.Biol.Chem.* 

Ford, E. S. & Mokdad, A. H. 2008, "Epidemiology of obesity in the Western Hemisphere", *J.Clin.Endocrinol.Metab*, vol. 93, no. 11 Suppl 1, p. S1-S8.

Fox, S. M., III & Haskell, W. L. 1968, "Physical activity and the prevention of coronary heart disease", *Bull.N.Y.Acad.Med.*, vol. 44, no. 8, pp. 950-965.

Franco, O. H., de Laet, C., Peeters, A., Jonker, J., Mackenbach, J., & Nusselder, W. 2005, "Effects of physical activity on life expectancy with cardiovascular disease", *Arch.Intern.Med*, vol. 165, no. 20, pp. 2355-2360.

Franz, M. J., VanWormer, J. J., Crain, A. L., Boucher, J. L., Histon, T., Caplan, W., Bowman, J. D., & Pronk, N. P. 2007, "Weight-loss outcomes: a systematic review and metaanalysis of weight-loss clinical trials with a minimum 1-year follow-up", *J.Am.Diet.Assoc.*, vol. 107, no. 10, pp. 1755-1767.

Fraser, A., Harris, R., Sattar, N., Ebrahim, S., Smith, G. D., & Lawlor, D. A. 2009, "Alanine aminotransferase, gamma glutamyltransferase and incident diabetes: The British Women's Heart and Health Study and meta-analysis", *Diabetes Care*.

Frayling, T. M., Timpson, N. J., Weedon, M. N., Zeggini, E., Freathy, R. M., Lindgren, C.
M., Perry, J. R., Elliott, K. S., Lango, H., Rayner, N. W., Shields, B., Harries, L. W., Barrett,
J. C., Ellard, S., Groves, C. J., Knight, B., Patch, A. M., Ness, A. R., Ebrahim, S., Lawlor, D.
A., Ring, S. M., Ben-Shlomo, Y., Jarvelin, M. R., Sovio, U., Bennett, A. J., Melzer, D.,
Ferrucci, L., Loos, R. J., Barroso, I., Wareham, N. J., Karpe, F., Owen, K. R., Cardon, L. R.,
Walker, M., Hitman, G. A., Palmer, C. N., Doney, A. S., Morris, A. D., Smith, G. D.,
Hattersley, A. T., & McCarthy, M. I. 2007, "A common variant in the FTO gene is associated
with body mass index and predisposes to childhood and adult obesity", *Science*, vol. 316, no. 5826, pp. 889-894.

Frayn, K. N. 1983, "Calculation of substrate oxidation rates in vivo from gaseous exchange", *J.Appl.Physiol*, vol. 55, no. 2, pp. 628-634.

Frayn, K. N., Arner, P., & Yki-Jarvinen, H. 2006, "Fatty acid metabolism in adipose tissue, muscle and liver in health and disease", *Essays Biochem.*, vol. 42, pp. 89-103.

Frayn, K. N., Coppack, S. W., & Humphreys, S. M. 1993, "Subcutaneous adipose tissue metabolism studied by local catheterization", *Int.J.Obes.Relat Metab Disord.*, vol. 17 Suppl 3, p. S18-S21.

Frayn, K. N., Coppack, S. W., Humphreys, S. M., Clark, M. L., & Evans, R. D. 1993, "Periprandial regulation of lipid metabolism in insulin-treated diabetes mellitus", *Metabolism*, vol. 42, no. 4, pp. 504-510.

Frayn, K. N. & Macdonald I.A. Assessment of substrate and energy metabolism in vivo. Draznin B, Rizza R Editors. Clinical Research in Diabetes and Obesity Part I: Methods, Assessment, and Metabolic Regulation. 101-124. 1997. Totowa NJ: Humana Press.

Freathy, R. M., Bennett, A. J., Ring, S. M., Shields, B., Groves, C. J., Timpson, N. J., Weedon, M. N., Zeggini, E., Lindgren, C. M., Lango, H., Perry, J. R., Pouta, A., Ruokonen, A., Hypponen, E., Power, C., Elliott, P., Strachan, D. P., Jarvelin, M. R., Smith, G. D., McCarthy, M. I., Frayling, T. M., & Hattersley, A. T. 2009, "Type 2 diabetes risk alleles are associated with reduced size at birth", *Diabetes*, vol. 58, no. 6, pp. 1428-1433.

Freeman, D. J. 2009, "Effects of maternal obesity on fetal growth and body composition: implications for programming and future health", *Semin.Fetal Neonatal Med.* 

Friedewald, W. T., Levy, R. I., & Fredrickson, D. S. 1972, "Estimation of the concentration of low-density lipoprotein cholesterol in plasma, without use of the preparative ultracentrifuge", *Clin.Chem.*, vol. 18, no. 6, pp. 499-502.

Frimel, T. N., Sinacore, D. R., & Villareal, D. T. 2008, "Exercise attenuates the weight-lossinduced reduction in muscle mass in frail obese older adults", *Med.Sci.Sports Exerc.*, vol. 40, no. 7, pp. 1213-1219.

Froidevaux, F., Schutz, Y., Christin, L., & Jequier, E. 1993, "Energy expenditure in obese women before and during weight loss, after refeeding, and in the weight-relapse period", *Am.J.Clin.Nutr.*, vol. 57, no. 1, pp. 35-42.

Frontoni, S., Bracaglia, D., Baroni, A., Pellegrini, F., Perna, M., Cicconetti, E., Ciampittiello, G., Menzinger, G., & Gambardella, S. 2003, "Early autonomic dysfunction in glucose-tolerant but insulin-resistant offspring of type 2 diabetic patients", *Hypertension*, vol. 41, no. 6, pp. 1223-1227.

Frost C, T. S. Correcting for regression dilution bias: comparison of methods for a single predictor variable. Journal of the Royal Statistical Society: Series A 163, 173-189. 2000.

Fruhbeck, G., Jebb, S. A., & Prentice, A. M. 1998, "Leptin: physiology and pathophysiology", *Clin.Physiol*, vol. 18, no. 5, pp. 399-419.

Furler, S. M., Gan, S. K., Poynten, A. M., Chisholm, D. J., Campbell, L. V., & Kriketos, A. D. 2006, "Relationship of adiponectin with insulin sensitivity in humans, independent of lipid availability", *Obesity.(Silver.Spring)*, vol. 14, no. 2, pp. 228-234.

Galgani, J. E., Heilbronn, L. K., Azuma, K., Kelley, D. E., Albu, J. B., Pi-Sunyer, X., Smith, S. R., & Ravussin, E. 2008, "Metabolic flexibility in response to glucose is not impaired in people with type 2 diabetes after controlling for glucose disposal rate", *Diabetes*, vol. 57, no. 4, pp. 841-845.

Galgani, J. E., Moro, C., & Ravussin, E. 2008, "Metabolic Flexibility and Insulin Resistance", *Am.J.Physiol Endocrinol.Metab*.

Galic, S., Oakhill, J. S., & Steinberg, G. R. 2010, "Adipose tissue as an endocrine organ", *Mol.Cell Endocrinol.*, vol. 316, no. 2, pp. 129-139.

Gallagher, D. & Song, M. Y. 2003, "Evaluation of body composition: practical guidelines", *Prim.Care*, vol. 30, no. 2, pp. 249-265.

Gan, S. K., Kriketos, A. D., Ellis, B. A., Thompson, C. H., Kraegen, E. W., & Chisholm, D. J. 2003, "Changes in aerobic capacity and visceral fat but not myocyte lipid levels predict increased insulin action after exercise in overweight and obese men", *Diabetes Care*, vol. 26, no. 6, pp. 1706-1713.

Garg, A. 2004, "Acquired and inherited lipodystrophies", *N.Engl.J.Med.*, vol. 350, no. 12, pp. 1220-1234.

Gautier, J. F., Milner, M. R., Elam, E., Chen, K., Ravussin, E., & Pratley, R. E. 1999, "Visceral adipose tissue is not increased in Pima Indians compared with equally obese Caucasians and is not related to insulin action or secretion", *Diabetologia*, vol. 42, no. 1, pp. 28-34.

Geer, E. B. & Shen, W. 2009, "Gender differences in insulin resistance, body composition, and energy balance", *Gend.Med.*, vol. 6 Suppl 1, pp. 60-75.

Geiger, P. C., Han, D. H., Wright, D. C., & Holloszy, J. O. 2006, "How muscle insulin sensitivity is regulated: testing of a hypothesis", *Am.J.Physiol Endocrinol.Metab*, vol. 291, no. 6, p. E1258-E1263.

Geiger, P. C., Wright, D. C., Han, D. H., & Holloszy, J. O. 2005, "Activation of p38 MAP kinase enhances sensitivity of muscle glucose transport to insulin", *Am.J.Physiol Endocrinol.Metab*, vol. 288, no. 4, p. E782-E788.

Gerstein, H. C., Yusuf, S., Bosch, J., Pogue, J., Sheridan, P., Dinccag, N., Hanefeld, M., Hoogwerf, B., Laakso, M., Mohan, V., Shaw, J., Zinman, B., & Holman, R. R. 2006, "Effect of rosiglitazone on the frequency of diabetes in patients with impaired glucose tolerance or impaired fasting glucose: a randomised controlled trial", *Lancet*, vol. 368, no. 9541, pp. 1096-1105.

Giallauria, F., Orio, F., Palomba, S., Lombardi, G., Colao, A., & Vigorito, C. 2008a, "Cardiovascular risk in women with polycystic ovary syndrome", *J.Cardiovasc.Med.(Hagerstown.)*, vol. 9, no. 10, pp. 987-992.

Giallauria, F., Palomba, S., Maresca, L., Vuolo, L., Tafuri, D., Lombardi, G., Colao, A., Vigorito, C., & Francesco, O. 2008b, "Exercise training improves autonomic function and inflammatory pattern in women with polycystic ovary syndrome (PCOS)", *Clin.Endocrinol.(Oxf)*, vol. 69, no. 5, pp. 792-798.

Gil, A., Maria, A. C., Gil-Campos, M., & Canete, R. 2007, "Altered signalling and gene expression associated with the immune system and the inflammatory response in obesity", *Br.J.Nutr.*, vol. 98 Suppl 1, p. S121-S126.

Gill, J. M., Al-Mamari, A., Ferrell, W. R., Cleland, S. J., Packard, C. J., Sattar, N., Petrie, J. R., & Caslake, M. J. 2004, "Effects of prior moderate exercise on postprandial metabolism and vascular function in lean and centrally obese men", *J.Am.Coll.Cardiol.*, vol. 44, no. 12, pp. 2375-2382.

Gill, J. M., Al-Mamari, A., Ferrell, W. R., Cleland, S. J., Perry, C. G., Sattar, N., Packard, C. J., Caslake, M. J., & Petrie, J. R. 2007, "Effect of prior moderate exercise on postprandial metabolism in men with type 2 diabetes: heterogeneity of responses", *Atherosclerosis*, vol. 194, no. 1, pp. 134-143.

Gill, J. M. & Malkova, D. 2006, "Physical activity, fitness and cardiovascular disease risk in adults: interactions with insulin resistance and obesity", *Clin.Sci.(Lond)*, vol. 110, no. 4, pp. 409-425.

Gill-Randall, R., Adams, D., Ollerton, R. L., Lewis, M., & Alcolado, J. C. 2004, "Type 2 diabetes mellitus--genes or intrauterine environment? An embryo transfer paradigm in rats", *Diabetologia*, vol. 47, no. 8, pp. 1354-1359.

Giorda, C. B., Avogaro, A., Maggini, M., Lombardo, F., Mannucci, E., Turco, S., Alegiani, S. S., Raschetti, R., Velussi, M., & Ferrannini, E. 2008, "Recurrence of cardiovascular events in patients with type 2 diabetes: epidemiology and risk factors", *Diabetes Care*, vol. 31, no. 11, pp. 2154-2159.

Giugliano, G., Nicoletti, G., Grella, E., Giugliano, F., Esposito, K., Scuderi, N., & D'Andrea, F. 2004, "Effect of liposuction on insulin resistance and vascular inflammatory markers in obese women", *Br.J.Plast.Surg.*, vol. 57, no. 3, pp. 190-194.

Gogg, S., Smith, U., & Jansson, P. A. 2009, "Increased MAPK activation and impaired insulin signaling in subcutaneous microvascular endothelial cells in type 2 diabetes: the role of endothelin-1", *Diabetes*, vol. 58, no. 10, pp. 2238-2245.

Goldfine, A. B., Beckman, J. A., Betensky, R. A., Devlin, H., Hurley, S., Varo, N., Schonbeck, U., Patti, M. E., & Creager, M. A. 2006, "Family history of diabetes is a major determinant of endothelial function", *J.Am.Coll.Cardiol.*, vol. 47, no. 12, pp. 2456-2461.

Goodpaster, B. H., He, J., Watkins, S., & Kelley, D. E. 2001, "Skeletal muscle lipid content and insulin resistance: evidence for a paradox in endurance-trained athletes", *J.Clin.Endocrinol.Metab*, vol. 86, no. 12, pp. 5755-5761.

Goodpaster, B. H., Katsiaras, A., & Kelley, D. E. 2003, "Enhanced fat oxidation through physical activity is associated with improvements in insulin sensitivity in obesity", *Diabetes*, vol. 52, no. 9, pp. 2191-2197.

Goodpaster, B. H., Thaete, F. L., & Kelley, D. E. 2000, "Thigh adipose tissue distribution is associated with insulin resistance in obesity and in type 2 diabetes mellitus", *Am.J.Clin.Nutr.*, vol. 71, no. 4, pp. 885-892.

Goran, M. I. & Poehlman, E. T. 1992, "Endurance training does not enhance total energy expenditure in healthy elderly persons", *Am.J.Physiol*, vol. 263, no. 5 Pt 1, p. E950-E957.

Goulet, E. D., Melancon, M. O., ubertin-Leheudre, M., & Dionne, I. J. 2005, "Aerobic training improves insulin sensitivity 72-120 h after the last exercise session in younger but not in older women", *Eur.J.Appl.Physiol*, vol. 95, no. 2-3, pp. 146-152.

Green, K. L., Cameron, R., Polivy, J., Cooper, K., Liu, L., Leiter, L., & Heatherton, T. 1997, "Weight dissatisfaction and weight loss attempts among Canadian adults. Canadian Heart Health Surveys Research Group", *CMAJ.*, vol. 157 Suppl 1, p. S17-S25.

Grefhorst, A., Hoekstra, J., Derks, T. G., Ouwens, D. M., Baller, J. F., Havinga, R., Havekes, L. M., Romijn, J. A., & Kuipers, F. 2005, "Acute hepatic steatosis in mice by blocking betaoxidation does not reduce insulin sensitivity of very-low-density lipoprotein production", *Am.J.Physiol Gastrointest.Liver Physiol*, vol. 289, no. 3, p. G592-G598.

Groop, L. C. 1999, "Insulin resistance: the fundamental trigger of type 2 diabetes", *Diabetes Obes.Metab*, vol. 1 Suppl 1, p. S1-S7.

Gruen, M. L., Hao, M., Piston, D. W., & Hasty, A. H. 2007, "Leptin requires canonical migratory signaling pathways for induction of monocyte and macrophage chemotaxis", *Am.J.Physiol Cell Physiol*, vol. 293, no. 5, p. C1481-C1488.

Grundy, S. M. 2002, "Obesity, metabolic syndrome, and coronary atherosclerosis", *Circulation*, vol. 105, no. 23, pp. 2696-2698.

Grundy, S. M., Cleeman, J. I., Daniels, S. R., Donato, K. A., Eckel, R. H., Franklin, B. A., Gordon, D. J., Krauss, R. M., Savage, P. J., Smith, S. C., Jr., Spertus, J. A., & Costa, F. 2006, "Diagnosis and management of the metabolic syndrome: an American Heart Association/National Heart, Lung, and Blood Institute scientific statement", *Curr.Opin.Cardiol*, vol. 21, no. 1, pp. 1-6.

Guerrero-Romero, F. & Rodriguez-Moran, M. 2006, "Prevalence of dyslipidemia in nonobese prepubertal children and its association with family history of diabetes, high blood pressure, and obesity", *Arch.Med.Res.*, vol. 37, no. 8, pp. 1015-1021.

Gustafson, B. 2010, "Adipose tissue, inflammation and atherosclerosis", *J.Atheroscler.Thromb.*, vol. 17, no. 4, pp. 332-341.

Gutierrez, D. A., Puglisi, M. J., & Hasty, A. H. 2009, "Impact of increased adipose tissue mass on inflammation, insulin resistance, and dyslipidemia", *Curr.Diab.Rep.*, vol. 9, no. 1, pp. 26-32.

Haffner, S. M., Lehto, S., Ronnemaa, T., Pyorala, K., & Laakso, M. 1998, "Mortality from coronary heart disease in subjects with type 2 diabetes and in nondiabetic subjects with and without prior myocardial infarction", *N.Engl.J Med*, vol. 339, no. 4, pp. 229-234.

Hall, L. M. L., Sattar, N., & Gill, J. M. R. 2008, "Risk of metabolic and vascular disease in South Asians: potential mechanisms for increased insulin resistance", *Future Lipidology*, vol. 3, no. 4, pp. 411-424.

Haller, H. 1977, "[Epidermiology and associated risk factors of hyperlipoproteinemia]", *Z.Gesamte Inn.Med*, vol. 32, no. 8, pp. 124-128.

Han, T. S., Williams, K., Sattar, N., Hunt, K. J., Lean, M. E., & Haffner, S. M. 2002, "Analysis of obesity and hyperinsulinemia in the development of metabolic syndrome: San Antonio Heart Study", *Obes.Res.*, vol. 10, no. 9, pp. 923-931.

Hanefeld, M., Koehler, C., Henkel, E., Fuecker, K., Schaper, F., & Temelkova-Kurktschiev, T. 2000, "Post-challenge hyperglycaemia relates more strongly than fasting hyperglycaemia with carotid intima-media thickness: the RIAD Study. Risk Factors in Impaired Glucose Tolerance for Atherosclerosis and Diabetes", *Diabet.Med.*, vol. 17, no. 12, pp. 835-840.

Haram, P. M., Adams, V., Kemi, O. J., Brubakk, A. O., Hambrecht, R., Ellingsen, O., & Wisloff, U. 2006, "Time-course of endothelial adaptation following acute and regular exercise", *Eur.J.Cardiovasc.Prev.Rehabil.*, vol. 13, no. 4, pp. 585-591.

Hattersley, A. T. & Tooke, J. E. 1999, "The fetal insulin hypothesis: an alternative explanation of the association of low birthweight with diabetes and vascular disease", *Lancet*, vol. 353, no. 9166, pp. 1789-1792.

Havel, P. J. 2004, "Update on adipocyte hormones: regulation of energy balance and carbohydrate/lipid metabolism", *Diabetes*, vol. 53 Suppl 1, p. S143-S151.

Hawley, J. A. & Lessard, S. J. 2008, "Exercise training-induced improvements in insulin action", *Acta Physiol (Oxf)*, vol. 192, no. 1, pp. 127-135.

Hegele, R. A. 2003, "Monogenic forms of insulin resistance: apertures that expose the common metabolic syndrome", *Trends Endocrinol.Metab*, vol. 14, no. 8, pp. 371-377.

Hegele, R. A., Zinman, B., Hanley, A. J., Harris, S. B., Barrett, P. H., & Cao, H. 2003, "Genes, environment and Oji-Cree type 2 diabetes", *Clin.Biochem.*, vol. 36, no. 3, pp. 163-170.

Heilbronn, L. K. & Campbell, L. V. 2008, "Adipose tissue macrophages, low grade inflammation and insulin resistance in human obesity", *Curr.Pharm.Des*, vol. 14, no. 12, pp. 1225-1230.

Helmrich, S. P., Ragland, D. R., Leung, R. W., & Paffenbarger, R. S., Jr. 1991, "Physical activity and reduced occurrence of non-insulin-dependent diabetes mellitus", *N.Engl.J.Med.*, vol. 325, no. 3, pp. 147-152.

Henriksen, E. J. 2002, "Invited review: Effects of acute exercise and exercise training on insulin resistance", *J.Appl.Physiol*, vol. 93, no. 2, pp. 788-796.

Herman, W. H., Hoerger, T. J., Brandle, M., Hicks, K., Sorensen, S., Zhang, P., Hamman, R. F., Ackermann, R. T., Engelgau, M. M., & Ratner, R. E. 2005, "The cost-effectiveness of lifestyle modification or metformin in preventing type 2 diabetes in adults with impaired glucose tolerance", *Ann.Intern.Med.*, vol. 142, no. 5, pp. 323-332.

Higgins, S., Gill, J. M., Janilionyte, R., Caslake, M. J., & Malkova, D. 2005, "Physical activity, dietary intake and metabolic risk factors in non-diabetic daughters of patients with type II diabetes", *Prev.Med*, vol. 40, no. 2, pp. 145-151.

Hill, R. J. & Davies, P. S. 2001, "The validity of self-reported energy intake as determined using the doubly labelled water technique", *Br.J.Nutr.*, vol. 85, no. 4, pp. 415-430.

Holloszy, J. O. 2005, "Exercise-induced increase in muscle insulin sensitivity", *J.Appl.Physiol*, vol. 99, no. 1, pp. 338-343.

Holloszy, J. O. 2009, "Skeletal muscle "mitochondrial deficiency" does not mediate insulin resistance", *Am.J.Clin.Nutr.*, vol. 89, no. 1, pp. 463S-466S.

Hotamisligil, G. S., Arner, P., Caro, J. F., Atkinson, R. L., & Spiegelman, B. M. 1995, "Increased adipose tissue expression of tumor necrosis factor-alpha in human obesity and insulin resistance", *J Clin.Invest*, vol. 95, no. 5, pp. 2409-2415.

Hotta, K., Funahashi, T., Arita, Y., Takahashi, M., Matsuda, M., Okamoto, Y., Iwahashi, H., Kuriyama, H., Ouchi, N., Maeda, K., Nishida, M., Kihara, S., Sakai, N., Nakajima, T., Hasegawa, K., Muraguchi, M., Ohmoto, Y., Nakamura, T., Yamashita, S., Hanafusa, T., & Matsuzawa, Y. 2000, "Plasma concentrations of a novel, adipose-specific protein, adiponectin, in type 2 diabetic patients", *Arterioscler.Thromb.Vasc.Biol.*, vol. 20, no. 6, pp. 1595-1599.

Houston, D. K., Nicklas, B. J., & Zizza, C. A. 2009, "Weighty Concerns: The Growing Prevalence of Obesity among Older Adults", *J.Am.Diet.Assoc.*, vol. 109, no. 11, pp. 1886-1895.

Howlett, K. F., Sakamoto, K., Yu, H., Goodyear, L. J., & Hargreaves, M. 2006, "Insulinstimulated insulin receptor substrate-2-associated phosphatidylinositol 3-kinase activity is enhanced in human skeletal muscle after exercise", *Metabolism*, vol. 55, no. 8, pp. 1046-1052.

Hu, F. B., Sigal, R. J., Rich-Edwards, J. W., Colditz, G. A., Solomon, C. G., Willett, W. C., Speizer, F. E., & Manson, J. E. 1999, "Walking compared with vigorous physical activity and risk of type 2 diabetes in women: a prospective study", *JAMA*, vol. 282, no. 15, pp. 1433-1439.

Hu, F. B., Stampfer, M. J., Haffner, S. M., Solomon, C. G., Willett, W. C., & Manson, J. E. 2002, "Elevated risk of cardiovascular disease prior to clinical diagnosis of type 2 diabetes", *Diabetes Care*, vol. 25, no. 7, pp. 1129-1134.

Hu, F. B., Stampfer, M. J., Solomon, C. G., Liu, S., Willett, W. C., Speizer, F. E., Nathan, D. M., & Manson, J. E. 2001, "The impact of diabetes mellitus on mortality from all causes and coronary heart disease in women: 20 years of follow-up", *Arch.Intern.Med.*, vol. 161, no. 14, pp. 1717-1723.

Hu, G., Lindstrom, J., Valle, T. T., Eriksson, J. G., Jousilahti, P., Silventoinen, K., Qiao, Q., & Tuomilehto, J. 2004, "Physical activity, body mass index, and risk of type 2 diabetes in patients with normal or impaired glucose regulation", *Arch.Intern.Med.*, vol. 164, no. 8, pp. 892-896.

Huggett, R. J., Hogarth, A. J., Mackintosh, A. F., & Mary, D. A. 2006, "Sympathetic nerve hyperactivity in non-diabetic offspring of patients with type 2 diabetes mellitus", *Diabetologia*, vol. 49, no. 11, pp. 2741-2744.

Hulver, M. W., Zheng, D., Tanner, C. J., Houmard, J. A., Kraus, W. E., Slentz, C. A., Sinha, M. K., Pories, W. J., MacDonald, K. G., & Dohm, G. L. 2002, "Adiponectin is not altered

with exercise training despite enhanced insulin action", *Am.J.Physiol Endocrinol.Metab*, vol. 283, no. 4, p. E861-E865.

Humphreys, P., McCarthy, M., Tuomilehto, J., Tuomilehto-Wolf, E., Stratton, I., Morgan, R., Rees, A., Owens, D., Stengard, J., Nissinen, A., & . 1994, "Chromosome 4q locus associated with insulin resistance in Pima Indians. Studies in three European NIDDM populations", *Diabetes*, vol. 43, no. 6, pp. 800-804.

Humphriss, D. B., Stewart, M. W., Berrish, T. S., Barriocanal, L. A., Trajano, L. R., Ashworth, L. A., Brown, M. D., Miller, M., Avery, P. J., Alberti, K. G., & Walker, M. 1997, "Multiple metabolic abnormalities in normal glucose tolerant relatives of NIDDM families", *Diabetologia*, vol. 40, no. 10, pp. 1185-1190.

Hutchinson, D. S., Summers, R. J., & Bengtsson, T. 2008, "Regulation of AMP-activated protein kinase activity by G-protein coupled receptors: potential utility in treatment of diabetes and heart disease", *Pharmacol.Ther.*, vol. 119, no. 3, pp. 291-310.

Imrie, H., Abbas, A., & Kearney, M. 2010, "Insulin resistance, lipotoxicity and endothelial dysfunction", *Biochim.Biophys.Acta*, vol. 1801, no. 3, pp. 320-326.

International Diabetes Federation. The IDF consensus worldwide definition of the metabolic syndrome [article online]. 2005. Ref Type: Generic

Itani, S. I., Ruderman, N. B., Schmieder, F., & Boden, G. 2002, "Lipid-induced insulin resistance in human muscle is associated with changes in diacylglycerol, protein kinase C, and IkappaB-alpha", *Diabetes*, vol. 51, no. 7, pp. 2005-2011.

Iwasaki, Y., Kambayashi, M., Asai, M., Yoshida, M., Nigawara, T., & Hashimoto, K. 2007, "High glucose alone, as well as in combination with proinflammatory cytokines, stimulates nuclear factor kappa-B-mediated transcription in hepatocytes in vitro", *J Diabetes Complications*, vol. 21, no. 1, pp. 56-62.

Jaap, A. J., Shore, A. C., & Tooke, J. E. 1997, "Relationship of insulin resistance to microvascular dysfunction in subjects with fasting hyperglycaemia", *Diabetologia*, vol. 40, no. 2, pp. 238-243.

Jackson, K. G., Wolstencroft, E. J., Bateman, P. A., Yaqoob, P., & Williams, C. M. 2005, "Acute effects of meal fatty acids on postprandial NEFA, glucose and apo E response: implications for insulin sensitivity and lipoprotein regulation?", *Br.J.Nutr.*, vol. 93, no. 5, pp. 693-700.

Jadhav, U. M. & Kadam, N. N. 2005, "Non-invasive assessment of arterial stiffness by pulsewave velocity correlates with endothelial dysfunction", *Indian Heart J.*, vol. 57, no. 3, pp. 226-232.

Jensen, M. D. 2008, "Role of body fat distribution and the metabolic complications of obesity", *J.Clin.Endocrinol.Metab*, vol. 93, no. 11 Suppl 1, p. S57-S63.

Jensen, T. E., Wojtaszewski, J. F., & Richter, E. A. 2009, "AMP-activated protein kinase in contraction regulation of skeletal muscle metabolism: necessary and/or sufficient?", *Acta Physiol (Oxf)*, vol. 196, no. 1, pp. 155-174.

Jeppesen, J., Hansen, T. W., Olsen, M. H., Rasmussen, S., Ibsen, H., Torp-Pedersen, C., Hildebrandt, P. R., & Madsbad, S. 2008, "C-reactive protein, insulin resistance and risk of cardiovascular disease: a population-based study", *Eur.J.Cardiovasc.Prev.Rehabil.* 

Jessani, S., Millane, T., & Lip, G. Y. 2009, "Vascular damage in impaired glucose tolerance: an unappreciated phenomenon?", *Curr.Pharm.Des*, vol. 15, no. 29, pp. 3417-3432.

Jia, W. P., Lu, J. X., Xiang, K. S., Bao, Y. Q., Lu, H. J., & Chen, L. 2003, "Prediction of abdominal visceral obesity from body mass index, waist circumference and waist-hip ratio in Chinese adults: receiver operating characteristic curves analysis", *Biomed.Environ.Sci.*, vol. 16, no. 3, pp. 206-211.

Jiang, L., Liu, C., Wang, W. Q., Ye, L., Zhu, N., Zhou, W. W., Su, T. W., Li, X. Y., & Ning, G. 2006, "[Leprechaunism: an inherited insulin resistance syndrome caused by the defect of insulin receptor]", *Zhonghua Nei Ke.Za Zhi.*, vol. 45, no. 9, pp. 730-733.

Johanson, E. H., Jansson, P. A., Lonn, L., Matsuzawa, Y., Funahashi, T., Taskinen, M. R., Smith, U., & Axelsen, M. 2003, "Fat distribution, lipid accumulation in the liver, and exercise capacity do not explain the insulin resistance in healthy males with a family history for type 2 diabetes", *J.Clin.Endocrinol.Metab*, vol. 88, no. 9, pp. 4232-4238.

Johansson, P., Ostenson, C. G., Hilding, A. M., Andersson, C., Rehnberg, C., & Tillgren, P. 2009, "A cost-effectiveness analysis of a community-based diabetes prevention program in Sweden", *Int.J.Technol.Assess.Health Care*, vol. 25, no. 3, pp. 350-358.

Johnson, N. A., Sachinwalla, T., Walton, D. W., Smith, K., Armstrong, A., Thompson, M. W., & George, J. 2009, "Aerobic exercise training reduces hepatic and visceral lipids in obese individuals without weight loss", *Hepatology*.

Jurca, R., Jackson, A. S., LaMonte, M. J., Morrow, J. R., Jr., Blair, S. N., Wareham, N. J., Haskell, W. L., van, M. W., Church, T. S., Jakicic, J. M., & Laukkanen, R. 2005, "Assessing cardiorespiratory fitness without performing exercise testing", *Am.J.Prev.Med.*, vol. 29, no. 3, pp. 185-193.

Kadoglou, N. P., Iliadis, F., Angelopoulou, N., Perrea, D., Ampatzidis, G., Liapis, C. D., & Alevizos, M. 2007, "The anti-inflammatory effects of exercise training in patients with type 2 diabetes mellitus", *Eur.J.Cardiovasc.Prev.Rehabil.*, vol. 14, no. 6, pp. 837-843.

Kahn, B. B., Alquier, T., Carling, D., & Hardie, D. G. 2005, "AMP-activated protein kinase: ancient energy gauge provides clues to modern understanding of metabolism", *Cell Metab*, vol. 1, no. 1, pp. 15-25.

Kahn, S. E. 2003, "The relative contributions of insulin resistance and beta-cell dysfunction to the pathophysiology of Type 2 diabetes", *Diabetologia*, vol. 46, no. 1, pp. 3-19.

Kahn, S. E., Haffner, S. M., Heise, M. A., Herman, W. H., Holman, R. R., Jones, N. P., Kravitz, B. G., Lachin, J. M., O'Neill, M. C., Zinman, B., & Viberti, G. 2006, "Glycemic durability of rosiglitazone, metformin, or glyburide monotherapy", *N.Engl.J.Med.*, vol. 355, no. 23, pp. 2427-2443.

Kannel, W. B. & McGee, D. L. 1979, "Diabetes and glucose tolerance as risk factors for cardiovascular disease: the Framingham study", *Diabetes Care*, vol. 2, no. 2, pp. 120-126.

Karpe, F. 1997, "Mechanisms of postprandial hyperlipidaemia--remnants and coronary artery disease", *Diabet.Med.*, vol. 14 Suppl 3, p. S60-S66.

Kelley, D. E., Goodpaster, B., Wing, R. R., & Simoneau, J. A. 1999, "Skeletal muscle fatty acid metabolism in association with insulin resistance, obesity, and weight loss", *Am.J.Physiol*, vol. 277, no. 6 Pt 1, p. E1130-E1141.

Kelley, D. E. & Mandarino, L. J. 2000, "Fuel selection in human skeletal muscle in insulin resistance: a reexamination", *Diabetes*, vol. 49, no. 5, pp. 677-683.

Kelley, D. E., Thaete, F. L., Troost, F., Huwe, T., & Goodpaster, B. H. 2000, "Subdivisions of subcutaneous abdominal adipose tissue and insulin resistance", *Am.J.Physiol Endocrinol.Metab*, vol. 278, no. 5, p. E941-E948.

Kempen, K. P., Saris, W. H., & Westerterp, K. R. 1995, "Energy balance during an 8-wk energy-restricted diet with and without exercise in obese women", *Am.J.Clin.Nutr.*, vol. 62, no. 4, pp. 722-729.

Kieffer, T. J. & Habener, J. F. 2000, "The adipoinsular axis: effects of leptin on pancreatic beta-cells", *Am.J Physiol Endocrinol.Metab*, vol. 278, no. 1, p. E1-E14.

Kilhovd, B. K., Juutilainen, A., Lehto, S., Ronnemaa, T., Torjesen, P. A., Birkeland, K. I., Berg, T. J., Hanssen, K. F., & Laakso, M. 2005, "High serum levels of advanced glycation end products predict increased coronary heart disease mortality in nondiabetic women but not in nondiabetic men: a population-based 18-year follow-up study", *Arterioscler.Thromb.Vasc.Biol.*, vol. 25, no. 4, pp. 815-820.

Kim, E. S., Im, J. A., Kim, K. C., Park, J. H., Suh, S. H., Kang, E. S., Kim, S. H., Jekal, Y., Lee, C. W., Yoon, Y. J., Lee, H. C., & Jeon, J. Y. 2007, "Improved insulin sensitivity and adiponectin level after exercise training in obese Korean youth", *Obesity.(Silver.Spring)*, vol. 15, no. 12, pp. 3023-3030.

Kim, H. S., Abbasi, F., Lamendola, C., McLaughlin, T., & Reaven, G. M. 2001a, "Effect of insulin resistance on postprandial elevations of remnant lipoprotein concentrations in postmenopausal women", *Am.J.Clin.Nutr.*, vol. 74, no. 5, pp. 592-595.

Kim, J. K., Kim, Y. J., Fillmore, J. J., Chen, Y., Moore, I., Lee, J., Yuan, M., Li, Z. W., Karin, M., Perret, P., Shoelson, S. E., & Shulman, G. I. 2001b, "Prevention of fat-induced insulin resistance by salicylate", *J.Clin.Invest*, vol. 108, no. 3, pp. 437-446.

King, H., Aubert, R. E., & Herman, W. H. 1998, "Global burden of diabetes, 1995-2025 - Prevalence, numerical estimates, and projections", *Diabetes Care*, vol. 21, no. 9, pp. 1414-1431.

King, N. A., Caudwell, P., Hopkins, M., Byrne, N. M., Colley, R., Hills, A. P., Stubbs, J. R., & Blundell, J. E. 2007, "Metabolic and behavioral compensatory responses to exercise interventions: barriers to weight loss", *Obesity*.(*Silver.Spring*), vol. 15, no. 6, pp. 1373-1383.

King, N. A., Hopkins, M., Caudwell, P., Stubbs, R. J., & Blundell, J. E. 2008, "Individual variability following 12 weeks of supervised exercise: identification and characterization of compensation for exercise-induced weight loss", *Int.J.Obes.(Lond)*, vol. 32, no. 1, pp. 177-184.

Klein, P. D., James, W. P., Wong, W. W., Irving, C. S., Murgatroyd, P. R., Cabrera, M., Dallosso, H. M., Klein, E. R., & Nichols, B. L. 1984, "Calorimetric validation of the doubly-labelled water method for determination of energy expenditure in man", *Hum.Nutr.Clin.Nutr.*, vol. 38, no. 2, pp. 95-106.

Klein, S., Fontana, L., Young, V. L., Coggan, A. R., Kilo, C., Patterson, B. W., & Mohammed, B. S. 2004, "Absence of an effect of liposuction on insulin action and risk factors for coronary heart disease", *N.Engl.J.Med.*, vol. 350, no. 25, pp. 2549-2557.

Klesges, R. C., Eck, L. H., Mellon, M. W., Fulliton, W., Somes, G. W., & Hanson, C. L. 1990, "The accuracy of self-reports of physical activity", *Med.Sci.Sports Exerc.*, vol. 22, no. 5, pp. 690-697.

Klimcakova, E., Polak, J., Moro, C., Hejnova, J., Majercik, M., Viguerie, N., Berlan, M., Langin, D., & Stich, V. 2006, "Dynamic strength training improves insulin sensitivity without altering plasma levels and gene expression of adipokines in subcutaneous adipose tissue in obese men", *J Clin.Endocrinol.Metab*.

Knowler, W. C., Barrett-Connor, E., Fowler, S. E., Hamman, R. F., Lachin, J. M., Walker, E. A., & Nathan, D. M. 2002, "Reduction in the incidence of type 2 diabetes with lifestyle intervention or metformin", *N.Engl.J Med*, vol. 346, no. 6, pp. 393-403.

Kobberling J & Tillil H. Empirical risk figures for first degree relatives of non-insulindependent diabetics. Kobberling J, Tattersall R eds. The genetics of diabetes mellitus. 201-209. 1982. Academic Press, London.

Kondo, T., Kobayashi, I., & Murakami, M. 2006, "Effect of exercise on circulating adipokine levels in obese young women", *Endocr.J.*, vol. 53, no. 2, pp. 189-195.

Koopman, R., Manders, R. J., Zorenc, A. H., Hul, G. B., Kuipers, H., Keizer, H. A., & van Loon, L. J. 2005, "A single session of resistance exercise enhances insulin sensitivity for at least 24 h in healthy men", *Eur.J.Appl.Physiol*, vol. 94, no. 1-2, pp. 180-187.

Koska, J., Stefan, N., Permana, P. A., Weyer, C., Sonoda, M., Bogardus, C., Smith, S. R., Joanisse, D. R., Funahashi, T., Krakoff, J., & Bunt, J. C. 2008a, "Increased fat accumulation in liver may link insulin resistance with subcutaneous abdominal adipocyte enlargement, visceral adiposity, and hypoadiponectinemia in obese individuals", *Am.J.Clin.Nutr.*, vol. 87, no. 2, pp. 295-302.

Koska, J., Stefan, N., Votruba, S. B., Smith, S. R., Krakoff, J., & Bunt, J. C. 2008b, "Distribution of subcutaneous fat predicts insulin action in obesity in sex-specific manner", *Obesity.(Silver.Spring)*, vol. 16, no. 9, pp. 2003-2009.

Kotronen, A., Vehkavaara, S., Seppala-Lindroos, A., Bergholm, R., & Yki-Jarvinen, H. 2007, "Effect of liver fat on insulin clearance", *Am.J.Physiol Endocrinol.Metab*, vol. 293, no. 6, p. E1709-E1715.

Koves, T. R., Ussher, J. R., Noland, R. C., Slentz, D., Mosedale, M., Ilkayeva, O., Bain, J., Stevens, R., Dyck, J. R., Newgard, C. B., Lopaschuk, G. D., & Muoio, D. M. 2008, "Mitochondrial overload and incomplete fatty acid oxidation contribute to skeletal muscle insulin resistance", *Cell Metab*, vol. 7, no. 1, pp. 45-56.

Kowalska, I., Straczkowski, M., Nikolajuk, A., Adamska, A., Karczewska-Kupczewska, M., Otziomek, E., Kinalska, I., & Gorska, M. 2008, "Insulin resistance, serum adiponectin, and proinflammatory markers in young subjects with the metabolic syndrome", *Metabolism*, vol. 57, no. 11, pp. 1539-1544.

Krassas, G. E., Tzotzas, T., Tsametis, C., & Konstantinidis, T. 2001, "Prevalence and trends in overweight and obesity among children and adolescents in Thessaloniki, Greece", *J.Pediatr.Endocrinol.Metab*, vol. 14 Suppl 5, pp. 1319-1326.

Kriketos, A. D., Greenfield, J. R., Peake, P. W., Furler, S. M., Denyer, G. S., Charlesworth, J. A., & Campbell, L. V. 2004, "Inflammation, insulin resistance, and adiposity: a study of first-degree relatives of type 2 diabetic subjects", *Diabetes Care*, vol. 27, no. 8, pp. 2033-2040.

Krssak, M., Falk, P. K., Dresner, A., DiPietro, L., Vogel, S. M., Rothman, D. L., Roden, M., & Shulman, G. I. 1999, "Intramyocellular lipid concentrations are correlated with insulin sensitivity in humans: a 1H NMR spectroscopy study", *Diabetologia*, vol. 42, no. 1, pp. 113-116.

Laakso, M., Edelman, S. V., Brechtel, G., & Baron, A. D. 1990, "Decreased effect of insulin to stimulate skeletal muscle blood flow in obese man. A novel mechanism for insulin resistance", *J.Clin.Invest*, vol. 85, no. 6, pp. 1844-1852.

Laaksonen, D. E., Lakka, H. M., Lynch, J., Lakka, T. A., Niskanen, L., Rauramaa, R., Salonen, J. T., & Kauhanen, J. 2003, "Cardiorespiratory fitness and vigorous leisure-time physical activity modify the association of small size at birth with the metabolic syndrome", *Diabetes Care*, vol. 26, no. 7, pp. 2156-2164.

Lago, F., Dieguez, C., Gomez-Reino, J., & Gualillo, O. 2007, "The emerging role of adipokines as mediators of inflammation and immune responses", *Cytokine Growth Factor Rev.*, vol. 18, no. 3-4, pp. 313-325.

Lakka, H. M., Laaksonen, D. E., Lakka, T. A., Niskanen, L. K., Kumpusalo, E., Tuomilehto, J., & Salonen, J. T. 2002, "The metabolic syndrome and total and cardiovascular disease mortality in middle-aged men", *JAMA*, vol. 288, no. 21, pp. 2709-2716.

Lakoski, S. G., Cushman, M., Criqui, M., Rundek, T., Blumenthal, R. S., D'Agostino, R. B., Jr., & Herrington, D. M. 2006, "Gender and C-reactive protein: data from the Multiethnic Study of Atherosclerosis (MESA) cohort", *Am.Heart J.*, vol. 152, no. 3, pp. 593-598.

Larcher, F., Del, R. M., Serrano, F., Segovia, J. C., Ramirez, A., Meana, A., Page, A., Abad, J. L., Gonzalez, M. A., Bueren, J., Bernad, A., & Jorcano, J. L. 2001, "A cutaneous gene therapy approach to human leptin deficiencies: correction of the murine ob/ob phenotype using leptin-targeted keratinocyte grafts", *FASEB J.*, vol. 15, no. 9, pp. 1529-1538.

Lattuada, G., Costantino, F., Caumo, A., Scifo, P., Ragogna, F., De, C. F., Del, M. A., Luzi, L., & Perseghin, G. 2005, "Reduced whole-body lipid oxidation is associated with insulin resistance, but not with intramyocellular lipid content in offspring of type 2 diabetic patients", *Diabetologia*, vol. 48, no. 4, pp. 741-747.

Laws, A. & Reaven, G. M. 1993, "Insulin resistance and risk factors for coronary heart disease", *Baillieres Clin.Endocrinol.Metab*, vol. 7, no. 4, pp. 1063-1078.
Lear, S. A., Humphries, K. H., Kohli, S., Chockalingam, A., Frohlich, J. J., & Birmingham, C. L. 2007, "Visceral adipose tissue accumulation differs according to ethnic background: results of the Multicultural Community Health Assessment Trial (M-CHAT)", *Am.J.Clin.Nutr.*, vol. 86, no. 2, pp. 353-359.

Lebovitz, H. E. 2006, "Insulin resistance--a common link between type 2 diabetes and cardiovascular disease", *Diabetes Obes.Metab*, vol. 8, no. 3, pp. 237-249.

Lee, Y., Yu, X., Gonzales, F., Mangelsdorf, D. J., Wang, M. Y., Richardson, C., Witters, L. A., & Unger, R. H. 2002, "PPAR alpha is necessary for the lipopenic action of hyperleptinemia on white adipose and liver tissue", *Proc.Natl.Acad.Sci.U.S.A*, vol. 99, no. 18, pp. 11848-11853.

Leonard, D., McDermott, R., Odea, K., Rowley, K. G., Pensio, P., Sambo, E., Twist, A., Toolis, R., Lowson, S., & Best, J. D. 2002, "Obesity, diabetes and associated cardiovascular risk factors among Torres Strait Islander people", *Aust.N.Z.J.Public Health*, vol. 26, no. 2, pp. 144-149.

Leonard, W. R. 2003, "Measuring human energy expenditure: what have we learned from the flex-heart rate method?", *Am.J.Hum.Biol.*, vol. 15, no. 4, pp. 479-489.

Levine, J. A., Eberhardt, N. L., & Jensen, M. D. 1999, "Role of nonexercise activity thermogenesis in resistance to fat gain in humans", *Science*, vol. 283, no. 5399, pp. 212-214.

Liese, A. D., Mayer-Davis, E. J., Tyroler, H. A., Davis, C. E., Keil, U., Duncan, B. B., & Heiss, G. 1997, "Development of the multiple metabolic syndrome in the ARIC cohort: joint contribution of insulin, BMI, and WHR. Atherosclerosis risk in communities", *Ann.Epidemiol.*, vol. 7, no. 6, pp. 407-416.

Lihn, A. S., Ostergard, T., Nyholm, B., Pedersen, S. B., Richelsen, B., & Schmitz, O. 2003, "Adiponectin expression in adipose tissue is reduced in first-degree relatives of type 2 diabetic patients", *Am.J.Physiol Endocrinol.Metab*, vol. 284, no. 2, p. E443-E448.

Lillioja, S., Foley, J., Bogardus, C., Mott, D., & Howard, B. V. 1986, "Free fatty acid metabolism and obesity in man: in vivo in vitro comparisons", *Metabolism*, vol. 35, no. 6, pp. 505-514.

Lindsay, R. S. 2008, "Commentary: Type 2 diabetes and birth weight--genetic and environmental effects", *Int.J.Epidemiol.*, vol. 37, no. 1, pp. 192-193.

Lindsay, R. S., Dabelea, D., Roumain, J., Hanson, R. L., Bennett, P. H., & Knowler, W. C. 2000, "Type 2 diabetes and low birth weight: the role of paternal inheritance in the association of low birth weight and diabetes", *Diabetes*, vol. 49, no. 3, pp. 445-449.

Liu, T. C., Liu, Y. Y., Lee, S. D., Huang, C. Y., Chien, K. Y., Cheng, I. S., Lin, C. Y., & Kuo, C. H. 2008, "Effects of short-term detraining on measures of obesity and glucose tolerance in elite athletes", *J.Sports Sci.*, vol. 26, no. 9, pp. 919-925.

Livingston, E. H. 2006, "Lower body subcutaneous fat accumulation and diabetes mellitus risk", *Surg.Obes.Relat Dis.*, vol. 2, no. 3, pp. 362-368.

Livingstone, M. B., Prentice, A. M., Coward, W. A., Ceesay, S. M., Strain, J. J., McKenna, P. G., Nevin, G. B., Barker, M. E., & Hickey, R. J. 1990, "Simultaneous measurement of freeliving energy expenditure by the doubly labeled water method and heart-rate monitoring", *Am.J.Clin.Nutr.*, vol. 52, no. 1, pp. 59-65.

Longstreet, D. A., Heath, D. L., Panaretto, K. S., & Vink, R. 2007, "Correlations suggest low magnesium may lead to higher rates of type 2 diabetes in Indigenous Australians", *Rural.Remote.Health*, vol. 7, no. 4, p. 843.

Lopez, A. D. & Murray, C. C. 1998, "The global burden of disease, 1990-2020", *Nat.Med*, vol. 4, no. 11, pp. 1241-1243.

Lorenzo, C., Williams, K., Stern, M. P., & Haffner, S. M. 2009, "Height, ethnicity, and the incidence of diabetes: the San Antonio Heart Study", *Metabolism*.

Luscombe, N. D., Clifton, P. M., Noakes, M., Farnsworth, E., & Wittert, G. 2003, "Effect of a high-protein, energy-restricted diet on weight loss and energy expenditure after weight stabilization in hyperinsulinemic subjects", *Int.J.Obes.Relat Metab Disord.*, vol. 27, no. 5, pp. 582-590.

Lynch, J., Helmrich, S. P., Lakka, T. A., Kaplan, G. A., Cohen, R. D., Salonen, R., & Salonen, J. T. 1996, "Moderately intense physical activities and high levels of cardiorespiratory fitness reduce the risk of non-insulin-dependent diabetes mellitus in middle-aged men", *Arch.Intern.Med.*, vol. 156, no. 12, pp. 1307-1314.

Ma, L., Hanson, R. L., Que, L. N., Guo, Y., Kobes, S., Bogardus, C., & Baier, L. J. 2008, "PCLO variants are nominally associated with early-onset type 2 diabetes and insulin resistance in Pima Indians", *Diabetes*, vol. 57, no. 11, pp. 3156-3160.

Macfarlane, D. J. 2001, "Automated metabolic gas analysis systems: a review", *Sports Med.*, vol. 31, no. 12, pp. 841-861.

Macfarlane, D. J. & Thomas, N. 2009, "Exercise and diet in weight management: updating what works", *Br.J.Sports Med.* 

Machann, J., Stefan, N., & Schick, F. 2008, "(1)H MR spectroscopy of skeletal muscle, liver and bone marrow", *Eur.J.Radiol.*, vol. 67, no. 2, pp. 275-284.

Macor, C., Ruggeri, A., Mazzonetto, P., Federspil, G., Cobelli, C., & Vettor, R. 1997, "Visceral adipose tissue impairs insulin secretion and insulin sensitivity but not energy expenditure in obesity", *Metabolism*, vol. 46, no. 2, pp. 123-129.

Magkos, F., Patterson, B. W., Mohammed, B. S., & Mittendorfer, B. 2007, "A single 1-h bout of evening exercise increases basal FFA flux without affecting VLDL-triglyceride and VLDL-apolipoprotein B-100 kinetics in untrained lean men", *Am.J.Physiol Endocrinol.Metab*, vol. 292, no. 6, p. E1568-E1574.

Magkos, F., Tsekouras, Y., Kavouras, S. A., Mittendorfer, B., & Sidossis, L. S. 2008, "Improved insulin sensitivity after a single bout of exercise is curvilinearly related to exercise energy expenditure", *Clin.Sci.(Lond)*, vol. 114, no. 1, pp. 59-64. Magni, P., Liuzzi, A., Ruscica, M., Dozio, E., Ferrario, S., Bussi, I., Minocci, A., Castagna, A., Motta, M., & Savia, G. 2005, "Free and bound plasma leptin in normal weight and obese men and women: relationship with body composition, resting energy expenditure, insulin-sensitivity, lipid profile and macronutrient preference", *Clin.Endocrinol.(Oxf)*, vol. 62, no. 2, pp. 189-196.

Major, G. C., Doucet, E., Trayhurn, P., Astrup, A., & Tremblay, A. 2007, "Clinical significance of adaptive thermogenesis", *Int.J.Obes.(Lond)*, vol. 31, no. 2, pp. 204-212.

Mak, K. H. & Haffner, S. M. 2003, "Diabetes abolishes the gender gap in coronary heart disease", *Eur.Heart J*, vol. 24, no. 15, pp. 1385-1386.

Malecki, M. T. 2005, "Genetics of type 2 diabetes mellitus", *Diabetes Res. Clin. Pract.*, vol. 68 Suppl1, p. S10-S21.

Malik, S., Wong, N. D., Franklin, S. S., Kamath, T. V., L'Italien, G. J., Pio, J. R., & Williams, G. R. 2004, "Impact of the metabolic syndrome on mortality from coronary heart disease, cardiovascular disease, and all causes in United States adults", *Circulation*, vol. 110, no. 10, pp. 1245-1250.

Maltezos, E., Papazoglou, D., Exiara, T., Papazoglou, L., Karathanasis, E., Christakidis, D., & Ktenidou-Kartali, S. 2002, "Tumour necrosis factor-alpha levels in non-diabetic offspring of patients with type 2 diabetes mellitus", *J.Int.Med.Res.*, vol. 30, no. 6, pp. 576-583.

Mansfield, M. W., Stickland, M. H., & Grant, P. J. 1997, "PAI-1 concentrations in first-degree relatives of patients with non-insulin-dependent diabetes: metabolic and genetic associations", *Thromb.Haemost.*, vol. 77, no. 2, pp. 357-361.

Manson, J. E., Nathan, D. M., Krolewski, A. S., Stampfer, M. J., Willett, W. C., & Hennekens, C. H. 1992, "A prospective study of exercise and incidence of diabetes among US male physicians", *JAMA*, vol. 268, no. 1, pp. 63-67.

Maraldi, N. M., Capanni, C., Mattioli, E., Columbaro, M., Squarzoni, S., Parnaik, W. K., Wehnert, M., & Lattanzi, G. 2007, "A pathogenic mechanism leading to partial lipodistrophy and prospects for pharmacological treatment of insulin resistance syndrome", *Acta Biomed.*, vol. 78 Suppl 1, pp. 207-215.

Marcell, T. J., McAuley, K. A., Traustadottir, T., & Reaven, P. D. 2005, "Exercise training is not associated with improved levels of C-reactive protein or adiponectin", *Metabolism*, vol. 54, no. 4, pp. 533-541.

Marfell-Jones M, Olds T, Stewart A, & Carter L. International Standards for anthropometric assessment. 2006. ISAK.

Margetic, S., Gazzola, C., Pegg, G. G., & Hill, R. A. 2002, "Leptin: a review of its peripheral actions and interactions", *Int.J Obes.Relat Metab Disord.*, vol. 26, no. 11, pp. 1407-1433.

Mari, A., Pacini, G., Murphy, E., Ludvik, B., & Nolan, J. J. 2001, "A model-based method for assessing insulin sensitivity from the oral glucose tolerance test", *Diabetes Care*, vol. 24, no. 3, pp. 539-548.

Marra, M., Scalfi, L., Covino, A., Esposito-Del, P. A., & Contaldo, F. 1998, "Fasting respiratory quotient as a predictor of weight changes in non-obese women", *Int.J.Obes.Relat Metab Disord.*, vol. 22, no. 6, pp. 601-603.

Martinez-Garcia, M. A., Luque-Ramirez, M., San-Millan, J. L., & Escobar-Morreale, H. F. 2009, "Body iron stores and glucose intolerance in premenopausal women: role of hyperandrogenism, insulin resistance, and genomic variants related to inflammation, oxidative stress, and iron metabolism", *Diabetes Care*, vol. 32, no. 8, pp. 1525-1530.

Maskarinec, G., Grandinetti, A., Matsuura, G., Sharma, S., Mau, M., Henderson, B. E., & Kolonel, L. N. 2009, "Diabetes prevalence and body mass index differ by ethnicity: the Multiethnic Cohort", *Ethn.Dis.*, vol. 19, no. 1, pp. 49-55.

Matsuda, M. & DeFronzo, R. A. 1999, "Insulin sensitivity indices obtained from oral glucose tolerance testing: comparison with the euglycemic insulin clamp", *Diabetes Care*, vol. 22, no. 9, pp. 1462-1470.

Matthews, D. R., Hosker, J. P., Rudenski, A. S., Naylor, B. A., Treacher, D. F., & Turner, R. C. 1985, "Homeostasis model assessment: insulin resistance and beta-cell function from fasting plasma glucose and insulin concentrations in man", *Diabetologia*, vol. 28, no. 7, pp. 412-419.

McAinch, A. J., Steinberg, G. R., Mollica, J., O'Brien, P. E., Dixon, J. B., Macaulay, S. L., Kemp, B. E., & Cameron-Smith, D. 2006, "Differential regulation of adiponectin receptor gene expression by adiponectin and leptin in myotubes derived from obese and diabetic individuals", *Obesity.(Silver.Spring)*, vol. 14, no. 11, pp. 1898-1904.

McBean, A. M., Li, S., Gilbertson, D. T., & Collins, A. J. 2004, "Differences in diabetes prevalence, incidence, and mortality among the elderly of four racial/ethnic groups: whites, blacks, hispanics, and asians", *Diabetes Care*, vol. 27, no. 10, pp. 2317-2324.

McBride, A. & Hardie, D. G. 2009, "AMP-activated protein kinase--a sensor of glycogen as well as AMP and ATP?", *Acta Physiol (Oxf)*, vol. 196, no. 1, pp. 99-113.

McCance, D. R., Pettitt, D. J., Hanson, R. L., Jacobsson, L. T., Knowler, W. C., & Bennett, P. H. 1994, "Birth weight and non-insulin dependent diabetes: thrifty genotype, thrifty phenotype, or surviving small baby genotype?", *BMJ*, vol. 308, no. 6934, pp. 942-945.

McEleavy, O. D., McCallum, R. W., Petrie, J. R., Small, M., Connell, J. M., Sattar, N., & Cleland, S. J. 2004, "Higher carotid-radial pulse wave velocity in healthy offspring of patients with Type 2 diabetes", *Diabet.Med*, vol. 21, no. 3, pp. 262-266.

McGarry, J. D. 2002, "Banting lecture 2001: dysregulation of fatty acid metabolism in the etiology of type 2 diabetes", *Diabetes*, vol. 51, no. 1, pp. 7-18.

McIntyre, E. A. & Walker, M. 2002, "Genetics of type 2 diabetes and insulin resistance: knowledge from human studies", *Clin.Endocrinol.(Oxf)*, vol. 57, no. 3, pp. 303-311.

McMurray, R. G. & Hackney, A. C. 2005, "Interactions of metabolic hormones, adipose tissue and exercise", *Sports Med.*, vol. 35, no. 5, pp. 393-412.

McNeill, A. M., Rosamond, W. D., Girman, C. J., Golden, S. H., Schmidt, M. I., East, H. E., Ballantyne, C. M., & Heiss, G. 2005, "The metabolic syndrome and 11-year risk of incident cardiovascular disease in the atherosclerosis risk in communities study", *Diabetes Care*, vol. 28, no. 2, pp. 385-390.

McTiernan, A., Sorensen, B., Irwin, M. L., Morgan, A., Yasui, Y., Rudolph, R. E., Surawicz, C., Lampe, J. W., Lampe, P. D., Ayub, K., & Potter, J. D. 2007, "Exercise effect on weight and body fat in men and women", *Obesity*.(*Silver.Spring*), vol. 15, no. 6, pp. 1496-1512.

Meex, R. C., Schrauwen-Hinderling, V. B., Moonen-Kornips, E., Schaart, G., Mensink, M., Phielix, E., van de, W. T., Sels, J. P., Schrauwen, P., & Hesselink, M. K. 2010, "Restoration of muscle mitochondrial function and metabolic flexibility in type 2 diabetes by exercise training is paralleled by increased myocellular fat storage and improved insulin sensitivity", *Diabetes*, vol. 59, no. 3, pp. 572-579.

Melanson, E. L., Donahoo, W. T., Dong, F., Ida, T., & Zemel, M. B. 2005, "Effect of lowand high-calcium dairy-based diets on macronutrient oxidation in humans", *Obes.Res.*, vol. 13, no. 12, pp. 2102-2112.

Micossi, P., Gallus, G., Valsania, P., Garancini, P., Radaelli, G., & Pozza, G. 1987, "First degree family history of diabetes and risk factors for cardiovascular disease in non-diabetic subjects in the National Telephone Company: a metabolic survey", *Eur.J.Epidemiol.*, vol. 3, no. 1, pp. 46-53.

Miettinen, H., Lehto, S., Salomaa, V., Mahonen, M., Niemela, M., Haffner, S. M., Pyorala, K., & Tuomilehto, J. 1998, "Impact of diabetes on mortality after the first myocardial infarction. The FINMONICA Myocardial Infarction Register Study Group", *Diabetes Care*, vol. 21, no. 1, pp. 69-75.

Minokoshi, Y., Kim, Y. B., Peroni, O. D., Fryer, L. G., Muller, C., Carling, D., & Kahn, B. B. 2002, "Leptin stimulates fatty-acid oxidation by activating AMP-activated protein kinase", *Nature*, vol. 415, no. 6869, pp. 339-343.

Miyatake, N., Nishikawa, H., Morishita, A., Kunitomi, M., Wada, J., Suzuki, H., Takahashi, K., Makino, H., Kira, S., & Fujii, M. 2002, "Daily walking reduces visceral adipose tissue areas and improves insulin resistance in Japanese obese subjects", *Diabetes Res. Clin. Pract.*, vol. 58, no. 2, pp. 101-107.

Mohammed, B. S., Cohen, S., Reeds, D., Young, V. L., & Klein, S. 2008, "Long-term effects of large-volume liposuction on metabolic risk factors for coronary heart disease", *Obesity.(Silver.Spring)*, vol. 16, no. 12, pp. 2648-2651.

Moller, D. E. 2000, "Potential role of TNF-alpha in the pathogenesis of insulin resistance and type 2 diabetes", *Trends Endocrinol.Metab*, vol. 11, no. 6, pp. 212-217.

Monajemi, H., Stroes, E., Hegele, R. A., & Fliers, E. 2007, "Inherited lipodystrophies and the metabolic syndrome", *Clin.Endocrinol.(Oxf)*, vol. 67, no. 4, pp. 479-484.

Montecucco, F., Steffens, S., & Mach, F. 2008, "Insulin resistance: a proinflammatory state mediated by lipid-induced signaling dysfunction and involved in atherosclerotic plaque instability", *Mediators.Inflamm.*, vol. 2008, p. 767623.

Montoye HJ, K. H. S. W. W. R. Measuring Physical Activity and Energy Expenditure. 1996. Human Kinetics, Champaign, IL.

Monzillo, L. U., Hamdy, O., Horton, E. S., Ledbury, S., Mullooly, C., Jarema, C., Porter, S., Ovalle, K., Moussa, A., & Mantzoros, C. S. 2003, "Effect of lifestyle modification on adipokine levels in obese subjects with insulin resistance", *Obes.Res.*, vol. 11, no. 9, pp. 1048-1054.

Moran, C. N., Barwell, N. D., Malkova, D., Cleland, S. J., McPhee, I., Packard, C. J., Zammit, V. A., & Gill, J. M. 2010, "Effects of diabetes family history and exercise training on the expression of adiponectin and leptin and their receptors", *Metabolism*.

Morris, J. N., Everitt, M. G., Pollard, R., Chave, S. P., & Semmence, A. M. 1980, "Vigorous exercise in leisure-time: protection against coronary heart disease", *Lancet*, vol. 2, no. 8206, pp. 1207-1210.

Mullen, K. L., Pritchard, J., Ritchie, I., Snook, L. A., Chabowski, A., Bonen, A., Wright, D., & Dyck, D. J. 2009, "Adiponectin resistance precedes the accumulation of skeletal muscle lipids and insulin resistance in high-fat-fed rats", *Am.J.Physiol Regul.Integr.Comp Physiol*, vol. 296, no. 2, p. R243-R251.

Murcia, A. M., Hennekens, C. H., Lamas, G. A., Jimenez-Navarro, M., Rouleau, J. L., Flaker, G. C., Goldman, S., Skali, H., Braunwald, E., & Pfeffer, M. A. 2004, "Impact of diabetes on mortality in patients with myocardial infarction and left ventricular dysfunction", *Arch.Intern.Med.*, vol. 164, no. 20, pp. 2273-2279.

Murri, M., Garcia-Fuentes, E., Garcia-Almeida, J. M., Garrido-Sanchez, L., Mayas, M. D., Bernal, R., & Tinahones, F. J. 2010, "Changes in oxidative stress and insulin resistance in morbidly obese patients after bariatric surgery", *Obes.Surg.*, vol. 20, no. 3, pp. 363-368.

Nagaya, T., Yoshida, H., Takahashi, H., & Kawai, M. 2005, "Increases in body mass index, even within non-obese levels, raise the risk for Type 2 diabetes mellitus: a follow-up study in a Japanese population", *Diabet.Med*, vol. 22, no. 8, pp. 1107-1111.

Nair, K. S., Bigelow, M. L., Asmann, Y. W., Chow, L. S., Coenen-Schimke, J. M., Klaus, K. A., Guo, Z. K., Sreekumar, R., & Irving, B. A. 2008, "Asian Indians have enhanced skeletal muscle mitochondrial capacity to produce ATP in association with severe insulin resistance", *Diabetes*, vol. 57, no. 5, pp. 1166-1175.

Nakanishi, S., Okubo, M., Yoneda, M., Jitsuiki, K., Yamane, K., & Kohno, N. 2004, "A comparison between Japanese-Americans living in Hawaii and Los Angeles and native Japanese: the impact of lifestyle westernization on diabetes mellitus", *Biomed.Pharmacother.*, vol. 58, no. 10, pp. 571-577.

Nassis, G. P., Papantakou, K., Skenderi, K., Triandafillopoulou, M., Kavouras, S. A., Yannakoulia, M., Chrousos, G. P., & Sidossis, L. S. 2005, "Aerobic exercise training improves insulin sensitivity without changes in body weight, body fat, adiponectin, and inflammatory markers in overweight and obese girls", *Metabolism*, vol. 54, no. 11, pp. 1472-1479.

NEEL, J. V. 1962, "Diabetes mellitus: a "thrifty" genotype rendered detrimental by "progress"?", *Am.J Hum.Genet.*, vol. 14, pp. 353-362.

Nielsen, N. B., Hojbjerre, L., Sonne, M. P., Alibegovic, A. C., Vaag, A., Dela, F., & Stallknecht, B. 2009, "Interstitial concentrations of adipokines in subcutaneous abdominal and femoral adipose tissue", *Regul.Pept.*, vol. 155, no. 1-3, pp. 39-45.

Nindl, B. C., Harman, E. A., Marx, J. O., Gotshalk, L. A., Frykman, P. N., Lammi, E., Palmer, C., & Kraemer, W. J. 2000, "Regional body composition changes in women after 6 months of periodized physical training", *J.Appl.Physiol*, vol. 88, no. 6, pp. 2251-2259.

Ning, F., Qiao, Q., Tuomilehto, J., Hammar, N., Ho, S. Y., Soderberg, S., Zimmet, P. Z., Shaw, J. E., Nakagami, T., Mohan, V., Ramachandran, A., Lam, T. H., Andersson, S. W., Janus, E. D., Boyko, E. J., Fujimoto, W. Y., & Pang, Z. C. 2010, "Does abnormal insulin action or insulin secretion explain the increase in prevalence of impaired glucose metabolism with age in populations of different ethnicities?", *Diabetes Metab Res.Rev.*, vol. 26, no. 4, pp. 245-253.

Nissen, S. E. & Wolski, K. 2007, "Effect of rosiglitazone on the risk of myocardial infarction and death from cardiovascular causes", *N.Engl.J.Med.*, vol. 356, no. 24, pp. 2457-2471.

Northoff, H., Symons, S., Zieker, D., Schaible, E. V., Schafer, K., Thoma, S., Loffler, M., Abbasi, A., Simon, P., Niess, A. M., & Fehrenbach, E. 2008, "Gender- and menstrual phase dependent regulation of inflammatory gene expression in response to aerobic exercise", *Exerc.Immunol.Rev.*, vol. 14, pp. 86-103.

Ntali, G., Koutsari, C., Karakike, K., Makras, P., Skopouli, F., Mc, M. M., Slater, C., Higgins, S., Zampelas, A., & Malkova, D. 2009, "Erythrocyte fatty acid composition and insulin sensitivity in daughters of type 2 diabetic patients and women with no family history of diabetes", *J.Endocrinol.Invest*.

Nyholm, B., Fisker, S., Lund, S., Moller, N., & Schmitz, O. 1997, "Increased circulating leptin concentrations in insulin-resistant first-degree relatives of patients with non-insulin-dependent diabetes mellitus: relationship to body composition and insulin sensitivity but not to family history of non-insulin-dependent diabetes mellitus", *Eur.J.Endocrinol.*, vol. 136, no. 2, pp. 173-179.

Nyholm, B., Nielsen, M. F., Kristensen, K., Nielsen, S., Ostergard, T., Pedersen, S. B., Christiansen, T., Richelsen, B., Jensen, M. D., & Schmitz, O. 2004, "Evidence of increased visceral obesity and reduced physical fitness in healthy insulin-resistant first-degree relatives of type 2 diabetic patients", *Eur.J Endocrinol.*, vol. 150, no. 2, pp. 207-214.

O'Gorman, D. J., Karlsson, H. K., McQuaid, S., Yousif, O., Rahman, Y., Gasparro, D., Glund, S., Chibalin, A. V., Zierath, J. R., & Nolan, J. J. 2006, "Exercise training increases insulin-stimulated glucose disposal and GLUT4 (SLC2A4) protein content in patients with type 2 diabetes", *Diabetologia*, vol. 49, no. 12, pp. 2983-2992.

O'Leary, V. B., Marchetti, C. M., Krishnan, R. K., Stetzer, B. P., Gonzalez, F., & Kirwan, J. P. 2006, "Exercise-induced reversal of insulin resistance in obese elderly is associated with reduced visceral fat", *J.Appl.Physiol*, vol. 100, no. 5, pp. 1584-1589.

Oberbach, A., Tonjes, A., Kloting, N., Fasshauer, M., Kratzsch, J., Busse, M. W., Paschke, R., Stumvoll, M., & Bluher, M. 2006, "Effect of a 4 week physical training program on

plasma concentrations of inflammatory markers in patients with abnormal glucose tolerance", *Eur.J.Endocrinol.*, vol. 154, no. 4, pp. 577-585.

Ohlson, L. O., Larsson, B., Bjorntorp, P., Eriksson, H., Svardsudd, K., Welin, L., Tibblin, G., & Wilhelmsen, L. 1988, "Risk factors for type 2 (non-insulin-dependent) diabetes mellitus. Thirteen and one-half years of follow-up of the participants in a study of Swedish men born in 1913", *Diabetologia*, vol. 31, no. 11, pp. 798-805.

Ohlson, L. O., Larsson, B., Svardsudd, K., Welin, L., Eriksson, H., Wilhelmsen, L., Bjorntorp, P., & Tibblin, G. 1985, "The influence of body fat distribution on the incidence of diabetes mellitus. 13.5 years of follow-up of the participants in the study of men born in 1913", *Diabetes*, vol. 34, no. 10, pp. 1055-1058.

Olek, K. 2006, "Maturity-onset diabetes of the young: an update", *Clin.Lab*, vol. 52, no. 11-12, pp. 593-598.

Olive, J. L. & Miller, G. D. 2001, "Differential effects of maximal- and moderate-intensity runs on plasma leptin in healthy trained subjects", *Nutrition*, vol. 17, no. 5, pp. 365-369.

Oliveira, L., Morais, J., Monteiro, P., Costa, M., Antunes, A., Pires, R., Morais, M. E., Freitas, M., & Providencia, L. 2002, "The worse hospital outcome of diabetic patients is not explained by the severity of underlying coronary artery disease", *Rev.Port.Cardiol.*, vol. 21, no. 7-8, pp. 819-832.

Olmedillas, H., Sanchis-Moysi, J., Fuentes, T., Guadalupe-Grau, A., Ponce-Gonzalez, J. G., Morales-Alamo, D., Santana, A., Dorado, C., Calbet, J. A., & Guerra, B. 2010, "Muscle hypertrophy and increased expression of leptin receptors in the musculus triceps brachii of the dominant arm in professional tennis players", *Eur.J.Appl.Physiol*, vol. 108, no. 4, pp. 749-758.

Orchard, T. J., Temprosa, M., Goldberg, R., Haffner, S., Ratner, R., Marcovina, S., & Fowler, S. 2005, "The effect of metformin and intensive lifestyle intervention on the metabolic syndrome: the Diabetes Prevention Program randomized trial", *Ann.Intern.Med.*, vol. 142, no. 8, pp. 611-619.

Osmond, C. & Barker, D. J. 2000, "Fetal, infant, and childhood growth are predictors of coronary heart disease, diabetes, and hypertension in adult men and women", *Environ.Health Perspect.*, vol. 108 Suppl 3, pp. 545-553.

Paffenbarger, R. S., Jr., Hyde, R. T., Wing, A. L., & Hsieh, C. C. 1986, "Physical activity, all-cause mortality, and longevity of college alumni", *N.Engl.J Med*, vol. 314, no. 10, pp. 605-613.

Paffenbarger, R. S., Jr., Lee, I. M., & Kampert, J. B. 1997, "Physical activity in the prevention of non-insulin-dependent diabetes mellitus", *World Rev.Nutr.Diet.*, vol. 82, pp. 210-218.

Pagliassotti, M. J., Gayles, E. C., & Hill, J. O. 1997, "Fat and energy balance", *Ann.N.Y.Acad.Sci.*, vol. 827, pp. 431-448.

Pan, X. R., Li, G. W., Hu, Y. H., Wang, J. X., Yang, W. Y., An, Z. X., Hu, Z. X., Lin, J., Xiao, J. Z., Cao, H. B., Liu, P. A., Jiang, X. G., Jiang, Y. Y., Wang, J. P., Zheng, H., Zhang,

H., Bennett, P. H., & Howard, B. V. 1997, "Effects of diet and exercise in preventing NIDDM in people with impaired glucose tolerance. The Da Qing IGT and Diabetes Study", *Diabetes Care*, vol. 20, no. 4, pp. 537-544.

Pannacciulli, N., De, P. G., Ciccone, M., Rizzon, P., Giorgino, F., & Giorgino, R. 2003, "Effect of family history of type 2 diabetes on the intima-media thickness of the common carotid artery in normal-weight, overweight, and obese glucose-tolerant young adults", *Diabetes Care*, vol. 26, no. 4, pp. 1230-1234.

Park, J. Y., Chong, A. Y., Cochran, E. K., Kleiner, D. E., Haller, M. J., Schatz, D. A., & Gorden, P. 2008, "Type 1 diabetes associated with acquired generalized lipodystrophy and insulin resistance: the effect of long-term leptin therapy", *J.Clin.Endocrinol.Metab*, vol. 93, no. 1, pp. 26-31.

Pasman, W. J., Westerterp, M. S., & Saris, W. H. 1999, "The effect of body weight changes and endurance training on 24h substrate oxidation", *Int.J.Obes.Relat Metab Disord.*, vol. 23, no. 12, pp. 1223-1232.

Patel, J. V., Lim, H. S., Gunarathne, A., Tracey, I., Durrington, P. N., Hughes, E. A., & Lip, G. Y. 2008, "Ethnic differences in myocardial infarction in patients with hypertension: effects of diabetes mellitus", *QJM*., vol. 101, no. 3, pp. 231-236.

Paul, D. S., Harmon, A. W., Winston, C. P., & Patel, Y. M. 2003, "Calpain facilitates GLUT4 vesicle translocation during insulin-stimulated glucose uptake in adipocytes", *Biochem.J.*, vol. 376, no. Pt 3, pp. 625-632.

Pedersen, B. K. 2006, "The anti-inflammatory effect of exercise: its role in diabetes and cardiovascular disease control", *Essays Biochem.*, vol. 42, pp. 105-117.

Perreault, L., Lavely, J. M., Bergman, B. C., & Horton, T. J. 2004, "Gender differences in insulin action after a single bout of exercise", *J.Appl.Physiol*, vol. 97, no. 3, pp. 1013-1021.

Perseghin, G., Ghosh, S., Gerow, K., & Shulman, G. I. 1997, "Metabolic defects in lean nondiabetic offspring of NIDDM parents: a cross-sectional study", *Diabetes*, vol. 46, no. 6, pp. 1001-1009.

Perseghin, G., Price, T. B., Petersen, K. F., Roden, M., Cline, G. W., Gerow, K., Rothman, D. L., & Shulman, G. I. 1996, "Increased glucose transport-phosphorylation and muscle glycogen synthesis after exercise training in insulin-resistant subjects", *N.Engl.J.Med.*, vol. 335, no. 18, pp. 1357-1362.

Petersen, A. M. & Pedersen, B. K. 2005, "The anti-inflammatory effect of exercise", *J.Appl.Physiol*, vol. 98, no. 4, pp. 1154-1162.

Petersen, K. F., Dufour, S., & Shulman, G. I. 2005, "Decreased insulin-stimulated ATP synthesis and phosphate transport in muscle of insulin-resistant offspring of type 2 diabetic parents", *PLoS.Med.*, vol. 2, no. 9, p. e233.

Petersen, K. F. & Shulman, G. I. 2006, "Etiology of insulin resistance", *Am.J Med*, vol. 119, no. 5 Suppl 1, p. S10-S16.

Pi-Sunyer, F. X. 2000, "Obesity: criteria and classification", *Proc.Nutr.Soc.*, vol. 59, no. 4, pp. 505-509.

Picot, J., Jones, J., Colquitt, J. L., Gospodarevskaya, E., Loveman, E., Baxter, L., & Clegg, A. J. 2009, "The clinical effectiveness and cost-effectiveness of bariatric (weight loss) surgery for obesity: a systematic review and economic evaluation", *Health Technol.Assess.*, vol. 13, no. 41, pp. 1-357, iii.

Piwonska, A., Piotrowski, W., Broda, G., Drygas, W., Gluszek, J., Zdrojewski, T., Kozakiewicz, K., Stepaniak, U., & Bandosz, P. 2008, "The relationship between resting heart rate and atherosclerosis risk factors", *Kardiol.Pol.*, vol. 66, no. 10, pp. 1069-1075.

Pladevall, M., Singal, B., Williams, L. K., Brotons, C., Guyer, H., Sadurni, J., Falces, C., Serrano-Rios, M., Gabriel, R., Shaw, J. E., Zimmet, P. Z., & Haffner, S. 2006, "A single factor underlies the metabolic syndrome: a confirmatory factor analysis", *Diabetes Care*, vol. 29, no. 1, pp. 113-122.

Polak, J., Klimcakova, E., Moro, C., Viguerie, N., Berlan, M., Hejnova, J., Richterova, B., Kraus, I., Langin, D., & Stich, V. 2006, "Effect of aerobic training on plasma levels and subcutaneous abdominal adipose tissue gene expression of adiponectin, leptin, interleukin 6, and tumor necrosis factor alpha in obese women", *Metabolism*, vol. 55, no. 10, pp. 1375-1381.

Polak, J., Kovacova, Z., Jacek, M., Klimcakova, E., Kovacikova, M., Vitkova, M., Kuda, O., Sebela, M., Samcova, E., & Stich, V. 2007, "An increase in plasma adiponectin multimeric complexes follows hypocaloric diet-induced weight loss in obese and overweight premenopausal women", *Clin.Sci.(Lond)*, vol. 112, no. 11, pp. 557-565.

Pollock, M. L., Gaesser, G. A., Butcher, J. D., Després, J. P., Dishman, R. K., Franklin, B. A., & Garber, C. E. 1998, "ACSM Position Stand: The Recommended Quantity and Quality of Exercise for Developing and Maintaining Cardiorespiratory and Muscular Fitness, and Flexibility in Healthy Adults", *Medicine & Science in Sports & Exercise*, vol. 30, no. 6.

Pols, M. A., Peeters, P. H., Kemper, H. C., & Collette, H. J. 1996, "Repeatability and relative validity of two physical activity questionnaires in elderly women", *Med.Sci.Sports Exerc.*, vol. 28, no. 8, pp. 1020-1025.

Popa, C., Netea, M. G., van Riel, P. L., van der Meer, J. W., & Stalenhoef, A. F. 2007, "The role of TNF-alpha in chronic inflammatory conditions, intermediary metabolism, and cardiovascular risk", *J.Lipid Res.*, vol. 48, no. 4, pp. 751-762.

Pradhan, A. D., Cook, N. R., Buring, J. E., Manson, J. E., & Ridker, P. M. 2003, "C-reactive protein is independently associated with fasting insulin in nondiabetic women", *Arterioscler.Thromb.Vasc.Biol.*, vol. 23, no. 4, pp. 650-655.

Pradhan, A. D., Manson, J. E., Rifai, N., Buring, J. E., & Ridker, P. M. 2001, "C-reactive protein, interleukin 6, and risk of developing type 2 diabetes mellitus", *JAMA*, vol. 286, no. 3, pp. 327-334.

Qu, Y., Yang, Z., Jin, F., Sun, L., Feng, J., Tang, L., Zhang, C., Zhu, X., Shi, X., Sun, H., Wang, B., & Wang, L. 2008, "The haplotype identified in LEPR gene is associated with type 2 diabetes mellitus in Northern Chinese", *Diabetes Res.Clin.Pract.* 

Raatz, S. K., Torkelson, C. J., Redmon, J. B., Reck, K. P., Kwong, C. A., Swanson, J. E., Liu, C., Thomas, W., & Bantle, J. P. 2005, "Reduced glycemic index and glycemic load diets do not increase the effects of energy restriction on weight loss and insulin sensitivity in obese men and women", *J.Nutr.*, vol. 135, no. 10, pp. 2387-2391.

Ramachandran, A., Snehalatha, C., Mary, S., Mukesh, B., Bhaskar, A. D., & Vijay, V. 2006, "The Indian Diabetes Prevention Programme shows that lifestyle modification and metformin prevent type 2 diabetes in Asian Indian subjects with impaired glucose tolerance (IDPP-1)", *Diabetologia*, vol. 49, no. 2, pp. 289-297.

Rattigan, S., Bradley, E. A., Richards, S. M., & Clark, M. G. 2006, "Muscle metabolism and control of capillary blood flow: insulin and exercise", *Essays Biochem.*, vol. 42, pp. 133-144.

Ravikumar, B., Gerrard, J., Dalla, M. C., Firbank, M. J., Lane, A., English, P. T., Cobelli, C., & Taylor, R. 2008, "Pioglitazone decreases fasting and postprandial endogenous glucose production in proportion to decrease in hepatic triglyceride content", *Diabetes*, vol. 57, no. 9, pp. 2288-2295.

Razak, F., Anand, S. S., Shannon, H., Vuksan, V., Davis, B., Jacobs, R., Teo, K. K., McQueen, M., & Yusuf, S. 2007, "Defining obesity cut points in a multiethnic population", *Circulation*, vol. 115, no. 16, pp. 2111-2118.

Reaven, G. M. 1988, "Banting lecture 1988. Role of insulin resistance in human disease", *Diabetes*, vol. 37, no. 12, pp. 1595-1607.

Reaven, G. M. 1992, "Syndrome X", Blood Press Suppl, vol. 4, pp. 13-16.

Regitz-Zagrosek, V., Lehmkuhl, E., & Weickert, M. O. 2006, "Gender differences in the metabolic syndrome and their role for cardiovascular disease", *Clin.Res.Cardiol.*, vol. 95, no. 3, pp. 136-147.

Reid, B. N., Ables, G. P., Otlivanchik, O. A., Schoiswohl, G., Zechner, R., Blaner, W. S., Goldberg, I. J., Schwabe, R. F., Chua, S. C., Jr., & Huang, L. S. 2008, "Hepatic overexpression of hormone-sensitive lipase and adipose triglyceride lipase promotes fatty acid oxidation, stimulates direct release of free fatty acids, and ameliorates steatosis", *J.Biol.Chem.*, vol. 283, no. 19, pp. 13087-13099.

Rizzoni, D., Porteri, E., Guelfi, D., Muiesan, M. L., Valentini, U., Cimino, A., Girelli, A., Rodella, L., Bianchi, R., Sleiman, I., & Rosei, E. A. 2001, "Structural alterations in subcutaneous small arteries of normotensive and hypertensive patients with non-insulin-dependent diabetes mellitus", *Circulation*, vol. 103, no. 9, pp. 1238-1244.

Robinson, S. M., Wheeler, T., Hayes, M. C., Barker, D. J., & Osmond, C. 1991, "Fetal heart rate and intrauterine growth", *Br.J.Obstet.Gynaecol.*, vol. 98, no. 12, pp. 1223-1227.

Roden, M., Stingl, H., Chandramouli, V., Schumann, W. C., Hofer, A., Landau, B. R., Nowotny, P., Waldhausl, W., & Shulman, G. I. 2000, "Effects of free fatty acid elevation on postabsorptive endogenous glucose production and gluconeogenesis in humans", *Diabetes*, vol. 49, no. 5, pp. 701-707. Romijn, J. A., Klein, S., Coyle, E. F., Sidossis, L. S., & Wolfe, R. R. 1993, "Strenuous endurance training increases lipolysis and triglyceride-fatty acid cycling at rest", *J.Appl.Physiol*, vol. 75, no. 1, pp. 108-113.

Ross, R., Dagnone, D., Jones, P. J., Smith, H., Paddags, A., Hudson, R., & Janssen, I. 2000, "Reduction in obesity and related comorbid conditions after diet-induced weight loss or exercise-induced weight loss in men. A randomized, controlled trial", *Ann.Intern.Med.*, vol. 133, no. 2, pp. 92-103.

Ruotsalainen, E., Salmenniemi, U., Vauhkonen, I., Pihlajamaki, J., Punnonen, K., Kainulainen, S., & Laakso, M. 2006, "Changes in inflammatory cytokines are related to impaired glucose tolerance in offspring of type 2 diabetic subjects", *Diabetes Care*, vol. 29, no. 12, pp. 2714-2720.

Ruotsalainen, E., Vauhkonen, I., Salmenniemi, U., Pihlajamaki, J., Punnonen, K., Kainulainen, S., Jalkanen, S., Salmi, M., & Laakso, M. 2008, "Markers of endothelial dysfunction and low-grade inflammation are associated in the offspring of type 2 diabetic subjects", *Atherosclerosis*, vol. 197, no. 1, pp. 271-277.

Rutter, M. K., Meigs, J. B., Sullivan, L. M., D'Agostino, R. B., Sr., & Wilson, P. W. 2005, "Insulin resistance, the metabolic syndrome, and incident cardiovascular events in the Framingham Offspring Study", *Diabetes*, vol. 54, no. 11, pp. 3252-3257.

Ryan, A. S., Nicklas, B. J., & Berman, D. M. 2002, "Racial differences in insulin resistance and mid-thigh fat deposition in postmenopausal women", *Obes.Res.*, vol. 10, no. 5, pp. 336-344.

Ryan, M. C., Abbasi, F., Lamendola, C., Carter, S., & McLaughlin, T. L. 2007, "Serum alanine aminotransferase levels decrease further with carbohydrate than fat restriction in insulin-resistant adults", *Diabetes Care*, vol. 30, no. 5, pp. 1075-1080.

Saely, C. H., Aczel, S., Marte, T., Langer, P., Hoefle, G., & Drexel, H. 2005, "The metabolic syndrome, insulin resistance, and cardiovascular risk in diabetic and nondiabetic patients", *J.Clin.Endocrinol.Metab*, vol. 90, no. 10, pp. 5698-5703.

Saely, C. H., Koch, L., Schmid, F., Marte, T., Aczel, S., Langer, P., Hoefle, G., & Drexel, H. 2006, "Adult Treatment Panel III 2001 but not International Diabetes Federation 2005 criteria of the metabolic syndrome predict clinical cardiovascular events in subjects who underwent coronary angiography", *Diabetes Care*, vol. 29, no. 4, pp. 901-907.

Sakamoto, K. & Goodyear, L. J. 2002, "Invited review: intracellular signaling in contracting skeletal muscle", *J.Appl.Physiol*, vol. 93, no. 1, pp. 369-383.

Salmenniemi, U., Ruotsalainen, E., Vanttinen, M., Vauhkonen, I., Pihlajamaki, J., Kainulainen, S., Punnonen, K., & Laakso, M. 2005, "High amount of visceral fat mass is associated with multiple metabolic changes in offspring of type 2 diabetic patients", *Int.J.Obes.(Lond)*, vol. 29, no. 12, pp. 1464-1470.

Saltevo, J., Kautiainen, H., & Vanhala, M. 2009, "Gender differences in adiponectin and lowgrade inflammation among individuals with normal glucose tolerance, prediabetes, and type 2 diabetes", *Gend.Med.*, vol. 6, no. 3, pp. 463-470. Saltevo, J., Vanhala, M., Kautiainen, H., Kumpusalo, E., & Laakso, M. 2008, "Gender differences in C-reactive protein, interleukin-1 receptor antagonist and adiponectin levels in the metabolic syndrome: a population-based study", *Diabet.Med.*, vol. 25, no. 6, pp. 747-750.

Samuel, V. T., Beddow, S. A., Iwasaki, T., Zhang, X. M., Chu, X., Still, C. D., Gerhard, G. S., & Shulman, G. I. 2009, "Fasting hyperglycemia is not associated with increased expression of PEPCK or G6Pc in patients with Type 2 Diabetes", *Proc.Natl.Acad.Sci.U.S.A*, vol. 106, no. 29, pp. 12121-12126.

Sarada, S., Himadri, P., Mishra, C., Geetali, P., Ram, M. S., & Ilavazhagan, G. 2008, "Role of oxidative stress and NFkB in hypoxia-induced pulmonary edema", *Exp.Biol.Med.(Maywood.)*, vol. 233, no. 9, pp. 1088-1098.

Sarlund, H., Pyorala, K., Penttila, I., & Laakso, M. 1992, "Early abnormalities in coronary heart disease risk factors in relatives of subjects with non-insulin-dependent diabetes", *Arterioscler.Thromb.*, vol. 12, no. 6, pp. 657-663.

Sato, F., Tamura, Y., Watada, H., Kumashiro, N., Igarashi, Y., Uchino, H., Maehara, T., Kyogoku, S., Sunayama, S., Sato, H., Hirose, T., Tanaka, Y., & Kawamori, R. 2007, "Effects of diet-induced moderate weight reduction on intrahepatic and intramyocellular triglycerides and glucose metabolism in obese subjects", *J. Clin. Endocrinol. Metab*, vol. 92, no. 8, pp. 3326-3329.

Sattar, N., Gaw, A., Scherbakova, O., Ford, I., O'Reilly, D. S., Haffner, S. M., Isles, C., Macfarlane, P. W., Packard, C. J., Cobbe, S. M., & Shepherd, J. 2003, "Metabolic syndrome with and without C-reactive protein as a predictor of coronary heart disease and diabetes in the West of Scotland Coronary Prevention Study", *Circulation*, vol. 108, no. 4, pp. 414-419.

Sattar, N. & Lowe, G. D. 2006, "High sensitivity C-reactive protein and cardiovascular disease: an association built on unstable foundations?", *Ann.Clin.Biochem.*, vol. 43, no. Pt 4, pp. 252-256.

Savage, D. B., Petersen, K. F., & Shulman, G. I. 2007, "Disordered lipid metabolism and the pathogenesis of insulin resistance", *Physiol Rev.*, vol. 87, no. 2, pp. 507-520.

Sawicki, P. T. & Kaiser, T. 2004, "Response to Chiasson et al.: Acarbose for the prevention of Type 2 diabetes, hypertension and cardiovascular disease in subjects with impaired glucose tolerance: facts and interpretations concerning the critical analysis of the STOP-NIDDM Trial data", *Diabetologia*, vol. 47, no. 6, pp. 976-977.

Schafer, S., Kantartzis, K., Machann, J., Venter, C., Niess, A., Schick, F., Machicao, F., Haring, H. U., Fritsche, A., & Stefan, N. 2007, "Lifestyle intervention in individuals with normal versus impaired glucose tolerance", *Eur.J. Clin.Invest*, vol. 37, no. 7, pp. 535-543.

Scheen, A. J. 2007, "Antidiabetic agents in subjects with mild dysglycaemia: prevention or early treatment of type 2 diabetes?", *Diabetes Metab*, vol. 33, no. 1, pp. 3-12.

Schenk, S., Harber, M. P., Shrivastava, C. R., Burant, C. F., & Horowitz, J. F. 2009, "Improved insulin sensitivity after weight loss and exercise training is mediated by a reduction in plasma fatty acid mobilization, not enhanced oxidative capacity", *J.Physiol*. Schenk, S. & Horowitz, J. F. 2006, "Coimmunoprecipitation of FAT/CD36 and CPT I in skeletal muscle increases proportionally with fat oxidation after endurance exercise training", *Am.J.Physiol Endocrinol.Metab*, vol. 291, no. 2, p. E254-E260.

Schoeller, D. A. & Webb, P. 1984, "Five-day comparison of the doubly labeled water method with respiratory gas exchange", *Am.J.Clin.Nutr.*, vol. 40, no. 1, pp. 153-158.

Schonfeld, G., Yue, P., Lin, X., & Chen, Z. 2008, "Fatty liver and insulin resistance: not always linked", *Trans.Am.Clin.Climatol.Assoc.*, vol. 119, pp. 217-223.

Schulz, L. O., Bennett, P. H., Ravussin, E., Kidd, J. R., Kidd, K. K., Esparza, J., & Valencia, M. E. 2006, "Effects of traditional and western environments on prevalence of type 2 diabetes in Pima Indians in Mexico and the U.S", *Diabetes Care*, vol. 29, no. 8, pp. 1866-1871.

Scuteri, A., Tesauro, M., Rizza, S., Iantorno, M., Federici, M., Lauro, D., Campia, U., Turriziani, M., Fusco, A., Cocciolillo, G., & Lauro, R. 2008, "Endothelial function and arterial stiffness in normotensive normoglycemic first-degree relatives of diabetic patients are independent of the metabolic syndrome", *Nutr.Metab Cardiovasc.Dis.*, vol. 18, no. 5, pp. 349-356.

Seematter, G., Binnert, C., Martin, J. L., & Tappy, L. 2004, "Relationship between stress, inflammation and metabolism", *Curr.Opin.Clin.Nutr.Metab Care*, vol. 7, no. 2, pp. 169-173.

Seidell, J. C., Muller, D. C., Sorkin, J. D., & Andres, R. 1992, "Fasting respiratory exchange ratio and resting metabolic rate as predictors of weight gain: the Baltimore Longitudinal Study on Aging", *Int.J.Obes.Relat Metab Disord.*, vol. 16, no. 9, pp. 667-674.

Seppala-Lindroos, A., Vehkavaara, S., Hakkinen, A. M., Goto, T., Westerbacka, J., Sovijarvi, A., Halavaara, J., & Yki-Jarvinen, H. 2002, "Fat accumulation in the liver is associated with defects in insulin suppression of glucose production and serum free fatty acids independent of obesity in normal men", *J.Clin.Endocrinol.Metab*, vol. 87, no. 7, pp. 3023-3028.

Serne, E. H., de Jongh, R. T., Eringa, E. C., Ijzerman, R. G., de Boer, M. P., & Stehouwer, C. D. 2006, "Microvascular dysfunction: causative role in the association between hypertension, insulin resistance and the metabolic syndrome?", *Essays Biochem.*, vol. 42, pp. 163-176.

Setter, S. M., Iltz, J. L., Thams, J., & Campbell, R. K. 2003, "Metformin hydrochloride in the treatment of type 2 diabetes mellitus: a clinical review with a focus on dual therapy", *Clin.Ther.*, vol. 25, no. 12, pp. 2991-3026.

Shaffer, E. A. 2006, "Gallstone disease: Epidemiology of gallbladder stone disease", *Best.Pract.Res.Clin.Gastroenterol.*, vol. 20, no. 6, pp. 981-996.

Shah, A. S., Dolan, L. M., Kimball, T. R., Gao, Z., Khoury, P. R., Daniels, S. R., & Urbina, E. M. 2009a, "Influence of Duration of Diabetes, Glycemic Control, and Traditional Cardiovascular Risk Factors on Early Atherosclerotic Vascular Changes in Adolescents and Young Adults with Type 2 Diabetes Mellitus", *J.Clin.Endocrinol.Metab*.

Shah, K., Stufflebam, A., Hilton, T. N., Sinacore, D. R., Klein, S., & Villareal, D. T. 2009b, "Diet and Exercise Interventions Reduce Intrahepatic Fat Content and Improve Insulin Sensitivity in Obese Older Adults", *Obesity.(Silver.Spring)*. Shah, P., Vella, A., Basu, A., Basu, R., Adkins, A., Schwenk, W. F., Johnson, C. M., Nair, K. S., Jensen, M. D., & Rizza, R. A. 2002, "Effects of free fatty acids and glycerol on splanchnic glucose metabolism and insulin extraction in nondiabetic humans", *Diabetes*, vol. 51, no. 2, pp. 301-310.

Shalitin, S., shkenazi-Hoffnung, L., Yackobovitch-Gavan, M., Nagelberg, N., Karni, Y., Hershkovitz, E., Loewenthal, N., Shtaif, B., Gat-Yablonski, G., & Phillip, M. 2009, "Effects of a twelve-week randomized intervention of exercise and/or diet on weight loss and weight maintenance, and other metabolic parameters in obese preadolescent children", *Horm.Res.*, vol. 72, no. 5, pp. 287-301.

Shaw, C. S., Sherlock, M., Stewart, P. M., & Wagenmakers, A. J. 2009, "Adipophilin distribution and colocalization with lipid droplets in skeletal muscle", *Histochem.Cell Biol.*, vol. 131, no. 5, pp. 575-581.

Shaw, K., Gennat, H., O'Rourke, P., & Del, M. C. 2006, "Exercise for overweight or obesity", *Cochrane.Database.Syst.Rev.* no. 4, p. CD003817.

Shen, B. J., Todaro, J. F., Niaura, R., McCaffery, J. M., Zhang, J., Spiro III, A., & Ward, K. D. 2003, "Are Metabolic Risk Factors One Unified Syndrome? Modeling the Structure of the Metabolic Syndrome X", *American Journal of Epidemiology*, vol. 157, no. 8, pp. 701-711.

Shojaee-Moradie, F., Baynes, K. C., Pentecost, C., Bell, J. D., Thomas, E. L., Jackson, N. C., Stolinski, M., Whyte, M., Lovell, D., Bowes, S. B., Gibney, J., Jones, R. H., & Umpleby, A. M. 2007, "Exercise training reduces fatty acid availability and improves the insulin sensitivity of glucose metabolism", *Diabetologia*, vol. 50, no. 2, pp. 404-413.

Siklova-Vitkova, M. P., Polak, J., Klimcakova, E., Vrzalova, J., Hejnova, J., Kovacikova, M., Kovacova, Z., Bajzova, M., Rossmeislova, L., Hnevkovska, Z., Langin, D., & Stich, V. P. 2009, "Effect of hyperinsulinemia and very-low-calorie diet on interstitial cytokine levels in subcutaneous adipose tissue of obese women", *Am.J.Physiol Endocrinol.Metab*.

Simmons, R. K., Griffin, S. J., Steele, R., Wareham, N. J., & Ekelund, U. 2008, "Increasing overall physical activity and aerobic fitness is associated with improvements in metabolic risk: cohort analysis of the ProActive trial", *Diabetologia*, vol. 51, no. 5, pp. 787-794.

Sjostrom, L., Lindroos, A. K., Peltonen, M., Torgerson, J., Bouchard, C., Carlsson, B., Dahlgren, S., Larsson, B., Narbro, K., Sjostrom, C. D., Sullivan, M., & Wedel, H. 2004, "Lifestyle, diabetes, and cardiovascular risk factors 10 years after bariatric surgery", *N.Engl.J.Med.*, vol. 351, no. 26, pp. 2683-2693.

Smith, C. J., Nelson, R. G., Hardy, S. A., Manahan, E. M., Bennett, P. H., & Knowler, W. C. 1996, "Survey of the diet of Pima Indians using quantitative food frequency assessment and 24-hour recall. Diabetic Renal Disease Study", *J Am.Diet.Assoc.*, vol. 96, no. 8, pp. 778-784.

Smith, S. R., Lovejoy, J. C., Greenway, F., Ryan, D., deJonge, L., de la, B. J., Volafova, J., & Bray, G. A. 2001, "Contributions of total body fat, abdominal subcutaneous adipose tissue compartments, and visceral adipose tissue to the metabolic complications of obesity", *Metabolism*, vol. 50, no. 4, pp. 425-435.

Sniderman, A. D., Bhopal, R., Prabhakaran, D., Sarrafzadegan, N., & Tchernof, A. 2007, "Why might South Asians be so susceptible to central obesity and its atherogenic consequences? The adipose tissue overflow hypothesis", *Int.J.Epidemiol.*, vol. 36, no. 1, pp. 220-225.

Snyder, K. A., Donnelly, J. E., Jabobsen, D. J., Hertner, G., & Jakicic, J. M. 1997, "The effects of long-term, moderate intensity, intermittent exercise on aerobic capacity, body composition, blood lipids, insulin and glucose in overweight females", *Int.J.Obes.Relat Metab Disord.*, vol. 21, no. 12, pp. 1180-1189.

Solomon, T. P., Sistrun, S. N., Krishnan, R. K., Del Aguila, L. F., Marchetti, C. M., O'Carroll, S. M., O'Leary, V. B., & Kirwan, J. P. 2008, "Exercise and diet enhance fat oxidation and reduce insulin resistance in older obese adults", *J.Appl.Physiol*, vol. 104, no. 5, pp. 1313-1319.

Soto, P. F., Herrero, P., Schechtman, K. B., Waggoner, A. D., Baumstark, J. M., Ehsani, A. A., & Gropler, R. J. 2008, "Exercise training impacts the myocardial metabolism of older individuals in a gender-specific manner", *Am.J.Physiol Heart Circ.Physiol*, vol. 295, no. 2, p. H842-H850.

Sowers, J. R. 2008, "Endocrine functions of adipose tissue: focus on adiponectin", *Clin.Cornerstone.*, vol. 9, no. 1, pp. 32-38.

Spalding, K. L., Arner, E., Westermark, P. O., Bernard, S., Buchholz, B. A., Bergmann, O., Blomqvist, L., Hoffstedt, J., Naslund, E., Britton, T., Concha, H., Hassan, M., Ryden, M., Frisen, J., & Arner, P. 2008, "Dynamics of fat cell turnover in humans", *Nature*, vol. 453, no. 7196, pp. 783-787.

Speakman, J. R., Selman, C., McLaren, J. S., & Harper, E. J. 2002, "Living fast, dying when? The link between aging and energetics", *J.Nutr.*, vol. 132, no. 6 Suppl 2, pp. 1583S-1597S.

Spurr, G. B. 1990, "Physical activity and energy expenditure in undernutrition", *Prog.Food Nutr.Sci.*, vol. 14, no. 2-3, pp. 139-192.

Spurr, G. B., Prentice, A. M., Murgatroyd, P. R., Goldberg, G. R., Reina, J. C., & Christman, N. T. 1988, "Energy expenditure from minute-by-minute heart-rate recording: comparison with indirect calorimetry", *Am.J.Clin.Nutr.*, vol. 48, no. 3, pp. 552-559.

St Onge, M. P., Janssen, I., & Heymsfield, S. B. 2004, "Metabolic syndrome in normalweight Americans: new definition of the metabolically obese, normal-weight individual", *Diabetes Care*, vol. 27, no. 9, pp. 2222-2228.

Stamler, J., Vaccaro, O., Neaton, J. D., & Wentworth, D. 1993, "Diabetes, other risk factors, and 12-yr cardiovascular mortality for men screened in the Multiple Risk Factor Intervention Trial", *Diabetes Care*, vol. 16, no. 2, pp. 434-444.

Stefanyk, L. E. & Dyck, D. J. 2010, "The interaction between adipokines, diet and exercise on muscle insulin sensitivity", *Curr.Opin.Clin.Nutr.Metab Care*, vol. 13, no. 3, pp. 255-259.

Steinberg, G. R. & Jorgensen, S. B. 2007, "The AMP-activated protein kinase: role in regulation of skeletal muscle metabolism and insulin sensitivity", *Mini.Rev.Med.Chem.*, vol. 7, no. 5, pp. 519-526.

Steinberg, G. R., McAinch, A. J., Chen, M. B., O'Brien, P. E., Dixon, J. B., Cameron-Smith, D., & Kemp, B. E. 2006, "The suppressor of cytokine signaling 3 inhibits leptin activation of AMP-kinase in cultured skeletal muscle of obese humans", *J.Clin.Endocrinol.Metab*, vol. 91, no. 9, pp. 3592-3597.

Steinberg, G. R., Parolin, M. L., Heigenhauser, G. J., & Dyck, D. J. 2002, "Leptin increases FA oxidation in lean but not obese human skeletal muscle: evidence of peripheral leptin resistance", *Am.J.Physiol Endocrinol.Metab*, vol. 283, no. 1, p. E187-E192.

Stewart, M. W., Humphriss, D. B., Mitcheson, J., Webster, J., Walker, M., & Laker, M. F. 1998, "Lipoprotein composition and serum apolipoproteins in normoglycaemic first-degree relatives of non-insulin dependent diabetic patients", *Atherosclerosis*, vol. 139, no. 1, pp. 115-121.

Steyn, N. P., Lambert, E. V., & Tabana, H. 2009, "Conference on "Multidisciplinary approaches to nutritional problems". Symposium on "Diabetes and health". Nutrition interventions for the prevention of type 2 diabetes", *Proc.Nutr.Soc.*, vol. 68, no. 1, pp. 55-70.

Stofkova, A. 2009, "Leptin and adiponectin: from energy and metabolic dysbalance to inflammation and autoimmunity", *Endocr.Regul.*, vol. 43, no. 4, pp. 157-168.

Stone, P. H., Muller, J. E., Hartwell, T., York, B. J., Rutherford, J. D., Parker, C. B., Turi, Z. G., Strauss, H. W., Willerson, J. T., Robertson, T., & . 1989, "The effect of diabetes mellitus on prognosis and serial left ventricular function after acute myocardial infarction: contribution of both coronary disease and diastolic left ventricular dysfunction to the adverse prognosis. The MILIS Study Group", *J.Am.Coll.Cardiol.*, vol. 14, no. 1, pp. 49-57.

Straczkowski, M. & Kowalska, I. 2008, "The role of skeletal muscle sphingolipids in the development of insulin resistance", *Rev.Diabet.Stud.*, vol. 5, no. 1, pp. 13-24.

Straczkowski, M., Kowalska, I., Dzienis-Straczkowska, S., Stepien, A., Skibinska, E., Szelachowska, M., & Kinalska, I. 2001, "Changes in tumor necrosis factor-alpha system and insulin sensitivity during an exercise training program in obese women with normal and impaired glucose tolerance", *Eur.J.Endocrinol.*, vol. 145, no. 3, pp. 273-280.

Straczkowski, M., Kowalska, I., Stepien, A., Dzienis-Straczkowska, S., Szelachowska, M., & Kinalska, I. 2002, "Increased plasma-soluble tumor necrosis factor-alpha receptor 2 level in lean nondiabetic offspring of type 2 diabetic subjects", *Diabetes Care*, vol. 25, no. 10, pp. 1824-1828.

Strasser, B., Spreitzer, A., & Haber, P. 2007, "Fat loss depends on energy deficit only, independently of the method for weight loss", *Ann.Nutr.Metab*, vol. 51, no. 5, pp. 428-432.

Stratford, S., Hoehn, K. L., Liu, F., & Summers, S. A. 2004, "Regulation of insulin action by ceramide: dual mechanisms linking ceramide accumulation to the inhibition of Akt/protein kinase B", *J.Biol.Chem.*, vol. 279, no. 35, pp. 36608-36615.

Strychar, I. 2006, "Diet in the management of weight loss", CMAJ., vol. 174, no. 1, pp. 56-63.

Stryer, L. Biochemistry. [3rd Edition], 63-64. 1988. W.H. Freeman and Company, New York.

Sydow, K., Mondon, C. E., & Cooke, J. P. 2005, "Insulin resistance: potential role of the endogenous nitric oxide synthase inhibitor ADMA", *Vasc.Med.*, vol. 10 Suppl 1, p. S35-S43.

Tan, G. D., Savage, D. B., Fielding, B. A., Collins, J., Hodson, L., Humphreys, S. M., O'Rahilly, S., Chatterjee, K., Frayn, K. N., & Karpe, F. 2008, "Fatty acid metabolism in patients with PPARgamma mutations", *J.Clin.Endocrinol.Metab*, vol. 93, no. 11, pp. 4462-4470.

Taskinen, M. R. 2005, "Type 2 diabetes as a lipid disorder", *Curr.Mol.Med*, vol. 5, no. 3, pp. 297-308.

Taylor, D. J., Thompson, C. H., Kemp, G. J., Barnes, P. R., Sanderson, A. L., Radda, G. K., & Phillips, D. I. 1995, "A relationship between impaired fetal growth and reduced muscle glycolysis revealed by 31P magnetic resonance spectroscopy", *Diabetologia*, vol. 38, no. 10, pp. 1205-1212.

Taylor, R. 2008, "Pathogenesis of type 2 diabetes: tracing the reverse route from cure to cause", *Diabetologia*, vol. 51, no. 10, pp. 1781-1789.

Taylor, S. I., Accili, D., & Imai, Y. 1994, "Insulin resistance or insulin deficiency. Which is the primary cause of NIDDM?", *Diabetes*, vol. 43, no. 6, pp. 735-740.

Tesauro, M., Rizza, S., Iantorno, M., Campia, U., Cardillo, C., Lauro, D., Leo, R., Turriziani, M., Cocciolillo, G. C., Fusco, A., Panza, J. A., Scuteri, A., Federici, M., Lauro, R., & Quon, M. J. 2007, "Vascular, metabolic, and inflammatory abnormalities in normoglycemic offspring of patients with type 2 diabetes mellitus", *Metabolism*, vol. 56, no. 3, pp. 413-419.

Thayer, J. F. 2009, "Vagal tone and the inflammatory reflex", *Cleve.Clin.J.Med.*, vol. 76 Suppl 2, p. S23-S26.

Thayer, J. F. & Fischer, J. E. 2009, "Heart rate variability, overnight urinary norepinephrine and C-reactive protein: evidence for the cholinergic anti-inflammatory pathway in healthy human adults", *J.Intern.Med.*, vol. 265, no. 4, pp. 439-447.

The DCCT Research Group 1993, "The effect of intensive treatment of diabetes on the development and progression of long-term complications in insulin-dependent diabetes mellitus. The Diabetes Control and Complications Trial Research Group", *N.Engl.J.Med.*, vol. 329, no. 14, pp. 977-986.

The DECODE Study Group 1999, "Glucose tolerance and mortality: comparison of WHO and American Diabetes Association diagnostic criteria. The DECODE study group. European Diabetes Epidemiology Group. Diabetes Epidemiology: Collaborative analysis Of Diagnostic criteria in Europe", *Lancet*, vol. 354, no. 9179, pp. 617-621.

The DECODE Study Group 2001, "Glucose tolerance and cardiovascular mortality: comparison of fasting and 2-hour diagnostic criteria", *Arch.Intern.Med*, vol. 161, no. 3, pp. 397-405.

Thomas, S. L., Hyde, J., Karunaratne, A., Kausman, R., & Komesaroff, P. A. 2008, ""They all work...when you stick to them": a qualitative investigation of dieting, weight loss, and physical exercise, in obese individuals", *Nutr.J.*, vol. 7, p. 34.

Thompson, P. D., Crouse, S. F., Goodpaster, B., Kelley, D., Moyna, N., & Pescatello, L. 2001, "The acute versus the chronic response to exercise", *Med.Sci.Sports Exerc.*, vol. 33, no. 6 Suppl, p. S438-S445.

Tomiyama, H. & Yamashina, A. 2010, "Non-invasive vascular function tests: their pathophysiological background and clinical application", *Circ.J.*, vol. 74, no. 1, pp. 24-33.

Tracey, K. J. 2002, "The inflammatory reflex", Nature, vol. 420, no. 6917, pp. 853-859.

Tsochatzis, E. A., Papatheodoridis, G. V., & Archimandritis, A. J. 2009, "Adipokines in nonalcoholic steatohepatitis: from pathogenesis to implications in diagnosis and therapy", *Mediators.Inflamm.*, vol. 2009, p. 831670.

Tuomilehto, J., Lindstrom, J., Eriksson, J. G., Valle, T. T., Hamalainen, H., Ilanne-Parikka, P., Keinanen-Kiukaanniemi, S., Laakso, M., Louheranta, A., Rastas, M., Salminen, V., & Uusitupa, M. 2001, "Prevention of type 2 diabetes mellitus by changes in lifestyle among subjects with impaired glucose tolerance", *N.Engl.J Med*, vol. 344, no. 18, pp. 1343-1350.

Turcotte, L. P. & Fisher, J. S. 2008, "Skeletal muscle insulin resistance: roles of fatty acid metabolism and exercise", *Phys.Ther.*, vol. 88, no. 11, pp. 1279-1296.

Tushuizen, M. E., Bunck, M. C., Pouwels, P. J., Bontemps, S., van Waesberghe, J. H., Schindhelm, R. K., Mari, A., Heine, R. J., & Diamant, M. 2007, "Pancreatic fat content and beta-cell function in men with and without type 2 diabetes", *Diabetes Care*, vol. 30, no. 11, pp. 2916-2921.

Tziomalos, K., Weerasinghe, C. N., Mikhailidis, D. P., & Seifalian, A. M. 2008, "Vascular risk factors in South Asians", *Int.J.Cardiol.*, vol. 128, no. 1, pp. 5-16.

UKPDS Study Group 1998, "Intensive blood-glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). UK Prospective Diabetes Study (UKPDS) Group", *Lancet*, vol. 352, no. 9131, pp. 837-853.

Unwin, N., Shaw, J., Zimmet, P., & Alberti, K. G. 2002, "Impaired glucose tolerance and impaired fasting glycaemia: the current status on definition and intervention", *Diabet.Med.*, vol. 19, no. 9, pp. 708-723.

Uotani, S., Abe, T., & Yamaguchi, Y. 2006, "Leptin activates AMP-activated protein kinase in hepatic cells via a JAK2-dependent pathway", *Biochem.Biophys.Res.Commun.*, vol. 351, no. 1, pp. 171-175.

Valencia, M. E., Bennett, P. H., Ravussin, E., Esparza, J., Fox, C., & Schulz, L. O. 1999, "The Pima Indians in Sonora, Mexico. [Review] [15 refs]", *Nutrition Reviews*, vol. 57, no. 5 Pt 2, p. S55-S57.

van Dam, R. M., Boer, J. M., Feskens, E. J., & Seidell, J. C. 2001, "Parental history of diabetes modifies the association between abdominal adiposity and hyperglycemia", *Diabetes Care*, vol. 24, no. 8, pp. 1454-1459.

van der, K. K. & Seidell, J. C. 1993, "Techniques for the measurement of visceral fat: a practical guide", *Int.J Obes.Relat Metab Disord.*, vol. 17, no. 4, pp. 187-196.

van Harmelen, V., Dicker, A., Ryden, M., Hauner, H., Lonnqvist, F., Naslund, E., & Arner, P. 2002, "Increased lipolysis and decreased leptin production by human omental as compared with subcutaneous preadipocytes", *Diabetes*, vol. 51, no. 7, pp. 2029-2036.

van, H. G., Steensberg, A., Sacchetti, M., Fischer, C., Keller, C., Schjerling, P., Hiscock, N., Moller, K., Saltin, B., Febbraio, M. A., & Pedersen, B. K. 2003, "Interleukin-6 stimulates lipolysis and fat oxidation in humans", *J.Clin.Endocrinol.Metab*, vol. 88, no. 7, pp. 3005-3010.

Vauhkonen, I., Niskanen, L., Haffner, S., Kainulainen, S., Uusitupa, M., & Laakso, M. 1998, "Insulin resistant phenotype is associated with high serum leptin levels in offspring of patients with non-insulin-dependent diabetes mellitus", *Eur.J.Endocrinol.*, vol. 139, no. 6, pp. 598-604.

Venables, M. C. & Jeukendrup, A. E. 2008, "Endurance training and obesity: effect on substrate metabolism and insulin sensitivity", *Med.Sci.Sports Exerc.*, vol. 40, no. 3, pp. 495-502.

Ventura, E., Davis, J., Byrd-Williams, C., Alexander, K., McClain, A., Lane, C. J., Spruijt-Metz, D., Weigensberg, M., & Goran, M. 2009, "Reduction in risk factors for type 2 diabetes mellitus in response to a low-sugar, high-fiber dietary intervention in overweight Latino adolescents", *Arch.Pediatr.Adolesc.Med.*, vol. 163, no. 4, pp. 320-327.

Vettor, R., Milan, G., Rossato, M., & Federspil, G. 2005, "Review article: adipocytokines and insulin resistance", *Aliment.Pharmacol.Ther.*, vol. 22 Suppl 2, pp. 3-10.

Vincent, M. A., Barrett, E. J., Lindner, J. R., Clark, M. G., & Rattigan, S. 2003, "Inhibiting NOS blocks microvascular recruitment and blunts muscle glucose uptake in response to insulin", *Am.J.Physiol Endocrinol.Metab*, vol. 285, no. 1, p. E123-E129.

Vincent, M. A., Clerk, L. H., Rattigan, S., Clark, M. G., & Barrett, E. J. 2005, "Active role for the vasculature in the delivery of insulin to skeletal muscle", *Clin.Exp.Pharmacol.Physiol*, vol. 32, no. 4, pp. 302-307.

Viollet, B., Lantier, L., vin-Leclerc, J., Hebrard, S., Amouyal, C., Mounier, R., Foretz, M., & Andreelli, F. 2009, "Targeting the AMPK pathway for the treatment of Type 2 diabetes", *Front Biosci.*, vol. 14, pp. 3380-3400.

Vogels, N. & Westerterp-Plantenga, M. S. 2007, "Successful long-term weight maintenance: a 2-year follow-up", *Obesity.(Silver.Spring)*, vol. 15, no. 5, pp. 1258-1266.

Volek, J. S., Phinney, S. D., Forsythe, C. E., Quann, E. E., Wood, R. J., Puglisi, M. J., Kraemer, W. J., Bibus, D. M., Fernandez, M. L., & Feinman, R. D. 2009, "Carbohydrate restriction has a more favorable impact on the metabolic syndrome than a low fat diet", *Lipids*, vol. 44, no. 4, pp. 297-309.

Voors, A. W., Webber, L. S., & Berenson, G. S. 1982, "Resting heart rate and pressure-rate product of children in a total biracial community: the Bogalusa Heart Study", *American Journal of Epidemiology*, vol. 116, no. 2, pp. 276-286.

Votruba, S. B., Atkinson, R. L., Hirvonen, M. D., & Schoeller, D. A. 2002, "Prior exercise increases subsequent utilization of dietary fat", *Med.Sci.Sports Exerc.*, vol. 34, no. 11, pp. 1757-1765.

Wagenmakers, A. J. 2005, "Insulin resistance in the offspring of parents with type 2 diabetes", *PLoS.Med.*, vol. 2, no. 9, p. e289.

Wagenmakers, A. J., van Riel, N. A., Frenneaux, M. P., & Stewart, P. M. 2006, "Integration of the metabolic and cardiovascular effects of exercise", *Essays Biochem.*, vol. 42, pp. 193-210.

Walker, G. E., Verti, B., Marzullo, P., Savia, G., Mencarelli, M., Zurleni, F., Liuzzi, A., & Di Blasio, A. M. 2007, "Deep subcutaneous adipose tissue: a distinct abdominal adipose depot", *Obesity.(Silver.Spring)*, vol. 15, no. 8, pp. 1933-1943.

Wang, X., Lyles, M. F., You, T., Berry, M. J., Rejeski, W. J., & Nicklas, B. J. 2008, "Weight regain is related to decreases in physical activity during weight loss", *Med.Sci.Sports Exerc.*, vol. 40, no. 10, pp. 1781-1788.

Wang, Y., Rimm, E. B., Stampfer, M. J., Willett, W. C., & Hu, F. B. 2005, "Comparison of abdominal adiposity and overall obesity in predicting risk of type 2 diabetes among men", *Am.J Clin.Nutr.*, vol. 81, no. 3, pp. 555-563.

Wannamethee, S. G., Lowe, G. D., Rumley, A., Cherry, L., Whincup, P. H., & Sattar, N. 2007, "Adipokines and risk of type 2 diabetes in older men", *Diabetes Care*, vol. 30, no. 5, pp. 1200-1205.

Warram, J. H., Martin, B. C., Krolewski, A. S., Soeldner, J. S., & Kahn, C. R. 1990, "Slow glucose removal rate and hyperinsulinemia precede the development of type II diabetes in the offspring of diabetic parents", *Ann.Intern.Med.*, vol. 113, no. 12, pp. 909-915.

Weber, J. L., Reid, P. M., Greaves, K. A., DeLany, J. P., Stanford, V. A., Going, S. B., Howell, W. H., & Houtkooper, L. B. 2001, "Validity of self-reported energy intake in lean and obese young women, using two nutrient databases, compared with total energy expenditure assessed by doubly labeled water", *Eur.J.Clin.Nutr.*, vol. 55, no. 11, pp. 940-950.

Westerterp, K. R. & Goris, A. H. 2002, "Validity of the assessment of dietary intake: problems of misreporting", *Curr.Opin.Clin.Nutr.Metab Care*, vol. 5, no. 5, pp. 489-493.

Westerterp, K. R. & Speakman, J. R. 2008, "Physical activity energy expenditure has not declined since the 1980s and matches energy expenditures of wild mammals", *Int.J.Obes.(Lond)*, vol. 32, no. 8, pp. 1256-1263.

Weststrate, J. A., Dekker, J., Stoel, M., Begheijn, L., Deurenberg, P., & Hautvast, J. G. 1990, "Resting energy expenditure in women: impact of obesity and body-fat distribution", *Metabolism*, vol. 39, no. 1, pp. 11-17.

Weyer, C., Funahashi, T., Tanaka, S., Hotta, K., Matsuzawa, Y., Pratley, R. E., & Tataranni, P. A. 2001, "Hypoadiponectinemia in obesity and type 2 diabetes: close association with insulin resistance and hyperinsulinemia", *J Clin.Endocrinol.Metab*, vol. 86, no. 5, pp. 1930-1935.

Weyer, C., Pratley, R. E., Salbe, A. D., Bogardus, C., Ravussin, E., & Tataranni, P. A. 2000, "Energy expenditure, fat oxidation, and body weight regulation: a study of metabolic adaptation to long-term weight change", *J.Clin.Endocrinol.Metab*, vol. 85, no. 3, pp. 1087-1094.

Whincup, P. H., Gilg, J. A., Owen, C. G., Odoki, K., Alberti, K. G., & Cook, D. G. 2005, "British South Asians aged 13-16 years have higher fasting glucose and insulin levels than Europeans", *Diabet.Med*, vol. 22, no. 9, pp. 1275-1277.

Whiteley, L., Padmanabhan, S., Hole, D., & Isles, C. 2005, "Should diabetes be considered a coronary heart disease risk equivalent?: results from 25 years of follow-up in the Renfrew and Paisley survey", *Diabetes Care*, vol. 28, no. 7, pp. 1588-1593.

WHO Consultation 1968, "Exercise tests in relation to cardiovascular function. Report of a WHO meeting", *World Health Organ Tech.Rep.Ser.*, vol. 388, pp. 1-30.

WHO Consultation. Definition, Diagnosis and Classification of Diabetes Mellitus and its Complications. World Health Organisation, Department of Non-Communicable Disease Surveillance, Geneva. 1-65. 1999.

Wild, S., Roglic, G., Green, A., Sicree, R., & King, H. 2004, "Global prevalence of diabetes: estimates for the year 2000 and projections for 2030", *Diabetes Care*, vol. 27, no. 5, pp. 1047-1053.

Wilkinson, I. B. & Webb, D. J. 2001, "Venous occlusion plethysmography in cardiovascular research: methodology and clinical applications", *Br.J.Clin.Pharmacol.*, vol. 52, no. 6, pp. 631-646.

Williamson, D. H. & Whitelaw, E. 1978, "Physiological aspects of the regulation of ketogenesis", *Biochem.Soc.Symp.* no. 43, pp. 137-161.

Wilmore, J. H. & Costill, D. L. 1973, "Adequacy of the Haldane transformation in the computation of exercise V O2 in man", *J.Appl.Physiol*, vol. 35, no. 1, pp. 85-89.

Wilmore, J. H., Despres, J. P., Stanforth, P. R., Mandel, S., Rice, T., Gagnon, J., Leon, A. S., Rao, D., Skinner, J. S., & Bouchard, C. 1999, "Alterations in body weight and composition consequent to 20 wk of endurance training: the HERITAGE Family Study", *Am.J.Clin.Nutr.*, vol. 70, no. 3, pp. 346-352.

Wilson, A. M., Ryan, M. C., & Boyle, A. J. 2006, "The novel role of C-reactive protein in cardiovascular disease: risk marker or pathogen", *Int.J.Cardiol.*, vol. 106, no. 3, pp. 291-297.

Winnick, J. J., Sherman, W. M., Habash, D. L., Stout, M. B., Failla, M. L., Belury, M. A., & Schuster, D. P. 2008, "Short-term aerobic exercise training in obese humans with type 2 diabetes mellitus improves whole-body insulin sensitivity through gains in peripheral, not hepatic insulin sensitivity", *J.Clin.Endocrinol.Metab*, vol. 93, no. 3, pp. 771-778.

Wojtaszewski, J. F., Hansen, B. F., Gade, Kiens, B., Markuns, J. F., Goodyear, L. J., & Richter, E. A. 2000, "Insulin signaling and insulin sensitivity after exercise in human skeletal muscle", *Diabetes*, vol. 49, no. 3, pp. 325-331.

Wojtaszewski, J. F. & Richter, E. A. 2006, "Effects of acute exercise and training on insulin action and sensitivity: focus on molecular mechanisms in muscle", *Essays Biochem.*, vol. 42, pp. 31-46.

Wolf, G. 2008, "Role of fatty acids in the development of insulin resistance and type 2 diabetes mellitus", *Nutr.Rev.*, vol. 66, no. 10, pp. 597-600.

Woodman, R. J., Watts, G. F., Playford, D. A., Best, J. D., & Chan, D. C. 2005, "Oxidized LDL and small LDL particle size are independently predictive of a selective defect in microcirculatory endothelial function in type 2 diabetes", *Diabetes Obes.Metab*, vol. 7, no. 5, pp. 612-617.

Wooten, J. S., Biggerstaff, K. D., & Anderson, C. 2008, "Response of lipid, lipoproteincholesterol, and electrophoretic characteristics of lipoproteins following a single bout of aerobic exercise in women", *Eur.J.Appl.Physiol*, vol. 104, no. 1, pp. 19-27.

Wright, D. C., Geiger, P. C., Holloszy, J. O., & Han, D. H. 2005, "Contraction- and hypoxiastimulated glucose transport is mediated by a Ca2+-dependent mechanism in slow-twitch rat soleus muscle", *Am.J.Physiol Endocrinol.Metab*, vol. 288, no. 6, p. E1062-E1066.

Yajnik, C. S., Fall, C. H., Vaidya, U., Pandit, A. N., Bavdekar, A., Bhat, D. S., Osmond, C., HALES, C. N., & Barker, D. J. 1995, "Fetal growth and glucose and insulin metabolism in four-year-old Indian children", *Diabet.Med.*, vol. 12, no. 4, pp. 330-336.

Yamauchi, T. & Kadowaki, T. 2008, "Physiological and pathophysiological roles of adiponectin and adiponectin receptors in the integrated regulation of metabolic and cardiovascular diseases", *Int.J.Obes.(Lond)*, vol. 32 Suppl 7, p. S13-S18.

Yamauchi, T., Kamon, J., Minokoshi, Y., Ito, Y., Waki, H., Uchida, S., Yamashita, S., Noda, M., Kita, S., Ueki, K., Eto, K., Akanuma, Y., Froguel, P., Foufelle, F., Ferre, P., Carling, D., Kimura, S., Nagai, R., Kahn, B. B., & Kadowaki, T. 2002, "Adiponectin stimulates glucose utilization and fatty-acid oxidation by activating AMP-activated protein kinase", *Nat.Med.*, vol. 8, no. 11, pp. 1288-1295.

Yamauchi, T., Kamon, J., Waki, H., Terauchi, Y., Kubota, N., Hara, K., Mori, Y., Ide, T., Murakami, K., Tsuboyama-Kasaoka, N., Ezaki, O., Akanuma, Y., Gavrilova, O., Vinson, C., Reitman, M. L., Kagechika, H., Shudo, K., Yoda, M., Nakano, Y., Tobe, K., Nagai, R., Kimura, S., Tomita, M., Froguel, P., & Kadowaki, T. 2001, "The fat-derived hormone adiponectin reverses insulin resistance associated with both lipoatrophy and obesity", *Nat.Med*, vol. 7, no. 8, pp. 941-946.

Yatagai, T., Nishida, Y., Nagasaka, S., Nakamura, T., Tokuyama, K., Shindo, M., Tanaka, H., & Ishibashi, S. 2003, "Relationship between exercise training-induced increase in insulin sensitivity and adiponectinemia in healthy men", *Endocr.J*, vol. 50, no. 2, pp. 233-238.

Ybarra, J., Blanco-Vaca, F., Fernandez, S., Castellvi, A., Bonet, R., Palomer, X., Ordonez-Llanos, J., Trius, A., Vila-Rovira, R., & Perez, A. 2008, "The effects of liposuction removal of subcutaneous abdominal fat on lipid metabolism are independent of insulin sensitivity in normal-overweight individuals", *Obes.Surg.*, vol. 18, no. 4, pp. 408-414. Yeckel, C. W., Gulanski, B., Zgorski, M. L., Dziura, J., Parish, R., & Sherwin, R. S. 2009, "Simple exercise recovery index for sympathetic overactivity is linked to insulin resistance", *Med.Sci.Sports Exerc.*, vol. 41, no. 3, pp. 505-515.

Yim, J. E., Heshka, S., Albu, J., Heymsfield, S., Kuznia, P., Harris, T., & Gallagher, D. 2007, "Intermuscular adipose tissue rivals visceral adipose tissue in independent associations with cardiovascular risk", *Int.J.Obes.(Lond)*, vol. 31, no. 9, pp. 1400-1405.

Young, D. A., Wallberg-Henriksson, H., Sleeper, M. D., & Holloszy, J. O. 1987, "Reversal of the exercise-induced increase in muscle permeability to glucose", *Am.J.Physiol*, vol. 253, no. 4 Pt 1, p. E331-E335.

Young, J., Morbois-Trabut, L., Couzinet, B., Lascols, O., Dion, E., Bereziat, V., Feve, B., Richard, I., Capeau, J., Chanson, P., & Vigouroux, C. 2005, "Type A insulin resistance syndrome revealing a novel lamin A mutation", *Diabetes*, vol. 54, no. 6, pp. 1873-1878.

Young, J. C., Garthwaite, S. M., Bryan, J. E., Cartier, L. J., & Holloszy, J. O. 1983, "Carbohydrate feeding speeds reversal of enhanced glucose uptake in muscle after exercise", *Am.J.Physiol*, vol. 245, no. 5 Pt 1, p. R684-R688.

Youngren, J. F., Keen, S., Kulp, J. L., Tanner, C. J., Houmard, J. A., & Goldfine, I. D. 2001, "Enhanced muscle insulin receptor autophosphorylation with short-term aerobic exercise training", *Am.J.Physiol Endocrinol.Metab*, vol. 280, no. 3, p. E528-E533.

Yu, C. M., Lau, C. P., Cheung, B. M., Fong, Y. M., Ho, Y. Y., Lam, K. B., & Li, L. S. 2000, "Clinical predictors of morbidity and mortality in patients with myocardial infarction or revascularization who underwent cardiac rehabilitation, and importance of diabetes mellitus and exercise capacity", *Am.J.Cardiol.*, vol. 85, no. 3, pp. 344-349.

Yu, M., Blomstrand, E., Chibalin, A. V., Wallberg-Henriksson, H., Zierath, J. R., & Krook, A. 2001, "Exercise-associated differences in an array of proteins involved in signal transduction and glucose transport", *J.Appl.Physiol*, vol. 90, no. 1, pp. 29-34.

Zahorska-Markiewicz, B., Olszanecka-Glinianowicz, M., Janowska, J., & Kocelak, P. 2008, "The effect of weight loss on serum concentrations of FAS and tumour necrosis factor alpha in obese women", *Endokrynol.Pol.*, vol. 59, no. 1, pp. 18-22.

Zhang, Q., Wang, Y., & Huang, E. S. 2009, "Changes in racial/ethnic disparities in the prevalence of Type 2 diabetes by obesity level among US adults", *Ethn.Health* pp. 1-19.

Ziccardi, P., Nappo, F., Giugliano, G., Esposito, K., Marfella, R., Cioffi, M., D'Andrea, F., Molinari, A. M., & Giugliano, D. 2002, "Reduction of inflammatory cytokine concentrations and improvement of endothelial functions in obese women after weight loss over one year", *Circulation*, vol. 105, no. 7, pp. 804-809.

Zierath, J. R. 2002, "Invited review: Exercise training-induced changes in insulin signaling in skeletal muscle", *J.Appl.Physiol*, vol. 93, no. 2, pp. 773-781.

Zierath, J. R., Krook, A., & Wallberg-Henriksson, H. 2000, "Insulin action and insulin resistance in human skeletal muscle", *Diabetologia*, vol. 43, no. 7, pp. 821-835.

Zurlo, F., Lillioja, S., Esposito-Del, P. A., Nyomba, B. L., Raz, I., Saad, M. F., Swinburn, B. A., Knowler, W. C., Bogardus, C., & Ravussin, E. 1990, "Low ratio of fat to carbohydrate oxidation as predictor of weight gain: study of 24-h RQ", *Am.J.Physiol*, vol. 259, no. 5 Pt 1, p. E650-E657.

# APPENDICES

- Appendix A: Information forms for patients and general practitioners, consent forms and screening questionnaires.
- Appendix B: Methods used to obtain subcutaneous abdominal adipose sample by liposuction biopsy technique. Exercise planner. Food diary. Physical activity diary
- Appendix C: 'Delayed Exercise' results. Absolute change in variables after seven-week period of 'usual lifestyle'.

# **APPENDIX A1**

# HEALTH SCREEN FOR STUDY VOLUNTEERS (version 3, 11/08/2004)

# Name: .....

It is important that volunteers participating in research studies are currently in good health and have had no significant medical problems in the past. This is to ensure (i) their own continuing well-being and (ii) to avoid the possibility of individual health issues confounding study outcomes.

Please complete this brief questionnaire to confirm fitness to participate:

1. At present, do you have any health problem for which you are:					
	(a) on medication, prescribed or otherwise	yes [	]	no [	]
	(b) attending your general practitioner	yes [	]	no [	]
	(c) on a hospital waiting list	yes [	]	no [	]
2.	<b>In the past two years,</b> have you had any illness which requ (a) consult your GP	ired you yes [	to: ]	no [	]
	(b) attend a hospital outpatient department	yes [	]	no [	]
	(c) be admitted to hospital	yes [	]	no [	]
3	Have you ever had any of the following.				
5.	(a) Convulsions/epilepsy	yes [	]	no [	]
	(b) Asthma	yes [	]	no [	]
	(c) Eczema	yes [	]	no [	]
	(d) Diabetes	yes [	]	no [	]
	(e) A blood disorder	yes [	]	no [	]
	(f) Head injury	yes [	]	no [	]
	(g) Digestive problems	yes [	]	no [	]
	(h) Hearing problems	yes [	]	no [	]

	(i) Problems with bones or joints	yes [	]	no [	]	
	(j) Disturbance of balance/co-ordination	yes [	]	no [	]	
	(k) Numbness in hands or feet	yes [	]	no [	]	
	(l) Disturbance of vision	yes [	]	no [	]	
	(m) Thyroid problems	yes [	]	no [	]	
	(n) Kidney or liver problems	yes [	]	no [	]	
	(o) Chest pain or heart problems	yes [	]	no [	]	
	(p) Any other health problems	yes [	]	no [	]	
4.	(a) Are you pregnant or think that you might be pregnant?	yes [	]	no [	]	
	(b) Do you take the contraceptive pill or other hormone-based	contrac	eptives?	yes [ ] n	0[]	
	(c) Are you postmenopausal?		yes [	]	no [	]
	(d) Are you receiving Hormone Replacement Therapy (HR	T)?	yes [	]	no [	]
5.	Have any of your family (parents, grandparents, brother	rs, sisteı	rs, child	ren, aur	nts, uncl	les,
	cousins) ever had any of the following: (if yes please give d	letails in	cluding	age of fi	rst diag	nosis)

	(a) Any heart problems	yes [	J	no [	]
	(b) Diabetes	yes [	]	no [	]
	(c) Stroke	yes [	]	no [	]
	(d) Any other family illnesses	yes [	]	no [	]
6.	Have you had any children ?	yes [	]	no [	]
	If so, how many ?				
	If so, when was your last child born ?				
	b) Diabetes yes [] no [   c) Stroke yes [] no [   d) Any other family illnesses yes [] no [   Have you had any children ? yes [] no [   If so, how many ?   If so, when was your last child born ?   If so, can you detail the birth weights of all your children				
					•••

7.	Do you currently smoke	yes [	]	no [	]	
	Have you ever smoked	yes [	]	no [	]	
	If so, for how long did you smoke and when did you stop?					

.....

8. How many units of alcohol do you typically drink in a week? .....

If YES to any question, please describe briefly if you wish (e.g. to confirm whether problem was short-lived, insignificant or well controlled.) (Use a separate sheet if necessary)

Name and address of GP

Blood pressure measured at screening	mm Hg

# **APPENDIX A2**

# Version 3, 8<sup>th</sup> June 2004

# **VOLUNTEER INFORMATION SHEET**

# Effects of a physical activity programme on insulin resistance and the function of

## adipose (fat) tissue in women with and without a family history of type 2 diabetes.

You are being invited to take part in a research study. Before you decide it is important for you to understand why the research is being done and what it will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information. Take time to decide whether or not you wish to take part.

Thank you for reading this.

## What is the purpose of the study?

The children of parents with type 2 (late-onset) diabetes are themselves at increased risk of developing diabetes and, although not diabetic, often have changes in their metabolism making insulin less effective at controlling their blood sugar levels. This can lead to an increased risk of heart disease and type 2 diabetes. This study will investigate whether a programme of moderate exercise can be effective in normalising the function of insulin in these individuals and will help us to understand how exercise can reduce someone's risk of developing type 2 diabetes and heart disease. The entire study will take 3 years but each person will be involved in the project for about 10 weeks.

#### Why have I been chosen?

You have been chosen because you are a healthy women aged between 20-45 years who either has at least one parent with type 2 diabetes or has no family history of type 2 diabetes.

#### Do I have to take part?

It is up to you to decide whether or not to take part. If you do decide to take part you will be given this information sheet to keep and be asked to sign a consent form. If you decide to take part you are still free to withdraw at any time and without giving a reason. A decision to withdraw at any time, or a decision not to take part, will not affect the standard of care you receive.

# What will happen to me if I take part?

This is a randomised trial to determine how effective exercise will be at improving the function of insulin in the body and reducing risk for late-onset diabetes and heart disease. You will be asked to either undertake a 7-week programme of moderate exercise (exercise group), <u>or to maintain your</u> normal lifestyle (i.e. undertake no planned exercise) for 7-weeks, before undertaking the same 7-week moderate exercise programme as the exercise group (delayed exercise group). Results from the exercise and normal lifestyle periods will then be compared. You will be assigned to the exercise or delayed exercise group by chance – i.e. you cannot choose which group you go into.

# What do I have to do?

1) In the first instance you will be asked to attend for a screening visit in which we will:

- discuss with you and complete confidential questionnaires regarding your health, family history, physical activity and diet
- measure your blood pressure
- provide an opportunity for you to ask questions
- take a small blood sample (20 ml or 4 teaspoons) to check the fat and sugar levels in your blood. We will also check for any abnormalities in the function of your liver, kidneys and thyroid gland to ensure that it is perfectly safe for you to participate in this study.

After this visit you will be assigned to the exercise or the sedentary group.

2) We will then ask you to undertake a number of tests to determine your level of health and fitness. These will include:

- Exercise tests The first test will involve walking on a treadmill with self-adhesive electrodes attached to your chest to monitor how hard your heart is working. This test will last a maximum of 12 minutes and will confirm whether it is safe for you to take part in this study. We will then ask you to perform a second test where you will walk or run on a treadmill (depending on your fitness level) at different speeds and we will monitor your heart rate and collect the air you breathe out to determine your fitness level. We will also take tiny 'finger prick' blood samples during this test, which will not involve a maximal effort and will last about 20 to 30 minutes.
- Monitoring physical activity and diet we will ask you to weigh and record everything that you eat and drink for a week and to record all your activities during this period in a diary. We will also ask you to wear a heart rate monitor during all your waking hours this week (except for in the shower/bath) and use this information to calculate how many calories you burn during normal daily living. To enable us to do this calculation, we will need to measure your heart rate and collect the air you breathe out during a range of activities (such as lying, sitting, standing and walking) in the laboratory. You should allow about 2 hours to complete these laboratory tests.
- Assessment of body composition we will measure your weight and height and measure around your waist and your hips. We will also perform a scan of your body to

calculate how much fat you are carrying. These tests will take about 15 minutes in total and can be carried out on the same day as other tests.

- **Glucose tolerance test** we will ask you to come to the laboratory after an overnight fast and drink a sugary drink. We will take blood samples before the drink and for two hours after the drink to assess how your body deals will this sugar. This will help us to determine how well the insulin in your body is working. This will be no more painful than a simple blood test as samples will be obtained from a tiny plastic tube called a 'cannula' placed in a forearm vein. We will take approximately 120 ml (about an eggcup full) of blood over the course of this test.
- Fat biopsy when you come to the laboratory for your glucose tolerance test, we will also take a small fat sample from your tummy to study how exercise influences the production of various factors influencing the metabolism of fat tissue. This may help us to understand the way in which exercise can the reduce risk of diabetes and heart disease. The fat biopsy will be taken by 'liposuction' and involves numbing the skin with a local anaesthetic then sucking up 1 to 2 grams of fat (about the size of a pen top) with a needle and syringe. This is no more painful than having a blood sample taken.

3) If you have been assigned to the exercise group, you will then begin your 7-week exercise programme. Your exercise programme will be individually designed according to your current level of fitness, starting with a small amount of exercise in the first week and building up to a maximum of 5 sessions per week by the end of the programme. The exercise sessions will all be at an intensity which is comfortable for you. You will perform your exercise sessions in the newly re-furbished Glasgow University sports facilities at the Stevenson Building and the Garscube Sports Complex. These are open from 7:30am until 10pm from Monday to Friday, from 9am until 6 pm on Saturdays and from 10am until 6pm on Sundays, so you can schedule your exercise sessions to fit in with your other time commitments. You will meet with an investigator at least once per week for a supervised exercise session and to discuss your exercise goals for the following week. It is important that you maintain your usual eating habits during your exercise programme. At the end of the programme you will undergo the same set of tests that you did at the start of the study, i.e. exercise tests, monitoring physical activity and diet, assessment of body composition, glucose tolerance test and fat biopsy. The glucose tolerance test and fat biopsy will be performed twice at the end of the programme; once on the day following your final exercise session and again three days later. This will help us to determine how long the benefits of exercise last when you stop exercising.

4) If you have been assigned to the <u>delayed exercise</u> group, we will ask you to maintain your usual dietary and exercise habits (i.e. undertake no planned exercise) for this 7-week period. <u>At the end of this period you will undergo the same set of tests that you did at the start of the study, i.e. exercise tests, monitoring physical activity and diet, assessment of body composition, glucose tolerance test and fat biopsy. You will then be asked to undertake the same 7-week exercise programme as the exercise group and ONE further set of tests will be performed at the end of this period.</u>

#### What are the possible disadvantages and risks of taking part?

• Exercise testing will not be at a maximal level but the possibility exists that, very occasionally, certain changes may occur during or shortly after the tests. They include abnormal blood pressure, fainting or a change in the normal rhythm of the heartbeat.

- Blood sampling via the cannula may cause minor bruising, an inflammation of the vein or haematoma (a small accumulation of blood under the skin). Good practice, however, minimises this risk. Some people may feel faint when they give blood.
- Taking the fat biopsy carries a small risk of minor bruising or an infection. Good sterile practice reduces this risk.
- The scan to determine your level of body fat uses a very small dose of radiation. Each scan is equivalent to about 1/20 of a chest X-ray or the amount of natural background radiation that we are exposed to in 4 hours. However, if you are pregnant, or think you might be pregnant, you should inform us as this radiation could harm an unborn baby.
- There is a small possibility that taking part in this study will reveal a health problem that you already have such as high cholesterol or high blood pressure. If such a problem is revealed, we will inform your GP to ensure that you receive appropriate treatment. If necessary, a consultant physician (Dr Steve Cleland) will supervise any follow-up care.

# What are the possible benefits of taking part?

Diabetes is a major and growing health problem in Scotland and understanding how we can prevent diabetes is important to individuals at risk of diabetes, the health service and the wider economy. This study aims to increase understanding about how exercise influences the function of insulin, which is an important factor in the development of diabetes. The findings will be published in scientific journals so that understanding of the way in which exercise decreases the risk of diabetes can be increased. This information may help make up better exercise guidelines, particularly for those with parents who have type 2 diabetes.

We will provide you with feedback about the main study findings and also about your own results and would be delighted to explain results and discuss the implications with you.

#### What if something goes wrong?

The chance of something going wrong are extremely small. All of the procedures involved in this study are low risk and our screening tests are designed to ensure that you will only participate if it is safe for you to do so. In the unlikely event that you are harmed due to someone's negligence, then you may have grounds for a legal action but you may have to pay for it. Regardless of this, if you wish to complain, or have any concerns about any aspect of the way you have been approached or treated during the course of this study, the normal National Health Service complaints mechanisms may be available to you.

#### Will my taking part in this study be kept confidential?

All information which is collected about you during the course of the research will be kept strictly confidential. Any information about you which leaves the University will have your name and address removed so that you cannot be recognised from it.

#### Who is organising and funding the research?

This study is being funded by the British Heart Foundation.

#### What will happen to my samples after the study has finished?

The blood and fat tissue samples that you provide for this study may be useful for future research into the prevention and treatment of diabetes and heart disease; this may involve analysis of certain genes associated with these diseases. Any use of your samples for future research will require further approval from a Research Ethics Committee and samples will be analysed in such a way that the results will not be directly traceable to you. If you do not wish your samples to be used for future research, please indicate this on the consent form.

## Who has reviewed the study?

This study has been reviewed and approved by the North Glasgow NHS Trust Research Ethics Committee.

# **Contact for Further Information**

Any questions about the procedures used in this study are encouraged. If you have any doubts or questions, please ask for further explanations by contacting <u>Dr Nicholas Barwell</u> on <u>07966 434471</u> /<u>0141 330 6588</u> or emailing <u>Nick.Barwell@tiscali.co.uk or N.Barwell@bio.gla.ac.uk</u>.

You will be given a copy of this information sheet and a signed consent form to keep for your records.

# **APPENDIX A3**



# CONSENT FORM

Title of Project: Effects of a physical activity programme on insulin resistance and the function of adipose (fat) tissue in women with and without a family history of type 2 diabetes.

Name of Researcher:\_\_\_\_\_

1.	I confirm that I have read and understand the information sheet dated <u>8</u> June 2004 (version 3) for the above study and		
	have had the opportunity to ask questions.		
2.	I understand that my participation is voluntary and that I am free to withdraw at any time, without giving any reason, without my medical		
	care or legal rights being affected.		
3.	I agree to take part in the above study.		
4.	I agree for my samples to be used for future research into the prevention and treatment of diabetes and heart disease. This may involve analysis	Yes	
	of genes associated with these diseases.	Г	
		No	



UNIVERSITY of GLASGOW

**Please initial box** 

Name of Patient	Date	Signature
Name of Person taking consent (if different fr	Date om researcher)	Signature
Researcher	Date	Signature

1 for patient; 1 for researcher; 1 to be kept with hospital notes
#### **APPENDIX A4**

Name

Address 1

Address 2

Date

<u>RE:</u>

Dear Dr

Your patient, Mrs ....., has kindly volunteered for a British Heart Foundation funded clinical trial based at the Institute of Biomedical and Life Sciences in the University of Glasgow. The title of this trial is 'Physical activity, insulin resistance and adipose tissue gene expression in the offspring of patients with type 2 diabetes'. We aim to assess whether a programme of moderate exercise will produce greater changes in insulin sensitivity and adipocyte gene expression in people with a family history of diabetes compared with matched control subjects. The findings may help inform future public health strategies for diabetes prevention.

I enclose a copy of the subject information sheet, and your copy of the patient's consent form. I also enclose details of the initial screening results for your records. I hope to be in contact with you as the study progresses, because some of the detailed physiological, biochemical and anthropometrical measurements that we make over the course of the study may be of some use to you.

If, in the course of screening and testing, we uncover factors which warrant medical input or follow up, these will be discussed with Dr Steve Cleland (Consultant Physician, Diabetes and Endocrinology – Stobhill Hospital) and we will notify you. If you have any questions or wish to contact me for further information, I would be delighted to assist by telephone, email, letter or in person.

Yours sincerely

Dr Nick Barwell MRCP

Adipose tissue was required to provide a sample for adipose tissue gene expression. Sampling was performed at between the hours of 09:00 and 10:00, (if the subject's metabolic test had commenced at 08:00) or between the hours of 11:00 and 12:00 (if the subject's metabolic test had commenced at 10:00). The subject was still fasting at the time of the biopsy, and had been reclining in the supine position for approximately 1 hour prior to the biopsy.

Before the biopsy was performed, subjects were asked about a previous history of allergic reaction to local anaesthetic, iodine, adhesive dressings and metals. The biopsy was performed by a qualified medical practitioner, the sample was processed by an assistant under the supervision of the medical practitioner. Having ensured no contraindication to the procedure, the subject's abdomen was exposed, draped with a sterile towel and the skin cleaned with Betadine solution (Seton Healthcare Group, Oldham, UK). An area of approximately 5 cm by 5 cm was anaesthetised with 10mls of 1% plain lignocaine (Braun Melsungen AG, Melsungen, Germany). The anaesthetised area was approximately 5 - 10 cm lateral to the umbilicus. After a 5 minute period to allow the anaesthetic to take effect, the adipose sample was taken from the anaesthetised area.

The sample was obtained using a 14G Braun Sterican needle (Braun Melsungen AG, Melsungen, Germany) by liposuction technique. This technique relies on the creation of a vacuum by the application of negative pressure to a 20ml syringe attached to the biopsy needle. After insertion of the needle subcutaneously, negative pressure was applied to the syringe, creating a constant suction pressure. The subcutaneous adipose tissue was released by a gentle rocking motion with the needle to break up the aggregates of adipose. The adipose sample was drawn up the needle into the syringe by the negative pressure.

325

Due to the narrow gauge of the biopsy needle, only a small amount of adipose could be obtained with one pass of the needle. Less than 2mls of sample tissue could be obtained from one pass. The sample was ejected from the needle on to a prepared tub covered with muslin. The assistant washed the sample with physiological saline (0.9% Saline for Irrigation, Baxter Healthcare, Glasgow, UK) and separated the adipose tissue from other tissues with forceps.

In order to increase the adipose yield, the same puncture site was used on three further occasions to attempt to obtain 1g of adipose tissue. Once washed and separated, the adipose tissue was divided into 3 portions and placed in labelled eppendorfs, then immediately snap frozen in liquid nitrogen. In all cases the optimum time from removal of the tissue from the body to snap freezing was less than 5 minutes.

After completion of the biopsy, a sterile dressing was applied to the puncture site, and pressure over the area applied with an ice pack. Pressure was maintained for 10 minutes to encourage the internal aspect of the area to seal and minimise bruising and bleeding. After snap freezing, the adipose samples were stored in a -80°C freezer.

## EXERCISE PLANNER

Week No. 6

Week Beginning 6.2.06

Subject Name

#### TARGET HEART RATE: 140 - 160

<u>Aims</u>

Achieve Target Heart Rate for \_\_\_\_\_60\_\_\_minutes, \_\_\_\_5\_\_\_ times in the next week.

Day	Time of Day	Exercise Performed
Monday		
Tuesday		
Wednesday		
Thursday		
Friday		
Saturday		
Sunday		

#### Remember!

Wear your heart rate monitor and watch for each exercise session.

Try to keep your heart rate in the target zone for the whole period of your exercise.

Remember to press the big red button twice to begin your exercise session.

Your next meeting with an exercise consultant is

ON :\_\_\_\_\_

AT :\_\_\_\_\_ WITH :\_\_\_\_\_

Any problems – phone Nick on 07966 434471

#### FOOD INVENTORY INSTRUCTIONS

It is important that you weigh and record everything that you eat and drink for each day of your monitoring week.

Please (i) start a separate page for each day.

(ii) start a separate line for each item.

Column 1

Record meal and time and place of eating.

Column 2

Describe each item as accurately as possible, stating where relevant:

- (i) type and brand
- (ii) whether food is fresh, dried, canned, frozen, salted, smoked, etc.
- (iii) whether food is cooked, if so give method of cooking e.g. fried, baked, etc.

#### Column 3

Record the weight of each item after cooking:

- (i) place scales on a level surface
- (ii) place plate or container on top of scales
- (iii) press 'ON/Reset' button to turn on scales
- (iv) once zero appears, add first item of food
- (v) record weight displayed
- (vi) press reset button before weighing next item

Wherever possible, record weights in grams. If this is not possible, record weights in household measures (e.g. sugar or jam in teaspoons, stating whether level, rounded, or heaped).

#### Column 4

Record the weight of any leftovers, such as food remaining on plate, weight of container in which food has been weighed, apple cores, etc.

Columns 5 and 6

Please leave blank.

If food consists of several items, please list each on a separate line i.e. instead of writing 'one cheese sandwich', record separately the weights of bread margarine, cheese, etc.

Please remember to record all drinks, as well as food, giving weights where possible, or volumes if these are known. Record separately the weights of added milk and sugar.

An example is shown overleaf.

# Food Inventory - Example

Name\_\_\_\_\_ Date \_\_\_\_\_

1.	2.	3. Weight of	4. Weight of	Leave
Time/Place	Description of food/drink	food/drink	container/	Blank
		(g)	leftovers (g)	
Breakfast	Cornflakes (Kelloggs)	28		
8:30am	Milk (Sainsbury's virtually fat-free)	48		
Home	Bread (Mothers Pride, large white	76		
	sliced, toasted)			
	Flora margarine	7		
	Robinsons lemon marmalade	12		
	Coffee (instant)	2		
	Milk (whole pasteurised)	10		
Lunch	Cheese (Cheddar)	55		
1:00pm	Bread (white, crusty)	76		
Pub	Butter	4		
	Chutney (2 teaspoons)			
Snack	Coffee (instant)	2		
3:30pm	Coffee-mate	6		
Office	Mars Bar	35		
	Apple	76	8 (core)	
Dinner	Turkey Fillet (frozen, grilled)	102		
6:30pm	Potatoes, old, boiled	320	74	
Home			(leftover)	
	Peas (Birds Eye, frozen, boiled)	50		
	Heinz tomato ketchup	14		
	Yoghurt (Ski strawberry thick and	162	10	
	creamy)		(carton)	
	Coffee, filter	148		
	Milk (Sainsbury's virtually fat-free)	8		
Snack	Banana	107		
7:45pm	Orange Tango (can)	330		
Home				

### **ACTIVITY DIARY**

The diary is split up into individual days. Each day is split into 15minute intervals. The boxes are large enough that you should be able to fill in a few things if required. From my point of view, the more detailed you can manage the better, but I would be happy if you can fill as much as possible. A spare day is provided.

You may not want to carry these diaries about with you if you are out. In this case feel free to write things down on smaller areas of paper and transfer them at the end of the day. As most people are sleeping at nights, the time periods are 30mins apart.

The first page is a sample morning. You can see there are lots of ways to let me know what you are doing, and cut corners if you're doing the same thing for a prolonged period.

Any questions – give me a call on 07966433471.

Thanks

Dr Nick Barwell

Time	ACTIVITY
0600	Sleep
0615	Sleep
0630	Sleep
0645	Sleep
0700	Got up, Showered, brushed teeth etc
0715	
0730	Dressed and made bed
0745	Breakfast in kitchen
0800	Tidy up flat
0815	Check bags and diary (sitting down)

Sample Day Date 23/9/04

0830	Watch news
0845	
0900	Walk to work
0915	Meeting with volunteer
0930	Tests
0945	Tests in lab (Standing and walking)
1000	
1015	
1030	
1045	Work at desk
1100	•
1115	
1130	Work at desk
1145	
1200	
1215	Toilet and drink break – walk to toilet
1230	Lunch at desk

#### **APPENDIX C**

**Table C.1** Absolute changes in physical characteristics and body composition of Control and Offspring after period of 'usual lifestyle'. Values are Mean ± SEM.

	Controls	Offspring	P - value
	( <b>n=16</b> )	( <b>n=13</b> )	
Body mass (kg) <sup>a</sup>	$0.18\pm0.25$	$0.52 \pm 0.30$	0.56
BMI $(kg.m^{-2})^a$	$0.066\pm0.091$	$0.191 \pm 0.111$	0.56
Total fat mass (kg)	$-0.08 \pm 0.29$	$-0.28 \pm 0.38$	0.67
Body fat (%)	$0.12\pm0.32$	$-0.82 \pm 0.26$	< 0.05
Trunk fat mass (kg) <sup>a</sup>	$0.10\pm0.17$	$-0.24 \pm 0.39$	0.24
Leg fat mass (kg) <sup>a</sup>	$-0.26 \pm 0.13$	$-0.12 \pm 0.15$	0.61
Arm fat mass (kg) <sup>a</sup>	$0.13 \pm 0.11$	$0.09\pm0.27$	0.42
Android fat mass (kg)	$0.02\pm0.05$	$0.01\pm0.08$	0.92
Gynoid fat mass (kg)	$-0.12 \pm 0.07$	$0.03\pm0.08$	0.24
Total lean mass (kg)	$-0.32 \pm 0.26$	$0.94 \pm 0.31^{*}$	< 0.01
Trunk lean mass (kg)	$-0.13 \pm 0.23$	$0.79 \pm 0.19^{*}$	< 0.01
Leg lean mass (kg)	$-0.08 \pm 0.12$	$0.10\pm0.14$	0.34
Arm lean mass (kg) <sup>a</sup>	$0.03\pm0.05$	$0.05\pm0.13$	0.90
Upper Body Fat (kg) <sup>a</sup>	$0.23\pm0.18$	$-0.14 \pm 0.37$	0.15
Lower Body Fat (kg) <sup>a</sup>	$-0.26 \pm 0.13$	$-0.12 \pm 0.15$	0.61
Waist Circ (cm) <sup>a</sup>	$0.36\pm0.46$	$\textbf{-0.98} \pm 0.43$	< 0.05
Hip Circ (cm) <sup>a</sup>	$-0.64 \pm 0.41$	$-0.36 \pm 0.85$	0.83
Biceps Skinfold (mm) <sup>a</sup>	$0.99\pm0.50$	$-1.90 \pm 1.98$	< 0.05
Triceps Skinfold (mm) <sup>a</sup>	$-0.42 \pm 0.46$	$1.70 \pm 1.59$	0.12
Subscapular Skinfold (mm)	$0.02\pm0.75$	$2.38 \pm 1.80$	0.46
Suprailiac Skinfold (mm) <sup>a</sup>	$0.59\pm0.79$	$-1.62 \pm 2.41$	0.30

	Controls	Offspring	P - value
	( <b>n=16</b> )	(n=13)	
Glucose			
Fasting (mmol.1 <sup>-1</sup> )	$-0.04 \pm 0.07$	$-0.03 \pm 0.08$	0.93
Mean during OGTT (mmol.l <sup>-1</sup> )	$-0.77 \pm 0.42$	$0.08\pm0.19$	0.16
Change from fasting during OGTT (mmol.l <sup>-1</sup> )	$-0.48 \pm 0.27$	$0.10 \pm 0.21$	0.18
Insulin			
Fasting <sup>a</sup> (mU.l <sup>-1</sup> )	$0.20\pm0.38$	$0.42\pm0.71$	0.44
Mean during OGTT <sup>a</sup> (mU.l <sup>-1</sup> )	$-0.58 \pm 4.01$	$-12.86 \pm 5.22$	0.20
Change from fasting during OGTT <sup>a</sup> (mU.l <sup>-1</sup> )	$-0.78 \pm 3.75$	$\textbf{-13.28} \pm 4.83$	0.17
NEFA			
Fasting (mmol.l <sup>-1</sup> )	$\textbf{-0.08} \pm 0.04$	$0.02\pm0.04$	0.05
Mean during OGTT <sup>a</sup> (mmol.l <sup>-1</sup> )	$\textbf{-0.03} \pm 0.03$	$\textbf{-}0.002\pm0.02$	0.36
Change from fasting during OGTT (mmol.l <sup>-1</sup> )	$0.07\pm0.02$	$\textbf{-}0.02\pm0.02$	0.20
3-ОНВ			
Fasting (mmol.l <sup>-1</sup> ) <sup>a</sup>	$\textbf{-}0.02\pm0.02$	$\textbf{-}0.03\pm0.03$	0.24
Mean during OGTT (mmol.l <sup>-1</sup> )	$-0.003 \pm 0.005$	$\textbf{-}0.01\pm0.02$	0.60
Change from fasting during OGTT(mmol.l <sup>-1</sup> )	$0.02\pm0.01$	$0.02\pm0.02$	0.99
Insulin sensitivity index <sup>a</sup>	$0.71\pm0.78$	$0.00\pm0.49$	0.25

**Table C.2** Absolute changes in glucose, insulin, NEFA and 3-OHB in the fasted state andduring OGTT and insulin sensitivity index after period of 'usual lifestyle'. Values are Mean  $\pm$ SEM.

	Controls	Offspring	P - value
	( <b>n=16</b> )	( <b>n=13</b> )	
Adiponectin (ng.ml <sup>-1</sup> ) <sup>a</sup>	$0.08 \pm 0.33$	$0.75 \pm 0.42$	0.55
Leptin (pg/ml <sup>-1</sup> ) <sup>a</sup>	$1.01 \pm 1.71$	$1.53 \pm 1.32$	0.43
TNF- $\alpha$ (pg.ml <sup>-1</sup> ) <sup>a</sup>	$0.02\pm0.04$	$0.05\pm0.07$	0.33
IL-6 $(pg.ml^{-1})^a$	$0.58\pm0.28$	$0.02\pm0.15$	0.14
Resistin (ng.ml <sup>-1</sup> )	$-0.25 \pm 0.73$	$0.70\pm0.64$	0.35
$CRP (mmol.l^{-1})^{a}$	$-0.03 \pm 0.31$	$0.83\pm0.37$	0.58

**Table C.3** Absolute change in fasting adipokines after period of 'usual lifestyle'. Values are Mean ± SEM.

Column P-values are for the interaction term in a 2-way ANOVA (group x trial) with repeated measures on the 'trial' factor. <sup>a</sup>statistical analysis performed on logarithmically transformed data. \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001 for difference from baseline within group.

**Table C.4** Absolute change in fasting liver enzymes and lipids after period of 'usual lifestyle'. Values are Mean ± SEM.

	Controls	Offspring	<b>P</b> -
	( <b>n=16</b> )	( <b>n=13</b> )	value
Fasting ALT (mmol.1 <sup>-1</sup> ) <sup>a</sup>	$-1.11 \pm 1.37$	$0.67 \pm 1.46$	0.52
Fasting GGT (mmol.1 <sup>-1</sup> ) <sup>a</sup>	$0.06 \pm 1.05$	$6.27 \pm 2.31^*$	0.06
Total cholesterol (mmol.l <sup>-1</sup> )	$-0.07 \pm 0.13$	$-0.03 \pm 0.10$	0.82
Fasting triglycerides (mmol.l <sup>-1</sup> ) <sup>a</sup>	$\textbf{-0.08} \pm 0.04$	$0.10\pm0.06$	< 0.05
HDL cholesterol (mmol.l <sup>-1</sup> )	$0.00\pm0.04$	$-0.01 \pm 0.04$	0.88
LDL cholesterol (mmol.l <sup>-1</sup> )	$-0.05 \pm 0.10$	$-0.06 \pm 0.09$	0.86

**Table C.5** Absolute change in metabolic rate, respiratory exchange ratio (RER), rates of fat and carbohydrate (CHO) oxidation in Controls and Offspring in the fasted state and 2 hours after glucose load, after period of 'usual lifestyle'. Values are Mean  $\pm$  SEM.

	Controls	Offspring	P - value
	( <b>n=16</b> )	(n=13)	
$RMR (kJ.kg^{-1}.day^{-1})$	$0.18\pm0.29$	$-0.07 \pm 0.34$	0.58
Fasting RER	$0.02\pm0.01$	$0.01\pm0.01$	0.75
Fasting fat oxidation rate (g.kg <sup>-1</sup> .day <sup>-1</sup> )	$-0.10 \pm 0.11$	$0.08\pm0.14$	0.32
Fasting CHO oxidation rate (g.kg <sup>-1</sup> .day <sup>-1</sup> )	$0.22\pm0.25$	$0.19\pm0.30$	0.92
Post-glucose metabolic rate (kJ.kg <sup>-1</sup> .day <sup>-1</sup> )	$0.13\pm0.24$	$0.05\pm0.33$	0.85
Post-glucose RER	$0.01\pm0.01$	$0.02\pm0.01$	0.40
Post-glucose fat oxidation rate (g.kg <sup>-1</sup> .day <sup>-1</sup> )	$-0.03 \pm 0.12$	$-0.17 \pm 0.11$	0.41
Post-glucose CHO oxidation rate (g.kg <sup>-1</sup> .day <sup>-1</sup> )	$0.03\pm0.29$	$0.73\pm0.35$	0.13

**Table C.6** Absolute changes in fasting systolic and diastolic blood pressure and peripheral PWV in fasted, and 2-hour post-glucose states, after period of 'usual lifestyle'. Values are Mean ± SEM.

	Controls	Offspring	P - value
	( <b>n=16</b> )	(n=13)	
Systolic blood pressure (mm Hg)	$1.19 \pm 2.66$	$-7.00 \pm 6.21$	0.21
Diastolic blood pressure (mm Hg)	$-6.38 \pm 2.27$	$-5.85 \pm 3.55$	0.90
Fasting PWV (m.s <sup>-1</sup> ) <sup>a</sup>	$-0.26 \pm 0.21$	$-0.21 \pm 0.37$	0.87
Post-glucose PWV (m.s <sup>-1</sup> ) <sup>a</sup>	$-0.72 \pm 0.40$	$0.34 \pm 0.90$	0.69

 Table C.7 Absolute change in cardiorespiratory fitness and indices of habitual physical

	Controls	Offspring	P - value
	( <b>n=16</b> )	(n=13)	
$VO_2max^a$ (ml.kg <sup>-1</sup> .min <sup>-1</sup> )	$-0.11 \pm 0.86$	$-1.58 \pm 0.59$	0.19
$VO_2$ at LT (ml.kg <sup>-1</sup> .min <sup>-1</sup> ) <sup>a</sup>	$-1.17 \pm 0.65$	$-0.49 \pm 1.12$	0.62
Resting HR <sup>b</sup> (b.min <sup>-1</sup> )	$0.82 \pm 1.15$	$1.80 \pm 1.35$	0.59
Average daily HR (b.min <sup>-1</sup> )	$-0.31 \pm 1.20$	$-1.07 \pm 1.04$	0.65
Average daily HR /resting HR	$\textbf{-}0.03\pm0.03$	$-0.06 \pm 0.03$	0.48
Time spent above 1.5 times RHR (min.day <sup>-1</sup> ) <sup>a</sup>	$17.90\pm28.02$	$-50.41 \pm 16.56$	0.20
Time spent above 2 times resting HR (min.day <sup>-1</sup> )	$3.73 \pm 6.17$	$-7.99 \pm 3.59$	0.15

activity after period of 'usual lifestyle'. Values are Mean  $\pm$  SEM.

**Table C.8** Absolute change in energy and macronutrient intake after period of 'usuallifestyle'. Values are Mean ± SEM.

	Controls	Offspring	P - value
	( <b>n=16</b> )	(n=13)	
Energy (kJ.kg <sup>-1</sup> .day <sup>-1</sup> )	$-4.26 \pm 2.80$	$2.83 \pm 3.32$	0.10
Total fat (g. kg <sup>-1</sup> .day <sup>-1</sup> )	$-0.13 \pm 0.10$	$0.10\pm0.10$	0.73
SFA $(g. kg^{-1}.day^{-1})^a$	$0.01 \pm 0.12$	$0.03\pm0.05$	0.85
$MUFA (g. kg^{-1}.day^{-1})$	$-0.04 \pm 0.03$	$0.03\pm0.04$	0.81
PUFA (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$0.00\pm0.02$	$0.02\pm0.01$	0.47
Total CHO (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$0.11 \pm 0.38$	$-0.55 \pm 0.41$	0.45
Sugar (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$-0.13 \pm 0.23$	$-0.44 \pm 0.17$	0.25
Starch (g. kg <sup>-1</sup> .day <sup>-1</sup> ) <sup>a</sup>	$\textbf{-0.07} \pm 0.28$	$0.24\pm0.17$	0.82
Protein $(g. kg^{-1}.day^{-1})^a$	$-0.17 \pm 0.10$	$0.00 \pm 0.05$	0.56